TEVA TEXAS STATE-WIDE OPIOID SETTLEMENT AND CONSENT JUDGMENT

I. Overview

Whereas, this Agreement sets forth the terms and conditions of a Texas State-Wide Opioid Settlement and Consent Judgment between and among the State of Texas (defined herein), all Texas Participating Subdivisions (defined herein), and Teva (defined herein) (collectively, "the Parties") to resolve opioid-related Claims against Teva (hereinafter, "Agreed Judgment").

Whereas, Teva has agreed to the below terms for the sole purpose of settlement, and nothing herein may be taken as or construed to be an admission or concession of any violation of law, rule, or regulation, or of any other matter of fact or law, or of any liability or wrongdoing, all of which Teva expressly denies. No part of this Agreement, including its statements and commitments, shall constitute evidence of any liability, fault, or wrongdoing by Teva. Unless the contrary is expressly stated, this Agreement is not intended for use by any third party for any purpose, including submission to any court for any purpose.

Whereas, this Agreed Judgment resolves, among other things, Teva's portion of: investigations by the Attorney General of the State of Texas relating to opioids; *County of Dallas v. Purdue Pharma, L.P., et al.*, MDL Pretrial Cause No. 2018-77098 and *County of Bexar v. Purdue Pharma, L.P., et al.*, MDL Pretrial Cause No. 2018-77066, both bellwether cases in *In re: Texas Opioid Litigation*, MDL No. 18-0358 (Harris County, Texas); *County of Harris v. Purdue Pharma, L.P., et al.*, Case No. 1:18-op-45677-DAP (N.D. Ohio); *Tarrant County v. Purdue Pharma, L.P., et al.*, Case No. 1:18-op-45274-DAP (N.D. Ohio); and cases brought by Participating Subdivisions.

The terms and conditions of the Agreed Judgment are set forth herein and below:

II. Definitions

- A. "Actions" means the investigations of Teva undertaken by the Attorney General of Texas relating to opioids and opioid-related claims brought by Participating Subdivisions against Teva, including but not limited to County of Dallas v. Purdue Pharma, L.P., et al., MDL Pretrial Cause No. 2018-77098 and County of Bexar v. Purdue Pharma, L.P., et al., MDL Pretrial Cause No. 2018-77066, both bellwether cases in In re: Texas Opioid Litigation, MDL No. 18-0358 (Harris County, Texas); County of Harris v. Purdue Pharma, L.P., et al., Case No. 1:18-op-45677-DAP (N.D. Ohio); and Tarrant County v. Purdue Pharma, L.P., et al., Case No. 1:18-op-45274-DAP (N.D. Ohio).
- B. "Agreement" or "Agreed Judgment" together or separately mean this term sheet and consent judgment together with the exhibits thereto.
- C. "Allergan plc" means (i) Allergan plc, (ii) all of its respective past and present direct or indirect parents, subsidiaries (for the elimination of doubt, including but not limited to Watson Laboratories, Inc.; Actavis LLC; Actavis Pharma, Inc. f/k/a Watson Pharma, Inc.; Warner Chilcott Co., LLC; Actavis South Atlantic LLC;

Actavis Elizabeth LLC; Actavis Mid Atlantic LLC; Actavis Totowa LLC; Actavis Kadian LLC; Actavis Laboratories UT, Inc. f/k/a Watson Laboratories Inc.-Salt Lake City; Actavis Laboratories FL, Inc. f/k/a Watson Laboratories, Inc.-Florida; and Anda, Inc.), divisions, affiliates, joint ventures, predecessors, successors, assigns, and insurers (in their capacity as such), and (iii) all of the foregoing respective past and present officers, directors, members, shareholders (solely in their capacity as shareholders of the foregoing entities), partners, trustees, employees, agents, attorneys, and insurers of the foregoing entities and persons referenced in clauses (i) and (ii) above for actions or omissions that occurred during and related to their work for, or employment with, any of the foregoing entities with respect to the Released Claims.

- D. "Bar" means either (1) a ruling by the highest court of the State setting forth the general principle that no Subdivisions or Special Districts in the State may maintain Released Claims against Released Entities, whether on the ground of the Agreement (or the release in it) or otherwise; (2) a law barring Subdivisions and Special Districts in the State from maintaining or asserting Released Claims against Released Entities (either through a direct bar or through a grant of authority to release claims and that authority is exercised in full); or (3) a Settlement Class Resolution in the State with full force and effect. For the avoidance of doubt, a law or ruling that is conditioned or predicated upon payment by a Released Entity (apart from payment of the Settlement Amount) shall not constitute a Bar.
- E. "Business" means each of or, as the context requires, any or all of the U.S. Generics Business, the International Generics Business, the OTC Business, the Transferred Brands Business, the Biostudy Research Business and the Transferred Owned Real Property of Seller Parent and its Controlled Affiliates, but excluding, for the avoidance of doubt, the Allergan Business, the Uteron Business, the Excluded Products and the Anda Business as these terms are defined in the Master Purchase Agreement, dated July 26, 2015, by and between Teva Pharmaceutical Industries Ltd. and Allergan plc, attached as Exhibit H.
- F. "Case-Specific Resolution" means either (1) a law barring specified Subdivisions or Special Districts from maintaining Released Claims against Released Entities (either through a direct bar or through a grant of authority to release claims and that authority is exercised in full); (2) a ruling by a court of competent jurisdiction over a particular Subdivision or Special District that has the legal effect of barring the Subdivision or Special District from maintaining any Released Claims at issue against Released Entities, whether on the ground of the Agreement (or the release in it) or otherwise; or (3) in the case of a Special District, a release consistent with Section VII below. For the avoidance of doubt, a law, ruling, or release that is conditioned or predicated upon a post-Effective Date payment by a Released Entity (apart from payment of the Settlement Amount) shall not constitute a Case-Specific Resolution.

- G. "Claim" means any past, present or future cause of action, claim for relief, crossclaim or counterclaim, theory of liability, demand, derivative claim, request, assessment, charge, covenant, damage, debt, lien, loss, penalty, judgment, right, obligation, dispute, suit, contract, controversy, agreement, parens patriae claim, promise, performance, warranty, omission, or grievance of any nature whatsoever, whether legal, equitable, statutory, regulatory or administrative, whether arising under federal, state or local common law, statute, regulation, guidance, ordinance or principles of equity, whether filed or unfiled, whether asserted or unasserted, whether known or unknown, whether accrued or unaccrued, whether foreseen, unforeseen or unforeseeable, whether discovered or undiscovered, whether suspected or unsuspected, whether fixed or contingent, and whether existing or hereafter arising, in all such cases, including but not limited to any request for declaratory, injunctive, or equitable relief, compensatory, punitive, or statutory damages, absolute liability, strict liability, restitution, subrogation, contribution, indemnity, apportionment, disgorgement, reimbursement, attorney fees, expert fees, consultant fees, fines, penalties, expenses, costs or any other legal, equitable, civil, administrative, or regulatory remedy whatsoever.
- H. "Covered Conduct" means any actual or alleged act, failure to act, negligence, statement, error, omission, breach of any duty, conduct, event, transaction, agreement, misstatement, misleading statement or other activity of any kind whatsoever from the beginning of time through the date of execution of this Agreement (and any past, present, or future consequence of any such act, failure to act, negligence, statement, error, omission, breach of duty, conduct, event, transaction, agreement, misstatement, misleading statement or other activity) arising from or relating in any way to (a) the discovery, development, manufacture, packaging, repackaging, marketing, promotion, advertising, labeling, recall, withdrawal, distribution, delivery, monitoring, reporting, supply, sale, prescribing, dispensing, physical security, warehousing, use or abuse of, or operating procedures relating to, any Product, or any system, plan, policy, or advocacy relating to any Product or class of Products, including but not limited to any unbranded promotion, marketing, programs, or campaigns relating to any Product or class of Products; (b) the characteristics, properties, risks, or benefits of any Product; (c) the reporting, disclosure, non-reporting, or non-disclosure to federal, state, or other regulators of orders for any Product placed with any Released Entity; (d) the selective breeding, harvesting, extracting, purifying, exporting, importing, applying for quota for, procuring quota for, handling, promoting, manufacturing, processing, packaging, supplying, distributing, converting, or selling of, or otherwise engaging in any activity relating to, precursor or component Products, including but not limited to natural, synthetic, semi-synthetic, or chemical raw materials, starting materials, finished active pharmaceutical ingredients, drug substances, or any related intermediate Products; or (e) diversion control programs or suspicious order monitoring related to any Product.

- I. "Consent Judgment" means a consent decree, order, judgment, or similar action (without any admission or finding of liability) as established through this Agreed Judgment.
- J. Except with respect to the Consent Judgment, "Court" means the Honorable Robert Schaffer, In Re: Texas Opioid Litigation, MDL No. 18-0358, Master File No. 2018-63587, in the 152nd Judicial District Court, Harris County, Texas. With respect to the Consent Judgment, "Court" means the court to which the Consent Judgment is presented for approval and/or entry.
- K. "Effective Date" means the date of entry of a final Consent Judgment, which shall be filed or otherwise submitted by the Parties to the Court no later than 30 days after the Initial Participation Date if the conditions set forth in Section III.C.1 are met.

L. "Finality" means:

- 1. the Agreement and the Consent Judgment have been approved and entered by the Court as to Teva, including the release of all Released Claims against Released Entities as provided in this Agreement;
- 2. for all lawsuits brought by the State against Released Entities for Released Claims, either previously filed or filed as part of the entry of the Consent Judgment, the Court has stated in the Consent Judgment or otherwise entered an order finding that all Released Claims against Released Entities asserted in the lawsuit have been resolved by agreement; and
- 3. (1) the time for appeal or to seek review of or permission to appeal from the approval and entry as described in subsection (a) hereof and entry of such order described in subsection (b) hereof has expired; or (2) in the event of an appeal, the appeal has been dismissed or denied, or the approval and entry described in (a) hereof and the order described in subsection (b) hereof have been affirmed in all material respects (to the extent challenged in the appeal) by the court of last resort to which such appeal has been taken and such dismissal or affirmance has become no longer subject to further appeal (including, without limitation, review by the United States Supreme Court).
- M. "Force Majeure Event" shall mean any event reasonably beyond the control of Teva, including wars, hostilities, revolution, riots, civil commotion, national emergency, unavailability of supplies, epidemics, fire, flood, earthquake, force of nature, explosion, terrorist act, embargo, or any act of God, or any law, proclamation, regulation, ordinance or other act or order of any court or governmental authority.

- N. "Initial Participation Date" means the date by which Subdivisions must join to become initial Participating Subdivisions. The Initial Participation Date shall be March 10, 2022.
- O. "Later Litigating Special District" means a Special District (or Special District official asserting the right of or for the Special District to recover for alleged harms to the Special District and/or the people thereof) that is not a Litigating Special District and that files a lawsuit bringing a Released Claim against a Released Entity, or that adds such a claim to a pre-existing lawsuit, after the execution date of this Agreement. It may also include a Litigating Special District whose claims were resolved by a judicial Bar or Case-Specific Resolution which is later revoked following the execution date of this Agreement, when such Litigating Special District takes any affirmative step in its lawsuit other than seeking a stay or removal.
- P. "Later Litigating Subdivision" means a Subdivision (or Subdivision official asserting the right of or for the Subdivision to recover for alleged harms to the Subdivision and/or the people thereof) that is not a Litigating Subdivision and that files a lawsuit bringing a Released Claim against a Released Entity, or that adds such a claim to a pre-existing lawsuit, after the Effective Date. It may also include a Litigating Subdivision whose claims were resolved by a judicial Bar or Case-Specific Resolution which is later revoked following the Effective Date, when such Litigating Subdivision takes any affirmative step in its lawsuit other than seeking a stay or removal.
- Q. "Litigating Special District" means a Special District (or Special District official) that brought any Released Claims against any Released Entities on or before the execution date of this Agreement that were not separately resolved prior to that date. Exhibit F includes Litigating Special Districts identified by the Parties as of the execution date but is subject to amendment in the event it proves to be incomplete and other entities that satisfy the definition for "Litigating Special District" are subsequently identified.
- R. "Litigating Subdivision" means a Subdivision (or Subdivision official asserting the right of or for the Subdivision to recover for alleged harms to the Subdivision and/or the people thereof) that brought any Released Claims against any Released Entities on or before the execution date that were not separately resolved prior to that date. Exhibit F includes Litigating Subdivisions identified by the Parties as of the Execution Date but is subject to amendment in the event it proves to be incomplete and other entities that satisfy the definition for "Litigating Subdivision" are subsequently identified.
- S. "Non-Litigating Special District" means a Special District that is neither a Litigating Special District nor a Later Litigating Special District.

- T. "Non-Litigating Subdivision" means a Subdivision that is neither a Litigating Subdivision nor a Later Litigating Subdivision.
- U. "*Non-Participating Subdivision*" means a Subdivision or Special District that is not a Participating Subdivision.
- V. "Participating Subdivision" means a Subdivision or Special District that signs the Election and Release Form annexed as Exhibit B and meets the requirements for becoming a Participating Subdivision¹ under subsection VIII.A. Outside Counsel for Dallas, Bexar, Harris and Tarrant Counties shall execute the Election and Release Form annexed as Exhibit B concurrently with their execution of this Agreement, subject to approval by the Commissioner's Court by the Initial Participation Date, and shall be Participating Subdivisions.
- W. "Product" means any chemical substance, whether used for medicinal or nonmedicinal purposes, and whether natural, synthetic, or semi-synthetic, or any finished pharmaceutical product made from or with such substance, that is an opioid or opiate, as well as any product containing any such substance. It also includes: 1) the following when used in combination with opioids or opiates: benzodiazepine, carisoprodol, zolpidem, or gabapentin; and 2) a combination or "cocktail" of any stimulant or other chemical substance prescribed, sold, bought, or dispensed to be used together that includes opioids or opiates. For the avoidance of doubt, "Product" does not include benzodiazepine, carisoprodol, zolpidem, or gabapentin when not used in combination with opioids or opiates. "Product" includes but is not limited to any substance consisting of or containing buprenorphine, codeine, fentanyl, hydrocodone, hydromorphone, meperidine, methadone, morphine, naloxone, naltrexone, oxycodone, oxymorphone, pentazocine, propoxyphene, tapentadol, tramadol, opium, heroin, carfentanil, any variant of these substances, or any similar substance. "Product" also includes any natural, synthetic, semisynthetic or chemical raw materials, starting materials, finished active pharmaceutical ingredients, drug substances, and any related intermediate products used or created in the manufacturing process for any of the substances described in the preceding sentence.
- X. "Qualified Settlement Fund" means the Texas Qualified Settlement Fund contemplated by this Agreement, into which all payments shall be made and Settlement Product provided by Teva, and which shall be established under the authority and jurisdiction of the Court.

6

¹ For purposes of this definition for Participating Subdivisions, it is the intent of the Parties to this Agreement that the Texas Opioid Council gives special consideration under the Texas Allocation Statute, including without limitation, Tex. Gov't Code §403.501(5) and §403.508(a)(2), as to whether any Special District was a litigating Special District prior to the date this Agreement was fully executed.

- Y. "Qualified Settlement Fund Administrator" means the Administrator appointed to administer the Texas Qualified Settlement Fund under the authority and jurisdiction of the Court.
- Z. "Released Claims" means any and all Claims that directly or indirectly are based on, arise out of, or in any way relate to or concern the Covered Conduct occurring prior to the Effective Date. "Released Claims" include any Claims that have been asserted against the Released Entities by the State or any of its Litigating Subdivisions or Litigating Special Districts in any federal, state or local action or proceeding (whether judicial, arbitral or administrative) based on, arising out of or relating to, in whole or in part, the Covered Conduct, or any such Claims that could be or could have been asserted now or in the future in those actions or in any comparable action or proceeding brought by the State, any of its Subdivisions or Special Districts, or any Releasor (whether or not such State, Subdivision, Special District, or Releasor has brought such action or proceeding). Released Claims also include all Claims asserted in any proceeding to be dismissed pursuant to the Agreement, whether or not such claims relate to Covered Conduct. Released Claims also includes (i) any and all Claims related to conduct of the Transferred Entities (including the Transferred Group or Business) for which Teva Pharmaceuticals Industries Ltd. assumed or became responsible, or for which any Released Entity is obligated to indemnify, defend, and hold Allergan plc harmless, pursuant to the terms of the Teva-Allergan Settlement Agreement, and (ii) Claims that are, directly or indirectly, jointly or severally, asserted against or imposed on Allergan plc to the extent such Claims are based on parent or control liability or a substantially similar theory in connection with any proceeding involving a member of the Transferred Group and a Product or the Business; however, Released Claims are not intended to and expressly do not include any claims against Allergan plc or, for the removal of doubt, any of its current or former subsidiaries in any way related to the development, manufacturing, marketing, promotion, sale, distribution, or dispensing of the branded opioid products Kadian, Norco, Fiorinal, and Combunox, in any of their formulations. The Parties intend that "Released Claims" be interpreted broadly. This Agreement does not release (i) Claims by individuals for damages for any alleged personal injuries arising out of their own use of any opioid product or (ii) Claims that any Releasor may have against Allergan plc (or, for the removal of doubt, any of its current or former subsidiaries) based upon the branded opioid drugs Kadian, Norco, Fiorinal, and Combunox. It is the intent of the Parties that such Claims by private individuals be treated in accordance with applicable law. This Agreement also does not release any Claims that the State has or may have against Released Entities based on state or federal antitrust violations. Released Claims is also used herein to describe Claims brought by a Later Litigating Subdivision or other non-party Subdivision or Special District that would have been Released Claims if they had been brought by a Releasor against a Released Entity.

- AA. "Released Entities" means (i) Teva, (ii) all of their respective past and present direct or indirect parents, subsidiaries, divisions, affiliates, joint ventures, predecessors, successors, assigns, and insurers (in their capacity as such), and (iii) all of the foregoing respective past and present officers, directors, members, shareholders (solely in their capacity as shareholders of the foregoing entities), partners, trustees, employees, agents, attorneys, and insurers of the foregoing entities and persons referenced in clauses (i) and (ii) above for actions or omissions that occurred during and related to their work for, or employment with, any of the foregoing entities with respect to the Released Claims. For the sake of clarity, Allergan plc, including its current and former subsidiaries, is not released with respect to any claims related to the development, manufacturing, marketing, promotion, sale, distribution, or dispensing of the branded opioid products Kadian, Norco, Fiorinal, or Combunox, in any of their formulations.
- BB. "Releasors" means (1) the State of Texas; (2) each Participating Subdivision, including Dallas, Bexar, Harris and Tarrant Counties; and (3) without limitation and to the maximum extent of the power of the State of Texas's Attorney General, and/or each Participating Subdivision to release Claims, (a) the State of Texas's and/or Participating Subdivision's departments, agencies, divisions, boards, commissions, Subdivisions, districts, instrumentalities of any kind and attorneys, including its Attorney General, and any person in their official capacity whether elected or appointed to serve any of the foregoing and any agency, person, or other entity claiming by or through any of the foregoing, (b) any public entities, public instrumentalities, public educational institutions, unincorporated districts, fire districts, irrigation districts, water districts, law enforcement districts, emergency services districts, school districts, hospital districts, and other Special Districts in the State, and (c) any person or entity acting in a parens patriae, sovereign, quasisovereign, private attorney general, qui tam, taxpayer, or other capacity seeking relief on behalf of or generally applicable to the general public with respect to the State of Texas or Subdivisions in the State, whether or not any of them participate in the Agreement. The inclusion of a specific reference to a type of entity in this definition shall not be construed as meaning that the entity is not a Subdivision. In addition to being a Releasor as provided herein, a Participating Subdivision shall also provide an Election and Release Form providing for a release to the fullest extent of the Participating Subdivision's authority, which shall be attached as an exhibit to the Agreement. The State of Texas's Attorney General represents that he or she has or has obtained the authority set forth in the Representation and Warranty Section.
- CC. "Settlement Amount" means the aggregate total sum to be paid pursuant to this Agreement by or on behalf of Teva and all Released Entities as specified in Section III.A.1 below. Except as provided in Section X and Section XII.D below, neither Teva nor any Released Entities shall be called upon to make any payments pursuant to this Agreement in addition to the amount set forth in Section III.A.1 below. Teva has no responsibility for any allocation of the Settlement Amount as set forth in this Agreement.

- DD. "Settlement Class Resolution" means a class action resolution in a court of competent jurisdiction in the State with respect to a class of Subdivisions and Special Districts in the State that (1) conforms with the State's statutes, case law, and/or rules of procedure regarding class actions; (2) is approved and entered as an order of a court of competent jurisdiction in the State and has achieved Finality; (3) is binding on all Non-Participating Subdivisions in the State (other than opt outs as permitted under the next sentence); (4) provides that all such Non-Participating Subdivisions may not bring Released Claims against Released Entities, whether on the ground of the Agreement (or the releases herein) or otherwise; and (5) does not impose any costs or obligations on Teva other than those provided for in the Agreement, or contain any provision inconsistent with any provision of the Agreement. If applicable State law requires that opt-out rights be afforded to members of the class, a class action resolution otherwise meeting the foregoing requirements shall qualify as a Settlement Class Resolution unless Subdivisions collectively representing 1% or more of the State's population opt out. In seeking certification of any Settlement Class, the applicable State and Participating Subdivisions shall make clear that certification is sought solely for settlement purposes and shall have no applicability beyond approval of the settlement for which certification is sought. Nothing in this Agreement constitutes an admission by any Party that class certification would be appropriate for litigation purposes in any case.
- EE. "Settlement Product" means "Naloxone Hydrochloride Nasal Spray" (4 mg strength) that is listed in Teva's then current generics catalog, which can be viewed at www.tevagenerics.com, and is provided to the State as part of the settlement, at no cost as set forth in Section III.B and Exhibit A.
- FF. "Special District" means a formal and legally recognized sub-entity of the State that is authorized by State law to provide one or a limited number of designated functions, such as school districts, fire districts, healthcare & hospital districts, and emergency services districts. Special Districts do not include sub-entities of the State that provide general governance for a defined area that would qualify as a Subdivision. A list of Texas Special Districts will be agreed to by the Parties.
- GG. "State" means the State of Texas, including all of its executive departments, agencies, divisions, boards, commissions, instrumentalities and officers, including the Attorney General.
- HH. "Subdivision(s)" means a formal and legally recognized sub-entity of the State that provides general governance for a defined area, including a county, city, town, village, or similar entity. Unless otherwise specified, "Subdivision" includes all functional counties and other functional levels of sub-entities of the State that provide general governance for a defined area. Historic, non-functioning sub-entities of the State are not Subdivisions, unless the entity has filed a lawsuit that includes a Released Claim against a Released Entity in a direct, parens patriae, or any other capacity. A list of Texas Subdivisions will be agreed to by the Parties.

- II. "Teva" means Teva Pharmaceutical Industries Ltd.; Teva Pharmaceuticals USA, Inc. ("Teva USA"); Cupric Holding Co., Inc.; Teva Pharmaceutical Holdings Cooperative U.A.; Teva Pharmaceuticals Europe B.V.; Cephalon, Inc.; Watson Laboratories, Inc.; Actavis LLC; Actavis Pharma, Inc. f/k/a Watson Pharma, Inc.; Warner Chilcott Co., LLC; Actavis South Atlantic LLC; Actavis Elizabeth LLC; Actavis Mid Atlantic LLC; Actavis Totowa LLC; Actavis Kadian LLC; Actavis Laboratories UT, Inc. f/k/a Watson Laboratories Inc.-Salt Lake City; Actavis Laboratories FL, Inc. f/k/a Watson Laboratories, Inc.-Florida; and Anda, Inc. and each of their current and former corporate parents, direct and indirect subsidiaries, predecessors, successors, affiliates, agents and current and former employees, officers and directors and any current or former related companies. "Teva" does not include Allergan plc or, for the elimination of doubt, its current or former subsidiaries, with respect to Claims related to the development, manufacturing, marketing, promotion, sale, distribution, or dispensing of the branded opioid products, Kadian, Norco, Fiorinal, or Combunox, in any of their formulations.
- JJ. "Teva-Allergan Settlement Agreement" means the January 31, 2018 Settlement Agreement and Mutual Releases executed by Teva Pharmaceutical Industries Ltd. and Allergan plc., included as Exhibit I to this Agreement.
- KK. "Transferred Entities" means the entities listed on Schedule 1.1(f) of the Master Purchase Agreement, dated July 26, 2015, by and between Teva Pharmaceutical Industries Ltd. and Allergan plc.
- LL. "*Transferred Group*" means, individually and collectively, each of the Transferred Entities and each of their direct and indirect subsidiaries.

III. Monetary Relief and Payments

- A. Remediation and Restitution Payments.
 - 1. Provided that (a) the Qualified Settlement Fund has been established under the authority and jurisdiction of the Court, and (b) Teva has received a W-9 and wire instructions for the Qualified Settlement Fund, Teva shall pay into the Texas Qualified Settlement Fund a total sum of One Hundred and Fifty Million Dollars (\$150,000,000) over a period of 15 years to be paid in accordance with the payment schedule set forth below:
 - a. Within 15 days after execution of this Agreement, Teva shall pay into the Texas Qualified Settlement Fund the sum of \$50,000,000.
 - b. In 2023 through 2033, Teva shall pay annually into the Texas Qualified Settlement fund the sum of \$5,000,000 on or before January 31 of each calendar year.

- c. In 2034 through 2036, Teva shall pay annually into the Texas Qualified Settlement fund the sum of \$15,000,000 on or before January 31 of each calendar year.
- 2. Release of these funds from the Qualified Settlement Fund is contingent upon the satisfaction of the conditions described in Section III.C.1 below. If the conditions described in Section III.C.1 below are not satisfied, the amount paid in Section III.A.1.a. shall revert forthwith to Teva and the amounts to be paid in Sections III. A.1.b and A.1.c shall not be disbursed.
- 3. The Settlement Amount shall be allocated in accordance with the Texas Opioid Abatement Fund Council and Settlement Allocation Term Sheet annexed hereto as Exhibit C and incorporated herein by reference (the "Texas Intrastate Term Sheet"). Accordingly, the Subdivision Share shall be \$19,730,437.50; the Texas Opioid Abatement Fund Share shall be \$92,075,375; the State Share shall be \$19,730,437.50; \$17,463,750 shall be allocated to the attorney's fee sub-fund within the Texas Qualified Settlement Fund pursuant to Section IX.A.1-2; and \$1,000,000 shall be allocated to the expenses sub-fund within the Texas Qualified Settlement Fund pursuant to Section IX.A.3.

B. Settlement Product.

- 1. Within 15 days after satisfaction of the conditions described in Section III.C.1 below and subject to Teva's good faith and reasonable efforts to meet the logistical requirements necessary to commence manufacturing of the needed increase in units, and continuing for a period of ten (10) years thereafter, the State may place orders with Teva USA for Settlement Product, to be supplied by Teva USA to one facility per order at no cost to the State ("Provision of the Settlement Product"), designated by the State, as more fully described in Exhibit A. The Settlement Product will be valued as follows:
 - a. For purposes of calculating the Settlement Product to be provided under the terms of this Agreement, the quantity of any order of Settlement Product shall be calculated based on the reference product holder's wholesale acquisition cost ("WAC") of the Settlement Product as of the Effective Date or the date of the order, whichever is lower. In the event that the WAC becomes lower following the Effective Date, the Parties agree to discuss in good faith any changes to the supply logistics for Settlement Product in Exhibit A as reasonably necessary.

- b. The total value of all orders placed by the State shall not exceed \$75,000,000 at WAC. Teva may reject any order that if fulfilled would cause the total value of all Settlement Product delivered to the State, calculated as described in Section III.B.1.a, to exceed this amount, in which case Teva shall have no obligation to fulfill or deliver the order, and the State shall reduce the order to an amount that does not exceed \$75,000,000 total for all orders by the State.
- c. In the event of a Force Majeure Event or other inability to supply any order made by the State for Settlement Product, Teva USA shall promptly provide written notice to the State. Teva USA and the State shall meet and confer within seven (7) days of such written notice to establish a commercially reasonable plan to resolve any inability to supply as quickly as reasonably possible.
- 2. Provision of the Settlement Product is contingent upon the satisfaction of the conditions described in Section III.C.1 below. If the conditions described in Section III.C.1 below are not satisfied, the Settlement Product shall not be provided.
- 3. For the avoidance of doubt, the Provision of the Settlement Product is neither money, funds, gifts, donation nor grants deposited into the Opioid Abatement Account under Tex. Gov't Code § 403.505(b) and § 403.507(a). Accordingly, the monetary value of the Provision of Settlement Product shall not be applied to the allocation of the Subdivision Share, Texas Opioid Abatement Fund Share, and/or the State Share. The Provision of the Settlement Product shall be distributed in accordance with Exhibit A.
- C. Release of Payment and Settlement Product for Full Joinder of Litigating Subdivisions and Special Districts and Support of Legislative Bar.
 - 1. If the Texas Attorney General notifies Teva by March 10, 2022, that (1) Litigating Subdivisions and Litigating Special Districts representing at least 96% of the population of Litigating Subdivisions have become Participating Subdivisions or had their claims released consistent with Section VII, and (2) all such Participating Subdivisions support the legislative enactment of a Bar as defined in Section II.D.2 and are using their best efforts to achieve enactment as soon as is practicable and will continue to employ such best efforts until such legislation is enacted, the Settlement Amount and the Settlement Product, less amounts withheld pursuant to Section IX below, shall be disbursed as provided in this Agreement in Sections III.A and III.B. The amounts withheld pursuant to Section IX below shall be disbursed in accordance with Section IX. The State and counsel for Dallas, Bexar, Harris and Tarrant Counties

acknowledge the materiality of their becoming Participating Subdivisions, release of claims consistent with Section VII, and their meaningful support for legislative enactment of a Bar to qualify for an accelerated payment under this subsection.

IV. Intra-State Allocation

A. The Settlement Amount shall be allocated according to this Agreement and the Texas Intrastate Term Sheet, and according to Texas law, including the guidelines established in Tex. Gov't Code Ann. Chapter 403, Subchapter R, Statewide Opioid Settlement.

V. Injunctive Relief

A. The State and Teva agree to the injunctive relief as specified in Exhibit D.

VI. Dismissal of Claims

A. Upon the execution of this Agreement, while awaiting formal approval of the Agreement by the Commissioners Courts of Dallas, Bexar, Harris and Tarrant Counties, the Parties agree to stay or extend all deadlines and proceedings in the Actions as to Teva and to jointly move for the claims against Teva to be severed from the Actions. It is the Parties' intent that all litigation activities in the Actions relating to the Claims against Teva shall immediately cease as of the date of the execution of this Agreement and that the Claims against Teva not be included in the trial of the Actions against the other defendants. To the extent that evidence produced by Teva is relevant to a claim against another defendant in the Actions, nothing in this Agreement precludes that evidence from being offered at trial. Concurrently with the execution of this Agreement, Dallas, Bexar, Harris and Tarrant Counties will execute an Agreed Motion to Dismiss the Released Claims with Prejudice and to sever Teva from the case, in the form annexed hereto as Exhibit E. The Parties will hold Dallas, Bexar, Harris and Tarrant Counties' Agreed Motion to Dismiss with Prejudice in escrow until the Counties' Commissioners Courts approve the Agreement or a resolution is passed satisfying the approval process of the Agreement and notice is given to Teva in accordance with Section III.C.1. Dallas, Bexar, Harris and Tarrant Counties and Teva shall promptly submit the executed Agreed Motion to Dismiss with Prejudice to the courts in which their actions are pending with a request that it be so ordered. In the event that Dallas, Bexar, Harris and/or Tarrant Counties' Commissioners Courts fail to approve the Agreement or the courts in which those Counties' Actions are pending decline to soorder the discontinuance of such Actions with prejudice as against Teva, Teva shall be entitled to terminate the Agreement and shall be excused from all obligations under it. Concurrently with the execution of this Agreement, Teva and the State will execute the Consent Judgment covering the State's claims against Teva. The

- Consent Judgment will be held in escrow until the Effective Date and shall be submitted to the Court with a request that it be so ordered.
- B. In the event that any Released Entity is brought back into any case by a Releasor following severance, any Releasor obtains a judgment against one or more of the Released Entities for Non-Released Claims, and Allergan refuses to assume contractual responsibility for that judgment, within 60 days of the judgment, Teva will seek indemnification for that judgment pursuant to the Teva-Allergan Settlement Agreement. Releasors agree that, to the extent that Teva is ultimately indemnified by Allergan through good-faith pursuance of their indemnity agreement, they will not seek to recover any judgment for Non-Released Claims against any Released Entity except in the amount that Allergan is ultimately determined to be liable or otherwise responsible to Teva under the Teva-Allergan Settlement Agreement in connection with the indemnification proceedings. Releasors agree that they will not make any attempt to execute any such judgment against any Released Entity until the relevant indemnification proceedings between Teva and Allergan are completed. Likewise, Releasors agree that they will not oppose any request or motion to suspend any requirement to post any bond in connection with any appeal by any Released Entity of the judgment, until the relevant indemnity proceedings between Teva and Allergan are completed. For the elimination of doubt, to the extent that Teva is unable to recover indemnity under the Teva-Allergan Settlement Agreement after good-faith effort, Releasors shall not attempt to otherwise enforce any judgment on any of the Released Entities related to Non-Released Claims. For the removal of doubt, a dismissal with prejudice provided pursuant to this Settlement Agreement shall not relieve Teva of the duty to seek indemnity under the Teva-Allergan Settlement Agreement. For purposes of this paragraph only, the term "Allergan" is limited to the definition of Allergan in the Teva-Allergan Settlement Agreement and the signatory to that Teva-Allergan Settlement Agreement.

VII. Release

A. Scope. As of the Effective Date, the Released Entities will be released and forever discharged from all of the Releasors' Released Claims. The State of Texas (for itself and its Releasors), Dallas, Bexar, Harris and Tarrant Counties (each for itself and its Releasors), and each Participating Subdivision (for itself and its Releasors) will, on or before the Effective Date, absolutely, unconditionally, and irrevocably covenant not to bring, file, or claim, or to cause, assist in bringing, or permit to be brought, filed, or claimed, or to otherwise seek to establish liability for any Released Claims against any Released Entity in any forum whatsoever. The releases provided for in the Agreement are intended by the Parties to be broad and shall be interpreted so as to give the Released Entities the broadest possible bar against any liability arising from or relating in any way to the Released Claims and extend to the full extent of the power of the State of Texas, its Attorney General,

and each Releasor to release claims. The Release shall be a complete bar to any Released Claim.

- B. Claim Over and Non-Party Settlement.
 - 1. *Statement of Intent*. It is the intent of the Parties that:
 - a. The payments made under this Agreement shall be the sole payments made by the Released Entities to the Releasors involving, arising out of, or related to the Released Claims;
 - b. Claims by Releasors against non-Parties should not result in additional payments by Released Entities for the Released Claims, whether through contribution, indemnification, or any other means; and
 - c. The Settlement meets the requirements of the Uniform Contribution Among Joint Tortfeasors Act and any similar state law or doctrine that reduces or discharges a released party's liability to any other parties.
 - d. The provisions of this subsection VII.B are intended to be implemented consistent with these principles. This Agreement and the releases and dismissals provided for herein are made in good faith.
 - 2. Releasors represent and warrant that this Agreement extinguishes whatever Claims, if any, that any Releasor may have against Allergan plc based upon or related in any way to generic opioid drugs; this Agreement expressly does not extinguish claims against Allergan plc or, for the elimination of doubt, its current or former subsidiaries, related to the development, manufacturing, marketing, promotion, sale, distribution, or dispensing of the branded opioid products Kadian, Norco, Fiorinal, or Combunox, in any formulation. Releasors further represent and warrant that if any Releasor enters into a separate settlement with or obtains any judgment against Allergan plc, any such settlement or judgment, will not involve any payment, loss, damages, expense, or relief for any Claims based upon or related in any way to generic opioid drugs. The representations and warranties in this subsection are a material term of this Agreement.
 - 3. Contribution/Indemnity Prohibited. No Released Entity shall seek to recover for amounts paid under this Agreement based on indemnification, contribution, or any other theory from a manufacturer, pharmacy, hospital, pharmacy benefit manager, health insurer, third-party vendor, trade association, distributor, or health care practitioner, provided that a Released

Entity shall be relieved of this prohibition with respect to any entity that asserts a Claim-Over against it. For the avoidance of doubt, nothing herein shall prohibit a Released Entity from recovering amounts owed pursuant to insurance contracts.

- 4. Non-Party Settlement. To the extent that, on or after the Effective Date, any Releasor enters into a Non-Party Settlement, including in any bankruptcy case or through any plan of reorganization (whether individually or as a class of creditors), related to Covered Conduct the Releasor will include (or in the case of a Non-Party Settlement made in connection with a bankruptcy case, will cause the debtor to include), unless prohibited from doing so under applicable law, in the Non-Party Settlement a prohibition on contribution or indemnity of any kind substantially equivalent to that required from Teva in subsection VII.B.3, or a release from such Non-Released Entity in favor of the Released Entities (in a form equivalent to the releases contained in this Agreement) of any Claim-Over. To the extent that a Releasor enters into any agreement with Allergan, plc resolving claims related to Covered Conduct, such a provision prohibiting contribution or indemnity may be limited, in the case of Teva, to a prohibition on contribution or indemnity for the Released Claims. The obligation to obtain the prohibition and/or release required by this subsection is a material term of this Agreement.
- 5. Claim-Over. In the event that any Releasor obtains a judgment with respect to Non-Party Covered Conduct against a Non-Released Entity that does not contain a prohibition like that in subsection VII.B.3 (including in any bankruptcy proceeding), or a Releasor is prevented for any reason from obtaining a prohibition/release in a Non-Party Settlement as provided in subsection VII.B.4 (including in any bankruptcy proceeding), and such Non-Released Entity asserts a Claim-Over against a Released Entity related to the Released Claims, that Releasor and Teva shall take the following actions to ensure that the Released Entities do not pay more with respect to the Released Claims to Releasors or to Non-Released Entities than the amounts owed under this Settlement Agreement by Teva:
 - a. Teva shall notify that Releasor of the Claim-Over within sixty (60) days of the assertion of the Claim-Over or sixty (60) days of the Effective Date of this Settlement Agreement, whichever is later;
 - b. Teva and that Releasor shall meet and confer concerning the means to hold Released Entities harmless and ensure that Released Entities are not required to make any payment with respect to the Released Claims (beyond the amounts and product provisions owed by Teva under this Settlement Agreement).

- c. That Releasor and Teva shall take steps sufficient and permissible under Texas law to hold Released Entities harmless from the Claim-Over with respect to Released Claims and ensure Released Entities are not required to make any payment with respect to the Released Claims (beyond the amounts and product provisions owed by Teva under this Settlement Agreement). Such steps may include, where permissible, filing of motions to dismiss or such other appropriate motion by Teva or Released Entities, and supported by Releasors, in response to any claim filed in litigation or arbitration or such other reasonable actions that ensure Teva is not required to pay more to Releasors with respect to Released Claims than the amounts owed or product provided by Teva under this Agreement.
- d. For the removal of doubt, Teva's payment and provision obligations under this agreement shall not be disrupted or delayed in the event of a Claim-Over, except by agreement of the parties to this Agreement.
- C. General Release. In connection with the releases provided for in the Agreement, the State of Texas (for itself and its Releasors), Dallas, Bexar, Harris and Tarrant Counties (each for itself and its Releasors), and each Participating Subdivision (for itself and its Releasors) will expressly waive, release, and forever discharge any and all provisions, rights, and benefits conferred by any law of any state or territory of the United States or other jurisdiction, or principle of common law, which is similar, comparable, or equivalent to § 1542 of the California Civil Code, which reads:

General Release; extent. A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release that, if known by him or her, would have materially affected his or her settlement with the debtor or released party.

A Releasor may thereafter discover facts other than or different from those which it knows, believes, or assumes to be true with respect to the Released Claims, but the State (for itself and its Releasors), Dallas, Bexar, Harris and Tarrant Counties (each for itself and its Releasors), and each Participating Subdivision (for itself and its Releasors) will expressly waive and fully, finally, and forever settle, release and discharge, upon the Effective Date, any and all Released Claims that may exist as of such date but which Releasors do not know or suspect to exist, whether through ignorance, oversight, error, negligence or through no fault whatsoever, and which, if known, would materially affect the State's decision to enter into the Agreement or the Participating Subdivisions' decision to participate in the Agreement.

- D. Cooperation. Releasors (i) will not encourage any person or entity to bring or maintain any Released Claim against any Released Entity and (ii) will reasonably cooperate with and not oppose any effort by a Released Entity to secure the prompt dismissal of any and all Released Claims. The State shall use its best efforts to secure releases consistent with this Section from all Litigating or Later Litigating Subdivisions and Special Districts.
- E. Res Judicata. Nothing in the Agreement shall be deemed to reduce the scope of the res judicata or claim preclusive effect that the settlement memorialized in the Agreement, and/or any Consent Judgment or other judgment entered on the Agreement, gives rise to under applicable law.
- F. Representation and Warranty. The signatories of this Agreement on behalf of the State of Texas and its Participating Subdivisions expressly represent and warrant that they will, on or before the Effective Date, have (or have obtained) the authority to settle and release, to the maximum extent of the State's power, all Released Claims of (1) the State of Texas, (2) all past and present executive departments, state agencies, divisions, boards, commissions and instrumentalities with the regulatory authority to enforce state and federal controlled substances acts, (3) any of the State of Texas's past and present executive departments, agencies, divisions, boards, commissions and instrumentalities that have the authority to bring Claims related to Covered Conduct seeking money (including abatement and/or remediation) or revocation of a pharmaceutical distribution license; and (4) any Participating Subdivisions. For the purposes of clause (3) above, executive departments, agencies, divisions, boards, commissions, and instrumentalities are those that are under the executive authority or direct control of the State's Governor. Also, for the purposes of clause (3), a release from the State's Governor is sufficient to demonstrate that the appropriate releases have been obtained.
- G. *Effectiveness*. The releases set forth in the Agreement shall not be impacted in any way by any dispute that exists, has existed, or may later exist between or among the Releasors. Nor shall such releases be impacted in any way by any current or future law, regulation, ordinance, or court or agency order limiting, seizing, or controlling the distribution or use of the Qualified Settlement Fund or any portion thereof, or by the enactment of future laws, or by any seizure of the Qualified Settlement Fund or any portion thereof.
- H. Non-Released Claims. Notwithstanding the foregoing or anything in the definition of Released Claims, the Agreement does not waive, release or limit any criminal liability, Claims for any outstanding liability under any tax or securities law, Claims against parties who are not Released Entities, Claims by individuals for damages for any alleged personal injuries arising out of their own use of any opioid product and any claims arising under the Agreement for enforcement of the Agreement.

VIII. Participation by Subdivisions

- A. Requirements for Becoming a Participating Subdivision: Litigating or Later Litigating Subdivisions or Litigating or Later Litigating Special Districts. A Litigating or Later Litigating Subdivision or Litigating or Later Litigating Special District in the State may become a Participating Subdivision either by executing an Election and Release Form and upon prompt dismissal of its legal action or by having its claims extinguished by operation of law or released by the State's Office of the Attorney General.
- B. *Notice*. The State's Office of the Attorney General shall send notice to all Non-Litigating Subdivisions in the State of Texas eligible to participate in the settlement and the requirements for participation. Such notice may include publication, email, and other standard forms of notification.
- C. Requirements for Becoming a Participating Subdivision: Non-Litigating Subdivisions and Non-Litigating Special Districts. A Non-Litigating Subdivision or a Non-Litigating Special District may become a Participating Subdivision either by executing an Election and Release Form specifying (1) that the Subdivision or Special District agrees to the terms of this Agreement pertaining to Subdivisions, (2) that the Subdivision or Special District releases all Released Claims against all Released Entities, and (3) that the Subdivision or Special District submits to the jurisdiction of the Court for purposes limited to the Court's role under the Agreement or by having their claims extinguished by operation of law or released by the State's Office of the Attorney General.
- D. *Non-Participating Subdivisions*. Non-Participating Subdivisions shall not directly receive any portion of the Texas Qualified Settlement Fund and the State may choose that its Non-Participating Subdivisions are ineligible for benefits from the fund.
- E. Representation With Respect to Participation Rate. The State of Texas represents and warrants for itself that it has a good faith belief that virtually all of Texas's Litigating Subdivisions and Litigating Special Districts will become Participating Subdivisions. The State acknowledges the materiality of the foregoing representation and warranty. Counsel for Dallas, Bexar, Harris and Tarrant Counties, in good faith, believe this is a fair Settlement. Therefore, counsel for Dallas, Bexar, Harris and Tarrant Counties will, in their best efforts, recommend this Settlement to their Subdivision clients within Texas. Further, the State of Texas and counsel for Dallas, Bexar, Harris and Tarrant Counties will use their best efforts to secure participation by all Subdivisions within Texas.
- F. Within 5 days of entry of the Notice of Dismissal per subsection VI, the Parties will seek to have entered the Case Management Order annexed hereto as Exhibit G.

And, further, Teva will participate in making motions to dismiss barred claims upon their release.

IX. Attorney Fee and Cost Payments

- A. The terms for attorney fee and cost payments to be allocated from the Settlement Amount are as follows:
 - 1. \$14,088,750, representing 9.3925% of the Settlement Amount, shall be allocated to the attorney fee sub-fund within the Texas Qualified Settlement Fund, to be available to reimburse Participating Subdivision attorney fees, upon application by eligible counsel who waive their contingency fees.
 - a. These fees shall be divided amongst Participating Subdivisions, including Dallas, Bexar, Harris and Tarrant Counties, as provided in the Texas Intrastate Term Sheet. Nothing in Section IX.A.1 is intended to limit the application of Sections C.5 and C.6 of the Texas Intrastate Term Sheet.
 - 2. \$3,375,000 shall be set aside within the Qualified Settlement Fund as the State's allocation for attorney's fees.
 - 3. The Qualified Settlement Fund Administrator shall withhold expenses of \$1,000,000, to be available to compensate Attorneys for Participating Subdivisions for costs and expenses arising out of representation of Participating Subdivisions related to their litigation against Teva. The costs and expenses shall be divided under the jurisdiction of the Court. No funds in the Litigating Subdivision Cost Fund may be used to compensate the costs incurred by Non-Participating Subdivisions, Non-Litigating Subdivisions or Non-Litigating Special Districts or costs and expenses arising out of representation of any such Subdivision or Special District.
 - 4. In addition to the compensation contemplated in the foregoing paragraph, the Qualified Settlement Fund Administrator shall allow reimbursement for reasonable costs and expenses as allowed by the Texas Intrastate Term Sheet from the Subdivision Share and Texas Abatement Fund Share, as provided in the Texas Intrastate Term Sheet, to be available to reimburse Participating Subdivision attorney costs and expenses upon application by eligible counsel who waive their contingency fees. These costs and expenses shall be divided under the jurisdiction and authority of the Court, amongst Participating Subdivisions, including Dallas, Bexar, Harris and Tarrant Counties, as provided in the Texas Intrastate Term Sheet. Any excess costs or expenses not allocated to reimburse Participating Subdivision attorney's costs and expenses pursuant to this Agreement under

- Exhibit C shall be replaced into to the Subdivision Share and Abatement Share Funds by the Qualified Settlement Fund Administrator.
- 5. Nothing in this agreement is intended to limit the application of the Texas Intrastate Term Sheet, which includes the calculation and process for allocation of fees and costs for Texas Political Subdivisions.
- 6. For the avoidance of doubt, nothing in this Section IX shall require any payment by Teva beyond the Settlement Amount.
- B. An attorney may not receive any reimbursement of fees, costs, or expenses unless the following eligibility criteria are met and annually certified by the Attorney:
 - 1. The attorney must expressly waive the enforcement against the Litigating Subdivision client of all Claims for fees, costs or expenses arising out of or related to any or all representations of any Participating Subdivision prior to applying for attorneys' fees in connection with this Agreement. All applications for attorneys' fees or costs in connection with this Agreement shall include an affirmation by the attorney of such waiver and notice to the client(s) of such waiver. For the avoidance of doubt, no attorney may recover fees or costs in connection with this Agreement unless the attorney expressly agrees not to seek fees, costs or expenses from each and every Participating Subdivision represented by that attorney.
 - 2. The attorney must represent that s/he has no present intent to represent or participate in the representation of any Later Litigating Subdivision or any Releasor with respect to Released Claims against Released Entities.
 - 3. The attorney must represent s/he will not charge or accept any referral fees for any Released Claims brought against Released Entities.
 - 4. The attorney may not have and must represent that s/he does not have a Claim for fees, costs or expenses related to a Later Litigating Subdivision.

X. Bankruptcy

A. Bankruptcy. On the Effective Date, this Agreed Judgment shall be filed in the Court as an Agreed Judgment as that term has meaning under Title 11 of the United States Code (the "Bankruptcy Code") and related jurisprudence. Nothing in this Agreement shall preclude the State or any Participating Subdivision from receiving a distribution from a potential bankruptcy of Teva to the extent that the State or Participating Subdivision has a right to receive a payment or distribution in accordance with this Agreement. Subject to the terms of this Agreement (including all releases, covenants and payment terms contained herein), all of the State's and the Participating Subdivisions' rights with respect to any bankruptcy case of Teva

are specifically reserved by the State and the Participating Subdivisions. If for any reason, the State or any Participating Subdivision must remit any portion of the Settlement Amount to a bankruptcy court or other party as a result of the commencement of a case with respect to Teva under Title 11 of the United States Code (the "Bankruptcy Code") then Teva shall make such payment to the State as soon as reasonably practicable.

XI. Enforcement and Dispute Resolution

- A. The terms of the Agreement are enforceable by the Participating Subdivisions before the Honorable Robert Schaffer, *In Re: Texas Opioid Litigation, MDL No. 18-0358*, Master File No. 2018-63587, in the 152nd Judicial District Court, Harris County, Texas, and by the State for the Consent Judgment in the court where the Consent Judgment is filed. Teva consents to the jurisdiction of the Texas MDL Court, and to the court in which the Consent Judgment is filed, limited to resolution of disputes identified in subsection XI.C.
- B. The parties to a dispute shall promptly meet and confer in good faith to resolve any dispute. If the parties cannot resolve the dispute informally, and unless otherwise agreed in writing, they shall follow the remaining provisions of this section to resolve the dispute.
- C. Disputes not resolved informally shall be resolved in the Court that entered the Consent Judgment for disputes with the Attorney General, or the Texas MDL Court for disputes with Participating Subdivisions. Texas law will apply.

XII. Miscellaneous

Taxes. Each of the Parties acknowledges, agrees, and understands that it is its A. intention that, for purposes of Section 162(f) of the Internal Revenue Code, the provision of the Settlement Amount and the Settlement Product by Teva (other than amounts directed to attorneys' fees and costs) constitutes restitution for damage or harm allegedly caused by the potential violation of a law and/or is an amount paid to come into compliance with the law. The Parties acknowledge, agree and understand that, other than the amounts directed to attorneys' fees and costs, no other portion of the Settlement Amount represents reimbursement to the State, any Participating Subdivision or other person or entity for the costs of any investigation or litigation, and no portion of the Settlement Amount represents or should properly be characterized as the payment of fines, penalties, or other punitive assessments. The State and Participating Subdivisions acknowledge, agree and understand that Teva intends to allocate the cost of the Settlement Amount among the Released Entities using a reasonable basis. If reasonably requested by Teva, the State and every Participating Subdivision shall complete and file Form 1098-F with the Internal Revenue Service, identifying the Settlement Amount and the Settlement Product (other than amounts directed to attorney fees and costs) as remediation/restitution amounts, and shall furnish Copy B of such Form 1098-F to Teva. Teva makes no warranty or representation to the State or any Participating Subdivision as to the tax consequences of the Settlement Amount or the Settlement Product or any portion thereof.

- B. Nothing in this Agreement shall be construed to authorize or require any action by Teva in violation of applicable federal, state, or other laws.
- C. Future Litigation Contracts. The State of Texas, by and through its Attorney General, represents that, to the extent permissible by law, it will not approve any future Subdivision or Special District outside counsel contracts for litigation arising out of or related to Covered Conduct against Teva.
- D. *Most Favored Nations*. If, after execution of this Agreement, there is a collective resolution—through settlement, bankruptcy or other mechanism—of substantially all claims against Teva brought by states, counties, and municipalities (a "Global Resolution") under which, but for this Agreement, the Texas allocation would be greater than the Settlement Amount on a net present value basis and the Settlement Product as valued in Section III.B.1, Teva shall pay the difference between the Settlement Amount and the Settlement Product as valued in Section III.B.1 and the amount that would have been allocated to Texas under the terms and in accordance with any such Global Resolution.
- E. *Modification*. This Agreement may be modified by a written agreement of the Parties or, in the case of the Consent Judgment, by court proceedings resulting in a modified judgment of the Court. For purposes of modifying this Agreement or the Consent Judgment, Teva must contact each of the Texas Attorney General and Counsel for Dallas, Bexar, Harris and Tarrant Counties for purposes of coordinating this process. Modifications must be in writing and agreed to by all of the parties to be enforceable.
- F. Any failure by any party to this Agreement to insist upon the strict performance by any other party of any of the provisions of this Agreement shall not be deemed a waiver of any of the provisions of this Agreement, and such party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Judgment.
- G. *Entire Agreement*. This Agreement represents the full and complete terms of the settlement entered into by the Parties hereto, except as provided herein. In any action undertaken by the Parties, no prior versions of this Agreement and no prior versions of any of its terms may be introduced for any purpose whatsoever.

- H. *Counterparts*. This Agreement may be executed in counterparts, and a facsimile or .pdf signature shall be deemed to be, and shall have the same force and effect as, an original signature.
- I. *Notice*. All notices under this Agreement shall be provided to the following via email and Overnight Mail:

For Defendant:

Teva Pharmaceuticals Attn: General Counsel's Office 400 Interpace Parkway Parsippany, NJ 07054

Copy to Teva's attorneys at:

Eric W. Sitarchuk Morgan, Lewis & Bockius LLP 1701 Market Street Philadelphia, PA 19103-2921 eric.sitarchuk@morganlewis.com

Rebecca J. Hillyer Morgan, Lewis & Bockius LLP 1701 Market Street 19103-2921 Philadelphia, PA rebecca.hillyer@morganlewis.com

For the Attorney General:

Stephanie Eberhardt
Assistant Attorney General
Office of the Attorney General
PO Box 12548
Austin, Texas 78711-2548
stephanie.eberhardt@oag.texas.gov

For Dallas County:

Jeffrey B. Simon Simon Greenstone Panatier, P.C. 1201 Elm Street, Suite 3400 Dallas, Texas 75270 Phone: (214) 276-7680 jsimon@sgptrial.com

For Bexar County:

Mikal C. Watts Watts Guerra LLC 4 Dominion Dr., Bldg 3, Suite 100 San Antonio, Texas 78257 Phone: (210) 447-0500 mcw@wattsguerra.com

For Harris County:

Dan Downey 1609 Shoal Creek Blvd., #100 Austin, Texas 78701 dan@dandowney.com

For Tarrant County:

Dara Hegar
The Lanier Law Firm P.C.
10940 West Sam Houston Pkwy N., Suite 100
Houston, Texas 77064
Phone: (713) 659-5200
Dara.Hegar@LanierLawFirm.com

Approved:

Dated:	2/4/2022	_				
		TEVA THARMACEUTICALS				
		By:				
		Бу				

Name: Eric W. Sitarchuk Rebecca J. Hillyer

Title: Morgan Lewis & Bockius LLP

Attorneys for Teva

THE STATE OF TEXAS
By:
Name: Brent Webster
Title: First Assistant Attorney General, Texas
THE COUNTY OF DALLAS, TEXAS
By:
Name: Jeffrey Simon
Title: Shareholder
Attorneys for the County of Dallas, Texas
THE COUNTY OF BEXAR, TEXAS
By:
Name: Mikal Watts
Title: Capital Partner
Attorneys for the County of Bexar, Texas
THE COUNTY OF HARRIS, TEXAS
By:
Name: Dan Downey
Title:
Attorneys for the County of Harris, Texas
THE COLDITAL OF TARRANT TENAC
THE COUNTY OF TARRANT, TEXAS
By:
Name: Dara Hegar
Title: Managing Attorney
Attorneys for the County of Tarrant, Texas

THE STATE OF TEXAS

Exhibit A

State Plan for Acceptance and Delivery of Settlement Product

Orders to TEVA

The Office of the Attorney General, on behalf of the State, shall place periodic orders, not to exceed four (4) quarterly orders per year, to Teva USA (Teva) for fulfillment of Settlement Product over a period of ten (10) years from the Effective Date. The total value of all orders placed by the State over the ten-year period shall not exceed \$75,000,000.

The total value of Settlement Product requested shall not exceed the following dollar amounts during a twelve-month period, as calculated based on the reference product holder's wholesale acquisition cost ("WAC") for the Settlement Product as of the Effective Date or the date of each order, whichever is lower:

O Years 1-6: \$10 million per year

O Years 7-10: \$3.75 million per year

The Settlement Product order from the State shall be in writing and directed to Teva USA's affiliate Anda, Inc., 2915 Weston Road, Weston, FL 33331, Attention: Patrick Cochrane, patrick.cochrane@andanet.com and Anthony Mihelich, anthony.mihelich@andanet.com. Each Settlement Product order must identify the reference product holder's WAC for the Settlement Product, the available annual amount remaining for fulfillment, and the total WAC value of Settlement Product delivered by Teva USA as of the date of the order. Teva USA shall respond to the State's order request within seven (7) calendar days confirming the order.

Teva USA will use its commercially reasonable efforts to ship the order directly to the facility designated by the State within six (6) months of the order at no cost to the State and shall provide the State with estimated delivery dates for receipt of the Settlement Product. Notwithstanding the foregoing, for each order from the State following the initial order, Teva USA agrees that it will use its good faith efforts to ship Settlement Product to the facility designated by the State within ninety (90) days of the order.

Delivery to State-Designated Facility

Delivery of the Settlement Product shall occur no more than five (5) business days after the shipment date. Should delivery not occur within this deadline, Teva USA agrees to notify the State in writing and to work in good faith to resolve shipping or delivery issues that may arise.

Shipping shall occur in the same manner that Teva USA regularly ships this Settlement Product, and any damages to the Settlement Product or other shipping damages or liability arising prior to receipt of the Settlement Product by the State shall be fully the responsibility of Teva USA. Should damage to Settlement Product occur during shipping, Teva USA agrees to re-ship the amount damaged promptly and at no cost to the State.

The State shall designate one location per order for delivery of Settlement Product by Teva USA and identify the location in the Settlement Product order. The Texas Department of Emergency Management (TDEM), located in Austin, Texas, will receive the Settlement Product on behalf of the State, but the State reserves the right to designate a different delivery location within Texas during the pendency of the settlement agreement at its discretion.

Should the State determine that an alternate state facility or agency will receive the Settlement Product during the pendency of the settlement, the State shall notify Teva USA and its affiliate Anda, Inc. in writing through the Settlement Product order.

Teva USA shall deliver to any facility in Texas identified by the State but may not be required to deliver Settlement Product to more than one location per order for any given delivery.

The State agrees to receive the Settlement Product in a location with appropriate storage accommodations and will comply with all applicable state and federal laws surrounding receipt of the Settlement Product.

The State shall inspect the Settlement Product within five (5) business days upon arrival at the state facility. If the State identifies damages to the Settlement Product during the inspection, Teva USA agrees to work in good faith to replace the damaged Settlement Product promptly.

Delivery of the Settlement Product is complete when Teva USA delivers all units of a particular order to the state facility and when both parties or their designees sign an invoice confirming the amount of units of Settlement Product received by the State.

Distribution by State

The State intends to distribute the Settlement Product to law enforcement agencies, first responders, and healthcare professionals throughout Texas ("Recipients"). The time, place, and manner of distribution of the Settlement Product by the State will be determined solely by the State. The State will require appropriate training on proper use of the Settlement Product by Recipients.

The State retains the right to alter its distribution plan according to the State's needs, including the right to store the Settlement Product at a state facility for any length of time. The State may distribute the Settlement Product as it deems best to address the opioid-related public health crisis in Texas, and alteration of distribution to Recipients shall be at the sole discretion of the State without regard to the preferences or recommendations of Teva USA.

Exhibit B

TEXAS SUBDIVISION AND SPECIAL DISTRICT ELECTION AND RELEASE FORM

This Election and Release Form for Texas Participating Subdivisions¹ resolves opioidrelated Claims against Teva under the terms and conditions set forth in the Teva Texas State-Wide Opioid Settlement Agreement between Teva, the State of Texas, and the Counties of Dallas, Bexar, Harris and Tarrant (the "Agreement"), the provisions of which are here incorporated by reference in their entirety. Upon executing this Election and Release Form, a Participating Subdivision agrees that, in exchange for the consideration described in the Agreement, the Participating Subdivision is bound by all the terms and conditions of the Agreement, including but not limited to the Release found in Section VII of the Agreement and the provisions concerning participation by Subdivisions or Special Districts in Section VIII, and the Participating Subdivision and its signatories expressly represent and warrant on behalf of themselves that they have, or will have obtained on or before the Effective Date or on or before the execution of this Election and Release Form if executed after the Effective Date, the authority to settle and release, to the maximum extent of the Subdivision's and Special District's power, all Released Claims related to Covered Conduct. If this Election and Release Form is executed on or before the Initial Participation Date, the Participating Subdivision shall dismiss the Released Claims with prejudice and sever Teva and all other Released Entities from all pending cases in which the Participating Subdivision has asserted Covered Claims against Teva or a Released Entity no later than the Initial Participation Date. If this Election and Release Form is executed after the Initial Participation Date, the Participating Subdivision shall dismiss the

¹ The Agreement defines a "Participating Subdivision" as a Subdivision or Special District that signs this Election and Release Form and meets the requirements for becoming a Participating Subdivision under subsection VIII.A. of the Agreement.

Released Claims with prejudice and sever Teva and all other Released Entities from all pending cases in which the Participating Subdivision has asserted Covered Claims against Teva or a Released Entity concurrently with the execution of this form. By executing this Election and Release Form, the Participating Subdivision submits to the jurisdiction of the Honorable Robert Schaffer, *In Re: Texas Opioid Litigation*, MDL No. 18-0358, Master File No. 2018-63587, in the 152nd Judicial District Court, Harris County, Texas.

Dated:			

[TX SUBDIVISION OR SPECIAL DISTRICT]

By:____

[COUNSEL]

[FIRM]

[ADDRESS]

[TELEPHONE] [EMAIL ADDRESS]

Counsel for [TX SUBDIVISION OR SPECIAL DISTRICT]

Exhibit C

TEXAS OPIOID SETTLEMENT SHARING AGREEMENT

TEXAS OPIOID ABATEMENT FUND COUNCIL AND SETTLEMENT ALLOCATION TERM SHEET

WHEREAS, the people of the State of Texas and its communities have been harmed through the National and Statewide epidemic caused by licit and illicit opioid use and distribution within the State of Texas; and now,

WHEREAS, the State of Texas, though its elected representatives and counsel, including the Honorable Ken Paxton, Attorney General of the State of Texas, and certain Political Subdivisions, through their elected representatives and counsel, are separately engaged in litigation seeking to hold those entities in the supply chain accountable for the damage caused; and now,

WHEREAS, the State of Texas, through its Attorney General and its Political Subdivisions, share a common desire to abate and alleviate the impacts of the epidemic throughout the State of Texas; and now,

THEREFORE, the State of Texas and its Political Subdivisions, subject to completing formal documents effectuating the Parties' agreements, enter into this State of Texas and Texas Political Subdivisions' Opioid Abatement Fund Council and Settlement Allocation Term Sheet (Texas Term Sheet) relating to the allocation and use of the proceeds of any Settlements as described.

A. Definitions

As used in this Texas Term Sheet:

- 1. "The State" shall mean the State of Texas acting through its Attorney General.
- 2. "Political Subdivision(s)" shall mean any Texas municipality and county.
- 3. "The Parties" shall mean the State of Texas, the Political Subdivisions, and the Plaintiffs' Steering Committee and Liaison Counsel (PSC) in the Texas Opioid MDL, *In Re: Texas Opioid Litigation*, MDL No. 2018-63587, in the 152d District Court of Harris County, Texas.
- 4. "Litigating Political Subdivision" means a Political Subdivision that filed suit in the state courts of the State of Texas prior to the Execution Date of this Agreement, whether or not such case was transferred to Texas Opioid MDL, or removed to federal court.
- 5. "National Fund" shall mean any national fund established for the benefit of the Texas Political Subdivisions. In no event shall any National Fund be used to create federal jurisdiction, equitable or otherwise, over the Texas Political Subdivisions or those similarly situated state-court litigants who are included in the state coalition, nor shall the National Fund require participating in a class action or signing a participation agreement as part of the criteria for participating in the National Fund.
- 6. "Negotiating Committee" shall mean a three-member group comprising four representatives for each of (1) the State; (2) the PSC; and (3) Texas'

Political Subdivisions (collectively, "Members"). The State shall be represented by the Texas Attorney General or his designees. The PSC shall be represented by attorneys Mikal Watts, Jeffrey Simon, Dara Hegar, Dan Downey, or their designees. Texas' Political Subdivisions shall be represented by Clay Jenkins (Dallas County Judge), Terrence O'Rourke (Special Assistant County Attorney, Harris County), Nelson Wolff (Bexar County Judge), and Nathaniel Moran (Smith County Judge) or their designees.

- 7. "Settlement" shall mean the negotiated resolution of legal or equitable claims against a Pharmaceutical Supply Chain Participant that includes the State and Political Subdivisions.
- 8. "Opioid Funds" shall mean monetary amounts obtained through a Settlement as defined in this Texas Term Sheet.
- 8. "Approved Purpose(s)" shall mean those uses identified in Exhibit A hereto.
- 9. "Pharmaceutical Supply Chain" shall mean the process and channels through which opioids or opioids products are manufactured, marketed, promoted, distributed, or dispensed.

- 10. "Pharmaceutical Supply Chain Participant" shall mean any entity that engages in or has engaged in the manufacture, marketing, promotion, distribution, or dispensing of an opioid analgesic.
- 11. "Texas Opioid Council" shall mean the Council described in Exhibit A hereto, which has the purpose of ensuring the funds recovered by Texas (through the joint actions of the Attorney General and the Texas Political Subdivisions) are allocated fairly and spent to remediate the opioid crisis in Texas, using efficient and cost-effective methods that are directed to the hardest hit regions in Texas while also ensuring that all Texans benefit from prevention and recovery efforts.

B. Allocation of Settlement Proceeds

1. All Opioid Funds distributed in Texas shall be divided with 15% going to Political Subdivisions ("Subdivision Share"), 70% to the Texas Opioid Abatement Fund through the Texas Opioid Council (Texas Abatement Fund Share) identified and described on Exhibits A and C hereto, and 15% to the Office of the Texas Attorney General as Counsel for the State of Texas ("State Share"). Out of the Texas Opioid Abatement Fund, reasonable expenses up to 1% shall be paid to the Texas Comptroller for the administration of the Texas Opioid Council pursuant to the Opioid

- Abatement Fund (Texas Settlement) Opioid Council Agreement, Exhibit A hereto.
- 2. The Subdivisions Share shall be allocated in accordance with the division of proceeds on Exhibit B hereto.
- 3. The Texas Abatement Fund Share shall be allocated to the Opioid Council to be apportioned in accordance with the guidelines of Exhibit A, and Exhibit C hereto.
- 4. In the event a Subdivision merges, dissolves, or ceases to exist, the allocation percentage for that Subdivision shall be redistributed as directed by the settlement document, and if not specified, equitably based on the composition of the successor Subdivision. If a Subdivision for any reason is excluded from a specific settlement, the allocation percentage for that Subdivision shall be redistributed as directed by the settlement document, and if not specified, equitably among the participating Subdivisions.
- 5. Funds obtained from parties unrelated to the Litigation, via grant, bequest, gift or the like, separate and distinct from the Litigation, may be directed to the Texas Opioid Council and disbursed as set forth below.
- 6. The Subdivision share shall be initially deposited and paid in cash directly to the Subdivision under the authority and guidance of the Texas MDL Court, who shall direct any Settlement funds to be held in trust in a

- segregated account to benefit the Subdivisions and to be promptly distributed as set forth herein and in accordance with Exhibit B.
- 7. Nothing in this Texas Term Sheet should alter or change any Subdivision's rights to pursue its own claim. Rather, the intent of this Texas Term Sheet is to join all parties to disburse settlement proceeds from one or more defendants to all parties participating in that settlement within Texas.
- 8. Opioid Funds from the Texas Abatement Fund Share shall be directed to the Texas Opioid Council and used in accordance with the guidelines as set out on Exhibit A hereto, and the Texas Abatement Fund Share shall be distributed to the Texas Opioid Council under the authority and guidance of the Texas MDL Court, consistent with Exhibits A and C, and the bylaws of the Texas Opioid Council documents and disbursed as set forth therein, including without limitation all abatement funds and the 1% holdback for expenses.
- 9. The State of Texas and the Political Subdivisions understand and acknowledge that additional steps may need to be undertaken to assist the Texas Opioid Council in its mission, at a predictable level of funding, regardless of external factors.

C. Payment of Counsel and Litigation Expenses

- 1. Any Master Settlement Agreement settlement will govern the payment of fees and litigation expenses to the Parties. The Parties agree to direct control of any Texas Political Subdivision fees and expenses to the "Texas Opioid Fee and Expense Fund," which shall be allocated and distributed by the Texas MDL Court, *In re: Texas Opioid Litigation*, MDL No. 2018-63587, in the 152nd District Court of Harris County, Texas, and with the intent to compensate all counsel for Texas Political Subdivisions who have not chosen to otherwise seek compensation for fees and expenses from any federal MDL common benefit fund.
 - 2. The Parties agree that no portion of the State of Texas 15% allocation share from any settlement shall be administered through the National Fund, the Texas MDL Court, or Texas Opioid Fee and Expense Fund, but shall be directed for payment to the State of Texas by the State of Texas.
 - 3. The State of Texas and the Texas Political Subdivisions, and their respective attorneys, agree that all fees whether contingent, hourly, fixed or otherwise owed by the Texas Political Subdivisions shall be paid out of the National Fund or as otherwise provided for herein to the Texas Opioid Fee and Expense Fund to be distributed by the 152nd

- District Court of Harris County, Texas pursuant to its past and future orders.
- 4. From any opioid-related settlements with McKesson, Cardinal Health, ABDC, and Johnson & Johnson, and for any future opioid-related settlements negotiated, in whole or in part, by the Negotiating Committee with any other Pharmaceutical Supply Chain Participant, the funds to be deposited in the Texas Opioid Fee and Expense Fund shall be 9.3925% of the combined Texas Political Subdivision and Texas Abatement Fund portions of each payment (annual or otherwise) to the State of Texas for that settlement, plus expenses from the National Fund, and shall be sought by Texas Political Subdivision Counsel initially through the National Fund. The Texas Political Subdivisions' percentage share of fees and expenses from the National Fund shall be directed to the Texas Opioid Fee and Expense Fund in the Texas MDL, as soon as is practical, for allocation and distribution in accordance with the guidelines herein.
- 5. If the National Fund share to the Texas Political Subdivisions is insufficient to cover the guaranteed 9.3925%, plus expenses from the National Fund, per subsection 4, immediately *supra*, or if payment from the National Fund is not received within 12 months after the date the

first payment is made by the Defendants pursuant to the settlement, then the Texas Political Subdivisions shall recover up to 12.5% of the Texas Political Subdivision Share to make up any difference.

6. If the National Fund and the Texas Political Subdivision share are insufficient to cover the guaranteed 9.3925%, plus expenses from the National Fund, or if payment from the National Fund is not received within 12 months after the date the first payment is made by the Defendants pursuant to the settlement, then the Texas Political Subdivisions shall recover up to 8.75% of the Abatement Fund Share to make up any difference. In no event shall the Texas Political Subdivision share exceed 9.3925% of the combined Texas Political Subdivision and Texas Abatement Fund portions of any settlement, plus expenses from the National Fund. In the event that any payment is received from the National Fund such that the total amount in fees and expenses exceeds 9.3925%, the Texas Political Subdivisions shall return any amounts received greater than 9.3925% of the combined Texas Political Subdivision and Texas Abatement Fund portions to those respective Funds.

- 7. For each settlement utilizing a National Fund, the Texas Political Subdivisions need only make one attempt at seeking fees and expenses there.
- 8. The total amount of the Texas Opioid Fee and Expense Fund shall be reduced proportionally, according to the agreed upon allocation of the Texas Subdivision Fund, for any Texas litigating Political Subdivision that (1) fails to enter the settlement; and (2) was filed in Texas state court, and was transferred to the Texas MDL (or removed before or during transfer to the Texas MDL) as of the execution date of this Agreement.

D. The Texas Opioid Council and Texas Abatement Fund

The Texas Opioid Council and Texas Abatement Fund is described in detail at Exhibit A, incorporated herein by reference.

E. Settlement Negotiations

1. The State and Negotiating Committee agree to inform each other in advance of any negotiations relating to a Texas-only settlement with a Pharmaceutical Supply Chain Participant that includes both the State and its Political Subdivisions and shall provide each other the opportunity to participate in all such negotiations. Any Texas-only Settlement agreed to with the State and Negotiating Committee shall be subject to the approval

of a majority of litigating Political Subdivisions. The Parties further agree to keep each other reasonably informed of all other global settlement negotiations with Pharmaceutical Supply Chain Participants and to include the Negotiating Committee or designees. Neither this provision, nor any other, shall be construed to state or imply that either the State or the Negotiating Committee is unauthorized to engage in settlement negotiations with Pharmaceutical Supply Chain Participants without prior consent or contemporaneous participation of the other, or that either party is entitled to participate as an active or direct participant in settlement negotiations with the other. Rather, while the State's and Negotiation Committee's efforts to achieve worthwhile settlements are to be collaborative, incremental stages need not be so.

- 2. Any Master Settlement Agreement (MSA) shall be subject to the approval and jurisdiction of the Texas MDL Court.
- 3. As this is a Texas-specific effort, the Committee shall be Chaired by the Attorney General. However, the Attorney General, or his designees, shall endeavor to coordinate any publicity or other efforts to speak publicly with the other Committee Members.
- 4. The State of Texas, the Texas MDL Plaintiff's Steering Committee representatives, or the Political Subdivision representatives may withdraw

from coordinated Settlement discussions detailed in this Section upon 10 business days' written notice to the remaining Committee Members and counsel for any affected Pharmaceutical Supply Chain Participant. The withdrawal of any Member releases the remaining Committee Members from the restrictions and obligations in this Section.

5. The obligations in this Section shall not affect any Party's right to proceed with trial or, within 30 days of the date upon which a trial involving that Party's claims against a specific Pharmaceutical Supply Chain Participant is scheduled to begin, reach a case specific resolution with that particular Pharmaceutical Supply Chain Participant.

F. Amendments

The Parties agree to make such amendments as necessary to implement the intent of this agreement.

Acknowledgment of Agreement

We, the undersigned, have participated in the drafting of the above Texas Term Sheet, including consideration based on comments solicited from Political Subdivisions. This document has been collaboratively drafted to maintain all individual claims while allowing the State and its Political Subdivisions to cooperate in exploring all possible means of resolution. Nothing in this agreement binds any party to any specific outcome. Any resolution under this document will require

acceptance by the State of Texas and a majority of the Litigating Political Subdivisions.

We, the undersigned, hereby accept the STATE OF TEXAS AND TEXAS POLITICAL SUBDIVISIONS' OPIOID ABATEMENT FUND COUNCIL AND SETTLEMENT ALLOCATION TERM SHEET. We understand that the purpose of this Texas Term Sheet is to permit collaboration between the State of Texas and Political Subdivisions to explore and potentially effectuate earlier resolution of the Opioid Litigation against Pharmaceutical Supply Chain Participants. We also understand that an additional purpose is to create an effective means of distributing any potential settlement funds obtained under this Texas Term Sheet between the State of Texas and Political Subdivisions in a manner and means that would promote an effective and meaningful use of the funds in abating the opioid epidemic throughout Texas.

EXHIBIT A

Opioid Abatement Fund (Texas) Settlement

Opioid Council

As part of the settlement agreement and upon its execution, the parties will form the Texas Opioid Council (Council) to establish the framework that ensures the funds recovered by Texas (through the joint actions of the Attorney General and the state's political subdivisions) are allocated fairly and spent to remediate the opioid crisis in Texas, using efficient and cost-effective methods that are directed to the hardest hit regions in Texas while also ensuring that all Texans benefit from prevention and recovery efforts.

I. Structure

The Council will be responsible for the processes and procedures governing the spending of the funds held in the Texas Abatement Fund, which will be approximately 70% of all funds obtained through settlement and/or litigation of the claims asserted by the State and its subdivisions in the investigations and litigation related to the manufacturing, marketing, distribution, and sale of opioids and related pharmaceuticals.

Money paid into the abatement fund will be held by an independent administrator, who shall be responsible for the ministerial task of releasing funds solely as authorized below by the Council, and accounting for all payments to and from the fund.

The Council will be formed when a court of competent jurisdiction enters an order settling the matter, including any order of a bankruptcy court. The Council's members must be appointed within sixty (60) days of the date the order is entered.

A. Membership

The Council shall be comprised of the following thirteen (13) members:

1. Statewide Members.

Six members appointed by the Governor and Attorney General to represent the State's interest in opioid abatement. The statewide members are appointed as follows:

- a. The Governor shall appoint three (3) members who are licensed health professionals with significant experience in opioid interventions;
- b. The Attorney General shall appoint three (3) members who are licensed professionals with significant experience in opioid incidences; and
- c. The Governor will appoint the Chair of the Council as a non-voting member. The Chair may only cast a vote in the event there is a tie of the membership.

2. Regional Members.

Six (6) members appointed by the State's political subdivisions to represent their designated Texas Health and Human Services Commission "HHSC" Regional Healthcare

Partnership (Regions) to ensure dedicated regional, urban, and rural representation on the Council. The regional appointees must be from either academia or the medical profession with significant experience in opioid interventions. The regional members are appointed as follows:

- a. One member representing Regions 9 and 10 (Dallas Ft-Worth);
- b. One member representing Region 3 (Houston);
- c. One member representing Regions 11, 12, 13, 14, 15, 19 (West Texas);
- d. One member representing Regions 6, 7, 8, 16 (Austin-San Antonio);
- e. One member representing Regions 1, 2, 17, 18 (East Texas); and
- f. One member representing Regions 4, 5, 20 (South Texas).

B. Terms

All members of the Council are appointed to serve staggered two-year terms, with the terms of members expiring February 1 of each year. A member may serve no more than two consecutive terms, for a total of four consecutive years. For the first term, four (4) members (two (2) statewide and two (2) for the subdivisions) will serve a three-year term. A vacancy on the Council shall be filled for the unexpired term in the same manner as the original appointment. The Governor will appoint the Chair of the Council who will not vote on Council business unless there is a tie vote, and the subdivisions will appoint a Vice-Chair voting member from one of the regional members.

C. Governance

1. Administration

The Council is attached administratively to the Comptroller. The Council is an independent, quasi-governmental agency because it is responsible for the statewide distribution of the abatement settlement funds. The Council is exempt from the following statutes:

- a. Chapter 316 of the Government Code (Appropriations);
- b. Chapter 322 of the Government Code (Legislative Budget Board);
- c. Chapter 325 of the Government Code (Sunset);
- d. Chapter 783 of the Government Code (Uniform Grants and Contract Management);
- e. Chapter 2001 of the Government Code (Administrative Procedure);
- f. Chapter 2052 of the Government Code (State Agency Reports and Publications);
- g. Chapter 2261 of the Government Code (State Contracting Standards and Oversight);
- h. Chapter 2262 of the Government Code (Statewide Contract Management);

- i. Chapter 262 of the Local Government Code (Purchasing and Contracting Authority of Counties); and
- j. Chapter 271 of the Local Government Code (Purchasing and Contracting Authority of Municipalities, Counties, and Certain Other Local Governments).

2. Transparency

The Council will abide by state laws relating to open meetings and public information, including Chapters 551 and 552 of the Texas Government Code.

- i. The Council shall hold at least four regular meetings each year. The Council may hold additional meetings on the request of the Chair or on the written request of three members of the council. All meetings shall be open to the public, and public notice of meetings shall be given as required by state law.
- ii. The Council may convene in a closed, non-public meeting:
 - a. If the Commission must discuss:
 - 1. Negotiation of contract awards; and
 - 2. Matters specifically exempted from disclosure by federal and state statutes.
 - b. All minutes and documents of a closed meeting shall remain under seal, subject to release only order of a court of competent jurisdiction.

3. Authority

The Council does not have rulemaking authority. The terms of each Judgment, Master Settlement Agreement, or any Bankruptcy Settlement for Texas control the authority of the Council and the Council may not stray outside the bounds of the authority and power vested by such settlements. Should the Council require legal assistance in determining their authority, the Council may direct the executive director to seek legal advice from the Attorney General to clarify the issue.

D. Operation and Expenses

The independent administrator will set aside up to one (1) percent of the settlement funds for the administration of the Council for reasonable costs and expenses of operating the foregoing duties, including educational activities.

1. Executive Director

The Comptroller will employ the executive director of the Council and other personnel as necessary to administer the duties of the Council and carry out the functions of the Council. The executive director must have at least 10 years of experience in government or public administration and is classified as a Director V/B30 under the State Auditor's State Classification. The Comptroller will pay the salaries of the Council employees from the

one (1) percent of the settlement funds set aside for the administration of the Council. The Comptroller will request funds from the Texas Abatement Fund Point of Contact.

2. Travel Reimbursement

A person appointed to the Council is entitled to reimbursement for the travel expenses incurred in attending Council duties. A member of the Council may be reimbursed for actual expenses for meals, lodging, transportation, and incidental expenses in accordance with travel rates set by the federal General Services Administration.

II. Duties/Roles

It is the duty of the Council to determine and approve the opioid abatement strategies and funding awards

A. Approved Abatement Strategies

The Council will develop the approved Texas list of abatement strategies based on but not limited to the existing national list of opioid abatement strategies (see attached Appendix A) for implementing the Texas Abatement Fund.

- 1. The Council shall only approve strategies which are evidence-informed strategies.
- 2. The Texas list of abatement strategies must be approved by majority vote. The majority vote must include a majority from both sides of the statewide members and regional members in order to be approved, e.g., at least four (4) of six (6) members on each side.

B. Texas Abatement Fund Point of Contact

The Council will determine a single point of contact called the Abatement Fund Point of Contact (POC) to be established as the sole entity authorized to receive requests for funds and approve expenditures in Texas and order the release of funds from the Texas Abatement Fund by the independent administrator. The POC may be an independent third party selected by the Council with expertise in banking or financial management. The POC will manage the Opioid Council Bank Account (Account). Upon a vote, the Council will direct the POC to contact the independent administrator to release funds to the Account. The Account is outside the State Treasury and not managed by any state or local officials. The POC is responsible for payments to the qualified entities selected by the Council for abatement fund awards. The POC will submit a monthly financial statement on the Account to the Council.

C. Auditor

An independent auditor appointed by the Council will perform an audit on the Account on an annual basis and report its findings, if any, to the Council.

D. Funding Allocation

The Council is the sole decision-maker on the funding allocation process of the abatement funds. The Council will develop the application and award process based on the parameters outlined below. An entity seeking funds from the Council must apply for funds; no funds will be awarded without an application. The executive director and personnel may assist the Council in gathering and compiling the applications for consideration; however, the Council members are the sole decision-makers of awards and funding determination. The Council will use the following processes to award funds:

- 1. Statewide Funds. The Council will consider, adopt and approve the allocation methodology attached as Exhibit C, based upon population health data and prevalence of opioid incidences, at the Council's initial meeting. Adoption of such methodology will allow each Region to customize the approved abatement strategies to fit its communities' needs. The statewide regional funds will account for seventy-five (75) percent of the total overall funds, less the one (1) percent administrative expense described herein.
- 2. *Targeted Funds*. Each Region shall reserve twenty-five (25) percent of the overall funds, for targeted interventions in the specific Region as identified by opioid incidence data. The Council must approve on an annual basis the uses for the targeted abatement strategies and applications available to every Region, including education and outreach programs. Each Region without approved uses for the targeted funds from the Council, based upon a greater percentage of opioid incidents compared to its population, is subject to transfer of all or a portion of the targeted funds for that Region for uses based upon all Regions' targeted funding needs as approved by the Council on an annual basis.
- 3. *Annual Allocation*. Statewide regional funds and targeted funds will be allocated on an annual basis. If a Region lapses its funds, the funds will be reallocated based on all Regions' funding needs.

E. Appeal Process

The Council will establish an appeal process to permit the applicants for funding (state or subdivisions) to challenge decisions by the Council-designated point of contact on requests for funds or expenditures.

- 1. To challenge a decision by the designated point of contact, the State or a subdivision must file an appeal with the Council within thirty (30) days of the decision. The Council then has thirty (30) days to consider and rule on the appeal.
- 2. If the Council denies the appeal, the party may file an appeal with the state district court of record where the final opioid judgment or Master Settlement Agreement is filed. The Texas Rules of Civil Procedure and Rules of Evidence will govern these proceedings. The Council may request representation from the Attorney General in these proceedings.

In making its determination, the state district court shall apply the same clear error standards contained herein that the Council must follow when rendering its decision.

- 3. The state district court will make the final decision and the decision is not appealable.
- 4. Challenges will be limited and subject to penalty if abused.
- 5. Attorneys' fees and costs are not recoverable in these appeals.

F. Education

The Council may determine that a percentage of the funds in the Abatement Fund from the targeted funds be used to develop an education and outreach program to provide materials on the consequences of opioid drug use, prevention and interventions. Any material developed will include online resources and toolkits for communities.

APPENDIX A

OPIOID ABATEMENT STRATEGIES

PART ONE: TREATMENT

A. TREAT OPIOID USE DISORDER (OUD)

- Expand availability of treatment for Opioid Use Disorder (OUD) and any cooccurring Substance Use Disorder or Mental Health (SUD/MH) issues, including all forms of Medication-Assisted Treatment (MAT) approved by the U.S. Food and Drug Administration.
- Support and reimburse services that include the full American Society of Addiction Medicine (ASAM) continuum of care for OUD and any co-occurring SUD/MH issues, including but not limited to:
 - a. Medication-Assisted Treatment (MAT);
 - b. Recruiting MAT Providers and Training;
 - c. Abstinence-based treatment;
 - d. Treatment, recovery, or other services provided by states, subdivisions, community health centers; non-for-profit providers; or for-profit providers; or
 - e. Treatment by providers that focus on OUD treatment as well as treatment by providers that offer OUD treatment along with treatment for other SUD/MH issues;
 - f. Recovery high schools
- 3. Expand telehealth to increase access to treatment for OUD and any co-occurring SUD/MH issues, including MAT, as well as counseling, psychiatric support, and other treatment and recovery support services.
- 4. "Support the establishment of the hub-and-spoke model of OUD treatment in all counties where possible, and across county lines where necessary."
- 5. Improve oversight of Opioid Treatment Programs (OTPs) to assure evidence-informed, promising, or emerging practices such as adequate methadone dosing.
- 6. Support mobile intervention, treatment, and recovery services, offered by qualified professionals, for persons with OUD and any co-occurring SUD/MH issues or persons who have experienced an opioid overdose.

- 7. Treatment of mental health trauma issues resulting from the traumatic experiences of the opioid user (e.g., violence, sexual assault, human trafficking, or adverse childhood experiences) and family members (e.g., surviving family members after an overdose or overdose fatality), and training of health care personnel to identify and address such mental health trauma.
- 8. Support detoxification (detox) services for persons with OUD and any cooccurring SUD/MH issues, including medical detox, referral to treatment, or connections to other services or supports.
- 9. Training on MAT for health care providers, students, or other supporting professionals, such as peer recovery coaches or recovery outreach specialists.
- 10. Support workforce development for addiction professionals who work with persons with OUD and any co-occurring SUD/MH issues.
- 11. Fellowships for addiction medicine specialists for direct patient care, instructors, and clinical research for treatments.
- 12. Scholarships and supports for certified addiction counselors and other mental and behavioral health providers involved in addressing OUD any co-occurring SUD/MH issues, including but not limited to training, scholarships, fellowships, loan repayment programs, or other incentives for providers to work in rural or underserved areas.
- 13. Provide training for clinicians to obtain a waiver under the federal Drug Addiction Treatment Act of 2000 (DATA 2000) to prescribe MAT for OUD, and provide technical assistance and professional support to clinicians who have obtained a DATA 2000 waiver.
- 14. Dissemination of web-based training curricula, such as the American Academy of Addiction Psychiatry's Provider Clinical Support Service-Opioids web-based training curriculum and motivational interviewing.
- 15. Development and dissemination of new curricula, such as the American Academy of Addiction Psychiatry's Provider Clinical Support Service for Medication-Assisted Treatment.
- 16. Support State or local learning collaboratives so that physicians involved in the care and treatment of those with OUD are kept abreast of the latest developments in evidence-based treatment.
- 17. Support State or local drop-in centers where those with OUD may go to seek assistance with recovery when they are ready to begin the process.

18. Support creation of teams in hospitals and emergency rooms to work with those with OUD and direct them to appropriate facilities for evidence-based treatment of OUD, including MAT.

B. SUPPORT PEOPLE IN TREATMENT AND RECOVERY

- 1. Provide the full continuum of care of recovery services for OUD and any cooccurring SUD/MH issues, including supportive housing, residential treatment, medical detox services, peer support services and counseling, community navigators, case management, and connections to community-based services.
- Provide counseling, peer-support, recovery case management and residential treatment with access to medications for those who need it to persons with OUD and any co-occurring SUD/MH issues.
- 3. Provide access to housing for people with OUD and any co-occurring SUD/MH issues, including supportive housing, housing assistance programs, or training for housing providers.
- 4. Provide community support services to assist in deinstitutionalizing persons with OUD and any co-occurring SUD/MH issues
- 5. Support or expand peer-recovery centers, which may include support groups, social events, computer access, or other services for persons with OUD and any co-occurring SUD/MH issues.
- 6. Provide or support transportation to treatment or recovery programs or services for persons with OUD and any co-occurring SUD/MH issues.
- 7. Provide employment training or educational services for persons in treatment for or recovery from OUD and any co-occurring SUD/MH issues.
- 8. Identify successful recovery programs such as physician, pilot, and college recovery programs, and provide support and technical assistance to increase the number and capacity of high-quality programs to help those in recovery.
- 9. Engage non-profits, the faith community, and community coalitions to support people in treatment and recovery and to support family members in their efforts to manage the opioid user in the family.
- 10. Training and development of procedures for government staff to appropriately interact and provide social and other services to current and recovering opioid users, including reducing stigma.
- 11. Support stigma reduction efforts regarding treatment and support for persons with OUD, including reducing the stigma on effective treatment.

C. CONNECT PEOPLE WHO NEED HELP TO THE HELP THEY NEED (CONNECTIONS TO CARE)

- 1. Ensure that health care providers are screening for OUD and other risk factors and know how to appropriately counsel and treat (or refer if necessary) a patient for OUD treatment.
- 2. Fund Screening, Brief Intervention and Referral to Treatment (SBIRT) programs and appropriate training for all health care providers to identify those with potential problems in order to reduce the transition from use to disorders.
- 3. Provide training and long-term implementation of SBIRT in key systems (health, schools, colleges, criminal justice, and probation), with a focus on youth and young adults when transition from misuse to opioid disorder is common.
- 4. Purchase automated versions of SBIRT and support ongoing costs of the technology.
- 5. Training for emergency room personnel treating opioid overdose patients on post-discharge planning, including community referrals for MAT, recovery case management or support services.
- 6. Support hospital programs that transition persons with OUD and any cooccurring SUD/MH issues, or persons who have experienced an opioid overdose, into community treatment or recovery services through a bridge clinic or similar approach.
- 7. Support crisis stabilization centers that serve as an alternative to hospital emergency departments for persons with OUD and any co-occurring SUD/MH issues or persons that have experienced an opioid overdose.
- 8. Support the work of Emergency Medical Systems, including peer support specialists, to connect individuals to treatment or other appropriate services following an opioid overdose or other opioid-related adverse event.
- 9. Provide funding for peer support specialists or recovery coaches in emergency departments, detox facilities, recovery centers, recovery housing, or similar settings; offer services, supports, or connections to care to persons with OUD and any co-occurring SUD/MH issues or to persons who have experienced an opioid overdose.
- 10. Provide funding for peer navigators, recovery coaches, care coordinators, or care managers that offer assistance to persons with OUD and any co-occurring SUD/MH issues or to persons who have experienced on opioid overdose.

- 11. Create or support school-based contacts that parents can engage with to seek immediate treatment services for their child; and support prevention, intervention, treatment, and recovery programs focused on young people.
- 12. Develop and support best practices on addressing OUD in the workplace.
- 13. Support assistance programs for health care providers with OUD.
- 14. Engage non-profits and the faith community as a system to support outreach for treatment.
- 15. Support centralized call centers that provide information and connections to appropriate services and supports for persons with OUD and any co-occurring SUD/MH issues.
- 16. Develop or support a National Treatment Availability Clearinghouse a multistate/nationally accessible database whereby health care providers can list locations for currently available in-patient and out-patient OUD treatment services that are accessible on a real-time basis by persons who seek treatment.

D. <u>ADDRESS THE NEEDS OF CRIMINAL-JUSTICE-INVOLVED PERSONS AND RURAL COUNTY UNATTENDED DEATHS</u>

- Address the needs of persons with OUD and any co-occurring SUD/MH issues who are involved or are at risk of becoming involved in the criminal justice system.
- Support pre-arrest or pre-arraignment diversion and deflection strategies for persons with OUD and any co-occurring SUD/MH issues, including established strategies such as:
 - a. Self-referral strategies such as the Angel Programs or the Police Assisted Addiction Recovery Initiative (PAARI);
 - b. Active outreach strategies such as the Drug Abuse Response Team (DART) model;
 - c. "Naloxone Plus" strategies, which work to ensure that individuals who have received naloxone to reverse the effects of an overdose are then linked to treatment programs or other appropriate services;
 - d. Officer prevention strategies, such as the Law Enforcement Assisted Diversion (LEAD) model; or
 - e. Officer intervention strategies.
- 3. Support pre-trial services that connect individuals with OUD and any cooccurring SUD/MH issues to evidence-informed treatment, including MAT, and related services.

- 4. Implementing or supporting pilot programs for the voluntary testing of individuals who enter local (city or county) criminal justice facilities, and for those identified with OUD, offer induction of evidence-based treatment, including MAT.
- 5. Support treatment and recovery courts for persons with OUD and any cooccurring SUD/MH issues, but only if they provide referrals to evidence-informed treatment, including MAT.
- 6. Provide evidence-informed treatment, including MAT, recovery support, harm reduction, or other appropriate services to individuals with OUD and any co-occurring SUD/MH issues who are incarcerated in jail or prison.
- 7. Provide evidence-informed treatment, including MAT, recovery support, harm reduction, or other appropriate services to individuals with OUD and any co-occurring SUD/MH issues who are leaving jail or prison have recently left jail or prison, are on probation or parole, are under community corrections supervision, or are in re-entry programs or facilities.
- Support critical time interventions (CTI), particularly for individuals living with dual-diagnosis OUD/serious mental illness, and services for individuals who face immediate risks and service needs and risks upon release from correctional settings.
- 9. Provide training on best practices for addressing the needs of criminal-justice-involved persons with OUD and any co-occurring SUD/MH issues to law enforcement, correctional, or judicial personnel or to providers of treatment, recovery, harm reduction, case management, or other services offered in connection with any of the strategies described in this section;
- 10. Provide training to Justices of the Peace on unattended deaths involving drug use and reimbursement of transfer to and costs or expenses of a Medical Examiner to enhance better death understanding, statistics and recording on overdose involved deaths.

E. ADDRESS THE NEEDS OF PREGNANT OR PARENTING WOMEN AND THEIR FAMILIES, INCLUDING BABIES WITH NEONATAL ABSTINENCE SYNDROME

- Support evidence-informed, promising, or emerging treatment, including MAT, recovery services and supports, and prevention services for pregnant women – or women who could become pregnant – who have OUD and any co-occurring SUD/MH issues.
- 2. Fund Screening, Brief Intervention and Referral to Treatment (SBIRT) programs and training for all health care providers to identify women with potential opioid

use disorder so that they might be given the option of referral to a proper treatment program.

- Training for obstetricians or other healthcare personnel that work with pregnant women and their families regarding treatment of OUD and any co-occurring SUD/MH issues.
- 4. Other measures to address Neonatal Abstinence Syndrome, including prevention, education, and treatment of OUD and any co-occurring SUD/MH issues.
- 5. Provide training to health care providers that work with pregnant or parenting women on best practices for compliance with federal requirements that children born with Neonatal Abstinence Syndrome get referred to appropriate services and receive a plan of safe care.
- 6. Child and family support for parenting women with OUD and any co-occurring SUD/MH issues.
- 7. Enhanced family supports and childcare services for parents with OUD and any co-occurring SUD/MH issues.
- 8. Provide enhanced support for children and family members suffering trauma as a result of addiction in the family; and offer trauma-informed behavioral health treatment for adverse childhood events.
- Offer home-based wrap-around services to persons with OUD and any cooccurring SUD/MH issues, including but not limited to parent skills training.
- 10. Support for Children's Services Fund additional positions and services, including supportive housing and other residential services, relating to children being removed from the home and/or placed in foster care due to custodial opioid use.
- 11. Provision of education and psychosocial support services to children born with Neonatal Abstinance Syndrome.
- 12. Support family and baby reunification in recovery housing.

PART TWO: PREVENTION

F. PREVENT OVER-PRESCRIBING AND ENSURE APPROPRIATE PRESCRIBING AND DISPENSING OF OPIOIDS

- 1. Trainingand continuing education of health care providers regarding safe and responsible opioid prescribing, dosing, and tapering patients off opioids.
- 2. Academic counter-detailing to educate prescribers on appropriate opioid prescribing.
- 3. Continuing Medical Education (CME) on appropriate prescribing of opioids.
- 4. Support for non-opioid pain treatment alternatives, including training providers to offer or refer to multi-modal, evidence-informed treatment of pain.
- 5. Support enhancements or improvements to Prescription Drug Monitoring Programs (PDMPs), including but not limited to improvements that:
 - a. Increase the number of prescribers using PDMPs;
 - Improve point-of-care decision-making by increasing the quantity, quality, or format of data available to prescribers using PDMPs, by improving the interface that prescribers use to access PDMP data, or both; or
 - c. Enable states to use PDMP data in support of surveillance or intervention strategies.
- 6. Development and implementation of a national PDMP Fund development of a multistate/national PDMP that permits information sharing while providing appropriate safeguards on sharing of private health information, including but not limited to:
 - Integration of PDMP data with electronic health records, overdose episodes, and decision support tools for health care providers relating to OUD.
 - b. Ensuring PDMPs incorporate available overdose/naloxone deployment data, including the United States Department of Transportation's Emergency Medical Technician overdose database.
- 7. Increase electronic prescribing to prevent diversion or forgery
- 8. Educate Dispensers on appropriate opioid dispensing.
- 9. Develop and train physicians on algorithm for proper evidence-based pain management.

- 10. Fund State or local hotline so health care providers with questions regarding proper pain management or opioid prescribing can call and have an expert answer their questions.
- 11. Support for health information systems consistent with State regulations.

G. PREVENT MISUSE OF OPIOIDS

- 1. Corrective advertising or affirmative public education campaigns.
- 2. Public education relating to drug disposal.
- 3. Drug take-back disposal or destruction programs.
- 4. Fund community anti-drug coalitions that engage in drug prevention efforts.
- 5. Support community coalitions in implementing evidence-informed prevention, such as reduced social access and physical access, stigma reduction including staffing, educational campaigns, support for people in treatment or recovery, or training of coalitions in evidence-informed implementation, including the Strategic Prevention Framework developed by the U.S. Substance Abuse and Mental Health Services Administration (SAMHSA).
- 6. Engage non-profits and faith community as a system to support prevention.
- 7. School and community education programs and campaigns for students, families, school employees, school athletic programs, parent-teacher and student associations, and others.
- 8. School-based or youth-focused programs or strategies that have demonstrated effectiveness in preventing drug misuse and seem likely to be effective in preventing the uptake and use of opioids.
- 9. Support evidence-informed programs or curricula to address mental health needs of young people who may be at risk of misusing opioids or other drugs, including emotional modulation and resilience skills.
- 10. Support greater access to mental health services and supports for young people, including services and supports provided by school nurses or other school staff, to address mental health needs in young people that (when not properly addressed) increase the risk of opioid or other drug misuse.
- 11. Support local law enforcement task forces aimed at disrupting and eliminating the manufacturers and distributers of illegal opioids.

H. PREVENT OVERDOSE DEATHS AND OTHER HARMS (HARM REDUCTION)

- Increase availability and distribution of naloxone and other drugs that treat overdoses for first responders, overdose patients, opioid users, families and friends of opioid users, schools, community navigators and outreach workers, drug offenders upon release from jail/prison, or other members of the general public.
- 2. Public health entities provide free naloxone and training to anyone in the community.
- 3. Training and education regarding naloxone and other drugs that treat overdoses for first responders, overdose patients, patients taking opioids, families, schools, and other members of the general public.
- 4. Enable school nurses and other school staff to respond to opioid overdoses, and provide them with naloxone, training, and support.
- 5. Expand, improve, or develop data tracking software and applications for overdoses/naloxone revivals.
- 6. Public education relating to emergency responses to overdoses.
- 7. Public education relating to immunity and Good Samaritan laws.
- 8. Educate first responders regarding the existence and operation of immunity and Good Samaritan laws.
- 9. Syringe service programs and other evidence-informed programs to reduce harms associated with intravenous drug use, including supplies, staffing, space, peer support services, referrals to treatment, connections to care, and the full range of harm reduction and treatment services provided by these programs.
- 10. Expand access to testing and treatment for infectious diseases such as HIV and Hepatitis C resulting from intravenous opioid use.
- 11. Support mobile units that offer or provide referrals to harm reduction services, treatment, recovery supports, health care, or other appropriate services to persons that use opioids or persons with OUD and any co-occurring SUD/MH issues.
- 12. Provide training in harm reduction strategies to health care providers, students, peer recovery coaches, recovery outreach specialists, or other professionals that provide care to persons who use opioids or persons with OUD and any co-occurring SUD/MH issues.

PART THREE: OTHER STRATEGIES

I. FIRST RESPONDERS

- 1. Law enforcement expenditures relating to the opioid epidemic.
- 2. Educate first responders regarding appropriate practices and precautions when dealing with fentanyl or other drugs.

J. <u>LEADERSHIP, PLANNING AND COORDINATION</u>

- 1. Community regional planning to identify goals for reducing harms related to the opioid epidemic, to identify areas and populations with the greatest needs for treatment intervention services, or to support other strategies to abate the opioid epidemic described in this opioid abatement strategy list.
- 2. A government dashboard to track key opioid-related indicators and supports as identified through collaborative community processes.
- 3. Invest in infrastructure or staffing at government and not-for-profit agencies to support collaborative, cross-system coordination with the purpose of preventing overprescribing, opioid misuse, or opioid overdoses, treating those with OUD and any co-occurring SUD/MH issues, supporting them in treatment or recovery, connecting them to care, or implementing other strategies to abate the opioid epidemic described in this opioid abatement strategy list.

K. TRAINING

- 1. Provide funding for staff training or networking programs and services to improve the capability of government, community, and not-for-profit entities to abate the opioid crisis.
- Support infrastructure and staffing for collaborative cross-system coordination
 to prevent opioid misuse, prevent overdoses, and treat those with OUD and any
 co-occurring SUD/MH issues, or implement other strategies to abate the opioid
 epidemic described in this opioid abatement strategy list (e.g., health care,
 primary care, pharmacies, PDMPs, etc.);
- 3. Medical Provider education;
- 4. Media Campaigns

L. RESEARCH

- 1. Support opioid abatement research, including but not limited to:
 - a. Monitoring, surveillance, and evaluation of programs and strategies described in this opioid abatement strategy list.
 - b. Research non-opioid treatment of chronic pain.
 - c. Research on improved service delivery for modalities such as SBIRT that demonstrate promising but mixed results in populations vulnerable to opioid use disorders.
 - d. Research on novel harm reduction and prevention efforts such as the provision of fentanyl test strips.
 - e. Research on innovative supply-side enforcement efforts such as improved detection of mail-based delivery of synthetic opioids.
 - f. Expanded research on swift/certain/fair models to reduce and deter opioid misuse within criminal justice populations that build upon promising approaches used to address other substances (e.g. Hawaii HOPE and Dakota 24/7).
 - g. Research on expanded modalities such as prescription methadone that can expand access to MAT;
 - h. Research on the effectiveness of Recovery High Schools and other educational interventions;
 - i. Research to track abatement progress in urban and rural areas.

M. MISCELLANEOUS

- 1. It is the intent of the Parties to the Texas Term Sheet in adopting the Abatement Strategies herein that the Council be guided by the allocation methodology in Exhibit C to the Texas Term Sheet in approving Regional strategies and that the Council consider the proportional share of the individual members in each Region when allocating the funds for approved abatement strategies within each Region.
- 2. It is the intent of the Parties to the Texas Term Sheet in adopting the Abatement Strategies herein that the Opioid Council have the flexibility to add, change or alter the Abatement Strategies herein as necessary to fulfill the intent that opioid abatement strategies best meet the needs of the Regions, subdivisions and intent of this document.

EXHIBIT B

[State to Add Allocations]

Exhibit D

Injunctive Relief

I. <u>Definitions Specific to this Exhibit</u>

- A. "Cancer-Related Pain Care" means care that provides relief from pain resulting from a patient's active cancer or cancer treatment as distinguished from treatment provided during remission.
- B. "End-of-Life Care" means care for persons with a terminal illness or at high risk for dying in the near future in hospice care, hospitals, long-term care settings, or at home.
- C. "Downstream Customer Data" shall mean transaction information that Teva collects relating to its direct customers' sales to downstream customers, including chargeback data tied to Teva providing certain discounts, "867 data" and IQVIA data.
- D. "Health Care Provider" shall mean any U.S.-based physician or other health care practitioner who is licensed to provide health care services and/or prescribe pharmaceutical products and any medical facility, practice, hospital, clinic or pharmacy.
- E. "Including but not limited to", when followed by a list or examples, shall mean that list or examples are illustrative instances only and shall not be read to be restrictive.
- F. "In-Kind Support" shall mean payment or assistance in the form of goods, commodities, services, or anything else of value.
- G. "Lobby" and "Lobbying" shall have the same meaning as "lobbying activities" and "lobbying contacts" under the federal lobbying disclosure act, 2 U.S.C. § 1602 *et seq.*, and any analogous state or local provisions governing the person or entity being lobbied in that particular state or locality. As used in this document, "Lobby" and "Lobbying" include Lobbying directly or indirectly, through grantees or Third Parties.
- H. "Opioid(s)" shall mean all natural, semi-synthetic, or synthetic chemicals that interact with opioid receptors and act like opium. For the avoidance of doubt, the term Opioid shall not include the opioid antagonists naloxone or naltrexone.
- I. "Opioid Product(s)" shall mean all current and future medications containing Opioids approved by the U.S. Food & Drug Administration ("FDA") and listed by the Drug Enforcement Administration ("DEA") as Schedule II, III, or IV pursuant to the Federal Controlled Substances Act (including but not limited to

buprenorphine, codeine, fentanyl, hydrocodone, hydromorphone, meperidine, methadone, morphine, oxycodone, oxymorphone, tapentadol, and tramadol). The term "Opioid Products(s)" shall not include (i) methadone, buprenorphine, or other substances when used exclusively to treat opioid abuse, addiction, or overdose; or (ii) raw materials, immediate precursors, and/or active pharmaceutical ingredients ("APIs") used in the manufacture or study of Opioids or Opioid Products, but only when such materials, immediate precursors, and/or APIs are sold or marketed exclusively to DEA-licensed manufacturers or DEA-licensed researchers.

- J. "OUD" shall mean opioid use disorder as defined in the *Diagnostic and Statistical Manual of Mental Disorders*, *Fifth Edition (DSM–5)*, as updated or amended.
- K. "Promote," "Promoting," "Promotion," and "Promotional" shall mean dissemination of information or other practices intended or reasonably anticipated to increase sales or prescriptions, or that attempts to influence prescribing practices of Health Care Providers in the United States.
- L. "Qualified Researcher" shall mean any researcher holding a faculty appointment or research position at an institution of higher education, a research organization, a nonprofit organization, or a government agency.
- M. "Suspicious Order" shall have the same meaning as provided by the Controlled Substances Act, 21 U.S.C. §§ 801-904, and the regulations promulgated thereunder and analogous Texas state laws and regulations.
- N. "Teva" means Teva Pharmaceuticals USA, Inc. ("Teva USA"); Cupric Holding Co., Inc.; Cephalon, Inc.; Watson Laboratories, Inc.; Actavis LLC; Actavis Pharma, Inc. f/k/a Watson Pharma, Inc.; Warner Chilcott Co., LLC; Actavis South Atlantic LLC; Actavis Elizabeth LLC; Actavis Mid Atlantic LLC; Actavis Totowa LLC; Actavis Kadian LLC; Actavis Laboratories UT, Inc. f/k/a Watson Laboratories Inc.-Salt Lake City; and Actavis Laboratories FL, Inc. f/k/a Watson Laboratories, Inc.-Florida.
- O. "Third Party" shall mean any person or entity other than Teva or a government entity.
- P. "Treatment of Pain" shall mean the provision of therapeutic modalities to alleviate or reduce pain.
- Q. "Unbranded Information" shall mean any information that does not identify a specific branded or generic product(s).

II. <u>Injunctive Relief</u>

A. General Provisions

- 1. Teva shall not make any written or oral statement about Opioids or any Opioid Product that is false, misleading, and/or deceptive as defined under the law of Texas.
- 2. Teva shall not represent that Opioids or any Opioid Products have approvals, characteristics, uses, benefits, or qualities that they do not have.

B. Ban on Promotion

- 1. Teva shall not engage in the Promotion of Opioids or Opioid Products including, but not limited to, by:
 - a. Employing or contracting with sales representatives or other persons to Promote Opioids or Opioid Products to Health Care Providers, patients, or persons involved in determining the Opioid Products included in formularies;
 - b. Using speakers, key opinion leaders, thought leaders, lecturers, and/or speaking events for Promotion of Opioids or Opioid Products;
 - c. Sponsoring, or otherwise providing financial support or In-Kind Support to, medical education programs relating to Opioids or Opioid Products;
 - d. Creating, sponsoring, operating, controlling, or otherwise providing financial support or In-Kind Support to any website, network, and/or social or other media account for the Promotion of Opioids or Opioid Products;
 - e. Creating, sponsoring, distributing, or otherwise providing financial support or In-Kind Support for materials Promoting Opioids or Opioid Products, including but not limited to brochures, newsletters, pamphlets, journals, books, and guides;
 - f. Creating, sponsoring, or otherwise providing financial support or In-Kind Support for advertisements that Promote Opioids or Opioid Products, including but not limited to internet advertisements or similar content, and providing hyperlinks or otherwise directing internet traffic to advertisements; or

- g. Engaging in Internet search engine optimization or other techniques designed to Promote Opioids or Opioid Products by improving rankings or making content appear among the top results in an Internet search or otherwise be more visible or more accessible to the public on the Internet.
- 2. Notwithstanding subsection II.B.1 directly above, Teva may:
 - a. Maintain a corporate website;
 - b. Maintain a website that contains principally the following content for any Opioid Product: the FDA-approved package insert, medication guide, and labeling, and a statement directing patients or caregivers to speak with a licensed Health Care Provider;
 - c. Provide information or support the provision of information as expressly required by law or any state or federal government agency with jurisdiction in the state where the information is provided. Teva may, in relation to its expressly required participation in the Transmucosal Immediate Release Fentanyl ("TIRF") Risk Evaluation and Mitigation Strategy ("REMS") Program, remain involved in the preparation of materials and training concerning the process for enrollment in the TIRF REMS Program;
 - d. Provide the following by mail, electronic mail, on or through Teva's corporate or product websites or through other electronic or digital methods: FDA-approved package insert, medication guide, approved labeling for Opioid Products or other prescribing information for Opioid Products that are published by a state or federal government agency with jurisdiction in the state where the information is provided;
 - e. Provide scientific and/or medical information in response to an unsolicited request by a Health Care Provider consistent with the standards set forth in the FDA's Draft Guidance for Industry, Responding to Unsolicited Requests for Off-Label Information About Prescription Drugs and Medical Devices (Dec. 2011), as updated or amended by the FDA, and Guidance for Industry, Good Reprint Practices for the Distribution of Medical Journal Articles and Medical or Scientific Reference Publications on Unapproved New Uses of Approved Drugs and Approved or Cleared Medical Devices (Jan. 2009), as updated or amended by the FDA;

- f. Provide a response to any unsolicited question or request from a patient or caregiver, directing the patient or caregiver to the FDA-approved labeling or to speak with a licensed Health Care Provider without describing the safety or effectiveness of Opioids or any Opioid Product or naming any specific provider or healthcare institution; or directing the patient or caregiver to speak with their insurance carrier regarding coverage of an Opioid Product;
- g. Provide Health Care Economic Information, as defined at 21 U.S.C. § 352(a), to a payor, formulary committee, or other similar entity with knowledge and expertise in the area of health care economic analysis consistent with standards set forth in the FDA's Draft Questions and Answers Guidance for Industry and Review Staff, *Drug and Device Manufacturer Communications With Payors, Formulary Committees, and Similar Entities* (Jan. 2018), as updated or amended by the FDA;
- h. Provide information relating solely to the pricing of any Opioid Product;
- i. Provide information, through a product catalog or similar means, related to an Opioid or Opioid Product, including, without limitation, pricing information, weight, color, shape, packaging size, type, reference listed drug, National Drug Code ("NDC") label, and such other descriptive information (including information set forth in a standard Healthcare Distribution Alliance Form or technical data sheet and the FDA approval letter) sufficient to identify the products available, to place an order for a product, and to allow the product to be loaded into a customer's inventory and ordering system or Third Party pricing compendia;
- j. Sponsor or provide financial support or In-Kind Support for an accredited or approved continuing medical education program required by either an FDA-approved REMS program or other federal or state law or regulation applicable in the state where the program is provided through an independent Third Party, which shall be responsible for the continuing medical education program's content without the participation of Teva;
- k. Provide information in connection with patient support information on co-pay assistance and managing pain in End-of-Life Care and/or Cancer-Related Pain Care relating to the use of Opioids for managing such pain, as long as the information identifies Teva as the source of the information; and

- 1. Provide rebates, discounts, and other customary pricing adjustments to DEA-registered customers and contracting intermediaries, such as Buying Groups, Group Purchasing Organizations, and Pharmacy Benefit Managers, except as prohibited by Section II.G.
- 3. Teva shall not engage in the following specific Promotional activity relating to any products indicated for the treatment of Opioid-induced side effects (for the avoidance of doubt, "Opioid-induced side effects" does not include addiction to Opioids or Opioid Products):
 - a. Employing or contracting with sales representatives or other persons to Promote products indicated for the treatment of Opioid-induced side effects to Health Care Providers or patients;
 - b. Using speakers, key opinion leaders, thought leaders, lecturers, and/or speaking events to Promote products indicated for the treatment of Opioid-induced side effects;
 - c. Sponsoring, or otherwise providing financial support or In-Kind Support to medical education programs that Promote products indicated for the treatment of Opioid-induced side effects; or
 - d. Creating, sponsoring, or otherwise providing financial support or In-Kind Support for advertisements that Promote products indicated for the treatment of Opioid-induced side effects, including but not limited to internet advertisements or similar content, and providing hyperlinks or otherwise directing internet traffic to advertisements.
- 4. Notwithstanding subsection II.B.3 directly above, Teva may Promote products for the treatment of Opioid-induced side effects (i) so long as such Promotion does not associate the product with Opioids or Opioid Products, or (ii) where such Promotion concerns a product's indication to reverse overdoses and/or treat Opioid addiction. Nothing herein shall prevent Teva from linking to the FDA label associated with a product.

5. Treatment of Pain

- a. Teva shall not, either through Teva or through Third Parties, engage in Promotion of the Treatment of Pain in a manner that encourages the utilization of Opioids or Opioid Products.
- b. Teva shall not, either through Teva or through Third Parties, Promote the concept that pain is undertreated in a manner that encourages the utilization of Opioids or Opioid Products.

- c. Teva shall not disseminate Unbranded Information, including Unbranded Information about a medical condition or disease state, that contains links to branded information about Opioid Products or otherwise Promotes Opioids or Opioid Products.
- 6. Notwithstanding subsection II.B.5 directly above, Teva may Promote or provide educational information about the Treatment of Pain with non-Opioid products or therapies, including Promoting or providing educational information about such non-Opioid products or therapies as alternatives to Opioid use, or as part of multimodal therapy which may include Opioid use, so long as such non-Opioid Promotional or educational information does not Promote Opioids or Opioid Products.

C. No Financial Reward or Discipline Based on Volume of Opioid Sales

- 1. Teva shall not provide financial incentives to its sales and marketing employees or discipline its sales and marketing employees based upon sales volume or sales quotas for Opioid Products. For the avoidance of doubt, this provision shall not prohibit financial incentives (*e.g.*, customary raises or bonuses) based on the performance of the overall company or business segment, as measured by EBITDA, revenue, cash flow, or other similar financial metrics.
- 2. Teva shall not offer or pay any remuneration (including any kickback, bribe, or rebate) directly or indirectly, to or from any person in return for the prescribing, sale, or use of an Opioid Product. For the avoidance of doubt, this provision shall not prohibit rebates or chargebacks to the extent permitted by other sections of this Consent Judgment.
- 3. Teva's compensation policies and procedures shall be designed to ensure compliance with this Consent Judgment and other legal requirements.

D. Ban on Funding/Grants to Third Parties

- 1. Teva shall not, directly or indirectly, provide financial support or In-Kind Support to any Third Party for Promotion of or education about Opioids, Opioid Products, or products indicated for the treatment of Opioid-induced side effects (subject to subsections II.B.2, 4 and 6). For the avoidance of doubt, this provision does not prohibit support expressly allowed by this Consent Judgment or required by a federal or state agency.
- 2. Teva shall not create, sponsor, provide financial support or In-Kind Support to, or otherwise operate or control any medical society or patient advocacy group that primarily engages in conduct that Promotes Opioids or Opioid Products.

- 3. Teva shall not provide links to any Third Party website or materials or otherwise distribute materials created by a Third Party for the purpose of Promoting Opioids, Opioid Products, or products indicated for the treatment of Opioid-induced side effects (subject to subsections II.B.2, 4 and 6).
- 4. Teva shall not use, assist, or employ any Third Party to engage in any activity that Teva itself would be prohibited from engaging in pursuant to this Consent Judgment.
- 5. Teva shall not enter into any contract or agreement with any person or entity or otherwise attempt to influence any person or entity in such a manner that has the purpose or reasonably foreseeable effect of limiting the dissemination of information regarding the risks and side effects of using Opioids.
- 6. Teva shall not compensate or provide In-Kind Support to Health Care Providers (other than Teva employees) or organizations to advocate for formulary access or treatment guideline changes for the purpose of increasing access to any Opioid Product through third-party payers, *i.e.*, any entity, other than an individual, that pays or reimburses for the dispensing of prescription medicines, including but not limited to managed care organizations and pharmacy benefit managers. Nothing in this provision, however, prohibits Teva from using independent contractors who operate under the direction of Teva to provide information to a payor, formulary committee, or other similar entity as permitted in subsection II.B.2 provided that any such persons are bound by the terms of this Consent Judgment. Nor does this provision prohibit the payment of customary rebates or other pricing concessions to third-party payers, including state Medicaid programs, as part of an overall pricing agreement.
- 7. No officer or executive management-level employee of Teva may concurrently serve as a director, board member, employee, agent, or officer of any entity other than Teva Pharmaceutical Industries Ltd. or a direct or indirect wholly-owned subsidiary thereof, that primarily engages in conduct that Promotes Opioids, Opioid Products, or products indicated for the treatment of Opioid-related side effects. For the avoidance of doubt, nothing in this provision shall preclude an officer or executive management-level employee of Teva from concurrently serving on the board of a hospital.
- 8. Teva shall play no role in appointing persons to the board, or hiring persons to the staff, of any entity that primarily engages in conduct that Promotes Opioids, Opioid Products, or products indicated for the treatment of Opioid-induced side effects. For the avoidance of doubt, nothing in this paragraph shall prohibit Teva from fully and accurately responding to unsolicited

requests or inquiries about a person's fitness to serve as an employee or board member at any such entity.

9. For the avoidance of doubt:

- a. Nothing in this Section II.D shall be construed or used to prohibit Teva from providing financial or In-Kind Support to:
 - (i) medical societies and patient advocate groups, who are principally involved in issues relating to (I) the treatment of OUD; (II) the prevention, education and treatment of opioid abuse, addiction, or overdose, including medication-assisted treatment for opioid addiction; and/or (III) rescue medications for opioid overdose; or
 - (ii) universities, medical institutions, or hospitals, for the purpose of addressing, or providing education on, issues relating to (I) the treatment of OUD; (II) the prevention, education and treatment of opioid abuse, addiction, or overdose, including medication-assisted treatment for opioid addiction; and/or (III) rescue medications for opioid overdose;
 - (iii) the American Medical Association (AMA), the American Cancer Society (ACS) or any other medical society solely dedicated to cancer treatment; or
 - (iv) trade associations including, without limitation, PhRMA (Pharmaceutical Research and Manufacturers of America), HDA (Healthcare Distribution Alliance), AAM (Association for Accessible Medications), PCMA (Pharmaceutical Care Management Association), and NACDS (National Association of Chain Drug Stores), or successor organizations to any of the foregoing.
- b. The prohibitions in this Section II.D shall not apply to engagement with Third Parties based on activities related to (i) medications with an FDA-approved label that lists only the treatment of opioid abuse, addiction, dependence and/or overdose as their "indications and usage," to the extent they are sold to addiction treatment facilities; (ii) raw materials, APIs and/or immediate precursors used in the manufacture or study of Opioids or Opioid Products, but only when such materials, APIs and/or immediate precursors are sold or marketed exclusively to DEA registrants or sold outside the United

- States or its territories; or (iii) education warning about drug abuse or promoting prevention or treatment of drug misuse.
- c. Teva will be in compliance with subsections II.D.2 and II.D.3 with respect to support of an individual Third Party to the extent that the State of Texas determines that such support does not increase the risk of the inappropriate use of Opioids and that Teva has not acted for the purpose of increasing the use of Opioids.

E. <u>Lobbying Restrictions</u>

- 1. Teva shall not Lobby for the enactment of any federal, state, or local legislative or regulatory provision that:
 - a. encourages or requires Health Care Providers to prescribe Opioids or sanctions Health Care Providers for failing to prescribe Opioids or failing to treat pain with Opioids; or
 - b. pertains to the classification of any Opioid or Opioid Product as a scheduled drug under the Controlled Substances Act.
- 2. Teva shall not Lobby against the enactment of any federal, state or local legislative or regulatory provision that supports:
 - a. The use of non-pharmacologic therapy and/or non-Opioid pharmacologic therapy to treat chronic pain over or instead of Opioid use, including but not limited to third party payment or reimbursement for such therapies;
 - b. The use and/or prescription of immediate release Opioids instead of extended release Opioids when Opioid use is initiated, including but not limited to third party reimbursement or payment for such prescriptions;
 - c. The prescribing of the lowest effective dose of an Opioid, including but not limited to third party reimbursement or payment for such prescription;
 - d. The limitation of initial prescriptions of Opioids to treat acute pain;
 - e. The prescribing and other means of distribution of naloxone to minimize the risk of overdose, including but not limited to third party reimbursement or payment for naloxone;

- f. The use of urine testing before starting Opioid use and annual urine testing when Opioids are prescribed, including but not limited to third party reimbursement or payment for such testing;
- g. Evidence-based treatment (such as using medication-assisted treatment with buprenorphine or methadone in combination with behavioral therapies) for OUD, including but not limited to third party reimbursement or payment for such treatment; or
- h. The implementation or use of Opioid drug disposal systems.
- 3. Teva shall not Lobby against the enactment of any federal, state or local legislative or regulatory provision expanding the operation or use of prescription drug monitoring programs ("PDMPs"), including but not limited to provisions requiring Health Care Providers to review PDMPs when Opioid use is initiated and with every prescription thereafter.
- 4. Notwithstanding the foregoing restrictions in subsections II.E.1-3, the following conduct is not restricted:
 - a. Lobbying against the enactment of any provision of any state, federal, municipal, or county taxes, fees, assessments, or other payments;
 - b. Challenging the enforcement of, or suing for declaratory or injunctive relief with respect to legislation, rules or regulations referred to in subsection II.E.1:
 - c. Communications made by Teva in response to a statute, rule, regulation, or order requiring such communication;
 - d. Communications by a Teva representative appearing before a federal or state legislative or administrative body, committee, or subcommittee as a result of a mandatory order or subpoena commanding that person to testify;
 - e. Responding, in a manner consistent with this Consent Judgment, to an unsolicited request for the input on the passage of legislation or the promulgation of any rule or regulation when such request is submitted in writing specifically to Teva from a government entity directly involved in the passage of that legislation or promulgation of that rule or regulation;
 - f. Lobbying for or against provisions of legislation or regulation that address other subjects in addition to those identified in subsections

- II.E.1-3, so long as Teva does not support specific portions of such legislation or regulation covered by subsection II.E.1 or oppose specific portions of such legislation or regulation covered by subsections II.E.2-3;
- g. Communicating with a federal or state agency in response to a Federal Register or similar notice or an unsolicited federal or state legislative committee request for public comment on proposed legislation;
- h. Responding to requests from the DEA, the FDA, or any other federal or state agency, and/or participating in FDA or other agency panels at the request of the agency; and
- i. Participating in meetings and other proceedings before the FDA, FDA advisory committee or other FDA committee in connection with the approval, modification of approval, or oversight of Teva's own products.
- 5. Teva shall provide notice of the prohibitions in Section II.E to all employees engaged in Lobbying; incorporate the prohibitions in Section II.E into trainings provided to Teva employees engaged in Lobbying; and certify that it has provided such notice and trainings to Teva employees engaged in Lobbying.

F. Monitoring and Reporting of Off-Label Use

- 1. Teva shall monitor for off-label prescribing of its branded Opioid Products in the United States as provided for in the TIRF REMS Program.
- 2. Upon request of one of the following, Teva shall provide the requestor with the data and analysis described in Subsection II.F.1, to be used for law enforcement, counter-detailing, academic, or medical research, or public health and other non-commercial purposes: Texas Attorney General or other law enforcement agency, Texas Medical Board, Texas Board of Pharmacy, Qualified Researchers, medical and pharmacy directors of health systems or clinics, medical associations, and other public health officials, including but not limited to city health authorities, county medical directors, and Texas public health authorities.
- 3. Teva shall provide the data and analysis described in Subsection VI.E.1 in chart format, including breakdown of prescriptions by year, diagnosis, and county.

G. Ban on High Dose Opioids

1. After any related commercial commitments existing on the Effective Date have expired, Teva shall not manufacture, promote, or distribute any oxycodone pill that exceeds 40 milligrams.

H. Ban on Prescription Savings Programs

- 1. Teva shall not directly or indirectly offer any discounts, coupons, rebates, or other methods which have the effect of reducing or eliminating a patient's co-payments or the cost of prescriptions (*e.g.*, free trial prescriptions) for any Opioid Product. This does not preclude Teva from offering discounts or rebates to commercial partners on entire portfolios of products, including providing discounts, coupons, rebates, or other methods for use by retail chain pharmacies, such as CVS, Walgreens, Rite Aid and the like.
- 2. Teva shall not directly or indirectly provide financial support to any Third Party for discounts, coupons, rebates, or other methods which have the effect of reducing or eliminating a patient's co-payments or the cost of prescriptions (*e.g.*, free trial prescriptions) for any Opioid Product.

I. Monitoring and Reporting of Direct and Downstream Customers

- 1. Teva shall operate an effective monitoring and reporting system in compliance with federal law, that shall include processes and procedures that:
 - a. Utilize all reasonably available transaction information to identify a Suspicious Order of an Opioid Product by a direct customer;
 - b. Utilize all reasonably available Downstream Customer Data to identify whether a downstream customer poses a material risk of diversion of an Opioid Product;
 - c. Utilize all information Teva receives that bears upon a direct customer's or a downstream customer's diversion activity or potential for diversion activity, including reports by Teva's employees, customers, Health Care Providers, law enforcement, state, tribal, or federal agencies, or the media; and
 - d. Upon request (unless otherwise required by law), report to the Texas Attorney General or State controlled substances regulatory agency any direct customer or downstream customer in Texas identified as part of the monitoring required by (a)-(c), above, and any customer relationship in such State terminated by Teva relating to diversion

or potential for diversion. These reports shall include the following information, to the extent known to Teva:

- (i) The identity of the downstream registrant and the direct customer(s) identified by Teva engaged in the controlled substance transaction(s), to include each registrant's name, address, business type, and DEA registration number;
- (ii) The dates of reported distribution of controlled substances by direct customers to the downstream registrant during the relevant time period;
- (iii) The drug name, drug family or NDC, and dosage amounts reportedly distributed;
- (iv) The transaction or order number of the reported distribution; and
- (v) A brief narrative providing a description of the circumstances leading to Teva's conclusion that there is a risk of diversion.
- 2. Teva shall not provide to any direct customer an Opioid Product to fill an order identified as a Suspicious Order unless Teva investigates and finds that the order is not suspicious.
- 3. Upon request, Teva shall provide cooperation and assistance to any federal, state or local law enforcement investigations of potential diversion or suspicious circumstances involving Opioid Products, including criminal law enforcement agencies, drug control agencies, professional licensing boards, and Attorney General's offices.
- 4. Teva agrees that it will refrain from providing an Opioid Product directly to a retail pharmacy or Health Care Provider.

J. Miscellaneous Terms

1. To the extent that any provision in this Consent Judgment conflicts with federal or relevant state law or regulation, the requirements of the law or regulation will prevail. To the extent that any provision in this Consent Judgment is in conflict with federal or relevant state law or regulation such that Teva cannot comply with both the law or regulation and the provision of this Consent Judgment, Teva may comply with such law or regulation.

- 2. Teva will enter into this Consent Judgment solely for the purpose of settlement, and nothing contained herein may be taken as or construed to be an admission or concession of any violation of law, rule, or regulation, or of any other matter of fact or law, or of any liability or wrongdoing, all of which Teva expressly denies. No part of this Consent Judgment, including its statements and commitments, shall constitute evidence of any liability, fault, or wrongdoing by Teva. This Consent Judgment is not intended for use by any Third Party for any purpose, including submission to any court for any purpose.
- 3. For the avoidance of doubt, this Consent Judgment shall not be construed or used as a waiver or limitation of any defense otherwise available to Teva in any action, and nothing in this Consent Judgment shall be construed or used to prohibit Teva in any way whatsoever from taking legal or factual positions with regard to any Opioid Product(s) in litigation or other legal or administrative proceedings.
- 4. Nothing in this Consent Judgment shall be construed to limit or impair Teva's ability (a) to communicate its positions and respond to media inquiries concerning litigation, investigations, reports, or other documents or proceedings relating to Teva or its Opioid Products, or (b) to maintain a website explaining its litigation positions and responding to allegations concerning its Opioid Products.
- 5. Nothing in this Consent Judgment shall prohibit Teva from divesting any Opioid or Opioid Product, in each case, including providing technical development services, transferring know-how and patents, and/or providing such other support services in connection therewith.
- 6. This Consent Judgment applies to the manufacture, sales, Promotion, marketing and distribution by Teva within the United States and its territories or involving Health Care Providers.
- 7. Upon the request of the Attorney General of the State of Texas, Teva shall provide the Attorney General of the State of Texas with copies of the following, within 30 days of the request:
 - a. Any litigation or civil or criminal law enforcement subpoenas or Civil Investigative Demands relating to Teva's Opioid Product(s); and
 - b. Warning or untitled letters issued by the FDA regarding Teva's Opioid Product(s) and all correspondence between Teva and the FDA related to such letters.

- 8. The parties by stipulation may agree to a modification of this Consent Judgment; provided that the parties may jointly agree to a modification only by a written instrument signed by or on behalf of both Teva and the Attorney General of the State of Texas.
- 9. If, after the Effective Date, Teva, or its distributor subsidiary Anda, Inc. enters into any collective resolution of substantially all opioid claims brought by states, counties, and municipalities (a "Global Resolution") that contains injunctive relief terms that are more favorable than the terms of this Consent Judgment, then this Consent Judgment will be revised to contain such more favorable injunctive relief terms. Teva and/or Anda shall provide the State a copy of any such Global Resolution within thirty (30) days of its effective date.

K. <u>Compliance with State Laws and Regulations Relating to the Sale, Promotion, and Distribution of Any Opioid Product</u>

- 1. Subject to subsection II.G.1 above, Teva shall continue to comply with all applicable state laws and regulations that relate to the sale, Promotion, distribution, and disposal of Opioids or Opioid Products, including but not limited to:
 - a. Texas Controlled Substances Act, including all guidance issued by the applicable state regulator(s);
 - b. Texas Consumer Protection Laws; and
 - c. Texas laws and regulations related to opioid prescribing, distribution, and disposal.

III. Clinical Data Transparency

A. Data to Be Shared

- 1. Teva shall continue to share truthful and balanced summaries of the results of all Teva-Sponsored Studies through its publicly available website (*see* https://www.tevapharm.com/teva-clinical-trials):
 - a. "Teva-Sponsored Studies" means pre-marketing clinical research and post-marketing clinical research that Teva "takes responsibility for and initiates" as "sponsor," as "sponsor" is defined in 21 C.F.R. § 312.3(b), and that involves an intervention with human subjects with an Opioid Product.

- b. The summaries may include redactions to protect personal identifying information, trade secret and confidential commercial information, and information that may provide a road map for defeating a product's abuse-deterrent properties.
- 2. With respect to any Teva-Sponsored Studies relating to any new Teva Opioid Product or new indication for an existing Teva Opioid Product, Teva shall, within 6 months after regulatory approval or 18 months after study completion, whichever occurs later, make the following clinical data that is reasonably accessible and in its possession, custody, and control available through a third-party data archive that makes clinical data available to Qualified Researchers with a bona fide scientific research proposal:
 - a. Fully analyzable data set(s) (including individual de-identified participant-level data);
 - b. The clinical study report(s) redacted for commercial or personal identifying information;
 - c. The full protocol(s) (including the initial version, final version, and all amendments); and
 - d. Full statistical analysis plan(s) (including all amendments and documentation for additional work processes).
 - e. Data related to Investigator Sponsored Studies are not subject to the requirements in Section III.

B. <u>Third-Party Data Archive</u>

- 1. The third-party data archive referenced above shall have a panel of reviewers with independent review authority to determine whether the researchers are qualified, whether a research application seeks data for bona fide scientific research, and whether a research proposal is complete.
- 2. The panel may exclude research proposals with a commercial interest.
- 3. Teva shall not interfere with decisions made by the staff or reviewers associated with the third-party data archive.
- 4. Any data sharing agreement with a Qualified Researcher who receives shared data via the third-party data archive shall contain contact information for Teva's pharmacovigilance staff. Every agreement shall require the lead Qualified Researcher to inform Teva's pharmacovigilance staff within 24 hours of any determination that research findings could bear on the risk-

benefit assessment regarding the product. The lead Qualified Researcher may also share findings bearing on the risk-benefit assessment regarding the product with regulatory authorities. Teva's pharmacovigilance staff shall take all necessary and appropriate steps upon receipt of such safety information, including but not limited to notifying the appropriate regulatory authorities or the public.

5. Teva shall bear all costs for making data and/or information available to the third-party data archive.

IV. Compliance

A. <u>Compliance Duration</u>

- 1. Sections II and III of this Exhibit shall be effective for fifteen (15) years from the Effective Date.
- 2. Nothing in this Consent Judgment shall relieve Teva of its independent obligation to fully comply with the laws of the State of Texas after expiration of the 15-year period specified in this subsection.

B. Compliance Deadlines

1. Teva must be in full compliance with the provisions included in this Consent Judgment by the Effective Date. Nothing herein shall be construed as permitting Teva to avoid existing legal obligations.

V. Enforcement

- A. If the State believes that Teva is not in compliance with any term of this Final Consent Order, then the State shall:
 - 1. Provide written notice specifying the reason(s) why the State believes Teva is not in compliance with this Final Consent Order; and
 - 2. Allow Teva at least thirty (30) days to attempt to cure such alleged non-compliance (the "Cure Period").
- B. The State may not commence a proceeding to enforce compliance with this Final Consent Order before the expiration of the Cure Period, provided that the State may take any action if the State believes that, because of the specific practice, a threat to health or safety of the public requires immediate action.
- C. Teva agrees to venue for any proceedings related to this paragraph in the Court in which the State of Texas files this Consent Judgment.

Exhibit E

AGREED MOTION TO DISMISS WITH PREJUDICE RELEASED CLAIMS AGAINST DEFENDANTS [INSERT APPLICABLE TEVA AND ANDA DEFENDANTS] AND TO SEVER CERTAIN DEFENDANTS FROM THE CASE

Plaintiff ______ and Defendants [INSERT APPLICABLE TEVA AND ANDA DEFENDANTS] (together "Defendants"), file this Agreed Motion to Dismiss with Prejudice Released Claims Against Defendants and to Sever Certain Defendants ("Motion") and, in support thereof, respectfully show the Court as follows:

Plaintiff and Defendants (collectively, the "Parties") have settled certain claims related to their dispute by mutual agreement ("Released Claims"). The Released Claims are defined by settlement agreement dated [INSERT] between the parties (the "Settlement"), and only those Released Claims should be dismissed with prejudice; to the extent other claims may exist, those non-released claims should be severed. For the avoidance of doubt and as set forth in the Settlement, all claims against Defendants [INSERT TEVA USA/CEPHALON/ANDA/TEVA LTD WHERE APPLICABLE] ("Teva Defendants") are dismissed with prejudice. With respect to

claims against [INSERT APPLICABLE DIVESTED ACTAVIS ENTITIES IN CASE] ("Acquired Entity Defendants") based upon or related in any way to generic opioid drugs, those claims related to generic opioid drugs are dismissed with prejudice. Those claims against the Acquired Entity Defendants related to branded opioid drugs Kadian, Norco, Fiorinal, and Combunox are not released or dismissed and should be severed and stayed until further notice from the Parties. Therefore, Plaintiff no longer desires to pursue the Released Claims against Defendants. Accordingly, the Parties jointly move that the Court enter an Order dismissing the Released Claims against Defendants with prejudice, severing the Acquired Entity Defendants from this case with respect to those claims that are not Released Claims, and staying any and all proceedings with respect to those severed claims until further notice from the Parties.

WHEREFORE, PREMISES CONSIDERED, Plaintiff [INSERT] and Defendants [LIST ALL APPLICABLE TEVA AND ANDA DEFENDANTS], respectfully request that this Court enter an Order: (a) granting this Agreed Motion; (b) dismissing the Released Claims with prejudice to the re-filing of same, including, but not limited to, all claims against [INSERT TEVA USA/CEPHALON/TEVA LTD WHERE APPLICABLE]; (c) severing [INSERT ACQUIRED ENTITY DEFENDANTS] Defendants from this case with respect to those claims that are not Released Claims; and (d) staying any and all proceedings with respect to those severed claims until further notice from the Parties.

Dated: , 202 Respectfully subn	nıtted,
--------------------------------	---------

/s/ Nancy L. Patterson

Nancy L. Patterson Texas Bar No. 15603520

nancy.patterson@morganlewis.com

Cullen Pick

Texas Bar No. 24098260

cullen.pick@morganlewis.com

1000 Louisiana Street, Suite 4000

Houston, Texas 77002

Telephone: (713) 890-5000 Facsimile: (713) 890-5001

Collie F. James, IV

collie.james@morganlewis.com

MORGAN, LEWIS & BOCKIUS LLP 600 Anton Boulevard, Suite 1800 Costa Mesa, CA 92626-7653

Telephone: (714) 830-0600 Facsimile: (714) 830-0700

Of Counsel:

Nick Merkley GableGotwals One Leadership Square, 15th Floor 211 N. Robinson Avenue Oklahoma City, Oklahoma 73102 Telephone: (405) 568-3311 NMerkley@gablelaw.com

Attorneys for Defendants

[SIGNATURE BLOCK OF COUNSEL FOR ANDA IF APPLICABLE]

[SIGNATURE BLOCK OF COUNSEL FOR SETTLING PLAINTIFF]

CERTIFICATE OF SERVICE

I hereby certify that counsel of record are being served with a copy of this document on
 in accordance with the Texas Rules of Civil Procedure.
s/ Nancy L. Patterson
Nancy L. Patterson

Exhibit F

List of Litigating Subdivisions and Litigating Special Districts

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Angelina County	Dies & Parkhurst, L.L.P.; Simon Greenstone Panatier Bartlett, P.C.		86,715
Bailey County	Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough		7,000
Bastrop County	Phipps Deacon Purnell PLLC		88,723
Bexar County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		2,003,554
Bexar County Hospital District	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs; Phipps Deacon Purnell PLLC	Bexar County	[TBD]
Bowie County	Simon Greenstone Panatier Bartlett, P.C.; Dunn Nutter & Morgan; Martin Walker, P.C.		93,245
Brazos County	Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough; Matthew S. Daniel		229,211
Brooks County	Phipps Deacon Purnell PLLC		7,093
Burleston County	Burleston County Attorney's Office; Fibich, Leebron, Copeland & Briggs, O'Neill Law LLP; The Gallagher Law Firm; Watts Guerra LLP		
Caldwell County	Phipps Deacon Purnell PLLC		43,664
Calhoun County	Phipps Deacon Purnell PLLC		21,290
Cameron County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		423,163
Camp County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		13,094
Castro County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		7,530

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Cherokee County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.; Richards Penn; Ament & Peacock		52,646
Childress County	Altman Legal Group; Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		7,306
City of Eagle Pass	Napoli Shkolnik PLLC; Luis R. Vera & Associates	Maverick County	29,684
City of Houston	The Lanier Law Firm; Law Office of Richard Schechter, P.C.; Reich and Binstock LLP; Baker Wotring LLP	Harris County	2,320,268
City of Laredo	Napoli Shkolnik PLLC; Luis R. Vera & Associates	Webb County	262,491
City of Leon Valley	Phipps Deacon Purnell PLLC; Denton Navarro Rocha Bernal & Zech P.C.	Bexar County	12,306
City of San Antonio	The Herrera Law Firm; Baron & Budd, P.C.; Greene, Ketchum, Farrell, Bailey & Tweel LLP; Hill Peterson Carper Bee & Deitzler PLLC; Levin, Papantonio, Thomas, Mitchell, Rafferty & Proctor, P.A.; McHugh Fuller; Powell & Majestro	Bexar County	1,547,253
Clay County	Harrison Davis Steakley Morrison Jones, P.C.; Altman Legal Group; Haley & Olson		10,471
Colorado County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		21,493
Coryell County	Simon Greenstone Panatier Bartlett, P.C.; Richard A. Dodd, LC		75,951
Dallas County	Simon Greenstone Panatier Bartlett, P.C.; Lanier Law Firm; Cochran Firm		2,635,516
Dallas County Hospital District	Burns Charest LLP; Harrison Davis Steakley Morrison Jones, P.C.; Barrett Law Group, P.A.; Cuneo Gilbert & Laduca, LLP; Taylor Martino, P.C.	J	[TBD]
Delta County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		5,331

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Duval County	The Barrera Law Firm; The Snapka Law Firm		11,157
El Campo Memorial Hospital	Beggs & Lane, RLLP; Frazer PLC; Law Officer of Grant D. Amey, LLC; The Bilek Law Firm, LLP; The Kykendall Group		[TBD]
El Paso County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		839,238
Ellis County	Motley Rice LLC; Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough		184,826
Falls County	Simon Greenstone Panatier Bartlett, P.C.; Richard A. Dodd, LC		17,297
Fort Bend County	Roy L. Cordes, Jr. (County Attorney)		811,688
Franklin County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		10,725
Freestone County	Simon Greenstone Panatier Bartlett, P.C.; The Beckham Group		19,717
Galveston County	Watts Guerra LLP; O'Neill Law		342,139
Gonzales Healthcare Systems	Beggs & Lane, RLLP; Frazer PLC; Law Office of Grant D. Amey, LLC; The Kykendall Group		[TBD]
Guadalupe County	Phipps Deacon Purnell PLLC		166,847
Guadulupe Valley Hospital	Burns Charest LLP; Harrison Davis Steakley Morrison Jones, P.C.; Barrett Law Group, P.A.; Cuneo Gilbert & Laduca, LLP; Taylor Martino, P.C.	Guadalupe County	[TBD]
Harris County	Gallagher Law Firm; Fibich Leebron Copeland Briggs; Dan Downey, P.C.		4,713,325
Harris County Hospital District	Gallagher Law Firm; Fibich Leebron Copeland Briggs; Dan Downey, P.C.	Harris County	[TBD]
Harrison County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs; Truelove Law Firm, PLLC		66,553

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Haskell County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		5,658
Hays County	Phipps Deacon Purnell PLLC		230,191
Henderson County	Motley Rice LLC; Fears Nachawati, PLLC; Mathew S. Daniel; Ferrer Poirot & Wansbrough		82,737
Hidalgo County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs; Hinojosa Law Firm, P.C.		868,707
Hopkins County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		37,084
Irving Independent School District	The Coffman Law Firm; Mitchell A. Toups, Ltd.	Dallas County	[TBD]
Jefferson County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		251,565
Jim Hogg County	The Barrera Law Firm, The Snapka Law Firm		5,200
Jim Wells County	The Barrera Law Firm; The Snapka Law Firm; The Gutierrez Law Firm; Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		40,482
Johnson County	Fears Nachawati, PLLC; McLean Law Firm, P.C.; Ferrer Poirot & Wansbrough		175,817
Jones County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		20,083
Kaufman County	Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough		136,154
Kendall County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.; The Armstrong Firm, PLLC		47,431
Kerr County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		52,600
Kinney County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		3,667
Kleberg County	Michael J. Krueger; The Snapka Law Firm		30,680

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Lamar County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		49,859
LaSalle County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		7,520
Leon County	Watts Guerra LLP; O'Neill Law; The Gallagher Law Firm		17,404
Liberty County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs; Abraham, Watkins, Nichols, Sorrels, Agosto & Aziz; Husain Law & Associates, PC; Law Offices of Randall P. Gunter, P.C.		88,219
Lubbock County	Phipps Deacon Purnell PLLC		310,569
Madison County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		14,284
Maverick County	Napoli Shkolnik PLLC		58,722
McLennan County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		256,623
Mitchell County	Harrison Davis Steakley Morris Jones, P.C.; Haley & Olson		8,545
Montgomery County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		607,391
Morris County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		12,388
Nolan County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		14,714
Nueces County	The Law Office of Richard Schechter, P.C.; The Lanier Law Firm; The Law Office of James B. Ragan; Reich & Binstock, LLP; The Purnell Law Firm; Phipps Anderson Deacon LLP		362,294
Nueces County Hospital District	The Law Office of Richard Schechter, P.C.; The Lanier Law Firm; The Law Office of James B. Ragan; Reich & Binstock, LLP; The Purnell Law Firm; Phipps Anderson Deacon LLP	Nueces County	[TBD]

SUBDIVISION OR SPECIAL	COUNSEL	WITHIN COUNTY OR	2019 POPULATION
DISTRICT NAME		MULTIPLE COUNTIES	ESTIMATE
Ochiltree County Hospital District	Beggs & Lane RLLP; Frazer PLC; The Bilek Law Firm, L.L.P.; The Kuykendall Group; Law Office of Grant D. Amey, LLC	Ochiltree County	[TBD]
Palo Pinto County Hospital District	Burns Charest LLP; Harrison Davis Steakley Morrison Jones, P.C.; Barrett Law Group, P.A.; Cuneo Gilbert & Laduca, LLP; Taylor Martino, P.C.	Palo Pinto County	[TBD]
Polk County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		51,353
Red River County	Simon Greenstone Panatier Bartlett, P.C.		12,023
Roberts County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		854
Rockwall County	Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough		104,915
Rusk County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.; Love Law Firm, P.C.; Adkison Law Firm		54,406
San Patricio County	Alexander Dubose Jefferson Townsend; Joel H. Thomas; Phipps Deacon Purnell LLP		66,730
San Saba County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		6,055
Shackelford County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		3,265
Smith County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		232,751
Socorro Independent School District	The Coffman Law Firm; Mitchell A. Toups, Ltd.	El Paso County	[TBD]
Stephens County	Fears Nachawati, PLLC; Ferrer Poirot & Wansbrough		9,366
Tarrant County	The Lanier Law Firm; Law Offices of Tom Hall		2,102,515
Tarrant County Hospital District	Wick Phillips Gould & Martin, LLP	Tarrant County	[TBD]
Terrell County	The Coffman Law Firm; Mitchell A. Toups, Ltd.		776
Texarkana Independent School District	The Coffman Law Firm; Mitchell A. Toups, Ltd.	Bowie County	[TBD]

SUBDIVISION OR SPECIAL DISTRICT NAME	COUNSEL	WITHIN COUNTY OR MULTIPLE COUNTIES	2019 POPULATION ESTIMATE
Throckmorton County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson		1,501
Titus County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		32,750
Travis County	Hendler Flores; Law Office of Richard Schechter, P.C.; Reich and Binstock LLP; The Lanier Law Firm		1,273,954
Trinity County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		14,651
Upshur County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.; Tefteller Law PLLC		41,753
Uvalde County	Phipps Deacon Purnell PLLC		26,741
Van Zandt County	Simon Greenstone Panatier Bartlett, P.C.; Martin Walker, P.C.		56,590
Walker County	Correro & Leisure, P.C.; Park & Durham; G. Allan Van Fleet, P.C.		72,791
Waller County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		55,246
Webb County	Cicala Law Firm; Sanford Heisler Sharp		276,652
West Wharton County Hospital District	Beggs & Lane RLLP; Frazer PLC; The Bilek Law Firm, L.L.P.; The Kuykendall Group; Law Office of Grant D. Amey, LLC	Wharton County	[TBD]
Wichita County	Harrison Davis Steakley Morrison Jones, P.C.; Haley & Olson; Altman Law Group		132,230
Williamson County	Watts Guerra LLP; The Gallagher Law Firm; Fibich Leebron Copeland Briggs		590,551
Wilson County	Phipps Deacon Purnell PLLC		51,070
Wilson County Memorial Hospital District	Phipps Deacon Purnell PLLC	Wilson County	[TBD]
Zavala County	Napoli Shkolnik PLLC; Luis R. Vera & Associates		11,840

Exhibit G

MASTER FILE NO. 2018-63587

IN RE: TEXAS OPIOID LITIGATION	§	IN THE DISTRICT COURT
IN RE. TEXAS OFIOID EITIGATION	8	HARRIS COUNTY, TEXAS
MDL No. 18-0358	o .	,
	§	152ND JUDICIAL DISTRICT
This Document Relates to All Cases	S	
	8	

CASE MANAGEMENT ORDER

This Case Management Order ("CMO") shall apply to all plaintiffs with cases pending as of [Date of Final Court Approval of Settlement] against Defendants and to all new plaintiffs filing cases after that date against Defendants (collectively, "Plaintiff" or "Plaintiffs"), whose claims are pending in this coordinated proceeding and not released by the Settlement Agreement in this action entered into on [settlement date] ("Settlement Agreement"). As used herein, "Defendants" refers to Teva Pharmaceutical Industries Ltd.; Teva Pharmaceuticals USA, Inc. ("Teva USA"); Cupric Holding Co., Inc.; Teva Pharmaceutical Holdings Cooperative U.A.; Teva Pharmaceuticals Europe B.V.; Cephalon, Inc.; Watson Laboratories, Inc.; Actavis LLC; Actavis Pharma, Inc. f/k/a Watson Pharma, Inc.; Warner Chilcott Co., LLC; Actavis South Atlantic LLC; Actavis Elizabeth LLC; Actavis Mid Atlantic LLC; Actavis Totowa LLC; Actavis Kadian LLC; Actavis Laboratories UT, Inc. f/k/a Watson Laboratories Inc.-Salt Lake City; Actavis Laboratories FL, Inc. f/k/a Watson Laboratories, Inc.-Florida; and Anda, Inc. Pursuant to the order of the Texas Multidistrict Litigation Panel, all subsequent related and tag-along proceedings filed in the State of Texas and transferred to this Multidistrict Litigation Proceeding, In re: Texas Opioid Litigation, MDL No. 18-0358, pending before this Court and docketed under Master File No. 2018-63587, shall be subject to the terms of this CMO.

Good cause appearing, it is ordered as follows:

A. Filing of Amended Petitions

- 1. Each Plaintiff with an existing case as of [Date of Final Court Approval of Settlement], shall file and serve on Defendants within ninety (90) days of that date an Amended Petition satisfying the requirements of the Texas Rules of Civil Procedure and this CMO, if that Plaintiff's case is not dismissed with prejudice prior to this deadline pursuant to the Settlement Agreement. Plaintiff's counsel shall comply with Texas Rule of Civil Procedure 65 when filing any such Amended Petition.
- 2. The time for Defendants to file a response to a new Petition or Amended Petition shall not begin to run until after the receipt by counsel for the Defendants of the Case-Specific Expert Report(s) required pursuant to this CMO, and after the claims process is concluded as described in Section B.3 below, whichever is later.

B. <u>Plaintiffs' Requirement to Produce Certain Specified Information About Their Claims</u>

- 1. **Plaintiffs' Production Requirements.** Each Plaintiff shall serve the following documents and/or information upon counsel for Defendants:
- a. Fact Sheet. If not already completed, executed, and served, each Plaintiff shall serve upon the Defendants within the deadlines specified herein a completed copy of the Fact Sheet, attached as Exhibit A to this Case Management Order. Each Plaintiff that has already completed, executed, and served a compliant Fact Sheet shall serve upon the Defendants within the deadlines specified herein an updated Fact Sheet reflecting any material change in the facts underlying the Plaintiff's claims or shall affirm that no such material change applies. Simultaneously with its service of its Fact Sheet or affirmation, each Plaintiff shall serve upon Defendants a verified statement under oath setting forth how each element of their claims has not

been resolved pursuant to the terms of the Settlement and the state and regional abatement fund provided therein.

b. **Record Production.**

- (i) Each Plaintiff shall produce all records establishing the existence of a public nuisance within the Plaintiff's territory or borders, including a definition of the nuisance and evidence to support its existence.
- (ii) Each Plaintiff shall produce all records supporting a claim for nuisance "abatement" relief within the Plaintiff's territory or borders, including a categorization and itemization of any requested nuisance abatement relief and evidence to support each component of such relief.
- (iii) Each Plaintiff shall produce all records supporting a claim of damages, including a categorization and itemization of claimed damages and calculations and evidence for each component of such damages. Each Plaintiff shall also specify whether the alleged amounts were paid or reimbursed through a grant, insurance, or other third-party source and provide records evidencing such payment or reimbursement.
- (iv) For any other relief involving the expenditure of money, including expenditures for the provision of services, each Plaintiff shall specify the entities that will make the expenditures, when and how long those entities will make the expenditures, and the nature of the expenditures, including how they will address any and all alleged harms. Each Plaintiff shall produce all documents relied upon in identifying or calculating the claimed relief.
- (v) Each Plaintiff seeking any form of relief based directly or indirectly upon allegedly unnecessary prescriptions shall identify those prescriptions, to whom and

by whom the prescriptions were written, the pharmacy that filled each such prescription, whether the Plaintiff was reimbursed for them, and the Plaintiff's basis for identifying the prescriptions.

- c. Affidavit. An affidavit signed by each Plaintiff and its counsel (i) attesting that the Plaintiff has complied with all requirements of the Fact Sheet attached as Exhibit A to this Case Management Order; (ii) attesting that records have been collected in compliance with this CMO; and (iii) attesting that all records collected have been produced pursuant to this CMO. If any of the documents or records described in this Section B do not exist, the signed affidavit by the Plaintiff and its counsel shall state that fact and the reasons, if known, why such materials do not exist.
- d. **Expert Reports.** Each Plaintiff shall serve on counsel for Defendants a case-specific expert report or reports executed by a qualified expert, under oath, and subject to the penalties of perjury (a "Case-Specific Expert Report"). The Case-Specific Expert Report shall include all matter required to comply with Texas Rule of Civil Procedure 195, Texas law, and at least:
 - (i) Plaintiff's Information. The Plaintiff's name;
- (ii) Expert's Information. The name, professional address, and curriculum vitae of the expert, including a list of all publications authored by the expert within the preceding ten (10) years, and the foundation for the expert's opinion in relation to the expert's professional experience;
- (iii) Plaintiff's Records. All records reviewed by the expert in preparation of the Case-Specific Expert Report;
- (iv) Reliance Materials. All materials relied on by the expert in preparation of the Case-Specific Expert Report;

- (v) Locations. If the Plaintiff is asserting a public nuisance claim, the location(s) where the Plaintiff alleges a public nuisance exists, including with specificity how Plaintiff has been affected by such public nuisance and copies of documents relied upon, if any, as evidence of such alleged effect.
- (vi) Subjects of Report(s). The Case-Specific Expert Report(s) must collectively include all matters on which the expert(s) intend to rely, including but not limited to the following:
- (1) Whether the Plaintiff's records reviewed by the expert(s) indicate that the Plaintiff suffered any injury or damage and, if so, the nature of the alleged injury or damage;
- (2) Whether the Plaintiff's records reviewed by the expert(s) indicate the existence of a nuisance and, if so, the nature of the nuisance;
- (3) Whether the Plaintiff's records reviewed by the expert(s) indicate that Defendants engaged in any wrongful conduct and, if so, the nature of that conduct;
- (4) An opinion that there is in fact a causal relationship between the individual Plaintiff's claims and Defendants' alleged conduct and the basis for that opinion;
- (5) An opinion quantifying the relief requested by the Plaintiff, including any "abatement" relief, damages, and statutory penalties, with specific calculations and evidence for each component of such relief, prepared and sworn/affirmed to by such expert and subject to the penalties of perjury.

2. **Deadline to comply.**

- a. For each Plaintiff with claims pending against Defendants as of the entry of this CMO, the items required by Section B.1 shall be produced no later than [DATE], or ninety (90) days after the date such Plaintiff elects not to settle its claims, whichever is sooner.
- b. For each Plaintiff with claims newly filed in or transferred to this proceeding against Defendants after the entry of this CMO, the items required by Section B.1 shall be produced no later than ninety (90) days after the case is filed in or transferred to this proceeding.

3. Failure to comply.

- a. Notice of Non-Compliance and Opportunity to Cure. If any Plaintiff fails to comply with any provision of this Order, Defendants shall provide Plaintiff written notice of such non-compliance ("Notice of Non-Compliance") specifying the non-compliance. Upon receipt of a Notice of Non-Compliance, Plaintiff shall have sixty (60) days to cure its non-compliance specified in the Notice of Non-Compliance. During the period wherein noncompliance has not yet been cured, all litigation deadlines applicable to Defendants, including without limitation deadlines for discovery or to file and serve a pleading or motion responsive to a Plaintiff's petition, shall be held in abeyance.
- b. Failure to Cure. If, after the passage of sixty (60) days of service of a Notice of Non-Compliance, a Plaintiff fails to cure its non-compliance, upon application by the Defendants, the Plaintiff's claims, as well as any derivative claim(s), will be dismissed with prejudice as against Defendants.
- c. Extensions of Time. The Court, on motion and for good cause shown, may order an extension of the time to comply with this Order.

C. <u>Discovery on Statute of Limitations and Other Time-Based Defenses</u>

1. Plaintiffs must, within the time frames established by Section B.2, serve upon counsel for the Defendants an affidavit signed by the Plaintiff and its counsel providing the following information: (1) the date the Plaintiff first learned that the harms alleged in its petition may be related to Defendants' conduct; (2) how the Plaintiff first learned the harms alleged in its petition may be related to Defendants' conduct; (3) the date the Plaintiff first spoke to or corresponded with an attorney about potential litigation in this matter; and (4) the date the Plaintiff first retained counsel for litigation in this matter. Defendants are permitted to serve written discovery on each Plaintiff related to these topics (and others), and each such Plaintiff must respond to the discovery prior to any depositions related to these topics, provided that the Plaintiff shall have at least thirty (30) days to respond to such discovery.

D. <u>Case-Specific Discovery and Related Dispositive Motion Practice</u>

- 1. If a Plaintiff complies with the production requirements outlined above in Sections B and C, then the Parties, as applicable, shall submit a proposed Scheduling Order to the Court that: (a) grants the Parties one-hundred and eighty (180) days from the entry of the Scheduling Order to conduct discovery on issues raised by the productions; and (b) sets a briefing schedule that gives the Parties forty-five (45) days from the close of discovery for the Parties to submit summary judgment motions and *Daubert/Robinson* motions, twenty-eight (28) days for responses, and twenty-eight (28) days for replies.
- 2. During such discovery, the Parties are permitted to: serve written discovery related to the issues raised by the productions specific to the Plaintiff and take the depositions of both fact and expert witnesses for the Plaintiff for up to seven hours each, with counsel for Defendants questioning first at each deposition. If a Plaintiff serves any written discovery upon

Defendants, the Parties shall meet and confer about an appropriate deadline for responding to such discovery, which deadline shall be at least sixty (60) days after service of such discovery. The Court's use of the term "specific to the Plaintiff" is intended to express the Court's intention not to permit additional "generic" discovery against the Defendant at this time. No other depositions may be taken during the expedited discovery period absent prior leave granted by the Court upon a showing of good cause.

3. If a case survives the Defendant's summary judgment motions, the Court will set a Case Management Conference to determine whether any non-duplicative discovery is necessary and to discuss other case management issues. Discovery with regard to any other defendants will be addressed at this time as well. The filing and briefing of summary judgment motions and *Daubert/Robinson* motions after the expedited discovery discussed above shall not prejudice or otherwise foreclose the opportunity for any Party or other defendant to file later, non-duplicative summary judgment and *Daubert/Robinson* motions after completing full fact and expert discovery. The Court's use of the term "non-duplicative" is intended to express the Court's intention not to permit later summary judgment motions concerning topics addressed in summary judgment motions filed at the conclusion of the expedited discovery period or *Daubert/Robinson* motions concerning witnesses addressed in *Daubert/Robinson* motions filed at the conclusion of the expedited discovery period.

	4.	The foregoing provisions do not preclude any Party or other defendant from
filing non-du	uplicative	e dispositive motions, including motions relating to personal jurisdiction.
SO ORDER	ED.	
Dated:		
		Hon. Robert K. Schaffer
		Presiding Judge

Exhibit H

MASTER PURCHASE AGREEMENT

dated as of July 26, 2015

by and between

ALLERGAN PLC

and

TEVA PHARMACEUTICAL INDUSTRIES LTD.

TABLE OF CONTENTS

		1 4 50
MASTER P	PURCHASE AGREEMENT	1
ARTICLE I	DEFINITIONS	1
1.1	Definitions	1
1.2	Construction	30
1.3	Control	30
1.4	Performance of Obligations by Affiliates	30
ARTICLE I	II PURCHASE AND SALE	31
2.1	Agreement to Purchase and Sell	31
2.2	Excluded Assets	34
2.3	Assumed Liabilities	35
2.4	Excluded Liabilities	35
2.5	Procedures for Assignments	36
2.6	Allocation of Global Purchase Price, Assumed Liabilities	37
2.7	Earn-Out	38
2.8	Withholding	42
ARTICLE I	III CLOSING	42
3.1	Closing	42
3.2	Transactions at Closing	43
3.3	Purchase Price Adjustments	44
ARTICLE I	V REPRESENTATIONS AND WARRANTIES OF SELLERS	45
4.1	Qualification, Organization, Subsidiaries, etc.	46
4.2	Corporate Authority Relative to this Agreement; No Violation	46
4.3	Reports and Financial Information	48
4.4	Compliance with Law; Permits	48
4.5	Title to Assets	48
4.6	Transferred Entities	49
4.7	Environmental Matters and Regulations	49
4.8	Employee Benefits	50
4.9	Absence of Certain Changes or Events	52
4.10	Business Contracts	52

	4.11	Investigations; Litigation	54
	4.12	Regulatory Matters	54
	4.13	Consents and Governmental Approvals	57
	4.14	Tax Matters	58
	4.15	Labor Matters	59
4	4.16	Intellectual Property	60
4	4.17	Real Property	63
4	4.18	Required Vote	63
	4.19	Insurance	63
	4.20	Acquired Assets	64
	4.21	Brokers	64
	4.22	No Undisclosed Liabilities	64
	4.23	Securities Matters	65
	4.24	Material Suppliers	65
	4.25	Transferred Inventory; Recalls	65
	4.26	Intercompany Arrangements	66
	4.27	Anda Companies	66
	4.28	Uteron Business	66
	4.29	No Other Representations	66
	4.30	Disclaimer	66
ARTICLE V REPRESENTATIONS AND WARRANTIES OF BUYER PARENT			66
:	5.1	Qualification, Organization, Subsidiaries, etc.	67
:	5.2	Share Capital	67
:	5.3	Corporate Authority Relative to this Agreement; No Violation	68
:	5.4	Reports and Financial Statements	69
;	5.5	No Undisclosed Liabilities	70
:	5.6	Compliance with Law	70
:	5.7	Absence of Certain Changes or Events	70
:	5.8	Investigations; Litigation	70
:	5.9	Consents and Governmental Approvals	70
:	5.10	Tax Matters	71
:	5.11	Required Vote	71
;	5.12	Financing	71
	5 1 3	No Other Representations	72

AR	TICLE VI (COVENANTS	73
	6.1	Conduct of the Business and Buyer Parent Prior to the Closing	73
	6.2	Antitrust Filings and Actions	78
	6.3	Reasonable Best Efforts	80
	6.4	Access; Confidentiality	82
	6.5	Special Contracts	82
	6.6	Shared Business Contracts	83
	6.7	Intercompany Arrangements	84
	6.8	Pre-Closing Reorganization	85
	6.9	Transaction Committee	85
ARTICLE VII EMPLOYEE MATTERS			
	7.1	Principles	86
	7.2	Transfer of Employees	86
	7.3	Compensation and Benefits	87
	7.4	Service Credit	87
	7.5	Welfare Plans	87
	7.6	Labor and Employment Law Matters	88
	7.7	Defined Contribution Plans	88
	7.8	Communications	88
	7.9	Treatment of Business Employee Equity Awards	89
ARTICLE VIII TAX MATTERS			91
	8.3	Tax Cooperation	93
	8.4	Post-Closing Conduct	93
	8.5	Section 338 Elections	93
	8.6	Property Taxes	94
	8.7	Transfer Taxes	94
	8.8	VAT on the Acquired Assets	94
	8.10	Tax-Sharing and Procedures Agreement	96
AR.	TICLE IX (OTHER AGREEMENTS	96
	9.1	Books and Records; Access; Assistance	96
	9.2	Privileged Matters	97
	9.3	Seller Non-Solicitation	0.8

9.4	Seller Parent's Financing Cooperation	99
9.5	Buyer Parent's Financing Obligation	101
9.6	Buyer Conduct; Non-Use	103
9.7	Wrong Pocket Assets	104
9.8	Release of Indemnity Obligations	104
9.9	Use of Name	105
9.10	Directors and Officers	106
9.11	Ancillary Agreements	107
9.12	Employee Non-Solicit	107
9.13	Divestitures	108
9.14	Financial Statements	108
9.15	Seller Debt	109
ARTICLE	E X CONDITIONS TO CLOSING	109
10.1	Conditions to Each Party's Obligations	109
10.2	Conditions to Sellers' Obligations	109
10.3	Conditions of Buyers' Obligations	110
10.4	Frustration of Closing Conditions	111
ARTICLE XI TERMINATION		
11.1	Termination	111
11.2	Buyer Break Fee	112
11.3	Seller Break Fee	113
11.4	Procedure and Effect of Termination	113
ARTICLE	E XII INDEMNIFICATION	114
12.1	Survival; Effect of Materiality Qualifiers	114
12.2	Indemnification by Sellers	114
12.3	Indemnification by Buyer	116
12.4	Third Party Claim Indemnification Procedures	117
12.5	Direct Claims	118
12.6	Adjustments to Losses	119
12.7	Payments	119
12.8		119
ARTICLE	E XIII MISCELLANEOUS	120
13.1	Assignment	120

13.2	Public Announcements	120
13.3	Confidentiality	120
13.4	Expenses	121
13.5	Severability	122
13.6	Entire Agreement; Amendment	122
13.7	No Third-Party Beneficiaries	122
13.8	Waiver	122
13.9	Governing Law	122
13.10	Consent to Jurisdiction; Waiver of Jury Trial; No Recourse to Financing Sources	123
13.11	Specific Performance	124
13.12	Representation by Counsel	124
13.13	Bulk Transfers	124
13.14	Headings	124
13.15	Counterparts; Signature Pages	124
13.16	Notices	125

EXHIBITS

Exhibit A – Form of Stockholders Agreement
Exhibit B – Terms of Manufacturing Agreement
Exhibit C – Pre-Closing Reorganization Principles

MASTER PURCHASE AGREEMENT

This Master Purchase Agreement (this "Agreement"), dated as of July 26, 2015, is entered into by and between Teva Pharmaceutical Industries Ltd., a company organized under the laws of Israel ("Buyer Parent") and Allergan plc, a public company limited by shares organized under the laws of Ireland ("Seller Parent"). Seller Parent and the Controlled Affiliates of Seller Parent that are party to any Ancillary Agreement are referred to in this Agreement each as a "Seller" and collectively as "Sellers." Sellers and Buyer Parent sometimes are referred to in this Agreement collectively as the "Parties" and individually as a "Party."

WHEREAS, Sellers and their respective Controlled Affiliates are engaged in, among other things, the Business;

WHEREAS, Seller Parent wishes to sell, and cause the other Sellers and their respective Controlled Affiliates to sell, to Buyer Parent (or one or more designees thereof), and Buyer Parent (or one or more designees thereof) wishes to purchase, the Business through the purchase from Sellers and their respective Controlled Affiliates of all of the Acquired Assets, and Buyers (or one or more designees thereof) wish to assume the Assumed Liabilities, each upon the terms and conditions set forth herein and in connection therewith the Parties wish to enter into the transactions contemplated by this Agreement and by the Ancillary Agreements (collectively, the "Transactions"); and

WHEREAS, prior to the Closing, Sellers will undertake an internal reorganization pursuant to a Pre-Closing Reorganization Plan (the "Pre-Closing Reorganization") to facilitate the Transactions.

NOW, THEREFORE, in consideration of the premises and mutual covenants, agreements and provisions herein contained, the Parties agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. In addition to the terms defined above and other terms defined in other Sections of this Agreement, the following capitalized terms have the following meanings when used herein:

"AA2S" means Actavis Acquisition 2 S.à r.l.

"Accounting Principles" means: (a) any amounts not in U.S. Dollars shall be converted into U.S. Dollars using the spot rate of exchange as quoted by Bloomberg at 12:00 pm, New York time on the Closing Date; and (b) U.S. GAAP as applied by Seller Parent in the financial statements set forth in its Annual Report on Form 10-K for the year ended December 31, 2014 as filed with the SEC on February 18, 2015.

"Acquired Assets" has the meaning set forth in Section 2.1 and for purposes of the representations and warranties herein shall be deemed to include the Transferred Group Assets.

- "Actavis Marks" has the meaning set forth in Section 9.9(b).
- "Adjustment Amount" has the meaning set forth in Section 3.3(g).
- "Adverse Law or Order" means (i) any Law shall have been enacted, promulgated, enforced or sought to be enforced by any Governmental Authority of competent jurisdiction which prohibits or makes illegal the consummation of the Transactions or (ii) any Order preventing the consummation of the Transactions, whether preliminary or final.
- "Affiliate" means, with respect to any Person, any other Person which, at the time of determination, directly or indirectly controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise. For purposes of this Agreement, Buyer Parent shall not be deemed to control, be controlled by or be under common control with or be an Affiliate of Seller Parent or any Seller, and Seller Parent shall not be deemed to control, be controlled by or be under common control with or be an Affiliate of Buyer Parent or any Buyer. For the avoidance of doubt, no member of the Transferred Group will be deemed an Affiliate of Buyer Parent prior to Closing but after the Closing each member of the Transferred Group shall be Affiliates of Buyer Parent.
- "AGH" means Actavis Group hf., a company incorporated in Iceland with registered number 500269-7319 and having its registered office at Reykjavíkurvegi 76-78, 220 Hafnarfirði, Iceland.
 - "AGH Intra Group Debt" means all rights of AA2S in, to and under all intra group loan agreements between AGH and AA2S.
- "Agreement" means this Master Purchase Agreement, including all Schedules and Exhibits hereto, as it may be amended from time to time in accordance with its terms.
- "Allergan Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted by Allergan, Inc. and its Subsidiaries immediately prior to its acquisition by Seller Parent on March 17, 2015.
 - "Allergan Marks" has the meaning set forth in Section 9.9(a).
 - "Alternative Financing" has the meaning set forth in Section 9.5(a).
- "Ancillary Agreements" means the Local Transfer Agreements, the Transfer Documents, the Stockholders Agreement, the IP Licensing Agreement and the Transition Agreements.
 - "ANDA" means an abbreviated new drug application submitted pursuant to 21 U.S.C. § 355(j).

- "Anda Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted by the Anda Companies, the revenues of which are not reflected in the Performance Financial Statement.
- "Anda Companies" means Anda Veterinary Supplies, Inc., Anda Marketing, Inc., Anda Pharmaceuticals, Inc., Anda, Inc. and their respective subsidiaries, as of the date of this Agreement.
 - "Anti-Corruption Laws" has the meaning set forth in Section 4.12(h).
 - "Antitrust Division" has the meaning set forth in Section 6.2(a).
- "Antitrust Laws" means any antitrust, competition or trade regulation Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening competition through merger or acquisition, including the HSR Act.
- "APH4" means Actavis Pharma Holding 4 ehf., a company incorporated in Iceland and having its registered office at Reykjavíkurvegi 76-78, 220 Hafnarfirði, Iceland.
 - "APH4 Intra Group Debt" means all rights of AA2S in, to and under all intra group loan agreements between APH4 and AA2S.
 - "Applicable SOL Period" has the meaning set forth in Section 4.12(h).
 - "Arrow" has the meaning set forth in Section 2.7(a)(i).
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - "Audited Financial Statements" has the meaning set forth in Section 9.14.
- "AUK" means Allergan UK LLP, a limited liability partnership incorporated under the laws of England and Wales with registered number OC400028 and whose registered office is at Whiddon Valley, Barnstaple, Devon EX32 8NS.
- "AUK Intra Group Debt" means all rights of AUK in, to and under the £558,744,481 zero coupon deep discounted security due 2025 issued by Allergan UK Group Limited pursuant to a zero coupon deep discounted security agreement between AUK and Allergan UK Group Limited dated as of May 26, 2015.
- "Base Working Capital Amount" means the average of the December 2014 Working Capital Amount and the September 2015 Working Capital Amount.
 - "Base Working Capital Maximum Amount" means the Base Working Capital Amount plus \$100,000,000.
 - "Base Working Capital Minimum Amount" means the Base Working Capital Amount less \$100,000,000.

"Biostudy Research Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted as of the date hereof and at Closing by Sellers or any of their respective Controlled Affiliates in connection with performing clinical and bioequivalent studies in patients for Generic Drugs for third parties and the Seller Parent and its Controlled Affiliates.

"Books and Records" means original or true and complete copies of all of the books, records, files, work papers, data and information (including customer, distributor and supplier lists, summaries of financial and accounting records to the extent related to the Business, purchase orders and invoices, sales orders and sales order log books, credit and collection records, inventory records, product specifications, cost and pricing information, quality control records and manuals, product development files, records and stability and clinical studies, manufacturing processes and equipment, correspondence and miscellaneous records with respect to customers, distributors, suppliers and all other general correspondence) to the extent related to the Business, the Acquired Assets, the Assumed Liabilities or the employment of the Business Employees.

"Brand R&D Assets" means any research and development assets of the Transferred Group principally relating to the products set forth in Schedule 1.1.

"Business" means each of or, as the context requires, any or all of the U.S. Generics Business, the International Generics Business, the OTC Business, the Transferred Brands Business, the Biostudy Research Business and the Transferred Owned Real Property of Seller Parent and its Controlled Affiliates, but excluding, for the avoidance of doubt, the Allergan Business, the Uteron Business, the Excluded Products and the Anda Business.

"Business Contracts" means (i) all of the IP Contracts and (ii) all of the Contracts (other than IP Contracts and Seller Benefit Plans) Related to the Business.

"Business Day" means any day, other than Saturday or Sunday, on which commercial banks in New York City, New York, Dublin, Ireland and Tel Aviv, Israel are generally open for business.

"Business Employee" means each person employed by Seller Parent or a Subsidiary of Seller who is Related to the Business. Not later than thirty (30) days after the date of this Agreement, Seller Parent shall provide Buyer Parent with a Schedule of Business Employees at the date of this Agreement, which Schedule shall be prepared and updated thereafter in accordance with this Agreement from time to time prior to, and shall be finalized at, the Closing, by mutual agreement of Buyer Parent and Seller Parent.

"Buyer" and "Buyers" means Buyer Parent or any Affiliate or Subsidiary of Buyer Parent that Buyer Parent designates to purchase any of the Acquired Assets and/or are a party to any Ancillary Agreement.

"Buyer Break Fee" has the meaning set forth in Section 11.2(c).

"Buyer Fundamental Representations" has the meaning set forth in Section 12.1(a).

"Buyer Group" means Buyer Parent and its Affiliates and, following Closing, the Transferred Group.

"Buyer Indemnified Parties" has the meaning set forth in Section 12.2(a).

"Buyer Material Adverse Effect" means any Effect that is, or would reasonably be expected to be, materially adverse to the business of the Buyer Group or the financial condition, liabilities, business or results of operations of the business of the Buyer Group, taken as a whole; provided, however, that no Effects resulting from the following shall be deemed to constitute a Buyer Material Adverse Effect or shall be taken into account when determining whether a Buyer Material Adverse Effect has occurred or would reasonably be likely to exist: (i) conditions (or changes therein) in any industry or industries in which the business of the Buyer Group operates that do not disproportionately affect the Buyer Group, (ii) general legal, economic, political or regulatory conditions (or changes therein) in the markets in which the Buyer Group operates, including any changes affecting financial, credit or capital market conditions that do not disproportionately affect the Buyer Group, (iii) any generally applicable change in Law or U.S. GAAP or interpretation of any of the foregoing that do not disproportionately affect the Buyer Group, (iv) the announcement of, or the pendency of, this Agreement or consummation of the Transactions to the extent shown by Buyer Parent to be the proximate cause therefrom, including to the extent so shown (A) the identity of Buyer Parent, (B) any departure or termination of any officers, directors, employees or independent contractors of the Buyer Group, (C) the termination or potential termination of (or the failure or potential failure to renew or enter into) any Contracts with customers, suppliers, distributors or other business partners (provided that this subclause (C) shall not apply with respect to the representation of Buyers set forth in Section 5.2(b) related to the execution, delivery and performance by each Buyer of this Agreement not resulting in a material breach or a material fault under any material Contract), and (D) any other negative development (or potential negative development) in the Buyer Groups' relationships with any of its customers, suppliers, distributors or other business partners, (v) conditions arising out of acts of terrorism or sabotage, war (whether or not declared), the commencement, continuation or escalation of a war, acts of armed hostility, weather conditions, natural disasters or other force majeure events that do not disproportionately affect the Buyer Group, (vi) any determination by, or delay of a determination by, the FDA or any other Governmental Authority, or any panel or advisory body empowered or appointed thereby, or any indication that any such entity, panel or body will make any determination or delay in making any determination, with respect to the approvability, labeling, contents of package insert, prescribing information, risk management profile, CMC matters, pre-approval inspection matters or requirements relating to the results of any pre-clinical or clinical testing that are part of the business of the Buyer Group, (vii) any actions taken or failure to take action, in each case, by Seller Parent or any of its Controlled Affiliates, or to which an officer of Seller Parent has consented, or which an officer of Seller Parent has requested, or the taking of any action required by the express terms of this Agreement (other than any requirement to operate in the ordinary course of business), or the failure to take any action prohibited by the express terms of this Agreement, (viii) the failure to meet any financial or other plan, estimate, budget or projection (provided that the underlying cause of such failure, to the extent not otherwise excluded herein, may be taken into account in making a determination as to whether a Buyer Material Adverse Effect has or would reasonably be expected to occur), (ix) changes in the price of the Buyer Parent Stock (provided that the underlying cause of such change, to the extent not otherwise excluded herein, may be taken into

account in making a determination as to whether a Buyer Material Adverse Effect has or would reasonably be expected to occur), (x) any Legal Proceedings made or brought by any of the current or former stockholders of the Buyer Parent (on their own behalf or on behalf of the Buyer Parent) in their capacity as stockholders against the Buyer Parent arising out of the Transactions; and (xi) any item set forth in the Buyer Parent Disclosure Letter to the extent the relevance of such disclosure is reasonably apparent.

- "Buyer Parent" has the meaning set forth in the recitals hereof.
- "Buyer Parent Capitalization Date" has the meaning set forth in Section 5.2(a).
- "Buyer Parent Disclosure Letter" has the meaning set forth in the introductory paragraph to Article V.
- "Buyer Parent Equity Award" means a Buyer Parent Option, a Buyer Parent Restricted Share Award, a Buyer Parent Time-Vesting RSU Award or a Buyer Parent Performance RSU Award.
 - "Buyer Parent Equity Plans" means Buyer Parent's 2010 Long-Term Equity-Based Incentive Plan, as amended.
 - "Buyer Parent Option" has the meaning set forth in Section 7.9(a).
 - "Buyer Parent Performance RSU Award" has the meaning set forth in Section 7.9(c).
- "Buyer Parent's Knowledge" means the actual knowledge of the persons listed on Schedule 1.1(y) or such knowledge that would reasonably be obtained in the course of diligently executing the responsibilities of their positions.
 - "Buyer Parent Restricted Share Award" has the meaning set forth in Section 7.9(b).
 - "Buyer Parent SEC Documents" has the meaning set forth in Section 5.4(a).
 - "Buyer Parent Stock" means fully paid, nonassessable ordinary shares of Buyer Parent, par value NIS 0.10 per share.
 - "Buyer Parent Subsidiaries" means the Subsidiaries of Buyer Parent.
 - "Buyer Parent Time-Vesting RSU Award" has the meaning set forth in Section 7.9(c).
- "Buyer Parent VWAP" means as of a certain date, the volume weighted average price for the Buyer Parent Stock (or American Depository Shares with respect thereto), as shown by Bloomberg (or, if Bloomberg ceases to publish such price, any successor service reasonably agreed by Seller Parent and Buyer Parent), for the immediately prior twenty (20) trading days.
 - "Buyer-Requested Modifications" has the meaning set forth in Section 6.8.
 - "Cap" has the meaning set forth in Section 12.2(b)(iii).

"Cash" means cash, cash equivalents, bank deposits and marketable securities of the Transferred Group at Closing, provided that cash, cash equivalents, bank deposits and marketable securities of the Transferred Group that are not freely available shall be excluded only to the extent not freely available due to the actions of the Transferred Group outside of the Ordinary Course of Business since March 31, 2015 (but shall exclude in all circumstances any Insurance Proceeds).

"Cash Consideration" means \$33,750,000,000, as adjusted in accordance with Section 3.3.

"Claim" means any claim, demand, cause of action, chose in action, right of recovery or off-set, suit, litigation, Proceeding, arbitration, hearing or investigation against any Person.

"Claim Notice" has the meaning set forth in Section 12.4(a).

"Closing" has the meaning set forth in Section 3.1.

"Closing Date" has the meaning set forth in Section 3.1.

"Closing Date Net Cash" has the meaning set forth in Section 3.3(b).

"Closing Date Net Working Capital Statement" means the net working capital statement that sets forth the Current Assets and Current Liabilities as of the Closing, prepared by Buyer Parent in accordance with Section 3.3 hereof and, in the event of a Seller's Objection, as adjusted by either the agreement of Buyer Parent and Seller Parent, or by the Reporting Accountants, as described in Section 3.3.

"Closing Date Working Capital" means (x) the total Current Assets shown on the Closing Date Net Working Capital Statement, minus (y) the total Current Liabilities shown on the Closing Date Net Working Capital Statement, as finally determined.

"Closing Date Working Capital Excess" means the amount, if any, by which the Closing Date Working Capital is in excess of the Base Working Capital Maximum Amount.

"Closing Date Working Capital Shortfall" means the amount, if any, by which the Closing Date Working Capital is less than the Base Working Capital Minimum Amount.

"Closing Net Cash" means any Net Cash as of the Closing.

"Closing Statements" has the meaning set forth in Section 3.3(b).

"Code" means the means the United States Internal Revenue Code of 1986, as amended.

"Collective Bargaining Agreements" has the meaning set forth in Section 4.15(a).

"Compliant" means, without giving effect to any supplements or updates, (a) the Required Information does not contain any untrue statement of a material fact or omit to state any material fact, in each case with respect to the Business, the Acquired Assets or Assumed

Liabilities, necessary in order to make such Required Information not misleading, (b) no audit opinion with respect to any financial statements contained in the Required Information shall have been withdrawn, amended or qualified, and Seller Parent's independent registered accounting firm has consented to or otherwise authorized the use in the Financing of their audit opinions related to the audited financial statements included in the Required Information and (c) (i) the financial statements and other financial information included in the Required Information that have been prepared by Seller Parent are, and remain throughout the Marketing Period, sufficient to permit the Financing Sources (including underwriters, placement agents or initial purchasers) to receive customary comfort letters with respect to such financial information (including customary negative assurance comfort with respect to periods following the end of the latest fiscal year and fiscal quarter for which historical financial statements are included) on any date during the Marketing Period and (ii) Seller Parent's independent registered accounting firm that has reviewed or audited such financial information has delivered drafts of customary comfort letters, including customary negative assurance comfort, and such independent registered accounting firm has confirmed they are prepared to issue such comfort letter upon any pricing date and the closing relating to the Debt Financing occurring during the Marketing Period.

"Confidentiality Agreement" means the letter agreement, dated October 6, 2014, by and between Seller Parent and Buyer Parent relating to confidentiality obligations, as amended.

"Consent" means any consent, approval, ratification, authorization, consultation, waiver, grant, agreement, license, certificate, exemption, order, registration, declaration, filing or notice of, with or to any Person.

"Consideration Allocation" has the meaning set forth in Section 2.6(b).

"Controlled Affiliate" means, with respect to any Person, any other Person which, at the time of determination, is controlled by such Person. For purposes of this definition, "controlled" as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise. For the avoidance of doubt, no member of the Transferred Group will be deemed a Controlled Affiliate of Buyer Parent prior to Closing but after the Closing each member of the Transferred Group shall be a Controlled Affiliate of Buyer Parent.

"Continuing Employees" has the meaning set forth in Section 7.3.

"Contracts" means any written or oral commitment, contract, agreement, lease, licenses and other agreements, consensual obligation, promise, instrument, note, indenture, legally binding commitment, license, sublicense, understanding and undertaking, in each case whether written or oral and whether express or implied.

"Current Assets" means accounts receivable (net), inventories, prepaid expenses and the other current assets of the Transferred Entities and the Acquired Assets as determined in accordance with the Accounting Principles, excluding (i) Cash, (ii) Insurance Proceeds (except as Insurance Proceeds are in respect of other Current Assets), (iii) income tax receivables, and (iv) Excluded Assets.

"Current Liabilities" means accounts payable, accrued expenses and deferred revenue of the Transferred Entities and the Acquired Assets (or otherwise constituting Assumed Liabilities), as determined in accordance with the Accounting Policies, excluding (i) the current portion of any Indebtedness to the extent included in the calculation of Net Cash, (ii) any income tax payables, (iii) deferred taxes, and (iv) Excluded Liabilities.

"De Minimis Amount" means \$10,000,000.

"Debt Commitment Letter" means one or more debt commitment letters among Buyer Parent and the financing sources party thereto (who shall be internationally recognized investment or commercial banks or financial institutions engaged in the business of providing debt financing in similar acquisition financings, which may include those financing sources listed on Schedule 1.1(z), in form and substance reasonably satisfactory to Buyer Parent and subject only to conditions precedent customary for debt commitments for similar acquisition financings (it being understood that the draft debt commitment letter referred to on Schedule 1.1(z) with the termination date thereof changed to the Outside Date is deemed customary), to be entered into after the date hereof, as amended, supplemented or replaced in compliance with this Agreement, pursuant to which the financing sources party thereto agree to provide debt financing to Buyer Parent or any other Buyer to pay a portion of the Cash Consideration and other amounts to be paid pursuant to this Agreement and associated costs and expenses of the Transactions on the Closing Date.

"Debt Financing" means the debt financing incurred or intended to be incurred pursuant to the Debt Commitment Letter, including the offering or private placement of debt securities or borrowing of loans contemplated by the Debt Commitment Letter and any related engagement letter and including any credit facilities or capital markets debt financing or equity or equity-related offerings undertaken in replacement of all or any portion of such financing.

"Debt Financing Conditions" means the conditions precedent set forth in the Debt Commitment Letter.

"Debt Financing Documents" means the agreements, documents and certificates contemplated by the Debt Financing, including (a) all credit agreements, loan documents, purchase agreements, underwriting agreements, indentures, debentures, notes and intercreditor agreements pursuant to which the Debt Financing will be governed or contemplated by the Debt Commitment Letter; (b) officer, secretary, solvency and perfection certificates, legal opinions, corporate organizational documents, good standing certificates, Lien searches, and resolutions contemplated by the Debt Commitment Letter or requested by the Financing Sources; and (c) all documentation and other information required under applicable "know-your-customer" and anti-money laundering rules and regulations, including the USA Patriot Act.

"December 2014 Balance Sheet" means the audited balance sheet as of December 31, 2014 of the Transferred Entities, the Acquired Assets and Assumed Liabilities delivered pursuant to Section 9.14, as determined or interpreted in accordance with the Accounting Principles.

"December 2014 Working Capital Amount" means the Current Assets less the Current Liabilities prepared from the associated line items set forth on the December 2014 Balance Sheet

(provided that to the extent the line items on the December 2014 Balance Sheet are different from the line items included in the definition of Current Assets or Current Liabilities contained herein, the intent is to include line items from the December 2014 Balance Sheet that are similar in nature to the line items included in the definitions of Current Assets and Current Liabilities).

- "Deductible" means \$200,000,000.
- "Direct Claim" has the meaning set forth in Section 12.5.
- "Distribution Agreement" means the distribution agreement to be entered into at Closing in a form to be mutually agreed upon by the Parties.
- "Divestiture Action" has the meaning set forth in Section 6.2(a).
- "Earn-out Payment" means, in respect of any Earn-out Period, the payment to be made by the Buyer Parent to the Seller Parent in respect of such Earn-out Period, calculated in accordance with Section 2.7.

"Earn-out Period" means:

- (A) the First Earn-out Period; and/or
- (B) the period commencing on the day following the end of the First Earn-out Period and ending on the date falling three (3) months after the end of the First Earn-out Period; and/or
- (C) each subsequent three month period.
- "Earn-out Report" has the meaning given in Section 2.7(c)(i).
- "Effect" means any change, effect, development, circumstance, condition or occurrence.

"Environmental Laws" means all applicable international, federal, state, or local Laws, statutes, ordinances, regulations, policies, guidance, rules, judgments, orders, court decisions or rule of common law, permits, restrictions and licenses, which (i) regulate or relate to the protection or cleanup of the environment; the use, treatment, storage, transportation, handling, disposal or release of Hazardous Materials, the preservation or protection of waterways, groundwater, drinking water, air, wildlife, plants or other natural resources; or the health and safety of persons or property, including protection of the health and safety of employees or (ii) impose Liability or responsibility with respect to any of the foregoing, including the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601 et seq.), the Occupational Safety and Health Act (29 U.S.C. § 651 et seq.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.) or any other Law of similar effect.

"Environmental Permits" means all Permits required under or issued pursuant to all applicable Environmental Laws.

"Equipment" has the meaning set forth in Section 2.1(b)(v).

- "Equity Participations" means (i) any share, quota, security, participation right and any other right entitling the holder, absolutely or contingently (through the exercise of any subscription, conversion, exchange, option or similar right), to participate in the revenues, dividends or equity appreciation of another Person, including capital stock, membership interests, units, performance units, options, warrants, company appreciation rights, interests in "phantom" stock plans, restricted or contingent stock or profits interests, voting securities, stock appreciation rights or equivalents, stock loan purchase plans, convertible debentures or stock bonus plans and (ii) commitments to issue any of the foregoing.
 - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
 - "ERISA Affiliate" has the meaning given in Section 4.8(d).
 - "ERISA Plan" means an employee benefit plan as defined in Section 3(3) of ERISA.
 - "Estimated Cash Consideration Adjustment Amount" has the meaning set forth in Section 3.3(a).
 - "Estimated Closing Date Net Debt Statement" has the meaning set forth in Section 3.3(a).
- "Estimated Closing Date Net Working Capital Statement" means an estimated net working capital statement that sets forth an estimate of the Current Assets and Current Liabilities as of the Closing, prepared by Seller Parent in accordance with Section 3.3.
- "Estimated Closing Date Working Capital Excess" means the amount, if any, by which the estimated Closing Date Working Capital is in excess of the Base Working Capital Maximum Amount.
- "Estimated Closing Date Working Capital Shortfall" means the amount, if any, by which the estimated Closing Date Working Capital is less than the Base Working Capital Minimum Amount.
 - "Estimated Net Cash" has the meaning set forth in Section 3.3(a).
 - "Estimated Statements" has the meaning set forth in Section 3.3(a).
 - "Exchange Act" means the United States Securities Exchange Act of 1934, as amended.
- "Exchange Ratio" means the quotient obtained by dividing the Seller Parent VWAP by the Buyer Parent VWAP, in each case as determined on the second Business Day prior to the Closing.
 - "Excluded Assets" has the meaning set forth in Section 2.2.
 - "Excluded Liabilities" has the meaning set forth in Section 2.4.

- "Excluded Products" means those products listed on Schedule 1.1(a) and any authorized generic product that has not been launched prior to Closing.
- "Expert" means a partner or principal of a widely recognized accounting or law firm with expertise in the relevant area of disagreement to be submitted to such partner or principal, provided that such partner or principal does not, and the accounting or law firm with which such partner or principal is associated does not, have a conflict of interest with respect to the determination of the dispute which is to be submitted to such partner or principal.
- "Expert Selection Process" means the selection by mutual agreement, as to the Expert, among the parties that are in disagreement over a matter to be submitted to the Expert or, in the absence of such mutual agreement, the appointment of the Expert by the International Chamber of Commerce Centre for Expertise to resolve such disagreement.
 - "FCPA" means the U.S. Foreign Corrupt Practices Act of 1977 (15 U.S.C. §§ 78dd-1, et seq.), as amended.
 - "FDA" means the U.S. Food and Drug Administration.
 - "FDCA" means the Federal Food, Drug and Cosmetic Act (21 U.S.C. §§ 301 et seq.), as amended from time to time and any successor statute.
 - "Final Cash Consideration Amount" means the amount determined pursuant to Section 3.3(f).
- "Final Determination" shall exist when (a) the parties to the dispute have reached an agreement in writing, (b) a court of competent jurisdiction shall have entered a final and non-appealable Order or judgment or (c) an arbitration or like panel shall have rendered a final non-appealable determination with respect to disputes the parties have agreed to submit thereto.
 - "Financial Statements" has the meaning set forth in Section 9.14.
 - "Financing" has the meaning set forth in Section 9.4.
- "Financing Failure Event" means any of the following: (a) the commitments with respect to all or any portion of the Debt Financing, once received, expiring or being terminated, (b) for any reason other than the funding of the Debt Financing or another Financing, all or any portion of the Debt Financing becoming unavailable or (c) a material breach or repudiation by any party to the Debt Commitment Letter, in each case, other than as contemplated by the Debt Commitment Letter.
- "Financing Sources" means the agents, arrangers, lenders and other entities that have committed or been engaged to provide or arrange the Debt Financing or other Financings in connection with the Transactions, including the parties to the Debt Commitment Letter, any joinder agreements, indentures or credit agreements entered pursuant thereto or relating thereto, together with their respective affiliates and their and their respective affiliates' current, former or future officers, directors, employees, partners, trustees, shareholders, equityholders, managers, members, limited partners, controlling Persons, agents and representatives and respective successors and assigns of the foregoing Persons.

"First Earn-out Period" means the period from the date of Closing until (to and including) the first to occur of March 31, June 30, September 30 or December 31 of the calendar year in which Closing occurs.

"Former Business Employee" means any individual whose employment with Seller Parent or a Subsidiary of Seller Parent terminated prior to the Closing and who was Related to the Business immediately prior to such individual's termination of employment.

"Foreign Antitrust Laws" has the meaning set forth in Section 4.12(h).

"FTC" has the meaning set forth in Section 6.2(a).

"Generic Drugs" means products submitted or approved pursuant to an ANDA or Section 505(b)(2) of the FDCA (and any applicable foreign marketing authorizations) which are marketed or in development to be marketed as generic products.

"Global Purchase Price" means (i) the Cash Consideration; and (ii) the Stock Consideration.

"Global Purchase Price Allocation" has the meaning set forth in Section 2.6(a).

"Government Antitrust Authority" has the meaning set forth in Section 6.2(d).

"Government Officials" has the meaning set forth in Section 4.12(h).

"Governmental Approvals" means all licenses, consents, permits, certificates, filings, registrations, notifications, franchise, concession, authorizations, approvals, ratification, permission, clearance, confirmation, endorsement, waiver, designation, rating or qualification issued, granted, given or otherwise made available by or under the authority of any Governmental Authority or pursuant to any requirement under the applicable Laws of any Governmental Authority.

"Governmental Authority" means (a) any nation or government, including any federal, state, local, foreign, municipality, principality, commonwealth, province, territory, county, district or other jurisdiction of any nature or other political subdivision thereof; or (b) any entity, department, commission, bureau, agency, authority, board, court, official or officer, domestic or foreign, exercising executive, judicial, regulatory, administrative, judicial, police, military, or taxing governmental functions.

"Hazardous Materials" means (a) any "hazardous substance," "pollutant," "contaminant," "hazardous waste," "regulated substance," "hazardous chemical" or "toxic chemical" as designated, listed or defined (whether expressly or by reference) in any Environmental Law; (b) any other pollutant, chemical, substance, toxic, infectious, carcinogenic, reactive, corrosive, ignitable or flammable chemical, or chemical compound, or hazardous substance, material or waste, whether solid, liquid or gas, that is subject to regulation, control or

remediation under any Environmental Laws, including any quantity of asbestos in any form, urea, formaldehyde, polychlorinated biphenyls (PCBs), radon gas, petroleum, waste oil, crude oil, or any fraction thereof, all forms of natural gas, petroleum products or by-products or derivatives; and (c) any compound, mixture, solution, product or other substance or material that contains any hazardous substance or material referred to in clause "(a)" and "(b)" above.

- "Health Care Laws" has the meaning set forth in Section 4.12(b).
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.
- "HSR Filing" has the meaning set forth in Section 6.2(a).
- "Indebtedness" means (a) all indebtedness, whether or not contingent, for borrowed money; (b) all obligations evidenced by notes, bonds, debentures, loan agreements or other similar instruments; (c) all obligations as lessee under leases that have been or should be, in accordance with U.S. GAAP, recorded as capital leases; (d) all payment obligations, contingent or otherwise, under acceptance, letter of credit, bank guarantees, surety bonds or similar facilities; (e) net obligations under any interest rate, swap, currency swap, forward currency or interest rate contracts or other interest rate or currency hedging arrangements; (f) all obligations to pay the deferred purchase price of property or services (other than trade accounts payable in the Ordinary Course of Business and included in the calculation of Closing Date Working Capital); (g) any obligations under conditional sale or other title retention agreements; (h) any guarantee (other than customary non-recourse carve-out or "badboy" guarantees) of any of the foregoing, whether or not evidenced by a note, mortgage, bond, indenture or similar instrument, provided that Indebtedness shall not include any performance guarantee or any other guarantee that is not a guarantee of other Indebtedness; and (i) all Indebtedness referred to in clauses (a) through (h) above secured by any Lien on the Acquired Assets or the Transferred Group Assets.
 - "Indemnified Party" has the meaning set forth in Section 12.3(a).
 - "Indemnifying Party" has the meaning set forth in Section 12.4(a).
 - "Information" has the meaning set forth in Section 13.3(b) and Section 13.3(c), as applicable.
- "Insurance Proceeds" shall mean (i) any cash received by Seller Parent or its Controlled Affiliates after the date of this Agreement in respect of any asset, right or property that is or, if the asset, right or property has been destroyed or lost, would have constituted a Transferred Asset or a Transferred Group Asset prior to the Closing and in respect of which an insurer has made payment and (ii) any receivables of the Transferred Group or receivables or Claims in respect of an asset, right or property that would have been a Transferred Asset or a Transferred Group Asset if it existed at the Closing, due from an insurer and receivables owed to or cash received by Seller Parent or any of its Controlled Affiliates from an insurer relating to an Assumed Liability.
- "Intellectual Property" means all intellectual property or other proprietary rights of every kind throughout the world, including all registered, unregistered and pending: (a) patents,

patent applications, including provisional applications, statutory invention registrations, inventions, discoveries and invention disclosures (whether or not patented), and all related continuations, continuations-in-part, divisions, reissues, re-examinations, substitutions, and extensions thereof, (b) trademarks, service marks, trade names, trade dress, corporate names, logos, Internet domain names, URLs, any other source identifiers of any kind or nature, in each case whether or not registered, together with all translations, adaptations, derivations and combinations thereof, and all common law rights thereto, and the goodwill associated with all of the foregoing, and any applications (including intent to use applications), registration and renewals for any of the foregoing, (c) published and unpublished works of authorship whether or not copyrightable, including computer software programs, applications, source code and object code, and databases, other compilations of information, manual and other documentation, copyrights in and to the foregoing, together with all common law rights and moral rights therein, and any applications and registrations therefor, including extensions, renewals, restorations, reversions, derivatives, translations, localizations, adaptations and combinations of the above, (d) mask works and any applications, registrations and renewals for any of the foregoing and (e) trade secret rights under applicable law and other rights, if any, under applicable law in any information, including inventions, discoveries and invention disclosures (whether or not patented), formulae, know-how, confidential or proprietary information, methods, processes, protocols, specifications, patterns, techniques, research in progress, algorithms, data, designs, drawings, schematics, blueprints, flow charts, models, strategies, prototypes, testing procedures and testing results, and other forms of technology (whether or not embodied in any tangible form and including all tangible embodiments of the foregoing, su

"Interim Financial Statements" has the meaning set forth in Section 9.14.

"IRS" means the United States Internal Revenue Service.

"International Generics Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted as of the date hereof and at Closing of Sellers or any of their respective Controlled Affiliates in connection with the (a) researching, developing, submitting for approval, manufacturing, labeling, distributing, packaging, storing, or transporting the International Generics Products; and (b) marketing, promoting, using, licensing, selling or offering for sale of the International Generics Products.

"International Generics Products" means, collectively, any and all pharmaceutical products of Sellers or any of their respective Controlled Affiliates that are sold or intended to be sold outside of the United States as Generic Drugs (including branded generic pharmaceutical products, but excluding biologics or other biosimilar products) and authorized generic pharmaceutical products that are being sold as of the date hereof and at Closing and all such products or other assets that (i) are being or have been researched, tested, developed, commercialized, manufactured, sold, promoted or distributed by Sellers or any of their respective Controlled Affiliates outside of the United States, including those listed on Schedule 1.1(b), or (ii) are owned by, licensed to or otherwise used pursuant to a Contract by the Sellers or any of their respective Controlled Affiliates or introduced into commerce by any of the Sellers or their respective Controlled Affiliates outside of the United States.

"Inventory" means, as of the Closing, (i) all inventory of finished Product owned by Sellers or any of their respective Controlled Affiliates, whether or not Labeled, (ii) all Product work-in-progress owned by Sellers or any of their respective Controlled Affiliates, (iii) all other inventory Related to the Business, including raw materials, active pharmaceutical ingredients, excipients, intermediaries, reagents, packaging, work-in-process, finished goods, spare parts and shop and production supplies, in each case whether imported, provided from contract manufacturers or otherwise and whether located at a facility of Sellers or any of their respective Controlled Affiliates, by a wholesaler or in transit and (iv) all other inventory located at manufacturing facilities acquired, directly or indirectly, by Buyer Parent at the Closing, except to the extent such inventory is primarily related to the Retained Business.

"IP Contracts" means a Contract granting or purporting to grant to Sellers rights in the Licensed Patents and the Licensed Marks.

"IP Licensing Agreement" means the licensing agreement, in a form to be agreed by the Parties prior to Closing, to be entered into at Closing whereby each Party licenses to the other (on mutual terms) such Intellectual Property of the Business or the Retained Business (as applicable) that is owned by Sellers or its Affiliates or the Transferred Group (as applicable) and used in the Business or the Retained Business (as applicable) immediately prior to the Closing for the other Party to operate the Business or the Retained Business (as applicable) post-Closing or otherwise in connection with the other businesses of each Party.

"IT Assets" means the Sellers' and their respective Controlled Affiliates' computers, computing hardware, computer software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, file servers, printers, and networking and all other information technology equipment, and all associated documentation that are owned or controlled by any Seller or any of their Controlled Affiliates.

"Know-How" means all non-public information, proprietary or otherwise, owned or held by or licensed to a Seller or any Affiliate of any Seller as of the date of this Agreement or at Closing.

"Labeling" (and the correlative terms "Label" and "Labeled") shall be as defined in Section 201(k), (m) of the FDCA (21 U.S.C. § 321(k), (m)) and other comparable foreign Law relating to the subject matter thereof, including the applicable Product's label, packaging and package inserts accompanying such Product, and any other written, printed, or graphic materials accompanying such Product, including patient instructions or patient indication guides.

"Law" means each provision of any national, supranational, federal, state, provincial, local, municipal or foreign, civil and criminal law, common law, constitution, statute, regulation, legislation, ordinance, Order, code, proclamation, treaty, convention, rule, ruling, directive, requirement, determination, decision, opinion or interpretation, promulgated, adopted, enacted, implemented, issued, passed, approved, or otherwise put into effect by or under the authority of any Governmental Authority, as well as any judgments, decrees, injunctions or agreements issued or entered into by any Governmental Authority.

"Lenalidomide" has the meaning set forth in Section 2.7(a)(ii).

- "Lenalidomide Agreement" has the meaning set forth in Section 2.7(a)(ii).
- "Liability" means, with respect to any Person, any debt, liability, duty or obligation of such Person, whether known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, due or to become due, vested or unvested, executory, determined, determinable or otherwise, and whether or not the same is required to be accrued on the financial statements of such Person, including those arising under any Law, Order or any Contract.
 - "Licensed Manufacturing IP" means the royalty free, worldwide non-exclusive license of the Manufacturing IP.
 - "Licensed Marks" has the meaning set forth in Section 2.1(b)(vii).
 - "Licensed Patents" has the meaning set forth in Section 2.1(b)(vii).
 - "Licensed Shared IP Sublicenses" has the meaning set forth in Section 6.6(b).
- "Lien" means any lien, encumbrance, mortgage, security interest, pledge, conditional sale agreement or other title retention agreement, or other charge or encumbrance of any nature whatsoever on any property or property interest.
- "Litigation Matters" means all Proceedings that have been or may be asserted by a third party against, or otherwise adversely affect, Seller or its Subsidiaries (other than the Transferred Group), on the one hand, and the Business, on the other hand.
- "Local Transfer Agreements" means the agreements, in a form to be agreed by the Parties as necessary to comply with applicable local governing Law to effect the transfer of the Acquired Assets and the assumption of the Assumed Liabilities located in a particular jurisdiction, which, subject to the terms and conditions of this Agreement, are to be signed at or prior to Closing, pursuant to which, among other things, for the consideration stated herein, certain Sellers shall grant, sell, transfer, convey, assign and deliver to certain Buyers, and such Buyers shall purchase and accept from such Sellers, all right, title, and interest of such Sellers in and to the specific items of Acquired Assets owned by such Sellers, and such Buyers agree to assume certain Assumed Liabilities of such Sellers, all in accordance with and pursuant to the terms and conditions therein and in accordance with the governing Law set forth in such Local Transfer Agreements.
- "Losses" means damages, diminution in value, Liabilities, Claims, payments, fines, fees, penalties, Taxes, charges, judgments, defaults, settlements, assessments, deficiencies, interest and costs and expenses (including removal costs, remediation costs, closure costs, fines, penalties and expenses of investigation and ongoing monitoring, reasonable attorneys', accountants' and other experts' fees and reasonable out-of-pocket disbursements) and other losses (excluding punitive and exemplary damages except to the extent required to be paid to a third party).

"Manufacturing Agreements" means the manufacturing and supply agreements to be entered into at Closing incorporating the terms set forth on Exhibit B and any other terms mutually agreed upon by the Parties in accordance with the terms hereof.

"Manufacturing Instructions" means those manufacturing, packaging and Labeling specifications for the Products used by or on behalf of a Seller or any Affiliate of any Seller in the production and supply of the Products, including the formulae and materials that are reasonably required or used for the manufacture, quality control, release and stability testing of the Products by or on behalf of a Seller or any Affiliate of any Seller immediately prior to the Closing.

"Manufacturing IP" means the Intellectual Property of the Seller Parent that the applicable member of the Transferred Group requires in order to be able to manufacture and supply products to the Buyer Parent pursuant to the Manufacturing and Supply Agreement.

"Marketing Period" means the first period of twelve (12) consecutive Business Days commencing on or after the date hereof throughout which and on the first and last day of which (a) Buyer Parent shall have received the Required Information and the Required Information is and remains Compliant and (b) the conditions set forth in Section 10.1 shall have been satisfied or waived (other than conditions which by their nature would be satisfied only at the Closing itself), and nothing has occurred and no condition exists that could reasonably be expected to cause any of the conditions set forth in Section 10.3 to fail to be satisfied assuming the Closing would be scheduled at any time during such twelve (12) consecutive Business Day period; provided that, (i) November 26, 2015 to November 28, 2015, March 25, 2016, May 30, 2016, May 31, 2016 and July 4, 2016 shall not constitute Business Days, (ii) if the Marketing Period has not been completed on or prior to August 20, 2016, then it may not commence until on or after September 6, 2016 and (iii) if the Marketing Period has not been completed on or prior to December 18, 2015, then it may not commence until on or after January 4, 2016; provided, further, that if Seller Parent shall in good faith reasonably believe that the Required Information has been delivered to Buyer Parent, it may deliver to Buyer Parent a written notice to that effect (stating that it believes that such delivery has been completed), in which case the Required Information shall be deemed to have been provided (and, if the other conditions set forth in this definition have been met, the Marketing Period commenced) on the first (1st) Business Day following the date that notice is received unless Buyer Parent in good faith reasonably believes the delivery of the Required Information has not been completed and, within three (3) Business Days of the delivery of such notice by Seller Parent, delivers a written notice to Seller Parent to that effect (stating with reasonable specificity which Required Information that Buyer Parent reasonably believes has not been delivered), in which case the Marketing Period shall be deemed to have not commenced and will only commence beginning on the date the Required Information is delivered to Buyer Parent and the other conditions set forth in this definition have been met. Notwithstanding the foregoing, the Marketing Period shall not commence and shall be deemed not to have commenced if, on or at any time prior to the completion of such twelve (12) consecutive Business Day period, (i) Seller Parent's independent registered accounting firm shall have withdrawn or qualified its authorization letter or audit opinion with respect to any financial statements contained in the Required Information, in which case the Marketing Period shall not be deemed to commence until the time at which, as applicable, a new authorization letter or unqualified audit opinion is issued with respect to the consolidated financial statements for the

applicable periods by the Seller Parent's independent registered accounting firm or another independent registered accounting firm acceptable to Buyer Parent or (ii) Seller Parent or its independent registered accounting firm indicates its intent to, or determined that it must, restate any financial statements or material financial information included in the Required Information, in which case the Marketing Period shall be deemed not to commence unless and until such restatement has been completed and the applicable Required Information has been amended or Seller Parent has announced that it has concluded that no restatement shall be required. If the Required Information is not Compliant throughout and on the first and the last day of such period, then a new twelve (12) consecutive Business Day period shall commence upon Buyer Parent receiving updated Required Information that is Compliant and the other conditions set forth in this definition having been met. Notwithstanding the foregoing, the Marketing Period shall end on any earlier date that is the date on which the full amount of the Debt Financing (together with any Financing replacing any portion of the Debt Financing) has been received by Buyer Parent if such date is prior to the end of the applicable twelve (12) consecutive Business Day period.

"Marks" has the meaning set forth in Section 2.1(b)(vii);

"Net Cash" means any Cash at Closing less Indebtedness of the Transferred Group and the Acquired Assets at Closing (except that any Indebtedness of the type described in clause (h) of the definition of Indebtedness shall not be included in such calculation), which for the avoidance of doubt, may be a negative number.

"Non-Business Employee" means each person employed by a Seller or a Subsidiary of a Seller who is not Related to the Business.

"Non-Shared Transfer Taxes" has the meaning set forth in Section 8.7.

"Non-U.S. Transferred Entity Benefit Plan" has the meaning set forth in Section 4.8(a).

"Notice of Disagreement" has the meaning given in Section 2.7(c)(ii).

"Notice Period" has the meaning set forth in Section 12.4(a).

"Order" means any writ, judgment, edict, fine, penalty, sanction, award, notice of deficiency, warning letter, decree, injunction, ruling, pronouncement, order, determination, decision, opinion, verdict, sentence, subpoena, writ or similar requirement, corporate integrity agreement, deferred prosecution agreement, settlement agreement or other binding obligation of any Governmental Authority (whether preliminary or final).

"Ordinary Course of Business" means the operation of the Business by Seller Parent and its Controlled Affiliates in the usual and customary way and consistent with their past practices through the date of this Agreement.

"Organizational Documents" means, with respect to any Person, collectively, its organizational documents, including any certificate of incorporation, notarial deed of incorporation, certificate of formation, articles of organization, articles of association, business rules and regulations, bylaws, operating agreement, certificate of limited partnership, partnership agreement, equityholders' agreement and/or certificates of existence, as applicable.

"OTC Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted as of the date hereof and at Closing of Sellers or any of their respective Controlled Affiliates in connection with the (a) researching, developing, submitting for approval, manufacturing, labeling, distributing, packaging, storing, or transporting the OTC Products; and (b) marketing, promoting, using, licensing, selling or offering for sale of the OTC Products.

"OTC Products" means, collectively, any and all over-the-counter (non-prescription) products of Sellers or any of their respective Controlled Affiliates, including all such products or other assets that (i) are being commercialized, manufactured, sold, promoted or distributed by Sellers or any of their respective Controlled Affiliates as of the date hereof and at Closing, including those listed in Schedule 1.1(c) and (ii) are owned by, licensed to or otherwise used pursuant to a Contract by the Sellers or any of their respective Controlled Affiliates or introduced into commerce by any of the Sellers or their respective Controlled Affiliates.

"Outside Date" has the meaning set forth in Section 11.1(b).

"Overlap Products" means (i) a Product that has an indication or targeted indication that is the same as, or substantially similar to, the indication or targeted indication of any commercial product or product in development of Buyer Group and (ii) a commercial product or product in development of Buyer Group that has an indication or targeted indication that is the same as, or substantially similar to, the indication or targeted indication of any Product, together in each case with any related assets that may be needed or required to be included by a Governmental Authority in connection with a Divestiture Action.

"Parties" or "Party" has the meaning set forth in the Preamble hereof.

"Patents" has the meaning set forth in Section 2.1(b)(vii);

"Permits" means all permits, licenses, registrations, certificates, franchises, variances, exemptions, orders and other Governmental Approvals, Consents and authorizations necessary or desirable for the past, present or anticipated conduct of the Business or Related to the Business, other than Regulatory Registrations.

"Permitted Lien" means any (a) mechanic's, materialmen's, and similar liens arising or incurred in the Ordinary Course of Business, (b) liens for Taxes not yet due and payable, (c) liens for Taxes that the taxpayer is contesting in good faith through appropriate Proceedings and for which adequate reserves have been established in accordance with U.S. GAAP, and (d) liens consisting of zoning or planning restrictions, easements, permits and other restrictions or limitations on the use of real property or irregularities in title thereto, which do not materially impair the value of such properties or the use of such property by a Seller in the operation of its business.

"Person" means any natural person, individual, corporation (including any non-profit corporation), partnership, joint venture, limited liability company, estate, trust, cooperative, foundation, society, political party, union, other unincorporated organization, joint stock company or Governmental Authority.

"Personally Identifiable Information" means any information that alone or in combination with other information held by the Sellers or any of their Affiliates can be used to specifically identify an individual person and any individually identifiable health information.

"Post-Closing Tax Period" means any Tax period beginning after the Closing and that portion of a Straddle Period beginning after the Closing Date.

"Potential Purchasers" has the meaning set forth in Section 9.13.

"Potential Sale Transaction" has the meaning set forth in Section 9.13.

"Pre-Closing Period" has the meaning set forth in Section 6.1(a).

"Pre-Closing Reorganization" has the meaning set forth in the Recitals.

"Pre-Closing Reorganization Plan" means the plan of actions to be undertaken by Parent Seller and its Controlled Affiliates to implement the Pre-Closing Reorganization on the basis of the principles and terms described on Exhibit C.

"Pre-Closing Tax Period" means any Tax period ending on (and including) or before the Closing Date and that portion of any Straddle Period ending on (and including) the Closing Date.

"Privileged Information" means, with respect to each Party, information regarding such Party or its Subsidiaries, or any of its operations, assets or liabilities (whether in documents or stored in any other form or known to its employees or agents) that is protected from disclosure pursuant to the attorney-client privilege, the work product doctrine or another applicable legal privilege, in each case that the other party or its Subsidiaries may come into possession of or obtain access in connection with this Agreement or the Ancillary Agreements.

"Proceeding" means any claim, action, arbitration, audit, hearing, inquiry, prosecution, contest, examination, proceeding, investigation, litigation, suit (whether civil, criminal, administrative, or investigative or appellate proceeding and any informal proceeding) commenced, brought, conducted, or heard by or before, or otherwise involving any Governmental Authority, arbitrator or arbitration panel.

"**Products**" means each of, or as the context requires, any or all of the U.S. Generic Products, the International Generics Products, the OTC Products or the Transferred Brand Products, including <u>Schedule 1.1(h)</u>.

"**Promotional Activities**" means those activities undertaken to encourage sales of the Products, including: journal advertising, broadcast advertising, direct mail programs, detailing, customer meetings, conventions and trade show exhibits, Product presentations, end user training, marketing plan development, ongoing post-market development, demand generation, symposia and other forms of advertising, promotion, sales and customer support.

"Property Taxes" means all real property Taxes, personal property Taxes and similar ad valorem Taxes.

"PTC" means Actavis Group PTC ehf., a company incorporated in Iceland and having its registered office at Reykjavíkurvegi 76-78, 220 Hafnarfirði, Iceland.

"PTC Intra Group Debt" means all rights of AA2S in, to and under all intra group loan agreements between PTC and AA2S.

"Puerto Rico Grant" means the Grant of Tax Exemption to Warner Chilcott Company, LLC by the Commonwealth of Puerto Rico, dated December 23, 2009, as set forth on Schedule 1.1(i).

"Regulatory Authority" means any Governmental Authority that is responsible for issuing technical, medical, scientific, Labeling and similar licenses, registrations, authorizations, permits, certifications, variances, exemptions, orders and approvals necessary for the manufacture, commercialization, Labeling, distribution, use, storage, import, export, transport, marketing or sale of the Products.

"Regulatory Documentation" means (a) all regulatory filings and supporting documents (including copies of all correspondence between any of Sellers and their Controlled Affiliates and the applicable Regulatory Authority), chemistry, manufacturing and controls data and documentation, preclinical and clinical studies and tests, (b) the ANDA and all regulatory files and foreign equivalents related thereto, current approved packaging and any other existing files and dossiers, including the underlying data or information used to support, maintain or obtain marketing authorization of the underlying Product, (c) all records maintained under record keeping or reporting Laws of the FDA or any other Governmental Authority including all applications, annual and safety reports, drug master files, FDA warming letters, FDA notices of adverse finding letters, FDA audit reports (including any responses to such reports), any correspondence with the Office of Prescription Drug Promotion, periodic safety update reports, complaint files, and annual product quality reviews, (d) the complete complaint, adverse event and medical inquiry filings with respect to the Products as required by applicable Laws and Related to the Business, including the Regulatory Registrations and (e) all equivalent, comparable or analogous documentation with respect to any other country outside the United States.

"Regulatory Registrations" means the premarket notifications or premarket approvals issued by the FDA, European Union Conformity Marking (CE marks) issued by a European Union Notified Body, and all other technical, medical, scientific, Labeling and similar licenses, registrations, authorizations, permits, certifications, franchises, variances, exemptions, orders, approvals, amendments and renewals of the Products (including marketing authorizations and Labeling approvals) issued by the Regulatory Authorities of any country and held or pending (including any applications) as of the Closing Date by Sellers or any of their Controlled Affiliates or third-party distributors (under rights of reservation of such Seller) that are required for the manufacture, commercialization, Labeling, distribution, use, storage, import, export, transport, marketing or sale of the Products within any country.

- "Related to the Business" means primarily related to or used primarily in connection with the Business as conducted by the Sellers and their respective Controlled Affiliates as of the date hereof and at Closing.
 - "Replacement Financing" has the meaning set forth in Section 9.5(b).
 - "Replacement Financing Documents" has the meaning set forth in Section 9.5(b).
 - "Reporting Accountants" means any of KPMG, Deloitte, PriceWaterhouseCoopers or Ernst & Young, reasonably acceptable to both Parties.
- "Representatives" means, with respect to any Person, any officers, directors, employees, Controlled Affiliates, attorneys, investment bankers, financial advisers, agents and other representatives of such Person.
 - "Required Information" has the meaning set forth in Section 9.4.
 - "Restricted Benefits" has the meaning set forth in Section 4.12(h).
 - "Restructuring Event" means a restructuring event for the purposes of Buyer Parent's restructuring policy.
- "Retained Business" means the assets, business, Know-How, operations and activities of the Seller Parent and its Subsidiaries not Related to the Business or not otherwise included in the definition of Acquired Assets, including, for the avoidance of doubt, the Allergan Business, the Uteron Business, the Excluded Products, the Anda Business and the Biologics Business.
 - "Retained Entity" means Seller Parent and each of its Subsidiaries not Related to the Business.
- "Reverse Transition Services Agreement" means the transition services agreement to be entered into at Closing in a form to be mutually agreed upon by the Parties.
 - "SEC" means the United States Securities and Exchange Commission.
 - "Securities Act" shall mean the United States Securities Act of 1933, as amended.
 - "Seller" and "Sellers" have the meaning set forth in the recitals hereof.
 - "Seller Benefit Plan" has the meaning set forth in Section 4.8(a).
 - "Seller Contract Notice" has the meaning set forth in Section 6.5.
 - "Seller Fundamental Representations" has the meaning set forth in Section 12.1(a).
 - "Seller Indemnified Parties" has the meaning set forth in Section 12.3(a).
- "Seller Material Adverse Effect" means any Effect that is, or would reasonably be expected to be, materially adverse to the Acquired Assets or the financial condition, business,

liabilities or results of operations of the Business taken as a whole; provided, however, that no Effects resulting from the following shall be deemed to constitute a Seller Material Adverse Effect or shall be taken into account when determining whether a Seller Material Adverse Effect has occurred or would reasonably be likely to exist: (i) conditions (or changes therein) in the generic pharmaceutical or over-the-counter pharmaceutical businesses that do not disproportionately affect the Acquired Assets or the Business, (ii) general legal, tax, economic, political or regulatory conditions (or changes therein) in the markets in which the acquired Business operates, including any changes affecting financial, credit or capital market conditions that do not disproportionately affect the Acquired Assets or the Business, (iii) any generally applicable change in Law or U.S. GAAP or interpretation of any of the foregoing that do not disproportionately affect the Acquired Assets or the Business, (iv) the announcement of, or the pendency of, this Agreement or consummation of the Transactions to the extent shown by Seller Parent to be the proximate cause therefrom, including to the extent so shown (A) the identity of Buyer Parent, (B) any departure or termination of any officers, directors, employees or independent contractors of the Seller Group, (C) the termination or potential termination of (or the failure or potential failure to renew or enter into) any Contracts with customers, suppliers, distributors or other business partners (provided that this subclause (C) shall not apply with respect to the representation of Sellers set forth in Section 4.2(b) related to the execution, delivery and performance by each Seller of this Agreement not resulting in a material breach or a material fault under any material Contract), and (D) any other negative development (or potential negative development) in the Seller Groups' relationships with any of its customers, suppliers, distributors or other business partners, (v) conditions arising out of acts of terrorism or sabotage, war (whether or not declared), the commencement, continuation or escalation of a war, acts of armed hostility, weather conditions, natural disasters or other force majeure events that do not disproportionately affect the Acquired Assets or the Business, (vi) any determination by, or delay of a determination by, the FDA or any other Governmental Authority, or any panel or advisory body empowered or appointed thereby, or any indication that any such entity, panel or body will make any determination or delay in making any determination, with respect to the approvability, labeling, contents of package insert, prescribing information, risk management profile, CMC matters, pre-approval inspection matters or requirements relating to the results of any pre-clinical or clinical testing that are part of the Business or Acquired Assets, (vii) any actions taken or failure to take action, in each case, by Buyer Parent or any of its Controlled Affiliates, or to which an officer of Buyer Parent has consented in writing, or which an officer of Buyer Parent has requested, or the taking of any action required by the express terms of this Agreement (other than any requirement to operate in the Ordinary Course of Business, or the failure to take any action prohibited by the express terms of this Agreement, (viii) the failure to meet any financial or other plan, estimate, budget or projection (provided that the underlying cause of such failure, to the extent not otherwise excluded herein, may be taken into account in making a determination as to whether a Seller Material Adverse Effect has or would reasonably be expected to occur) and (ix) any Proceedings made or brought by any of the current or former stockholders of the Seller Parent (on their own behalf or on behalf of the Seller Parent) in their capacity as stockholders against the Seller Parent arising out of the Transactions, and (x) any item set forth in the Sellers Disclosure Letter to the extent the relevance of such disclosure is reasonably apparent.

"Seller Material Contract" has the meaning set forth in Section 4.10(b).

- "Seller Parent" has the meaning set forth in the Preamble hereof.
- "Seller Parent Reference Balance Sheet" has the meaning set forth in Section 4.22.
- "Seller Parent Cash Incentive Award" means an incentive award payable in cash granted under a Seller Parent Incentive Plan.
- "Seller Parent Certificate of Incorporation" has the meaning set forth in Section 4.1(a).
- "Seller Parent Incentive Plan" means all equity incentive award plans maintained by Seller Parent or its Subsidiaries.
- "Seller Parent Incentive Award" means a Seller Parent Option, a Seller Parent Restricted Share Award, a Seller Parent Time-Vesting RSU Award, a Seller Parent Performance RSU Award or a Seller Parent Cash Incentive Award.
- "Seller Parent's Knowledge" means the actual knowledge of the persons listed on Schedule 1.1(d) or such knowledge that would reasonably be obtained in the course of diligently executing the responsibilities of their positions.
 - "Seller Parent Option" means an option to purchase shares of Seller Parent Stock granted under a Seller Parent Incentive Plan.
 - "Seller Parent Permitted Lien" has the meaning set forth in Section 4.17(a).
 - "Seller Parent Restricted Share Award" means an award of restricted shares of Seller Parent Stock granted under a Seller Parent Incentive Plan.
- "Seller Parent RSU Award" means an award of restricted stock units or performance units covering shares of Seller Parent Stock granted under a Seller Parent Incentive Plan.
 - "Seller Parent Stock" means the ordinary shares of Seller Parent, \$0.0001 per share.
- "Seller Parent VWAP" means as of a certain date, the volume weighted average price for the Seller Parent Stock, as shown by Bloomberg (or, if Bloomberg ceases to publish such price, any successor service reasonably agreed by Seller Parent and Buyer Parent), for the immediately prior twenty (20) trading days.
 - "Seller Related Parties" has the meaning set forth in Section 13.10(d).
 - "Seller-Requested Section 338 Election" has the meaning set forth in Section 8.5.
 - "Seller's Objection" has the meaning set forth in Section 3.3(c).
 - "Sellers Disclosure Letter" has the meaning set forth in the introductory paragraph to Article IV.
 - "Separated Contract" has the meaning set forth in Section 6.6(a).

"September 2015 Balance Sheet" means the unaudited balance sheet as of September 30, 2015 of the Transferred Entities and the Acquired Assets delivered pursuant to Section 9.14, and as amended or interpreted in accordance with the Accounting Principles.

"September 2015 Working Capital Amount" means the Current Assets less the Current Liabilities prepared from the associated line items set forth on the September 2015 Balance Sheet (provided that to the extent the line items on the September 2015 Balance Sheet are different from the line items included in the definition of Current Assets or Current Liabilities, the intent is to include line items from the September 2015 Balance Sheet that are similar in nature to the line items included in the definitions of Current Assets and Current Liabilities).

"Shared Business Contracts" means each Business Contract that relates to both the Business and the Retained Business other than any wholesale supply agreement for pharmaceutical products in the United States or Europe which shall be retained by the Seller Parent.

"Shared IP License Agreements" means each Contract to which Seller Parent or any of its Affiliates is a party under which Intellectual Property that is used and/or anticipated to be used in both the Business and the Retained Business is licensed (a) from Seller Parent or any of its Controlled Affiliates to any other Person or (b) to Seller Parent or any of its Controlled Affiliates from any other Person.

"Shared Transfer Taxes" has the meaning set forth in Section 8.7.

"Significant Subsidiary" means any Subsidiary of Seller Parent or Buyer Parent, as applicable, that is material or constitutes a "significant subsidiary" of Seller Parent or Buyer Parent, as applicable, within the meaning of Rule 1-02 of Regulation S-X promulgated under the Securities Act.

"Special Contract" has the meaning set forth in Section 6.5.

"Specialty Products" means the following products of Buyer Parent: Copaxone, Azilect, Nuvigil, Treanda and Fentura.

"Stock Consideration" means a number of shares of Buyer Parent Stock equal to the quotient of \$6,750,000,000 divided by the Buyer Parent VWAP of Buyer Parent Stock as of the end of the New York Stock Exchange trading day on July 31, 2015.

"Stockholders Agreement" means the stockholders agreement to be entered into by Seller Parent and Buyer Parent at the Closing in the form attached hereto as Exhibit A.

"Straddle Period" means any Tax period beginning before or on and ending after the Closing Date. With respect to Taxes relating to a Straddle Period, the portion of any Tax that is allocable to the taxable period that is deemed to end on the Closing Date will be: (i) in the case of Property Taxes, deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of calendar days of such Straddle Period in the Pre-Closing Tax Period and the denominator of which is the number of calendar days in the entire Straddle Period, and (ii) in the case of all other Taxes, except Transfer Taxes, determined as though the taxable year of the Company terminated at the close of business on the Closing Date.

"Subsidiary" or "Subsidiaries" means, with respect to any Person, any other Person of which (i) if a corporation, a majority of the total voting power of shares of capital stock entitled (without regard to the occurrence of any contingency) to vote in the matters to be approved by equity holders of such corporation, including the election of directors, managers, or trustees thereof, is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, (ii) if a limited liability company, partnership, association or other business entity (other than a corporation), either the managing member or general partner or a majority of partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more other Subsidiaries of that Person or a combination thereof and for this purpose, a Person or Persons owns a majority ownership interest in such a business entity (other than a corporation) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be any managing director or general partner of such business entity (other than a corporation) or control any managing director or general partner of such business entity (other than a corporation) or (iii) is otherwise contractually entitled to direct and control.

"Tax Contest Claim" has the meaning set forth in Section 8.2.

"Tax-Sharing Agreement" means any arrangement, agreement, or other binding obligation which provides for the allocation, sharing, indemnity or reimbursement of Taxes (other than any customary Tax indemnification provisions in ordinary course commercial arrangements, agreements or other binding obligations that are not primarily related to Taxes).

"Tax Return" means any report, return, certificate, declaration, election, report, Claim for refund or information return or statement required to be filed with any Governmental Authority or domestic or foreign taxing authority and with respect to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Taxes" means any taxes, levies, duties, tariffs, imposts and other similar charges, assessments and fees imposed by any Governmental Authority or domestic or foreign taxing authority, including U.S. federal, state, local or non-U.S. income, gross receipts, branch profits, license, payroll, employment, excise, severance, stamp, notarization, registration, land value gains, occupation, premium, windfall profits, escheat, environmental, customs duties, capital stock, franchise, profits, withholding, social security and aligned contribution, unemployment, disability, real property, personal property, sales, use, transfer, registration, ad valorem, value added, alternative or add-on minimum or estimated tax, including any interest, penalty or addition thereto, whether disputed or not.

"Technical Information" means all documents, drawings, specifications and any other documented information of a technical nature regarding the Products wherever located which, in each case: (i) are owned or controlled by a Seller or any Affiliate of any Seller or to which Seller or such Affiliate has regular access or use and (ii) are related to the use, design, development, validation, materials and components, biological compatibility, manufacture, processing, testing, storage, packaging, Labeling, regulations, safety, quality or performance of the Products.

- "Third Party Claim" has the meaning set forth in Section 12.4(a).
- "Transactions" has the meaning set forth in the recitals hereof.
- "Transfer Documents" means, collectively, any and all agreements, assignments, deeds, notarial forms, certificates and other instruments in forms attached as exhibits to the Local Transfer Agreements and other instruments of sale, conveyance, transfer, assignment and/or assumption, as the case may be, between a Seller and a Buyer as necessary under the Law of the relevant jurisdiction or contemplated by this Agreement and the applicable Local Transfer Agreement in order to transfer all right, title and interest of such Seller in and to the Acquired Assets, and for the Assumed Liabilities to be effectively assumed by and transferred to such Buyer, in accordance with the terms hereof and, to the extent not inconsistent with the terms hereof, necessary for the applicable Local Transfer Agreement, including any and all bills of sale, assignment and assumption agreements and patent, trademark and copyright assignments.
 - "Transfer Taxes" has the meaning set forth in Section 8.7.
 - "Transferred Assets" has the meaning set forth in Section 2.1(a)
- "Transferred Brand Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted as of the date hereof and at Closing of Sellers or any of their respective Affiliates in connection with: (a) the marketing, promoting, using, licensing, selling or offering for sale of the Transferred Brand Products; and (b) packaging, storing or transporting of Transferred Brand Products for any third parties.
 - "Transferred Brand Products" means those products listed on Schedule 1.1(e).
 - "Transferred Entities" means the entities listed on Schedule 1.1(f).
 - "Transferred Entity Benefit Plan" has the meaning set forth in Section 4.8(a).
 - "Transferred Group" means, individually and collectively, each of the Transferred Entities and each of their direct and indirect Subsidiaries.
 - "Transferred Group Assets" has the meaning set forth in Section 4.5.
 - "Transferred Intellectual Property" has the meaning set forth in Section 2.1(b)(vii).
 - "Transferred Leased Real Property" has the meaning set forth in Section 4.17(b).
 - "Transferred Owned Real Property" has the meaning set forth in Section 4.17(a).
- "Transferred Receivables" means the AUK Intra Group Debt, APH4 Intra Group Debt, the PTC Intra Group Debt and the AGH Intra Group Debt, none of which will be deemed Current Assets for purposes of any working capital calculation.
 - "Transferred Shares" means all of the outstanding capital stock of the Transferred Entities.

- "Transition Agreements" means the Transition Services Agreement, the Manufacturing Agreements and the Distribution Agreement.
- "Transition Period" has the meaning set forth in Section 9.9(b).
- "Transition Services Agreement" means the transition services agreement to be entered into at Closing in a form to be mutually agreed upon by the Parties.
 - "Treasury Regulations" means the regulations promulgated under the Code.
 - "United States" and "U.S." mean the United States of America (including its territories and possessions).
 - "U.S. GAAP" means accounting principles generally accepted in the United States, consistently applied.
- "U.S. Generics Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted as of the date hereof and at Closing of Sellers or any of their respective Affiliates in connection with the (a) researching, developing, submitting for approval, manufacturing, labeling distributing, packaging, storing, or transporting the U.S. Generics Products; and (b) marketing, promoting, using, licensing, selling or offering for sale of the U.S. Generics Products.
- "U.S. Generics Products" means, collectively, any and all pharmaceutical products of Sellers or any of their respective Controlled Affiliates that as of the date hereof and at Closing are sold or intended to be sold in the United States as Generic Drugs (including branded generic pharmaceutical products and generic over the counter products, but excluding biologics or other biosimilar product), including authorized generic pharmaceutical products that have been launched or available for launch on or prior to Closing and all such products or other assets that (i) are being or have been researched, tested, developed, commercialized, manufactured, sold, promoted or distributed by Sellers or any of their respective Affiliates in the United States as of the date hereof and at Closing, including those listed on Schedule 1.1(g), or (ii) are owned by, licensed to or otherwise used pursuant to a Contract by the Sellers or any of their Affiliates or introduced into commerce by any of the Sellers or their respective Affiliates as of the date hereof and at Closing.
- "UK Group Relief" means (i) group relief capable of being surrendered or claimed pursuant to Part 5 of the UK Corporation Tax Act 2010, or (ii) the notional transfer of an asset or reallocation of a gain or loss pursuant to section 171A or section 179A of the UK Taxation of Chargeable Gains Act 1992 and the notional reallocation of a gain pursuant to section 792 of the UK Corporation Tax Act 2009.
 - "U.S. Trade Laws" has the meaning set forth in Section 4.12(h).
- "Uteron Business" means the assets, business, Know-How, operations and activities (including related personnel/employees) conducted at any time by Uteron Pharma SPRL and each of its Subsidiaries.

"VAT" means value added tax imposed in any member state of the European Union pursuant to the VAT Directive, any other sales or turnover tax of a similar nature imposed in any country that is not a member of the European Union or any tax of a similar nature which may be substituted for or levied in addition to it, together with all penalties or interest thereon.

"VAT Directive" means EC Council Directive 2006/112/EC of 28 November 2006 on the common system of value added tax and national legislation implementing that Directive or any predecessor to it or supplemental to that Directive.

- 1.2 Construction. Unless expressly specified otherwise, whenever used in this Agreement, the terms "Article," "Exhibit," "Schedule" and "Section" refer to articles, exhibits, schedules and sections of this Agreement (and, for the avoidance of doubt, do not refer to appendices, articles, sections, schedules and exhibits of any Ancillary Agreement). Whenever used in this Agreement, the terms "hereby," "hereof," "herein" and "hereunder" and words of similar import refer to this Agreement as a whole, including all articles, sections, schedules and exhibits hereto. Whenever used in this Agreement, the terms "include," "includes" and "including" mean "include, without limitation," "includes, without limitation" and "including, without limitation," respectively. Whenever the context of this Agreement permits, the masculine, feminine or neuter gender, and the singular or plural number, are each deemed to include the others. "Days" means calendar days unless otherwise specified. Unless expressly specified otherwise, all payments to be made in accordance with or under this Agreement (or any Ancillary Agreement) shall be made in U.S. Dollars (USD\$). References in this Agreement to particular sections of a Law shall be deemed to refer to such sections or provisions as they may be amended after the date of this Agreement. The Parties have participated jointly in the negotiation and drafting of this Agreement and in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party (or any Affiliate thereof) by virtue of the authorship of any of the provisions of this Agreement.
- 1.3 <u>Control</u>. In the event of a conflict between this Agreement and any Local Transfer Agreement, unless such Local Transfer Agreement expressly provides otherwise with reference to this <u>Section 1.3</u>, this Agreement shall control to the extent permitted by applicable Law, and to the extent not so permitted, the applicable Local Transfer Agreement shall control.
- 1.4 <u>Performance of Obligations by Affiliates</u>. Any obligation of a Seller under or pursuant to this Agreement may be satisfied, met or fulfilled, in whole or in part, at Seller Parent's sole and exclusive option, either by Seller Parent directly or by any Affiliate or designee of Seller Parent that Seller Parent causes to satisfy, meet or fulfill such obligation, in whole or in part. Any obligation of Buyer Parent under or pursuant to this Agreement may be satisfied, met or fulfilled, in whole or in part, at Buyer Parent's sole and exclusive option, either by Buyer Parent directly or by any Affiliate or designee of Buyer Parent that Buyer Parent causes to satisfy, meet or fulfill such obligation, in whole or in part. With respect to any particular action, the use of the words "Seller Parent shall," "Seller Parent agrees to," "Sellers shall," a "Seller shall," and any similar variation with respect to any action, also means "Seller Parent shall," and any similar variation with respect to any action, also means "Buyer Parent shall," and any similar variation with respect to any action, also means "Buyer Parent shall cause" the particular action to be performed,

because Seller Parent and Buyer Parent each understand, agree and acknowledge that they are entering into this Agreement on behalf of themselves and certain of their respective Affiliates. Seller Parent guarantees the performance of all actions, agreements and obligations to be performed by any Subsidiaries or Affiliates of Seller Parent under the terms and conditions of this Agreement and any applicable Local Transfer Agreement, and Buyer Parent guarantees the performance of all actions, agreements and obligations to be performed by any Affiliates of Buyer Parent under the terms and conditions of this Agreement, regardless of whether or not Buyer Parent and/or Seller Parent are a party thereto. In case Seller Parent undertakes any actions, agreements and obligations that should be performed by any Subsidiaries or Affiliates of Seller Parent under the terms and conditions of this Agreement and any applicable Local Transfer Agreement, Seller Parent shall be deemed to be acting on behalf of such Subsidiaries or Affiliates. Buyer Parent shall reasonably promptly identify the applicable Buyer to be a party to each applicable Local Transfer Agreement prior to the execution of such Local Transfer Agreement and such Buyer, to the extent required under applicable Law, shall be an entity qualified to do business, or otherwise organized, in the applicable jurisdiction, at the closing of the applicable Local Transfer Agreement.

ARTICLE II PURCHASE AND SALE

- 2.1 Agreement to Purchase and Sell. At the Closing, in accordance with and pursuant to the terms and conditions of this Agreement, for the consideration stated in Section 3.2(a)(ii) and Section 2.7, Sellers shall, and shall cause their respective Affiliates to, grant, sell, transfer, convey, assign and deliver to Buyers, and Buyers shall purchase and accept from Sellers or any of their respective Affiliates, all right, title, and interest of Sellers and their respective Affiliates, as of the Closing, in and to the following (collectively, the "Acquired Assets"):
 - (a) the Transferred Shares, free and clear of all Liens; and
 - (b) the Transferred Assets, free and clear of all Liens other than Permitted Liens, comprising of:
 - (i) all Products;
 - (ii) all Inventory;
 - (iii) the Transferred Receivables;
 - (iv) the Transferred Owned Real Property and the Transferred Leased Real Property;

(v) all of the rights to the fixed and other tangible personal property and equipment, including materials, prototypes, tools, supplies, vehicles, furniture, fixtures, improvements to the property and other tangible assets Related to the Business whether owned or leased by Sellers or any of their Affiliates (collectively, the "Equipment"); provided, however, that any such Equipment necessary for Seller Parent to provide any services under a Transition Agreement shall be transferred to Buyers at such time(s) set forth in the applicable Transition Agreement;

(vi) all IT Assets Related to the Business; provided that rights to such IT Assets do not affect rights to the data or information that may be contained in or be processed by or using such IT Assets;

(vii) the patents and patent applications owned by Sellers or their Controlled Affiliates which are Related to the Business including those listed on Schedule 2.1(b)(vii) according to owner on the date hereof (the "Patents"), (ii) the patents in-licensed by Sellers or their Affiliates which are Related to the Business according to licensee on the date hereof (the "Licensed Patents"), (iii) the internet domain names, trademarks and service marks, and all applications and registrations for the foregoing, owned by Sellers or their Controlled Affiliates and Related to the Business, together with all common law rights associated with the trademarks and service marks which are the subject of such registrations and applications and the goodwill associated therewith (the "Marks"), (iv) the trademarks and service marks in-licensed by Sellers or their Controlled Affiliates and Related to the Business according to licensee on the date hereof, together with all common law rights associated with the trademarks and service marks which are the subject of any and all registrations and applications and the goodwill associated therewith (the "Licensed Marks"), (v) copyrights in (A) all design history files described in Section 2.1(b)(x), (B) the Manufacturing Instructions, (C) the Technical Information and (D) all Promotional Activities, (vi) trade dress, logos, packaging design, and slogans, copyrights in both published and unpublished works, including all compilations, in each case, solely if Related to the Business or primarily related to the Products, (vii) customized databases and customized computer programs used to operate Equipment, manuals and other documentation and all copyrights and applications thereof, and all derivatives, translations, adaptations and combinations thereof, in each case, if Related to the Business or primarily related to the Acquired Assets, (viii) the Know-How, (ix) all other Intellectual Property Related to the Business or primarily related to the Acquired Assets, (x) all copies and tangible embodiments thereof of each of the foregoing (in whatever form or medium), and (xi) all rights to sue at law or in equity for all Claims or causes of actions arising out of or related to any past, present or future infringement, misappropriation or violation of any of the foregoing, including the right to receive all proceeds and damages therefrom; (xii) the Licensed Manufacturing IP (all of the foregoing in this Section 2.1(b) (vii), together with the Manufacturing Instructions and the Technical Information, the "Transferred Intellectual Property");

(viii) subject to Section 2.5, the Business Contracts, and all rights, benefits and interests thereunder; provided, however, that (i) such Contracts which are Shared Business Contracts shall be subject to the provisions of Section 6.6 and (ii) such Business Contracts which Seller Parent requires in order to provide transition services to Buyers pursuant to a Transition Agreement shall be transferred or assigned to Buyers at such time(s) set forth in the applicable Transition Agreement;

(ix) (i) the Permits (including any applications that are in process), and (ii) the Regulatory Registrations (including any applications that are in process), supported by and including: (x) the original documents under the possession of Sellers and their Controlled Affiliates evidencing the Regulatory Registrations issued to and held by Sellers and their Controlled Affiliates by the Regulatory Authorities (or if the original is not available, copies of the portions thereof related to the Products); (y) all related Regulatory Documentation; provided, however, that if any Regulatory Documentation also covers the manufacturing, marketing or sale

of other products of Sellers or their Affiliates, Sellers may elect to redact those portions of Regulatory Documentation that pertain to such other products, or deliver copies of such materials unredacted but subject to the confidentiality provisions of this Agreement; and (z) all of Sellers' and their Controlled Affiliates' rights of reservation in any Regulatory Registrations under any agreement pursuant to which any Regulatory Registrations are held in the name of a third party; provided, however, that in each of (i) and (ii) above, any Permit or Regulatory Registration that is necessary for Seller Parent to provide any services under a Transition Agreement shall be transferred to Buyers at such time(s) set forth in the applicable Transition Agreement. Sellers shall deliver to Buyers the originals or, if applicable, copies of the Regulatory Registrations issued to and held directly by Sellers and their Controlled Affiliates (x) as soon as possible after the Closing Date, with respect to the Regulatory Registrations issued to and held directly by Sellers that Sellers do not need to retain to perform their respective obligations under the Ancillary Agreements; and (y) at such time(s) set forth in the applicable Transition Agreement with respect to Regulatory Registrations issued to and held directly by Sellers that Sellers need to retain to perform their respective obligations under the Transition Agreements;

(x) copies of the design history files with respect to the Products; provided, however, that if any design history files also covers the design history files of other products of Sellers or any of their Controlled Affiliates, Sellers may elect to redact only those portions that pertain only to such other products and not to the Products, or deliver copies of the design history files unredacted but subject to the confidentiality provisions of this Agreement;

(xi) the Manufacturing Instructions and Technical Information, and Sellers shall deliver to Buyers copies of the Manufacturing Instructions and Technical Information as soon as practicable after the Closing Date, but in any event no later than sixty (60) days after the Closing Date;

(xii) all Books and Records, provided, however, that if any Books and Records contain any information of Sellers or any of their Affiliates not related to the Business or the employment of the Business Employees, Sellers may elect to redact those portions of such Books and Records to the extent pertaining to such other information or, in Sellers' sole and absolute discretion, Sellers may deliver un-redacted copies of such Books and Records containing information not related to the Business or the employment of Business Employees but such information shall be subject to the confidentiality provisions of this Agreement, shall remain the property of Sellers, and Buyers shall have no rights with respect to such information;

(xiii) each human clinical trial study report, if any, conducted or sponsored by Seller or any Affiliate of Seller or submitted by Seller or any Affiliate of Seller to the FDA or similar Regulatory Authority with respect to the Products;

(xiv) product Labeling, product advertising, marketing and promotional materials, sales training materials and all other materials Related to the Business;

(xv) all Claims (including under any express or implied warranties, guarantees or indemnities), causes of action, choses in action, rights of recovery and rights of set-off of any kind (including the right to sue and recover for past infringements or misappropriations of Transferred Intellectual Property), in each case to the extent arising from the Business or related to any Acquired Asset or Assumed Liability;

(xvi) any Insurance Proceeds;

(xvii) Cash, to the extent included in the calculation of Closing Net Cash;

(xviii) all assets related to the Transferred Entity Benefit Plans (including any assets underlying any Transferred Entity Benefit Plans that are defined benefit pension plans);

- (xix) all goodwill of the Business as a going concern;
- (xx) the Current Assets at Closing;
- (xxi) those assets listed on Schedule 2.1(b)(xxi); and

(xxii) any other asset, property or right of Sellers and their respective Controlled Affiliates Related to the Business, whether tangible or intangible, real, personal or mixed,

in each case only to the extent such items are not already transferred by way of the sale of the Transferred Shares.

- 2.2 Excluded Assets. Notwithstanding anything to the contrary in this Agreement, Sellers shall not, nor shall Sellers cause any of their Affiliates to, sell, transfer or assign, and Buyers and Buyers' Affiliates shall not, nor shall Buyers or Buyers' Affiliates have any right to, purchase or otherwise acquire, any right, title or interest in any of the following assets, properties, rights or interests of Sellers or any of Sellers' Affiliates assets, which are expressly excluded from the Acquired Assets and are not to be acquired by Buyer Parent pursuant to this Agreement (the "Excluded Assets"):
 - (a) any assets expressly excluded from the definition set forth in Sections 2.1(b)(i) through 2.1(b)(xxii);
 - (b) the Retained Business;

Transactions:

- (c) any shares or other equity or other ownership interests in any Affiliates of Seller Parent other than the Transferred Group;
- (d) rights of Sellers and Sellers' Affiliates arising under this Agreement, the Ancillary Agreements or from the consummation of the
- (e) rights to refunds of Taxes related to the Transferred Assets for any Pre-Closing Tax Period;
 - (f) any wholesale supply agreement for pharmaceutical products;

- (g) cash, cash equivalents, bank deposits and marketable securities on hand and in transit of the Seller Parent and its Affiliates (excluding the Transferred Group);
- (h) all assets, information and Know-how relating to the research and development activities of the Transferred Group to the extent relating to the Brand R&D Assets; and
 - (i) any assets listed on Schedule 2.2(i).
- 2.3 <u>Assumed Liabilities</u>. Upon the Closing, in accordance with and pursuant to the terms and conditions of this Agreement and the Ancillary Agreements, Buyers shall assume only the following Liabilities and Claims of Seller Parent and its Controlled Affiliates, other than Excluded Liabilities referred to in <u>Section 2.4(a)</u> through (h) (the "Assumed Liabilities"):
 - (a) all Liabilities and Claims of the Transferred Group;
 - (b) all Liabilities and Claims of the Business and the Acquired Assets arising after the Closing;
 - (c) the Current Liabilities at Closing;
- (d) all Liabilities and Claims which would not be imposed on Seller Parent, Sellers, or any of their Affiliates but for the Buyer-Requested Modifications, including any associated costs, fees, Taxes or expenses relating to the Buyer-Requested Modifications;
- (e) all Liabilities for (i) Taxes with respect to the Acquired Assets for any Post-Closing Tax Period, including Property Taxes allocable to Buyer pursuant to Section 8.6, (ii) Taxes with respect to any Divestiture Action, and (ii) Buyers' share of Transfer Taxes as described in Section 8.7;
 - (f) all Liabilities or Claims listed on Schedule 2.3(f); and
- (g) all Liabilities and Claims to the extent related to (i) the Transferred Entity Benefit Plans (including, for the avoidance of doubt, to the extent related to Former Business Employees), (ii) to any Business Employee or (iii) to the extent solely related to the Business (as mutually agreed upon in good faith by Buyer Parent and Seller Parent) and to the extent reflected in the Financial Statements any Former Business Employee, other than Liabilities set forth in Section 2.4(f).
- 2.4 Excluded Liabilities. Neither Buyers nor any of their Affiliates shall assume, nor shall they be or become responsible for, any Liabilities and Claims of the Business or of Sellers or any of Sellers' Affiliates other than the Assumed Liabilities (collectively, the "Excluded Liabilities"). Without limiting the generality of the foregoing, the following shall constitute the Excluded Liabilities notwithstanding any other provision of this Agreement:
- (a) all Liabilities for (i) Taxes with respect to the Transferred Assets for any Pre-Closing Tax Period, including Property Taxes allocable to Sellers pursuant to Section 8.6, (ii) Taxes imposed on any member of the Transferred Group pursuant to Treasury Regulations

Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Law) by reason of the Liability of any Person other than any such member, and (iii) Sellers' share of Transfer Taxes as described in Section 8.7;

- (b) all Liabilities and Claims of the Retained Business;
- (c) all Liabilities and Claims which would not be imposed on Buyer Parent, Buyers, or any of their Affiliates (including any members of the Transferred Group) but for the Pre-Closing Reorganization and any Seller-Requested Section 338 Election, including any associated costs, fees, Taxes or expenses relating to the Pre-Closing Reorganization and any Seller-Requested Section 338 Election;
- (d) Liabilities and Claims of Transferred Group to the extent not Related to the Business or (ii) that arise out of, relate to or are in connection with matters arising under the securities laws or Claims brought by its securityholders or in the name of a Transferred Entity by its securityholders, in each case with respect to or to the extent relating to periods prior to the Closing;
- (e) all Liabilities and Claims related to the Seller Benefit Plans or to any employees of the Sellers or their Subsidiaries other than the Business Employees or, to the extent specified in Section 2.3(f), the Former Business Employees;
 - (f) all Liabilities under the Transferred Entity Benefit Plans with respect to Non-Business Employees;
 - (g) all Liabilities and Claims that are transaction expenses arising out of this Agreement and the Transactions; and
 - (h) Liabilities and Claims set forth on Schedule 2.4(h).

2.5 Procedures for Assignments.

- (a) Notwithstanding anything to the contrary contained herein, neither this Agreement nor any Local Transfer Agreement shall constitute an agreement to assign or transfer any Transferred Asset if an assignment or transfer thereof, without the Consent of a Person, would constitute a breach or violation thereof and such Consent is not obtained at or prior to the Closing. If the Parties are not successful in obtaining any Consent at or prior to the Closing, then the parties agree that on and after the Closing, the Sellers will use reasonable best efforts to obtain such Consent; provided, however, that in no event shall any Seller be required to pay any monies to obtain such Consent after the Closing other than filing, recordation or similar fees. The fact of a failure to obtain any such Consent shall not result in a breach of this Agreement in any manner.
- (b) If the Parties are not successful in transferring or assigning any Transferred Asset, right, benefit or obligation thereunder or resulting therefrom at or prior to the Closing in accordance with Section 2.5(a), and the Closing proceeds without such transfer or assignment, then (i) following the Closing and pending the receipt of such Consent, the Parties shall cooperate with each other in any mutually agreeable, reasonable and lawful arrangements

designed to provide Buyer Parent with the benefits of Transferred Asset and any right, benefit or obligation thereunder or resulting therefrom as if the appropriate Consent had been obtained, including provision of the consideration and other economic benefits to be received by Buyer Parent in and under every Transferred Asset and right, benefit or obligation thereunder or resulting therefrom, which consideration shall be held for the benefit of, and shall be promptly delivered to, Buyer Parent; and refrain from agreeing to any amendment, supplement, waiver or other modification of any Transferred Asset, right, benefit or obligation thereunder or resulting therefrom, as applicable, without the prior written consent of Buyer Parent. To the extent that the transfer of any Transferred Asset, right, benefit or obligation thereunder requires payment of additional fees, costs or expenses to third parties, such fees, costs and expenses shall be borne by Seller Parent. Once the applicable Consent for the assignment or transfer of any such asset not assigned or transferred at the Closing is obtained, Sellers shall, or shall cause their respective relevant Controlled Affiliates to, assign and transfer such asset to Buyer Parent at no additional cost. To the extent that any such asset cannot be transferred or the full benefits of use of any such asset cannot be provided to Buyer Parent following the Closing pursuant to this Section 2.5, then Buyer Parent and Seller Parent shall enter into such arrangements (including subleasing, sublicensing or subcontracting) to provide to the Parties the economic (taking into account Tax costs and benefits) and operational equivalent, to the extent permitted, of obtaining such Consent (provided that, with respect to any such asset that is a Contract, such obligation shall continue for only so long as the applicable Contract is in effect). Seller Parent shall hold in trust for and pay to Buyer promptly upon receipt thereof, all income, proceeds and other monies received by Sellers or any of its Affiliates in connection with its use of any asset in connection with the arrangements under this Section 2.5. To the extent that Buyer Parent is provided the benefits of any Transferred Asset, right, benefit or obligation thereunder or resulting therefrom referred to herein (whether from Sellers or otherwise) as if the appropriate Consent had been obtained, Buyer Parent shall arrange to discharge and perform the liabilities and obligations of Sellers thereunder or in connection therewith, as applicable, to the same extent as if the appropriate Consent had been obtained.

2.6 Allocation of Global Purchase Price, Assumed Liabilities, Cash Consideration and Stock Consideration.

(a) The Global Purchase Price, as adjusted pursuant to Section 3.3, plus the Assumed Liabilities (to the extent treated as consideration for the Acquired Assets for applicable Tax purposes) shall be allocated among the Acquired Assets consistent with applicable law applied in the manner initially determined by Seller Parent (the "Global Purchase Price Allocation"). As soon as reasonably practicable following the Closing Date, and in any event within ninety (90) days after the purchase price adjustments described in Section 3.3 have been determined, Seller Parent shall deliver to Buyer Parent the Global Purchase Price Allocation for Buyer Parent's review and comment. If Buyer Parent has comments to such proposed Global Purchase Price Allocation, it shall deliver written notice of such comments to Seller Parent within thirty (30) Business Days after the delivery of such proposed Global Purchase Price Allocation, setting forth in reasonable detail the basis for such comments, with such comments not to be unreasonably rejected by Seller Parent.

(b) The Cash Consideration and the Stock Consideration shall be allocated among the Acquired Assets consistent with applicable Law, applied in the manner initially

determined by Seller Parent (the "Consideration Allocation"). As soon as reasonably practicable following the Closing Date, and in any event within ninety (90) days after the Closing, Seller Parent shall deliver to Buyer Parent the Consideration Allocation for Buyer Parent's review and comment. If Buyer Parent has comments to such proposed Consideration Allocation, it shall deliver written notice of such comments to Seller Parent within fifteen (15) Business Days after the delivery of such proposed Consideration Allocation, setting forth in reasonable detail the basis for such comments, with such comments not to be unreasonably rejected by Seller Parent.

- (c) Any adjustments to any amounts treated as purchase price for applicable Tax purposes after the Global Purchase Price Allocation has been determined pursuant to Section 2.6(a) and any Earn-out Payment shall be allocated among the Acquired Assets in the manner consistent with the principles set forth on the Global Purchase Price Allocation.
- (d) Buyers and Sellers shall timely file all Tax Returns required to be filed (including any supplemental filings to reflect any revisions to the Global Purchase Price Allocation) and issue all necessary invoices consistent with the Global Purchase Price Allocation, as adjusted for any contingent payments made pursuant to Section 2.6. No Buyer or Seller shall take any Tax position inconsistent with such Global Purchase Price Allocation and no Buyer or Seller shall agree to any proposed adjustment to the Global Purchase Price Allocation by any Governmental Authority without first giving the other party prior written notice; provided, however, if any Governmental Authority disputes the Global Purchase Price Allocation, the party receiving notice of the dispute shall promptly notify the other party hereto, and the parties shall cooperate in good faith in responding to such dispute in order to preserve the effectiveness of the Global Purchase Price Allocation; provided, further, that nothing contained herein shall prevent any Buyer or Seller from settling any proposed deficiency or adjustment by any Governmental Authority based upon or arising out of the Global Purchase Price Allocation, and no Buyer or Seller shall be required to litigate before any court any proposed deficiency or adjustment by any Governmental Authority challenging such Global Purchase Price Allocation. Sellers and Buyers shall deliver to the other, upon reasonable request, if applicable, any draft Tax Return form reporting results from the Global Purchase Price Allocation for the other's approval, which approval shall not be unreasonably withheld, conditioned or delayed. Each Party shall have fifteen (15) days to approve any such draft Tax Return.

2.7 Earn-Out.

- (a) <u>Earn-Out Payment</u>. The Earn-out Payment payable by Buyer Parent to the Sellers in respect of each Earn-out Period shall be an amount equal to 50% of all Total Lenalidomide Net Sales during such Earn-out Period. For the purposes of this <u>Section 2.7</u>, the following definitions shall apply:
 - (i) "Arrow" means Arrow International Limited;
- (ii) "Lenalidomide" means each of (i) "Product," as defined in the Lenalidomide Agreement; and/or (ii) any other product sold or commercialized by the Buyer or its Affiliates pursuant to ANDA No. 201452 (and any and all amendments supplements thereto);

(iii) "Lenalidomide Agreement" means the Product Supply Agreement between Arrow International Limited and Natco Pharma Limited, dated as of November 2009;

- (iv) "Natco Lenalidomide Net Sales" shall, for any Earn-out Period, be an amount equal to (i) the share of "Profit" (as calculated in accordance with the Lenalidomide Agreement); plus (ii) the share of proceeds (whether cash or fair market value of all non-cash consideration) received pursuant to any settlement to be allocated to Arrow pursuant to Clause 8.5 of the Lenalidomide Agreement minus (iii) any Litigation Costs borne by Arrow in accordance with the Lenalidomide Agreement minus (iv) any litigation expenses, including legal fees and disbursements, costs, awards, judgments, claims, settlements and decisions borne by Arrow arising out of any claim (other than claims covered by clause (iii)) arising from the marketing, promotion or sale of Lenalidomide in the Territory (as defined in the Lenalidomide Agreement);
 - (v) "Other Lenalidomide" means Lenalidomide, excluding "Product", as defined in the Lenalidomide Agreement;
- (vi) "Other Lenalidomide Net Sales" means (i) gross sales proceeds of Other Lenalidomide less such costs as would have been deducted from such sales as if they had been sales of "Product" under the Lenalidomide Agreement or otherwise would have been deducted pursuant to clause (iv) above (but disregarding the exclusion for claims covered by clause (iii) retained therein); plus (ii) the share of proceeds (whether cash or fair market value of all non-cash consideration) received by Buyer or its Affiliates pursuant to any other settlement received with respect to Other Lenalidomide; and
- (vii) "**Total Lenalidomide Net Sales**" means Natco Lenalidomide Net Sales and Other Lenalidomide Net Sales, *provided*, that in no event shall the amount of the Earn-out Payment be less than zero. Notwithstanding the foregoing, in the event that the Earn-out Payment would be less than zero for any particular Earn-out Period, then the absolute value of such negative amount shall be deducted from future Earn-out Amounts, if any.
- (b) <u>Payment</u>. Buyer Parent will make Earn-Out Payments to Seller Parent under this <u>Section 2.7</u> on a quarterly basis beginning with the First Earn-Out Period. Earn-Out Payments due with respect to an Earn-Out Period shall be paid to the Seller Parent within sixty (60) days following the end of that Earn-Out Period. Each Earn-Out Payment will be accompanied by a report setting forth, for each month by stock-keeping unit the volume of Lenalidomide sold, the gross sales proceeds of Lenalidomide and the Lenalidomide Net Sales (broken down by category) with respect to the applicable Earn-Out Period and, in each case, to the extent the Buyer Parent or its Affiliates are permitted to give such information. Any Earn-Out Payments paid to Seller Parent shall be deemed to be an increase in the Global Purchase Price.

(c) Financial Records; Audits.

(i) The Buyer Parent will, and will procure that the Buyer Group will, keep complete and accurate financial records in sufficient detail to permit the Buyer Parent to

confirm the accuracy of all Earn-Out Payments, and such records will be open (in such form as may be available or reasonably requested) to inspection for three (3) years following the end of an Earn-Out Period to which they pertain. Seller Parent will have the right, at its own expense to have an independent, certified public accountant, selected by it to perform a review of the financial records of the Buyer Parent and, to the extent the Buyer Parent or its Affiliates are permitted access under the Lenalidomide Agreement, any financial records provided pursuant to the Lenalidomide Agreement, in each case as applicable to Earn-Out Payments. The report of such accountant (the "Earn-Out Report") will be made available to both the Buyer Parent and the Seller Parent promptly upon its completion.

(ii) If within twenty (20) Business Days after receipt of the Earn-Out Report, Seller Parent or Buyer Parent notifies the other in writing of any disagreement or difference of opinion relating to the Earn-Out Report (the "Notice of Disagreement"), the Parties shall be deemed to have accepted the Earn-Out Report which shall become final and binding on the parties.

(iii) If either Party delivers a Notice of Disagreement in relation to any Earn-out Report, the Seller Parent and Buyer Parent shall negotiate in good faith to seek to reach agreement on the items and amounts identified in such Notice of Disagreement, and, if agreement in writing is reached between Seller Parent and Buyer Parent on all such items and amounts, then any relevant Earn-out Payments shall be adjusted in accordance with such agreement. If Buyer and Seller Parent do not reach agreement as to the disagreement or difference of opinion set out in a Notice of Disagreement, in each case within twenty (20) Business Days of the delivery of any Notice of Disagreement, either Buyer Parent or Seller Parent may, by notice to the other, require that the issues identified in the Notice of Disagreement be referred to the Reporting Accountants.

(iv) Where a dispute is referred to the Reporting Accountants under <u>Section 2.7(c)(iii)</u>, the Reporting Accountants shall be engaged by the Seller Parent and Buyer Parent:

A. on the terms set out in this Section 2.7;

B. to make a final determination within twenty (20) Business Days of the date on which the foregoing disputed matters are submitted to it and for this purpose shall submit a written report to the parties;

C. to give Buyer Parent and Seller Parent reasonable opportunity to make written representations and require that each of them supply the other with a copy of any written representations at the same time as they are made;

D. to permit each of Buyer Parent and Seller Parent to be present while oral submissions (if any) are being made by the other;

E. to act as experts (and not as arbitrators) in making their determination and their determination of any matter falling within their jurisdiction shall be final and binding on each of Buyer Parent and Seller

Parent save in the event of manifest error (when the relevant part of their determination shall be void and the matter shall be resubmitted to the Reporting Accountants by either party for correction as soon as reasonably practicable);

- F. not be entitled to determine the scope of their own jurisdiction;
- G. direct how their charges and expenses (including VAT) shall be borne as they shall direct at the time they make any determination; and
 - H. otherwise on such terms as shall be agreed between Seller Parent, Buyer Parent and the Reporting Accountants.

Notwithstanding the foregoing, any determination with respect to the calculation of "Profits" or "Litigation Costs" under the Lenalidomide Agreement shall be binding on Buyer Parent and Seller Parent for purposes of this Agreement, provided that any amendment or waiver of the operative provisions thereof and any agreement by Arrow, Buyer Parent or any Affiliate of Buyer with respect to the calculation of "Profits" or "Litigation Costs" pursuant to the Lenalidomide Agreement shall only be binding on Seller Parent if it has consented thereto.

(d) Covenants.

- (i) Buyer Parent undertakes to Seller Parent to procure that during the Earn-out Periods the Buyer Group shall use their commercially reasonable efforts consistent with generic pharmaceutical industry practices to maximize the amount of sales of Lenalidomide.
- (ii) Buyer Parent undertakes to Seller Parent not to, and to procure that no member of the Buyer Group shall, following Closing and during the Earn-out Periods: (a) agree to any termination of, amendment of, waiver of any provision of or right under; or (b) take any actions which could reasonably be expected to give rise to a right of Natco Pharma Limited to terminate, the Lenalidomide Agreement, in each case, without the prior written consent of the Seller Parent.
- (iii) Buyer Parent undertakes to Seller Parent to, and to procure each member of the Buyer Group shall, following Closing and during the Earn-out Periods, comply in all material respects with the provisions of the Lenalidomide Agreement and shall exercise any right to extend the term of the Lenalidomide which any member of the Buyer Group may have.
- (iv) In the event that Buyer Parent or any of its Subsidiaries disposes of its rights (whether by way of sale, transfer, license or other disposition) with respect to Lenalidomide (including by disposing the business of selling Lenalidomide or any material part thereof), Buyer Parent shall pay to Seller Parent an amount equal to 50% of the amount received by Buyer Parent or any of its Subsidiaries from the disposition of such rights, less any expenses incurred in connection with such transaction. To the extent that the amounts received in the transaction include non-cash consideration, Buyer Parent and Seller Parent shall negotiate in

good faith to agree on the fair value of such non-cash consideration. To the extent that the amounts received in the transaction include an earnout or other contingent consideration, Buyer Parent shall pay to Seller Parent an amount equal to 50% of all such amounts, if and when received.

2.8 Withholding. Buyers, Sellers and any of their respective Affiliates shall be entitled to deduct and withhold, or cause to be deducted and withheld, from amounts otherwise payable pursuant to this Agreement, any amounts as are required to be withheld or deducted with respect to such amounts under the Code, or any applicable provisions of state, local or non-U.S. Tax Law. Each Party shall provide commercially reasonable notice to the other Party upon becoming aware of any such withholding obligation and shall cooperate with such other Party to the extent reasonable to obtain reduction of or relief from such withholding or deduction. Except as required by Law, neither Buyers nor any of their respective Affiliates (including any of the members of the Transferred Group) shall take, or cause to be taken, any action with respect to any Earn-out Payment that would result in any additional withholding or deduction (other than resulting from a change in Law in the jurisdiction from which the Earn-out Payment is made immediately after the Closing Date) from such Earn-out Payment (any such amounts withheld or deducted, an "Incremental Withholding"), which would not have resulted absent a Buyer or any of its respective Affiliates (including any of the members of the Transferred Group) taking, or causing to be taken, such action. In the case of any Incremental Withholding resulting from such action, Buyers shall increase the Earn-out Payment otherwise payable pursuant to this Agreement as necessary so that after such Incremental Withholding has been made (including such withholdings and deductions applicable to additional sums under this Section 2.8) the recipient receives an amount equal to the Earn-out Payment it would have received had no such Incremental Withholding been made. To the extent that amounts are so withheld and timely remitted to the appropriate Governmental Authority, such withholding was made.

ARTICLE III

CLOSING

3.1 Closing. The closing of the Transactions (the "Closing") shall take place in accordance with and pursuant to the terms and conditions of this Agreement and the Local Transfer Agreements at the offices of Latham & Watkins LLP, 53rd at Third, 885 Third Avenue, New York, New York 10022-4834 no later than ten (10) Business Days following the satisfaction or waiver of all of the conditions set forth in Article X (other than conditions with respect to actions to be taken at the Closing itself) or at such other time, date or place as Seller Parent and Buyer Parent may mutually agree in writing (the "Closing Date"); provided, however, that Buyer Parent shall not be obligated to effect the Closing prior to the third Business Day following the final day of the Marketing Period or such earlier date as Buyer Parent shall request on two (2) Business Days' prior written notice to Seller Parent (but subject in each case to the satisfaction or waiver of all of the conditions set forth in Article X (other than conditions with respect to conditions to be satisfied at the Closing itself, but subject to such conditions)); provided, further, that the closing of the transactions pursuant to any Local Transfer Agreement shall occur contemporaneously with the Closing of this Agreement unless otherwise agreed in

writing by the Parties and shall be deemed to occur on the same calendar day as the Closing Date in the United States. The Closing of the transactions in the United States and pursuant to those jurisdictions in which the applicable Local Transfer Agreement closes contemporaneously herewith, shall be deemed to occur at 12:01 a.m., local time, in each applicable jurisdiction unless a different time is provided for in the applicable Local Transfer Agreement. The Local Transfer Agreements that do not close contemporaneously with the Closing shall close at such dates and times as agreed to by the Parties.

3.2 Transactions at Closing.

(a) At the Closing, subject to the terms and conditions hereof:

(i) <u>Sellers' Actions and Deliveries</u>. In accordance with and pursuant to the terms and conditions of this Agreement and each Local Transfer Agreement, Seller Parent shall: (i) deliver to Buyer Parent share certificates representing the Transferred Shares free and clear of all Liens which certificates shall be duly endorsed in blank or accompanied by duly executed stock powers, or, in those jurisdictions where applicable, notarized deeds of transfer reasonably acceptable to Buyer Parent; (ii) execute and deliver to Buyer Parent (or to such Affiliates of Buyer Parent as instructed by Buyer Parent in writing prior to the Closing Date) such bills of sale, endorsements, assignments and other documents as may (in the reasonable judgment of Buyer Parent is counsel) be necessary or appropriate to transfer and convey, or cause to be transferred and conveyed, to Buyer Parent (or to such Affiliates of Buyer Parent as instructed by Buyer Parent in writing prior to the Closing Date) all of the Transferred Assets free and clear of all Liens, (iii) execute and deliver, or cause to be executed and delivered, to Buyer Parent (or to such Affiliates of Buyer Parent as instructed in writing by Buyer Parent prior to the Closing Date) the Ancillary Agreements that call for a Seller's signature, and (iv) deliver to Buyer Parent a properly executed affidavit prepared in accordance with Treasury Regulations Section 1.1445-2(b) certifying each applicable Seller's non-foreign status for U.S. federal income Tax purposes.

(ii) <u>Buyers' Actions and Deliveries</u>. In consideration for the transfer of the Acquired Assets, and in accordance with and pursuant to the terms and conditions of each Local Transfer Agreement, Buyer Parent shall (i) pay, or cause to be paid, the Cash Consideration, as adjusted by the Estimated Cash Consideration Adjustment Amount, to Seller Parent by a wire transfer of immediately available funds in accordance with written instructions provided by Seller Parent at least five (5) Business Days prior to the Closing Date, (ii) issue to Seller Parent (or its designee) the Stock Consideration and (iii) execute and deliver, or cause to be executed and delivered, to Seller Parent (and/or to such Affiliates of Seller Parent as instructed by Seller Parent in writing to Buyer Parent prior to the Closing Date) the Ancillary Agreements that call for a Buyer's signature and (iv) deliver a general release and discharge from the Buyer Group, executed and delivered to Seller Parent, in agreed form, releasing and discharging, to the extent the applicable member of the Transferred Group would be permitted to do so under applicable Law, each past and present director or officer of each member of the Transferred Group and their Subsidiaries from any and all Liability to the Buyer Group in connection with or arising out of any act or omission of any such director or officer acting in his or her capacity as such, at or prior to the Closing; provided that there shall be no release or discharge for criminal acts.

3.3 Purchase Price Adjustments.

(a) No later than five (5) Business Days prior to the Closing Date, Seller Parent shall deliver to Buyer Parent (i) the Estimated Closing Date Net Working Capital Statement, which shall set forth an estimate of the Current Assets and the Current Liabilities, and (ii) a statement (the "Estimated Closing Date Net Cash Statement") which shall set forth an estimate of the Net Cash (the "Estimated Net Cash" and together with the Estimated Closing Date Net Working Capital Statement, the "Estimated Statements"), in each case as of the Closing. The Estimated Statements shall be prepared in accordance with the Accounting Principles and as otherwise expressly contemplated by this Agreement. Upon completion of the Estimated Statements, Seller Parent shall derive: (i) the estimated Closing Date Working Capital Shortfall; (ii) the Estimated Closing Date Working Capital Shortfall; and (iv) the Estimated Closing Date Working Capital Excess, if any. The amount of cash to be paid at the Closing (the "Estimated Cash Consideration Adjustment Amount") shall be equal to the Cash Consideration minus (A) the Estimated Closing Date Working Capital Shortfall, if any, plus (B) the Estimated Closing Date Working Capital Excess, if any, and plus (C) the Estimated Net Cash.

(b) Within 90 calendar days after the Closing, Buyer Parent shall prepare and deliver to Seller Parent the Closing Date Net Working Capital Statement, which shall set forth the Current Assets and the Current Liabilities as of Closing, and a statement of Closing Net Cash (the "Closing Date Net Cash" and together with the Closing Date Net Working Capital Statement, the "Closing Statements"), in each case as of the Closing. The Closing Statements shall be prepared in accordance with the Accounting Principles. Upon completion of the Closing Statements, Buyer Parent shall derive the Closing Date Working Capital and the Closing Net Cash and provide such calculations to Seller Parent.

(c) Seller Parent shall complete its review of the Closing Statements within 60 days after delivery thereof by Buyer Parent and shall notify Buyer Parent in writing of its acceptance or dispute of any amounts reflected on the Closing Statements prior to the end of such period (such notice, the "Seller's Objection"). The Seller's Objection shall specify, with a reasonably detailed explanation, those items or amounts as to which Seller Parent disagrees (and shall include Seller Parent's proposed changes to the Closing Statements, Closing Date Net Working Capital and Closing Net Cash). Seller Parent shall be deemed to have agreed with all items and amounts included in the Closing Statements that Seller Parent does not dispute. If no Seller's Objection is timely received, the Closing Statements shall be deemed final.

(d) If a Seller's Objection is delivered, Buyer Parent and Seller Parent shall negotiate in good faith to reconcile their differences and any resolution by them as to any disputed amounts shall be final, binding and conclusive on the Parties. If Buyer Parent and Seller Parent are unable to reach a resolution to such effect within thirty (30) calendar days after Buyer Parent's receipt of the Seller's Objection, Buyer Parent and Seller Parent shall submit the amounts remaining in dispute for resolution to the Reporting Accountants. The Reporting Accountants shall be directed to, within thirty (30) calendar days after submission of the dispute, determine and report to the parties upon such remaining disputed amounts with respect to the Closing Statements, and such report shall be final, binding and conclusive on the Parties hereto and shall constitute an arbitral award upon which a judgment may be entered in any court having

jurisdiction thereof. The Reporting Accountants shall address only those items in dispute. Buyer Parent shall bear and pay a percentage of the fees and disbursements of the Reporting Accountants that is equal to the percentage of the total amount of changes proposed to the Closing Statements by Seller Parent that are successful, and Seller Parent shall bear and pay a percentage of the fees and disbursements of the Reporting Accountants that is equal to the percentage of the total amount of changes proposed to the Closing Statements by Seller Parent that are not successful, in each case as determined by the Reporting Accountants. The Parties, on behalf of themselves and their Controlled Affiliates, agree that the procedure set forth in this Section 3.3 for resolving disputes with respect to adjustments of the Global Purchase Price under this Section 3.3 shall be the sole and exclusive method for resolving any such disputes; provided, however, that this provision shall not prohibit either Party from instituting litigation to enforce any ruling of the Reporting Accountants; provided, further, that the foregoing shall not impair the right to make indemnity claims hereunder.

- (e) Seller Parent shall provide to Buyer Parent and its accountants reasonable access to the books and records related to the Business and the Transferred Entities for the period prior to the Closing and to any other information, including work papers of its accountants (to the extent permitted by such accountants), and to any employees during regular business hours and on reasonable advance notice, to the extent necessary for Buyer Parent to prepare the Closing Statements. Seller Parent and its accountants shall have reasonable access to relevant information used by Buyer Parent in preparing the Closing Statements, including the work papers of its accountants (to the extent permitted by such accountants).
- (f) The "Final Cash Consideration Amount" shall be equal to the Cash Consideration minus (A) the Closing Date Working Capital Shortfall, if any, plus (B) the Closing Date Working Capital Excess, if any, as the case may be, plus the Closing Net Cash.
- (g) The "Adjustment Amount" shall be equal to the Final Cash Consideration Amount minus the Estimated Cash Consideration Adjustment Amount. If the Adjustment Amount is positive, cash in the amount of such difference shall be payable by Buyer Parent to Seller Parent and if the Adjustment Amount is negative, cash in an amount equal to the absolute value of such difference shall be payable by Seller Parent to Buyer Parent.
- (h) Any amounts due pursuant to Section 3.3(g) shall be paid promptly by the relevant party (and in any event within five Business Days) after the final determination thereof in cash, plus interest from the Closing Date to, but not including, the date of payment at LIBOR calculated on a 365-day basis, in U.S. Dollars by wire transfer of immediately available funds to an account designated by Seller Parent or Buyer Parent (as applicable).

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF SELLERS

Except as disclosed in the Seller Parent SEC Documents and forms, documents and reports of Seller Parent, in each case publicly filed or furnished with the SEC since December 31, 2014 and prior to the date of this Agreement (including exhibits and other information incorporated by reference therein provided that such exhibits and other information

incorporated by reference therein is publicly available) to the extent the relevance of such disclosure is reasonably apparent (but excluding any disclosures set forth in any "risk factors" section, any disclosures in any "forward looking statements" section and any other disclosures included therein to the extent they are predictive, general, cautionary or forward-looking in nature) or in the applicable section of the disclosure letter delivered by Seller Parent to Buyer Parent immediately prior to the execution of this Agreement (the "Sellers Disclosure Letter") (it being agreed that disclosure of any item in any section of the Sellers Disclosure Letter shall be deemed disclosure with respect to any other section of this Agreement to which the relevance of such item is reasonably apparent), Sellers represent and warrant to Buyer Parent as set forth below.

4.1 Qualification, Organization, Subsidiaries, etc.

(a) Each Seller and each member of the Transferred Group is a legal entity duly organized, validly existing and, where relevant, in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so qualified or, where relevant, in good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect. Seller Parent has filed with the SEC, prior to the date of this Agreement, complete and accurate copies of the Certificate of Incorporation of Seller Parent as amended to the date hereof (the "Seller Parent Certificate of Incorporation"). The Seller Parent Certificate of Incorporation is in full force and effect and Seller Parent is not in violation of the Seller Parent Certificate of Incorporation.

(b) All the issued and outstanding shares of capital stock of, or other equity interests in, each Transferred Entity have been validly issued and are fully paid and nonassessable and are wholly owned, directly or indirectly, by Sellers free and clear of all Liens.

4.2 Corporate Authority Relative to this Agreement; No Violation.

(a) Seller Parent has all necessary corporate power and authority to execute, deliver, perform its obligations under and consummate the transactions contemplated by this Agreement and the Ancillary Agreements, to the extent it will be a party thereto. The consummation of the transactions contemplated hereby and thereby and the execution and delivery of this Agreement and the Ancillary Agreements, to the extent it will be a party thereto, and the performance of all of its obligations hereunder and thereunder have been duly authorized by Seller Parent. The execution, delivery and performance by Seller Parent of this Agreement and the Ancillary Agreements, to the extent it will be a party thereto, are not prohibited or limited by, and shall not result in a breach of or a default under, any provision of the Organizational Documents of Seller Parent, or a material breach or a material default under any material Contract binding on Seller Parent, or of any applicable Order, and shall not result in any Lien (other than as may arise as a result of an action taken, or contract entered into, by Buyers or their Affiliates or other than Permitted Liens) on any of the Acquired Assets. This Agreement

has been duly executed and delivered by Seller Parent, and the Ancillary Agreements will, at the Closing, be duly executed and delivered by Sellers to the extent Sellers are party thereto, and, assuming due and valid authorization, execution and delivery by each other Party thereto (other than any other Seller), this Agreement constitutes, and when executed and delivered by Seller Parent, to the extent Sellers are party thereto, the Ancillary Agreements will constitute, legal, valid and binding obligations of Sellers, as applicable enforceable against Sellers in accordance with their respective terms, except that the enforcement hereof or thereof may be limited by (x) bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors' rights generally and (y) general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law). Seller Parent has all necessary corporate power and authority to cause each Seller that is a party to any Ancillary Agreement to perform such Seller's obligations thereunder and to consummate the Transactions, including the transactions contemplated by the applicable Ancillary Agreement.

(b) Each Seller, as applicable, has all necessary corporate power and authority to execute, deliver and perform its obligations under and consummate the transactions contemplated by this Agreement and the Ancillary Agreements to which it is a party. The execution and delivery of such agreement(s) and the performance of all of its obligations thereunder and the consummation of the transactions contemplated thereunder have been duly authorized by each such Seller. The execution, delivery and performance by each Seller of this Agreement and the Ancillary Agreement(s) to which it is a party are not prohibited or limited by, and shall not result in a breach of or a default under, any provision of the Organizational Documents of any Seller that is a Significant Subsidiary, or a material breach or a material default under any material Contract binding on such Seller, or of any applicable Order, and shall not result in any Lien (other than as may arise as a result of an action taken, or contract entered into, by Buyers or their Affiliates or other than Permitted Liens) on any of the Acquired Assets or any Transferred Group Assets. The Ancillary Agreements, upon their delivery at or prior to Closing, will have been duly executed and delivered by each Seller that is a party thereto and constitute the legal, valid and binding obligation of each Seller that is a party thereto (other than any other Seller), enforceable against each such Seller in accordance with their respective terms, except as enforceability may be limited or affected by applicable bankruptcy, insolvency, moratorium, reorganization or other Laws of general application relating to or affecting creditors' rights generally. Except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, or materially delay or impair the Transactions, none of the execution, delivery or performance of this Agreement and the Ancillary Agreements by Seller Parent will: (a) result in a material modification, violation or breach of, or constitute (with or without notice or lapse of time or both) a material default (or give rise to any right, including, but not limited to, any right of termination, amendment, cancellation or acceleration of any material obligation or loss of any material benefit) under, any of the terms, conditions or provisions of any Business Contract; (b) violate in any material respect any Permit, Regulatory Registration, Order or Law applicable to the Business or the Acquired Assets or give any Governmental Authority or other Person the right to challenge any of the Transactions or to exercise any remedy or obtain any relief under, any Law or any Order to which any Seller or any of the Acquired Assets, is subject; or (c) result in the imposition or creation of any Liens upon or with respect to any Acquired Assets or any Transferred Group.

4.3 Reports and Financial Information.

- (a) Set forth on Schedule 4.3 is the unaudited Performance Financial Statement of the Business and associated support activities for the year ended December 31, 2014. The Performance Financial Statement (i) has been prepared on a performance basis consistent with Seller Parent's accounting policies, which accounting policies are in accordance with U.S. GAAP, and Seller Parent's internal management reporting policies and procedures; (ii) has been prepared from the Books and Records; and (iii) presents fairly in all material respects the results of operations of the Business and associated support activities for the period presented.
- (b) Seller Parent maintains a system of "internal control over financial reporting" (as defined in Rule 13a-15(f) of the Exchange Act) sufficient to provide reasonable assurance (i) that transactions are made in accordance with management's authorization, (ii) that transactions are recorded as necessary to permit the preparation of Seller Parent's consolidated financial statements in conformity with U.S. GAAP and (iii) regarding prevention or timely detection of the unauthorized acquisition, use or disposition of Seller Parent's properties or assets.
- (c) Seller Parent's "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) are reasonably designed to ensure that information required to be disclosed by Seller Parent in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time period specified in the rules and forms of the SEC, and that all such information is accumulated and communicated to Seller Parent's principal executive officer and principal financial officer as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the principal executive officer and principal financial officer of Seller Parent required under the Exchange Act with respect to such reports.

4.4 Compliance with Law; Permits.

Sellers are, and since January 1, 2014 have been in compliance with and are not in default under or in violation of any Laws, applicable to the Acquired Assets or the Business, except where such non-compliance, default or violation would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.

- (a) Notwithstanding anything contained in this <u>Section 4.4</u>, no representation or warranty shall be deemed to be made in this <u>Section 4.4</u> in respect of the matters referenced in <u>Section 4.14</u>, or in respect of environmental, Tax, employee benefits or labor Law matters.
- 4.5 <u>Title to Assets</u>. Sellers and their Subsidiaries collectively own or have a valid leasehold interest in, are in possession of, and have good, valid and, to the extent such concept is recognized in the jurisdiction where the Transferred Assets and the assets of the Transferred Entities (the "**Transferred Group Assets**") are located, marketable title to all of the Transferred Assets and the Transferred Group Assets except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect. Upon execution and delivery by Sellers of this Agreement and the Ancillary Agreements and upon Closing, Buyers will become at Closing the true and lawful owner of, and will receive good title to, the Transferred Assets material to the Business, free and clear of all Liens, other than Permitted Liens. All of the

Transferred Group Assets are free and clear of all Liens, other than Permitted Liens or except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect.

4.6 Transferred Entities.

- (a) Seller Parent or one of its Subsidiaries beneficially owns one hundred percent (100%) of the outstanding Equity Participations of each Transferred Entity. All of the Transferred Shares are validly issued, fully paid and nonassessable and free and clear of any and all Liens. Upon transfer of the Transferred Shares to Buyer Parent at the Closing, Buyer Parent shall own all outstanding Equity Participations of each Transferred Entity.
- (b) Each Seller has the full right to sell, convey, transfer, assign and deliver the Transferred Shares owned by it to the applicable Buyer and, upon the Closing, such Buyer will have good and valid title to all such Transferred Shares, free and clear of all Liens (other than as may result from the action of the Buyers). Other than the Transferred Shares, there are (i) no Equity Participations in any Transferred Entity issued or outstanding, (ii) no Contracts with respect to the issuance, sale or transfer of Equity Participations by any Transferred Entity, (iii) no Contracts with respect to the voting of any capital stock of any Transferred Entity, (iv) no preemptive rights, rights of participation, rights of maintenance or any similar rights with respect to Equity Participations in any Transferred Entity and (v) no Contracts with respect to the voting or registration of, or restricting any Person from, purchasing, selling, pledging or otherwise disposing of (or granting any option or similar right with respect to), any Equity Participations in any Transferred Entity.
- (c) <u>Schedule 4.6(c)</u> sets forth a complete and accurate list of the Persons in the Transferred Group (other than any new Persons which may be incorporated or formed to the extent necessary for the Pre-Closing Reorganization and which may form part of the Transferred Group upon Closing).
- 4.7 Environmental Matters and Regulations. Except for such matters as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect: (i) Sellers and their respective Subsidiaries, in respect of the Business and Acquired Assets are now and have been since January 1, 2013 in compliance with all, and have not violated any, applicable Environmental Laws; (ii) no Transferred Owned Real Property, Transferred Leased Real Property or third party site occupied by the Transferred Group (including soils, groundwater, surface water, buildings or other structures), is contaminated with any Hazardous Materials in a manner that is or is reasonably likely to be required to be remediated or removed, that is in violation of any Environmental Law, or that is reasonably likely to give rise to any Liability; (iii) Sellers have all of the Environmental Permits necessary for the conduct and operation of the Business as now being conducted, and all such Environmental Permits are in good standing; (iv) since January 1, 2013, neither Seller Parent nor any of its Subsidiaries has received any written notice, demand letter, claim, Proceeding or request for information alleging that Seller Parent or any of its Subsidiaries, in respect of the Business and the Acquired Assets, may be in violation of or subject to liability under any Environmental Law or are allegedly subject to any removal, remedial or response actions; and (v) Sellers and their respective Subsidiaries, with respect to the Business and the Acquired

Assets, are not subject to any order, decree, injunction or agreement with any Governmental Authority, or any indemnity or other agreement with any third party, imposing liability or obligations relating to any Environmental Law or any Hazardous Materials.

4.8 Employee Benefits.

(a) No later than forty five (45) days after the date of this Agreement, Sellers shall deliver to Buyer Parent Schedule 4.8(a)(1) which shall set forth a list of each material Seller Benefit Plan and Schedule 4.8(a)(2) which shall set forth a list of each material Transferred Entity Benefit Plan and separately identify each material Transferred Entity Benefit Plan that is maintained primarily for the benefit of employees outside of the United States (a "Non-U.S. Transferred Entity Benefit Plan"). For purposes of this Agreement, (1) "Benefit Plan" means any benefit or compensation plan, program, policy, practice, agreement, contract, arrangement or other obligation (whether or not written), including, but not limited to, each ERISA Plan, whether or not subject to ERISA, and each bonus, phantom equity, stock, stock option or other equity-based compensation arrangement or plan, incentive, deferred compensation, retirement or supplemental retirement, severance, employment, consulting, change-in-control, profit sharing, pension, vacation, cafeteria, dependent care, medical care, employee assistance program, education or tuition assistance programs, and each insurance and other similar fininge or employee benefit plan, program or arrangement, (2) "Seller Benefit Plan" means a Benefit Plan that is sponsored or maintained by Seller Parent or any of its Subsidiaries (other than the Transferred Group), in which any Business Employees participate, and (3) "Transferred Entity Benefit Plan" means a Benefit Plan sponsored or maintained by a Transferred Entity or Subsidiary of a Transferred Entity for the benefit of current or former Business Employees (or any dependent or beneficiary thereof). With respect to each material Transferred Entity Benefit Plan and Seller Benefit Plan, no later than forty-five (45) days following the date of this Agreement, Sellers shall cause to be made available to Buyer Parent correct and complete copies of (or, to the extent no such copy exists, a description of), in each case, to the extent ap

(b) Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect, (1) each of the Seller Benefit Plans and the Transferred Entity Benefit Plans has been established, operated and administered in compliance in accordance with its terms and all applicable Laws, including, but not limited to, ERISA, the Code and in each case the regulations thereunder, (2) all contributions or other amounts payable with respect to each Seller Benefit Plan and Transferred Entity Benefit Plan in respect of current or prior plan years have been paid or accrued in accordance with U.S. GAAP and (3) there are no pending or, to the Seller Parent's Knowledge, threatened claims (other than routine claims for benefits) or proceedings by a Governmental Authority by, on behalf of or against any Seller Benefit Plan or Transferred Entity Benefit Plan or, in each case, any trust related thereto.

(c) Except as would not reasonably be expected to have a Seller Material Adverse Effect, each Transferred Entity Benefit Plan that is an ERISA Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the Internal Revenue Service to be qualified under Section 401(a) of the Code and, to the Seller Parent's Knowledge, nothing has occurred that would reasonably be expected to adversely affect the qualification or tax exemption of any such Transferred Entity Benefit Plan. Except as would not reasonably be

expected to have a Seller Material Adverse Effect, with respect to any Transferred Entity Benefit Plan that is an ERISA Plan, neither the Sellers nor any of their respective Subsidiaries has engaged in a transaction in connection with which the Sellers nor any of their respective Subsidiaries reasonably could be subject to either a civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a tax imposed pursuant to Section 4975 or 4976 of the Code.

(d) Neither the Sellers nor any of their Subsidiaries (including the Transferred Group) has or is expected to incur any liability under subtitles C or D of Title IV of ERISA with respect to any ongoing, frozen or terminated "single-employer plan", within the meaning of Section 4001(a)(15) of ERISA, currently or formerly maintained by any of them or any ERISA Affiliate that would, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect. Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect. Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect, with respect to any Transferred Entity Benefit Plan subject to the minimum funding requirements of Section 412 of the Code or Title IV of ERISA, (1) no such plan is, or is expected to be, in "at-risk" status (within the meaning of Section 303(i)(4)(A) of ERISA or Section 430(i)(4)(A) of the Code), (2) as of the last day of the most recent plan year ended prior to the date hereof, the actuarially determined present value of all "benefit liabilities" within the meaning of Section 4001(a)(16) of ERISA did not exceed the then current value of assets of such Transferred Entity Benefit Plan or, if such liabilities did exceed such assets, the amount thereof was properly reflected on the financial statements of the Sellers or the applicable Subsidiary previously provided to Buyer Parent, (3) no unsatisfied liability (other than for premiums to the Pension Benefit Guaranty Corporation) under Title IV of ERISA has been, or is expected to be, incurred by the Sellers or any of their Subsidiaries (including the Transferred Group), (4) the Pension Benefit Guaranty Corporation has not instituted Proceedings to terminate any such Transferred Entity Benefit Plan and (5) no "reportable event" within the meaning of Section 4043 of ERISA) has occurred, nor has any event described in Sections 4062, 4063 or 4041 of ERISA occ

(e) Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect, each Seller Benefit Plan and Transferred Entity Benefit Plan that is a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code) is in documentary compliance with, and has been operated and administered in all respects in compliance with, Section 409A of the Code and the guidance issued by the Internal Revenue Service provided thereunder.

(f) Neither the Sellers nor any ERISA Affiliate have maintained, established, participated in or contributed to, or is or has been obligated to contribute to, or has otherwise incurred any material obligation or material liability (including any contingent liability) under, any "multiemployer plans" (within the meaning of Section 3(37) of ERISA) in the last six (6) years.

(g) Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect, all Non-U.S. Transferred Entity Benefit Plans

comply with applicable local Law, all such plans that are intended to be funded and/or book-reserved are funded and/or book reserved, as appropriate, based upon reasonable actuarial assumptions, and there is no pending or threatened material litigation relating to any Non-U.S. Transferred Entity Benefit Plan.

- (h) Except as set forth on Schedule 4.8(h), neither the Sellers nor any of their respective Subsidiaries has any obligation to provide to a current or former Business Employee, and no Transferred Entity Benefit Plan or other agreement provides any individual with the right to, a gross up, indemnification, reimbursement or other payment for any excise or additional taxes, interest or penalties incurred pursuant to Section 409A or Section 4999 of the Code or due to the failure of any payment to be deductible under of Section 280G of the Code.
- (i) Except as set forth on Schedule 4.8(i) (provided prior to or on the date of this Agreement with respect to the top 30 highest paid Business Employees and no later than forty-five (45) days after the date of this Agreement with respect to any other person), neither the execution and delivery of this Agreement nor the consummation of the Transactions (either alone or in conjunction with any other event) will (i) result in any payment (including severance, unemployment compensation, "excess parachute payment" (within the meaning of Section 280G of the Code), forgiveness of Indebtedness or otherwise) becoming due to any current or former Business Employee under any Seller Benefit Plan or Transferred Entity Benefit Plan, (ii) increase any benefits otherwise payable to any Business Employee under any Seller Benefit Plan or (iii) result in any acceleration of the time of payment, funding or vesting of any such benefits.

4.9 Absence of Certain Changes or Events.

- (a) From December 31, 2014 through the date of this Agreement, there has not occurred any event, development, occurrence, or change that has had, or would reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.
- (b) From December 31, 2014 through the date of this Agreement, (i) the Business has been conducted in the ordinary course and (ii) Seller Parent and its Subsidiaries have not taken any action that would constitute a breach of Section 6.1(b) had such action been taken after the execution of this Agreement.

4.10 Business Contracts.

- (a) Except as filed as exhibits to any documents filed by Seller Parent with the SEC prior to the date hereof and as set forth on Schedule 4.10(a) (other than Contracts described in Section 4.10(a)(ii)), none of the Business Contracts:
 - (i) is a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC);
- (ii) contains any non-compete or exclusivity provisions with respect to the Business or the Transferred Group, or upon consummation of the Transactions, Buyers or their Subsidiaries, in each case, which materially restricts the conduct of the Business or use of the Acquired Assets or Transferred Group Assets in the aggregate;

- (iii) relates to a partnership, joint venture or similar arrangement, unless, in each case, immaterial to the Business;
- (iv) is a Contract (other than a Contract with respect to API) for sale of goods or services that involved the payment of more than \$50,000,000 in 2014 or which Seller Parent reasonably anticipates will involve the payment of more than \$50,000,000 in 2015 under each such Contract;
- (v) is a Contract (other than a Contract with respect to API) for (A) the purchase of services, materials, supplies or equipment or (B) any royalty, profit sharing or contingent payment right, which, in each case, involved the payment of more than \$50,000,000 in 2014 or Seller Parent reasonably anticipates will involve the payment of more than \$50,000,000 in 2015 and has a remaining term in excess of twelve (12) months, excluding, in each case such Contracts that are terminable on less than 90 days' notice by Seller Parent or its Affiliates without penalty or other material impact;
- (vi) relates to any lease of real property which has an unexpired term of greater than 12 months and requires future annual payments in excess of \$1,000,000 or is a lease of any machinery, equipment, vehicle or other tangible personal property (other than in respect of office equipment or equipment used in the research and development of products) which, in each case, require future annual payments in excess of \$5,000,000;
- (vii) is a Contract for capital expenditures or the acquisition or construction of fixed assets which requires aggregate future payments in excess of \$100,000,000; or
 - (viii) is for any Indebtedness that may be directly or indirectly transferred to Buyer in connection with the Transactions.
- (b) Each Contract of the type described above in Section 4.10(a) and each IP Contract is referred to herein as a "Seller Material Contract." Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, or as set forth on Schedule 4.10(b):
- (i) each Seller Material Contract is valid and binding on the applicable Seller, to the extent it is a party thereto, and, to Seller Parent's Knowledge, each other party thereto, as applicable;
- (ii) no Seller is in, or alleged to be in, breach or default under any of the Seller Material Contracts, and to Seller Parent's Knowledge, no other party to any of the Seller Material Contracts has breached or defaulted thereunder; and
- (iii) since January 1, 2015, no Seller has received any notice or other communication regarding any actual or possible violation or breach of, or default under, any Seller Material Contract.
- (c) No Seller is currently renegotiating any material term of any Seller Material Contract or paying liquidated damages in lieu of performance thereunder.
- (d) Seller Parent has prior to the date of this Agreement delivered to, or made available to, Buyer Parent or its Representatives, true and complete copies of each Seller Material Contract, other than Contracts described in Section 4.10(a)(ii).

4.11 <u>Investigations</u>: <u>Litigation</u>. As of the date hereof, (a) there is no investigation or review pending (or, to Seller Parent's Knowledge, threatened) by any Governmental Authority with respect to the Acquired Assets or the Business, and (b) other than ANDA litigation in the Ordinary Course of the Business, there are no Claims pending (or, to Seller Parent's Knowledge, threatened) against Sellers or any of their respective Controlled Affiliates with respect to the Acquired Assets or the Business, and there are no Orders which, in the case of clause (a) or (b), (i) would reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect or (ii) would have a reasonable possibility of resulting in (A) monetary liability of the Business in excess of \$50,000,000, individually or in the aggregate, with any related claims, (B) material non-monetary relief against the Business or (C) a criminal violation or criminal liability by the Business.

4.12 Regulatory Matters.

(a) Except as set forth in Schedule 4.12(a) or as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, Sellers and their respective Subsidiaries hold and are operating in material compliance with all Permits, including the Regulatory Registrations, that are necessary for the conduct of the Business now being conducted by Sellers and their respective Subsidiaries and all Regulatory Registrations. There are no Proceedings pending or, to Seller Parent's Knowledge, threatened which would reasonably be expected to result in the material limitation, material adverse modification, revocation, cancellation or suspension of any material Permits or Regulatory Registration. Notwithstanding anything contained in this Section 4.12, no representation or warranty shall be deemed to be made in this Section 4.12 in respect of Environmental Permits which are addressed in Section 4.7.

(b) Except for those matters which, individually or in the aggregate, would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, the Sellers and their respective Subsidiaries are conducting and have conducted the Business in compliance, in all material respects, with all material Laws, including all Laws applicable to the nonclinical and clinical testing, manufacturing, ownership, operation, storage, import, export, distribution, marketing, pricing, sale, promotion, warehousing, packaging, Labeling, handling and/or testing of the Products, which include the federal Food, Drug, and Cosmetic Act, 21 U.S.C. § 301 et seq., the federal Anti-Kickback Law, 42 U.S.C. § 1320a-7b, the federal False Claims Act, 31 U.S.C. § 3279, et seq., the federal Civil Monetary Penalties Law, 42 U.S.C. § 1320a-7a, the Health Insurance Portability and Accountability Act of 1996, (42 U.S.C. § § 1320d et seq., as amended by the Health Information Technology for Economic and Clinical Health Act, 42 U.S.C. § 8 17921 et seq. HIPAA false statement provisions, 18 U.S.C. § 1035, and the HIPAA health care fraud provisions, 18 U.S.C. § 1347, Medicare, Title XVIII of the Social Security Act, Medicaid, Title XIX of the Social Security Act, EEA Member States laws implementing EU Directive 2001/83/EC on the Community code relating to medicinal products for human use, as last amended, or EU Regulation 726/2004 laying down Community procedures for the authorization and supervision of medicinal products for human and veterinary

use and establishing a European Medicines Agency, and the FDA's current Good Manufacturing Practice (cGMP) regulations as set forth in 21 C.F.R. Parts 210 and 211 (all such applicable Laws, collectively, the "Health Care Laws"). Except as set forth in Schedule 4.12(b), within the three (3) years preceding the date hereof, Sellers and their respective Subsidiaries have not received any written notice, including any warning letter, notice of adverse finding, or notice of deficiency, or similar communication from the FDA or any other Governmental Authority, (i) contesting the Regulatory Registrations for the Labeling and promotion of any of the Products, (ii) alleging that any of the Products or the ownership, manufacturing, operation, storage, import, export, distribution, marketing, pricing, sale, promotion, warehousing, packaging, Labeling, handling and/or testing thereof is in violation of any applicable Health Care Law, Regulatory Registration or Permit, (iii) otherwise alleging any violation of any Health Care Laws by any Seller or any of their respective Subsidiaries with respect to the Business and/or Products, or (iv) alleging that any such violation, if any, has not been remedied; except, in each case, to the extent that such violation has been remedied or that such violation has not been, and would not be, either individually or in the aggregate, material to Sellers and their respective Subsidiaries. No Seller or Subsidiary of any Seller is subject to any enforcement, regulatory or administrative Proceedings by the FDA or other Governmental Authority alleging that any operation or activity of Sellers and their respective Subsidiaries relating to the Business and/or Products is in material violation of any Health Care Law, and, to Seller Parent's Knowledge, no such Proceedings have been threatened. To Seller Parent's Knowledge, no act, omission, event or circumstance has occurred that would reasonably be expected to give rise to, or lead to, any such a Proceeding or a material Liability relating to the B

(c) All material documents, reports and notices required to be maintained or filed with any Governmental Authority by Sellers and their respective Subsidiaries with respect to the Business or any Product have been so maintained or filed on a timely basis, and were complete and accurate in all material respects as of the date of filing, or were subsequently updated, changed, corrected, or modified prior to the date of this Agreement. To Seller Parent's Knowledge, no such filing with any Governmental Authority contains any materially false, misleading or otherwise inaccurate statements or information, whether express or due to omission of material information, as of the date of filing. To Seller Parent's Knowledge, no action has been taken or statements made or failed to be made by either Sellers, their respective Subsidiaries or an employee, consultant, contractor, agent or other representative of Sellers with respect to the Business or any Products that could reasonably be expected to provide a basis for the FDA or other Regulatory Authority to invoke its Application Integrity Policy or similar governmental policy or regulation (including non-U.S. policies or regulations), rule or law.

(d) Except as set forth in Schedule 4.12(d) and except for those matters which, individually or in the aggregate, would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, during the period of three (3) years immediately preceding the date hereof, Sellers have not introduced into commercial distribution any Products manufactured by or on behalf of Sellers which were upon their shipment by Sellers adulterated or misbranded in violation of 21 U.S.C. § 331 or equivalent foreign country Health Care Law. Without limiting the generality of the foregoing, and except as listed in Schedule 4.12(d), (i) Sellers have not voluntarily or involuntarily issued or caused to be issued, and to Seller Parent's Knowledge, there are no facts that would require Sellers under any applicable Health Care Law to issue or cause to be issued, any recall notice, market withdrawal notice, safety notice, or other

similar notice or action disclosing an alleged material defect or lack of safety or efficacy of any Product and (ii) each Product in current commercial distribution is designed, manufactured, prepared, assembled, packaged, Labeled, sterilized, stored, installed, serviced, and processed in compliance in all material respects with, FDA current Good Manufacturing Practices regulations as set forth in 21 C.F.R. Parts 210 and 211 and equivalent Health Care Laws or standards in other jurisdictions.

- (e) To the Seller Parent's Knowledge, the Sellers and their respective Subsidiaries have marketed, sold, priced and promoted the Products in material compliance with all Health Care Law, and Sellers' and Sellers' Subsidiaries' financial relationships with distributors and customers, including health care professionals and entities, are maintained, documented and reported in compliance with all Health Care Law and its policies and procedures regarding product samples, conflicts of interest, payments and transfers of value, and remuneration arrangements. To the Seller Parent's Knowledge, the Sellers and their respective Subsidiaries have not made or caused to be made any bribe, kickback, influence payment, payoff or other unlawful payment to any Person in the conduct of the Business. With respect to the Business, no officer or employee of any Seller or Subsidiary of any Seller, or to Seller Parent's Knowledge, no agent of any Seller or Subsidiary of any Seller has: (A) made any untrue statement of material fact or fraudulent statement to the FDA or any other Governmental Authority; (B) failed to disclose a material fact required to be disclosed to the FDA or any other Governmental Authority; or (C) committed an act, made a statement, or failed to make a statement that would reasonably be expected to provide the basis for the FDA to invoke FDA's policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities," as set forth in 56 Fed. Reg. 46191 (September 10, 1991), or any equivalent Health Care Law. To Seller Parent's Knowledge, no officer, employee or agent of Sellers or their respective Subsidiaries has been convicted of any crime or engaged in any conduct for which debarment is mandated or permitted by 21 U.S.C. § 335a.
- (f) Seller Parent established and maintains a corporate compliance program that (i) addresses all material requirements of applicable Health Care Laws and all Governmental Authorities having jurisdiction over the Business, and (ii) has been structured to account for the guidance issued by the U.S. Department of Health and Human Services regarding characteristics of effective corporate compliance programs. Sellers and their respective Subsidiaries are in compliance in all material respects with all Orders to which they are subject, including any corporate integrity agreement, including all programmatic, operational and reporting requirements. Each Seller has designated an executive employee thereof as its chief compliance officer.
- (g) Neither any Seller, nor to Seller Parent's Knowledge any employee or contractor of any Seller, has made any voluntary or self-disclosure to any Governmental Authority regarding any potential non-compliance with any Health Care Law applicable to the Business or Products.
- (h) Without any limitation to the foregoing, except for those matters which, (i) individually or in the aggregate, would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, with respect only to the Business or (ii) would have a reasonable possibility of resulting in a criminal violation or criminal liability by the

Business, and except for those matters which, individually or in the aggregate, would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, to Seller Parent's Knowledge, no Seller, Subsidiary of Seller or any owner (in addition to Sellers), director, officer, employee, agent, or other contractor and subcontractor of Sellers (i) has in the last two (2) years or during any applicable statute of limitations period that has yet to expire or is currently tolled ("Applicable SOL Period") violated, has caused other parties to be in violation of, or has knowingly participated in a violation of the FCPA, the U.S. Travel Act, any U.S. domestic or commercial bribery statute, the UK Bribery Act of 2010, as amended, or any other applicable anticorruption law or regulation (collectively "Anti-Corruption Laws"); (ii) has in the last two (2) years or during any Applicable SOL Period directly or indirectly (through third parties) paid, provided, promised, offered, or authorized the payment or provision of any money or anything of value to (a) an official, employee, or agent of any government, military, public international organization, state-owned or affiliated entity, or instrumentality thereof (collectively "Government Officials"), (b) a political party or candidate, or (c) any other individual, entity, or organization, for purposes of obtaining, retaining, or directing businesses or another improper advantage from, to or for any Person, including Sellers or their respective Subsidiaries; (iii) has in the last two (2) years or during any Applicable SOL Period otherwise offered, promised, authorized, provided, or incurred any bribe, kickback, or other corrupt or unlawful payment, expense, contribution, gift, entertainment, travel or other benefit, or advantage (collectively, "Restricted Benefits") to or for the benefit of any Government Official, political party or candidate, or any other individual, entity, or organization; (iv) has in the last two (2) years or during any Applicable SOL Period solicited, accepted, or received any Restricted Benefits; (v) has established or maintained any unlawful or unrecorded funds or has otherwise violated the books, records, and internal controls requirements of the FCPA or similar Anti-Corruption Laws; (vi) has or is engaged in business directly or indirectly in Iran, Cuba, North Korea, Sudan, Syria or any other country in violation of comprehensive or selective U.S. sanctions; (vii) has otherwise violated, has caused other parties to be in violation of, or has knowingly participated in a violation of any Laws relating to the export or re-export of commodities, technologies, or services, including, but not limited to, the United States Export Administration Act of 1979, the United States Export Administration Regulations, the United States International Emergency Economic Powers Act, the United States Trading with the Enemy Act, the United States Arms Export Control Act, the United States Foreign Asset Control Regulations, the United States International Traffic in Arms Regulations, the International Boycott Provisions of Section 999 of the Code, U.S. customs laws and regulations and, in each case, any regulations or orders issued thereunder (all as amended from time to time) (collectively, "U.S. Trade Laws"); or (viii) is or has in the last two (2) years or during any Applicable SOL Period been the subject of any allegation, voluntary disclosure, subpoena or other information request, investigation, prosecution, settlement or other enforcement action related to any Anti-Corruption Law, U.S. Trade Law or other applicable law or regulation.

4.13 <u>Consents and Governmental Approvals</u>. Assuming the truth and accuracy of the representations and warranties made by Buyer Parent in <u>Article V</u>, and except for (a)(i) any HSR Filing with the FTC or the Antitrust Division or (ii) any filing under any other foreign antitrust, competition or similar Laws ("Foreign Antitrust Laws"), in each case to the extent required, and (b) as would not be reasonably be expected to have, individually and in the aggregate, a Seller Material Adverse Effect, no Consent from any Governmental Authority in connection with the execution and delivery of this Agreement or any Ancillary Agreements or the consummation or performance of the Transactions is required for Sellers to consummate the Transactions.

- 4.14 Tax Matters. Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect:
- (a) all Tax Returns that are required to be filed by or with respect to the Acquired Assets, any member of the Transferred Group and the Business have been timely filed (taking into account any extension of time within which to file), and all such Tax Returns are true, complete and accurate;
- (b) all Taxes due and owing with respect to the Acquired Assets, any member of the Transferred Group and the Business (whether or not shown on any Tax Return) have been paid, including any Taxes required to be withheld from amounts owing to any employee, creditor, or third party (in each case, whether or not shown on any Tax Return), other than Taxes for which adequate reserves have been established in accordance with U.S. GAAP on the financial statements of Seller Parent, Sellers or their respective Affiliates;
- (c) there is no notice, Claim, audit, action, suit, Proceeding, or investigation now pending or, to Seller Parent's Knowledge, threatened in writing against or with respect to any Taxes relating to or involving the Acquired Assets, any member of the Transferred Group or the Business;
- (d) neither any Seller nor any member of the Transferred Group has waived any statute of limitations with respect to any Taxes related to the Acquired Assets, any member of the Transferred Group or the Business or agreed to any extension of time with respect to any Tax assessment or deficiency related to the Acquired Assets, any member of the Transferred Group or the Business;
- (e) since July 1, 2013, no member of the Transferred Group has constituted a "distributing corporation" or a "controlled corporation" (in each case, within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code (or any similar provision of state, local, or non-U.S. Law);
- (f) no member of the Transferred Group is a party to any Tax-Sharing Agreement or has any liability for Taxes of any Person (other than Seller Parent, any Seller or any other member of the Transferred Group) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Law) or as transferee or successor;
 - (g) there are no Liens for Taxes upon any of the Acquired Assets, except for Permitted Liens;
- (h) neither any Seller nor any member of the Transferred Group has received any written claim in the past three years from a Governmental Authority in a jurisdiction in which such Seller or such member, as the case may be, does not file Tax Returns to the effect that (i) in the case of such Seller, it is or may be subject to taxation by that jurisdiction with respect to the Acquired Assets, any member of the Transferred Group or the Business or (ii) in the case of such member, it is or may be subject to taxation by that jurisdiction;

(i) no member of the Transferred Group has entered into any "listed transactions" within the meaning of Treasury Regulations Section 1.6011-4(b)(2) (or any similar provision of state, local or non-U.S. Law);

- (j) no member of the Transferred Group will be required, as a result of (i) a change in accounting method for a Tax period beginning on or after a date that is five years before the Closing Date, (ii) any "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or non-U.S. Tax Law), (iii) any installment sale or open transaction disposition made on or prior to the Closing Date, (iv) any election under Section 108(i) of the Code (or any similar provision of state, local or non-U.S. Tax Law) or (v) the application of Treasury Regulations Section 1.1502-13 (or any similar provision of state, local or non-U.S. Tax Law), other than with respect to transactions in the Ordinary Course of Business or as required under the Agreement among Seller Parent and its Affiliates, to include any item of income in or exclude any item of deduction from taxable income for any Tax period ending after the Closing Date;
- (k) no "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or non-U.S. Tax Law), private letter rulings or, technical advice memoranda, or similar agreements or rulings with respect to Taxes have been entered into or issued by any Governmental Authority with respect to the Acquired Assets, any member of the Transferred Group or the Business;
- (l) no Business Contract is a Tax-Sharing Agreement that would, in any manner, bind, obligate or restrict any Buyer or any member of the Transferred Group after the Closing;
- (m) no member of the Transferred Group that is an entity organized under the laws other than those of the United States or any state thereof is treated as organized under the laws of the United States for United States federal income tax purposes; and
 - (n) each applicable Affiliate of Seller has complied with the terms and conditions of the Puerto Rico Grant.

4.15 Labor Matters.

(a) No later than forty-five (45) days after the date of this Agreement, Sellers shall provide Buyer Parent Schedule 4.15(a) which shall set forth a complete and accurate list as of the date of this Agreement of each material collective bargaining, works council or labor union contract or labor arrangement currently covering any Business Employee or to which any member of the Transferred Group is otherwise a Party currently in effect (the "Collective Bargaining Agreements"). No later than forty-five (45) days after the date of this Agreement, Sellers shall cause to be made available to Buyer Parent a true and complete copy of each Collective Bargaining Agreement, other than any publicly-available regional or industry-wide Collective Bargaining Agreement identified as such on Schedule 4.15(a).

(b) No member of the Transferred Group is subject to any labor dispute, lockout, slowdown, strike, work stoppage or grievance, nor, to Seller Parent's Knowledge, are any such actions threatened, except as would not have, individually or in the aggregate, a Seller Material Adverse Effect. To Seller Parent's Knowledge, there are no organizational efforts with respect to the formation of a collective bargaining unit presently being made or threatened involving employees of any member of the Transferred Group, except for those the formation of which would not have or reasonably be expected to have, individually or in the aggregate, a Buyer Material Adverse Effect.

- (c) Except as would not reasonably be expected to have a Seller Material Adverse Effect, the Business is being conducted in compliance with all applicable Laws pertaining to the privacy, data protection and information security of employee information.
- (d) Except as would not, individually or in the aggregate, reasonably be expected to have a Seller Material Adverse Effect, each Seller and its Subsidiaries (i) is and has been in compliance with all applicable Laws regarding employment and employment practices and those Laws relating to terms and conditions of employment, classification of employees, wages and hours, occupational safety and health and workers' compensation, and (ii) has no charges or complaints relating to unfair labor practices or unlawful employment practices pending or, to Seller Parent's Knowledge, threatened against it before any Governmental Authority.
- (e) Neither the execution and delivery of this Agreement nor the Transactions will require the consent of, or advance notification to, any works councils, unions or similar labor organizations with respect to any employees of any Seller or the Acquired Assets, other than any such consents the failure of which to obtain or advance notifications the failure of which to provide as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.

4.16 Intellectual Property.

(a) The Patents, the Licensed Patents, the Marks and the Licensed Marks, together with the Intellectual Property owned by the Transferred Group, include all of the patents, patent applications, internet domain names, trade names, registered and unregistered trademarks and service marks that are owned by or licensed to Sellers and are material to, or necessary for, the Business as conducted immediately prior to the Closing including all trade names, registered and unregistered trademarks and service marks and logos used in connection with the Products. At Closing, the Transferred Group shall solely and exclusively (beneficially, and of record where applicable) own all the Patents and the Marks, free and clear of all Liens (other than Permitted Liens). The Transferred Intellectual Property, together with the Intellectual Property owned by the Transferred Group and Intellectual Property which is licensed to the Transferred Group by the Seller and its Controlled Affiliates, is all the Intellectual Property that is material to, or necessary for, the conduct of the Business as currently conducted and as anticipated to be conducted immediately prior to the Closing, including all Intellectual Property that is used or required for use in the use, design, development, manufacturing, quality control, packaging, storage, registration, marketing, distribution or sale of the Products.

- (b) (i) Sellers have the right to grant a sublicense (as contemplated under Section 6.6(b) hereof) under any and all Intellectual Property licensed to Seller under the Shared IP License Agreements, and (ii) no Intellectual Property licensed to or from Sellers under any Shared IP License Agreement relates to, or is used in, the design, development, validation, materials and components, biological compatibility, manufacture, processing, testing, storage, packaging, labeling, regulations, safety, quality or performance of the Products.
- (c) Sellers have sufficient rights to use the Licensed Patents and the Licensed Marks, in each case, pursuant to valid and enforceable written Contracts except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect. Except as set forth in the IP Contracts, no Seller is obligated to pay, and Buyer will not be obligated to pay with respect to the Business, any royalties, fees (other than official registering fees and similar fees), commissions or other amounts (i) for the use, licensing or sublicensing of any Transferred Intellectual Property or (ii) for the development, manufacture or commercialization of any Product.
- (d) To the Seller Parent's Knowledge, the Patents and Marks are subsisting, valid and enforceable. Sellers have taken no action pursuant to which such Patents and Marks are reasonably likely to cease to be valid and enforceable. The Transferred Intellectual Property which is owned by the Sellers together with the Intellectual Property owned by the Transferred Group is not, and, so far as the Seller Parent is aware, the other Transferred Intellectual Property is not, subject to any outstanding injunction, judgment, order, decree, or ruling except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.
- (e) Other than Proceedings arising in the Ordinary Course of Business (including ANDA litigation), to Seller Parent's Knowledge no Proceedings are currently pending and Sellers have not, since January 1, 2012, received any written notice or Claim by any Person alleging that (i) any Product, the Business or the Transferred Intellectual Property, or any act by Sellers related thereto or the use thereof, infringes or misappropriates (or in the past infringed or misappropriated) any Intellectual Property of any Person or (ii) any of the issued Patents or the registered Marks are invalid or unenforceable except, in each case, as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect. To Seller Parent's Knowledge, neither the operation of the Business currently conducted, nor the making, use, import, offer for sale or other disposition of any Product infringes or misappropriates, or has infringed or misappropriated, any Intellectual Property of any Person except, in each case, as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.
- (f) To Seller Parent's Knowledge, (i) there is no, nor has there been any, infringement by any Person of any of the material rights of any Seller in or to the Patents, Licensed Patents, Marks or Licensed Marks or other material Intellectual Property owned by the Transferred Group, and (ii) there is no, nor has there been any, misappropriation by any Person of any of the material Know-How, Manufacturing Instructions or Technical Information.
- (g) Other than Proceedings arising in the Ordinary Course of Business (including ANDA litigation) or Proceedings that, if resolved against each Seller, would not

reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, there are no Proceedings or actions pending before any Governmental Authority challenging the scope, ownership, validity or enforceability of the Patents and Marks or, to Seller Parent's Knowledge, the Licensed Patents and Licensed Marks and, to Seller Parent's Knowledge, no such Proceedings, which would reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, have been threatened in writing with respect to any of the Patents, Marks, Licensed Patents or Licensed Marks.

- (h) To the Seller Parent's Knowledge, the Trade Secrets that are owned, used or held by the Sellers primarily in connection with the Business have not been used, disclosed to or discovered by any Person except pursuant to written non-disclosure and/or license agreements which have not, to the Seller Parent's Knowledge, been breached.
- (i) Except as would not reasonably be expected to have a Seller Material Adverse Effect, the IT Assets operate and perform in all material respects in accordance with their documentation and functional specifications and otherwise as required by the Sellers in connection with the Business and have not materially malfunctioned or failed within the past two years in a manner that has had a material impact on the Business. To the Seller Parent's Knowledge, no Person has gained unauthorized access to the IT Assets.
- (j) Except as would not reasonably be expected to have a Seller Material Adverse Effect, Sellers have complied with all applicable Laws and contractual and fiduciary obligations relating to the collection, storage, use, transfer and any other processing of any Personally Identifiable Information collected or used by the Sellers in connection with the Business, in any manner, or to the Seller Parent's Knowledge, maintained by third parties having authorized access to such information. The Sellers have at all times taken reasonably necessary steps (including implementing and monitoring compliance with adequate measures with respect to technical and physical security) to ensure that all Personally Identifiable Information is protected against loss and against unauthorized access, use, modification or disclosure, and there has been no unauthorized access to or misuse of such Personally Identifiable Information.
- (k) Schedule 4.16(k) sets forth a complete and accurate list to the best of the Seller's knowledge (corresponding to the following subsections) of (i) each Product for which Seller or any of its Controlled Affiliates, prior to the Closing, potentially could commence sales without a valid license under any unexpired patent that covers or is alleged to cover such Product ("At-Risk"); (ii) each Paragraph IV Product that is the subject of a thirty-month statutory "stay" period under 21 U.S.C. § 355(j)(5)(B)(iii) (the "Stay Period") believed to be expiring in 2015 or 2016; and (iii) each Paragraph IV Product for which Seller or any of its Controlled Affiliates potentially could make a claim of entitlement to 180 days of "first to file" market exclusivity in accordance with 21 U.S.C. § 355 (j)(5)(B)(iv).
- (l) The consummation of the Transactions will not adversely affect any Seller's rights to any Transferred Intellectual Property, and all such Transferred Intellectual Property will be owned or available for use by Buyers on identical terms and conditions immediately subsequent to the consummation of the Transactions, without the payment of any additional consideration in connection therewith except as would reasonably be expected to cause a Seller Material Adverse Effect.

4.17 Real Property.

- (a) With respect to the real property owned by Sellers, their Subsidiaries or the Transferred Entities and Related to the Business (such property collectively, the "Transferred Owned Real Property", including such real property listed on Schedule 4.17(a)), except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, Sellers or the Transferred Entities have good and valid title to such Transferred Owned Real Property, free and clear of all Liens, other than any such Lien (i) for Taxes or governmental assessments, charges or claims of payment not yet due and payable, being contested in good faith or for which adequate accruals or reserves have been established, (ii) which is a carriers', warehousemen's, mechanics', materialmen's, repairmen's or other similar Lien arising in the Ordinary Course of Business, (iii) which is disclosed on the most recent consolidated balance sheet of Seller Parent or notes thereto or securing liabilities reflected on such balance sheet, (iv) which was incurred in the Ordinary Course of Business since the date of the most recent consolidated balance sheet of Seller Parent or (v) which would not reasonably be expected to materially impair the continued use of the applicable property for the purposes for which the property is currently being used (any such Lien described in any of clauses (i) through (v), "Seller Parent Permitted Lien"). Neither Sellers nor the Transferred Entities has received notice of any pending, and to Seller Parent's Knowledge there is no threatened, condemnation proceeding with respect to any Transferred Owned Real Property, except proceedings which would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.
- (b) Except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, (i) each lease, sublease and other agreement Related to the Business (the "Transferred Leased Real Property"), is valid, binding and in full force and effect, except that (A) enforcement may be subject to applicable bankruptcy, insolvency, examinership, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors' rights generally and (B) equitable remedies of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought and (ii) no uncured default of a material nature on the part of Sellers or the Transferred Entities, as applicable, or, to Seller Parent's Knowledge, the landlord thereunder exists with respect to any Transferred Leased Real Property. Except as would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect, Sellers or the Transferred Entities, as applicable, has a good and valid leasehold interest in or contractual right to use or occupy, subject to the terms of the lease, sublease or other agreement applicable thereto, the Transferred Leased Real Property, free and clear of all Liens, except for Seller Parent Permitted Liens.
 - 4.18 Required Vote. No vote of the holders of securities of Seller Parent is required for any Sellers to enter into or consummate the Transactions.
- 4.19 <u>Insurance</u>. Seller Parent and Sellers have all material policies of insurance covering the Business and the Business Employees, including policies of property, fire, workers' compensation, products liability, directors' and officers' liability, and other casualty and liability insurance, and such policies are in a form and amount which, to Seller Parent's Knowledge, is adequate for the operation of the Business, except, in each case, as would not reasonably be

expected to have, individually or in the aggregate, a Seller Material Adverse Effect. All such insurance policies are in full effect, no notice (in writing or otherwise) of cancellation has been received by any Seller under such policies, and there is no existing default or event which, with the giving of notice of lapse or time or both, has not had and would not reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect.

4.20 Acquired Assets.

- (a) The Acquired Assets constitute the entire right, title and interest owned by Sellers or any of their Controlled Affiliates in assets which are primarily related to, or primarily used in relation to, the Business, as at the date hereof.
- (b) The Acquired Assets, together with and the services to be provided under the Transition Services Agreement to be entered into upon Closing and other than any wholesale agreements which may be required, will constitute all assets and rights (i) necessary and sufficient for the conduct of the Business immediately following the Closing in all material respects as conducted by Sellers and their Controlled Affiliates as at the date hereof and as of the Closing and (ii) that generated the income and revenues included in the income statement included on Schedule 4.3 in all material respects and taking into account sales of Inventory in the Ordinary Course of Business and obsolete equipment.
- 4.21 <u>Brokers</u>. No broker, investment banker, agent, finder or other intermediary acting on behalf of Sellers or under the authority of Sellers is or shall be entitled to any broker's or finder's fee or any other commission or similar fee directly or indirectly in connection with any of the Transactions other than J.P. Morgan, the fees and expenses of which shall be the sole responsibility of Seller Parent.
- 4.22 No Undisclosed Liabilities. Except (a) as disclosed, reflected or reserved against in Seller Parent's consolidated balance sheet (or the notes thereto) as of December 31, 2014 included in the Seller Parent SEC Documents filed or furnished prior to the date hereof (the "Seller Parent Reference Balance Sheet"), (b) for liabilities of the Transferred Group or the Acquired Assets as of December 31, 2014 that were not required to be disclosed, reflected or reserved against in the Seller Parent Reference Balance Sheet under U.S. GAAP, (c) for liabilities of the Transferred Group or the Acquired Assets incurred in the Ordinary Course of Business since December 31, 2014, (d) for liabilities of the Transferred Group or the Acquired Assets expressly contemplated by this Agreement to be incurred and (e) for liabilities of the Transferred Group or the Acquired Assets which have been discharged or paid in full in the Ordinary Course of Business, there are no Liabilities of the Business other than those which, individually or in the aggregate, would not reasonably be expected to have a Seller Material Adverse Effect. For purposes of this Section 4.22, the term "liabilities of the Business" shall not include obligations of Seller Parent or any Seller Parent Subsidiary to perform under or comply with any applicable Law, action, judgment or Contract, but would include such liabilities and obligations if there has been a default or failure to perform or comply by Seller Parent or any Seller Parent Subsidiary with any such Law, action, judgment or Contract if such default or failure would, with or without the giving of notice or passage of time or both, reasonably be expected to result in a monetary obligation.

4.23 Securities Matters.

- (a) Seller Parent has such knowledge, sophistication and experience in financial and business matters that it is capable of evaluating the merits and risks of the receipt of the Stock Consideration and of protecting its interests in connection herewith. Seller Parent has the ability to bear the economic risk of this investment, including complete loss of the investment.
- (b) Seller Parent is acquiring the Stock Consideration for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, any distribution thereof, and has no present intention of selling, granting any participation in or otherwise distributing the same. Seller Parent understands that the Stock Consideration has not been registered under the Securities Act, by reason of a specific exemption from the registration provisions of the Securities Act which depends upon, among other things, the bona fide nature of the investment intent and the accuracy of Seller Parent's representations as expressed in this Section 4.23.
- (c) Seller Parent understands that the Stock Consideration is characterized as "restricted securities" under the U.S. federal securities laws inasmuch as they are being acquired from Buyer Parent in a transaction not involving a public offering and that under such laws and applicable regulations the Stock Consideration may be resold without registration under the Securities Act only in certain limited circumstances. Seller Parent acknowledges that the Stock Consideration must be held indefinitely unless a sale of the Stock Consideration is subsequently registered under the Securities Act or an exemption from such registration is available. Seller Parent is aware of the provisions of Rule 144 promulgated under the Securities Act which permit limited resale of shares purchased in a private placement or shares owned by certain Persons associated with Buyer Parent subject to the satisfaction of certain conditions. Seller Parent is aware that the Stock Consideration is subject to restrictions on resale under Israeli securities laws.
- 4.24 <u>Material Suppliers</u>. Since December 31, 2014, except as would not reasonably be expected to have a Seller Material Adverse Effect, no supplier of materials or supplies to the Business whose aggregate projected payments by the Business for such items for calendar year ending December 31, 2015 is expected to exceed \$50 million has notified the Sellers in writing that it will cease providing, or will be unable to fulfill in any material respect, orders for materials or supplies to the Business.

4.25 Transferred Inventory; Recalls.

- (a) Except as would not reasonably be expected to have a Seller Material Adverse Effect, as of the Closing, the transferred Inventory and all other Inventory that would be transferred Inventory but for the fact that it is held by a member of the Transferred Group, manufactured by Seller Parent or its Affiliates shall be in good and merchantable condition in all material respects, shall have been manufactured, stored and handled by Seller Parent and its Affiliates prior to Closing in compliance with applicable cGMPs and shall conform to the specifications for the manufacture, storage and handling of such Inventory.
- (b) Since January 1, 2014, there have been no material recalls or market withdrawals or replacements (voluntary or involuntary) with respect to any Product.

- 4.26 <u>Intercompany Arrangements</u>. At Closing, there shall be no agreements or arrangements, written or unwritten, of any kind, between any Seller or any of their Affiliates (excluding the Transferred Entities and their Subsidiaries), on the one hand, and the Transferred Entities and their Subsidiaries or the Business, on the other hand (other than the Transferred Receivables or as otherwise contemplated hereby).
- 4.27 <u>Anda Companies</u>. The business of the Anda Companies is primarily distribution of products on behalf of the Business, the Retained Business and third party suppliers.
 - 4.28 <u>Uteron Business</u>. The business of the Uteron Business is primarily related to women's health products.
- 4.29 No Other Representations. Except for the representations and warranties contained in Article V, Seller Parent acknowledges that neither Buyer Parent nor any Representative of Buyer Parent makes, and Seller Parent acknowledges that it has not relied upon or otherwise been induced by, any other express or implied representation or warranty with respect to Buyer Parent or any of its Affiliates.
- 4.30 Disclaimer. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS ARTICLE IV OR IN A LOCAL TRANSFER AGREEMENT, SELLERS MAKE NO REPRESENTATION OR WARRANTY AS TO THE BUSINESS, THE ACQUIRED ASSETS OR THE PRODUCTS, AND BUYER PARENT ACKNOWLEDGES AND AGREES THAT, EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS ARTICLE IV, SELLERS ARE SELLING AND CONVEYING THE BUSINESS, THE ACQUIRED ASSETS AND THE PRODUCTS ON AN "AS IS, WHERE IS" BASIS, WITHOUT RECOURSE (OTHER THAN IN THE CASE OF FRAUD) AND WITHOUT ANY REPRESENTATION OR WARRANTY OF MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE, NONINFRINGEMENT OR ANY OTHER IMPLIED OR EXPRESS WARRANTIES WHATSOEVER.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF BUYER PARENT

Except as disclosed in the Buyer Parent SEC Documents and forms, documents and reports of Buyer Parent, in each case filed or furnished with the SEC since December 31, 2014 and prior to the date of this Agreement (including exhibits and other information incorporated by reference therein provided that such exhibits and other information incorporated by reference therein is publicly available) to the extent the relevance of such disclosure is reasonably apparent (but excluding any disclosures set forth in any "risk factors" section, any disclosures in any "forward looking statements" section and any other disclosures included therein to the extent they are predictive, general, cautionary or forward-looking in nature) or in the applicable section of the disclosure letter delivered by Buyer Parent to Sellers immediately prior to the execution of

this Agreement (the "Buyer Parent Disclosure Letter") (it being agreed that disclosure of any item in any section of the Buyer Parent Disclosure Letter shall be deemed disclosure with respect to any other section of this Agreement to which the relevance of such item is reasonably apparent), Buyer Parent represents and warrants to Sellers as set forth below.

5.1 Qualification, Organization, Subsidiaries, etc. Each of Buyer Parent and its Subsidiaries is a legal entity duly organized, validly existing and, where relevant, in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so qualified or, where relevant, in good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to have a Buyer Material Adverse Effect. Buyer Parent has filed with the SEC, prior to the date of this Agreement, complete and accurate copies of the Articles of Association of Buyer Parent as amended to the date hereof.

5.2 Share Capital.

(a) The authorized share capital of Buyer Parent consists of 2,500,000,000 shares of NIS 0.10 par value each. As of July 26, 2015 (the "Buyer Parent Capitalization Date"), (i)(A) 850,349,891 shares of Buyer Parent Stock were issued and outstanding and (B) 110,016,092 shares of Buyer Parent Stock were held in treasury, (ii) 44,009,508 shares of Buyer Parent Stock were reserved for issuance pursuant to the Buyer Parent Equity Plans, and (iii) 3,956,000 shares of Buyer Parent were reserved for issuance pursuant to Buyer Parent's 0.25% convertible senior debentures due 2026 and (iv) no Buyer Parent Preferred Shares were issued and outstanding. All the outstanding Buyer Parent Stock are, and all shares of Buyer Parent Stock reserved for issuance as noted above shall be, when issued in accordance with the respective terms thereof, duly authorized, validly issued, fully paid and non-assessable and free of pre-emptive rights.

(b) Except as set forth in Section 5.2(a) above, as of the date hereof: (i) Buyer Parent does not have any shares of capital stock issued or outstanding other than shares of Buyer Parent Stock that have become outstanding after the Buyer Parent Capitalization Date, but were reserved for issuance as set forth in Section 5.2(a) above, and (ii) there are no outstanding subscriptions, options, warrants, puts, calls, exchangeable or convertible securities or other similar rights, agreements or commitments relating to the issuance of shares of capital stock to which Buyer Parent or any of Buyer Parent's Subsidiaries is a party obligating Buyer Parent or any of Buyer Parent's Subsidiaries to (A) issue, transfer or sell any shares of capital stock or other equity interests of Buyer Parent or securities convertible into or exchangeable for such shares or equity interests (in each case other than to Buyer Parent or a wholly owned Subsidiary of Buyer Parent); (B) grant, extend or enter into any such subscription, option, warrant, put, call, exchangeable or convertible securities or other similar right, agreement or commitment in respect of equity interests of Buyer Parent; or (C) redeem or otherwise acquire any shares of capital stock of Buyer Parent or other equity interests of Buyer Parent.

- (c) Neither Buyer Parent nor any Buyer Parent Subsidiary has outstanding bonds, debentures, notes or other similar obligations, the holders of which have the right to vote (or which are convertible into or exercisable for securities having the right to vote) with the shareholders of Buyer Parent on any matter.
- (d) There are no voting trusts or other agreements or understandings to which Buyer Parent or any of its Subsidiaries is a party with respect to the voting of the shares or other equity interest of Buyer Parent.
- (e) The Stock Consideration shall be, at the time of issuance, duly authorized, validly issued, fully paid and non-assessable and free of pre-emptive rights.

5.3 Corporate Authority Relative to this Agreement; No Violation.

- (a) Buyer Parent has all necessary corporate power and authority to execute, deliver, perform its obligations under and consummate the transactions contemplated by this Agreement and the Ancillary Agreements, to the extent it will be a party thereto. The consummation of the transactions contemplated hereby and thereby and the execution and delivery of this Agreement and the Ancillary Agreements, to the extent it will be a party thereto, and the performance of all of its obligations hereunder and thereunder have been duly authorized by Buyer Parent. The execution, delivery and performance by Buyer Parent of this Agreement and the Ancillary Agreements, to the extent it will be a party thereto, are not prohibited or limited by, and shall not result in a breach of or a default under, any provision of the Organizational Documents of Buyer Parent, or a material breach or a material default under any material Contract binding on Buyer Parent, or of any applicable Order. This Agreement has been duly executed and delivered by Buyer Parent, and the Ancillary Agreements will, at the Closing, be duly executed and delivered by Buyers to the extent Buyers are party thereto, and, assuming due and valid authorization, execution and delivery by each other Party thereto (other than any other Buyer), this Agreement constitutes, and when executed and delivered by Buyer Parent, to the extent Buyers are party thereto, the Ancillary Agreements will constitute, legal, valid and binding obligations of Buyers, as applicable enforceable against Buyers in accordance with their respective terms, except that the enforcement hereof or thereof may be limited by (x) bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors' rights generally and (y) general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law). Buyer Parent has all necessary corporate power and authority to cause each Buyer that is a party to any Ancillary A
- (b) Each Buyer, as applicable, has all necessary corporate power and authority to execute, deliver and perform its obligations under and consummate the transactions contemplated by this Agreement and the Ancillary Agreements to which it is a party. The execution and delivery of such agreement(s) and the performance of all of its obligations thereunder and the consummation of the transactions contemplated thereunder have been duly authorized by each such Buyer. The execution, delivery and performance by each Buyer of this Agreement and the Ancillary Agreement(s) to which it is a party are not prohibited or limited by, and shall not result in a breach of or a default under, any provision of the Organizational

Documents of such Buyer, or a material breach or a material default under any material Contract binding on such Buyer, or of any applicable Order, except for such breaches and defaults of Contracts which would not, individually or in the aggregate, reasonably be expected to prevent or materially delay the consummation of the Transactions. The Ancillary Agreements, upon their delivery at or prior to Closing, will have been duly executed and delivered by each Buyer that is a party thereto and constitute the legal, valid and binding obligation of each Buyer that is a party thereto (other than any other Buyer), enforceable against each such Buyer in accordance with their respective terms, except as enforceability may be limited or affected by applicable bankruptcy, insolvency, moratorium, reorganization or other Laws of general application relating to or affecting creditors' rights generally. Except as would not reasonably be expected to have, individually or in the aggregate, a Buyer Material Adverse Effect (disregarding clause (iv) of the proviso of the definition of Buyer Material Adverse Effect), or materially delay or impair the Transactions, none of the execution, delivery or performance of this Agreement and the Ancillary Agreements by the applicable Buyer, will: (a) result in a modification, violation or breach of, or constitute (with or without notice or lapse of time or both) a default (or give rise to any right, including, but not limited to, any right of termination, amendment, cancellation or acceleration of any obligation or loss of any benefit) under, any of the terms, conditions or provisions of any Contract of Buyer Parent; or (c) violate any Permit, Regulatory Registration, Order or Law applicable to such Buyer or give any Governmental Authority or other Person the right to challenge any of the Transactions or to exercise any remedy or obtain any relief under, any Law or any Order to which any Buyer is subject.

5.4 Reports and Financial Statements.

(a) From January 1, 2014 through the date of this Agreement, Buyer Parent has filed or furnished all forms, documents and reports required to be filed or furnished prior to the date hereof by it with the SEC (the "Buyer Parent SEC Documents"). As of their respective dates, or, if amended, as of the date of (and giving effect to) the last such amendment, the Buyer Parent SEC Documents complied in all material respects with the requirements of the Exchange Act and the applicable rules and regulations promulgated thereunder, and none of the Buyer Parent SEC Documents contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

(b) The consolidated financial statements (including all related notes and schedules) of Buyer Parent included in the Buyer Parent SEC Documents when filed complied as to form in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto in effect at the time of such filing and fairly present in all material respects the consolidated financial position of Buyer Parent and its consolidated Subsidiaries, as at the respective dates thereof, and the consolidated results of their operations and their consolidated cash flows for the respective periods then ended (subject, in the case of the unaudited statements, to normal year-end audit adjustments and to any other adjustments described therein, including the notes thereto) in conformity with U.S. GAAP applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto).

- 5.5 No Undisclosed Liabilities. Except (a) as disclosed, reflected or reserved against in Buyer Parent's consolidated balance sheet (or the notes thereto) as of March 31, 2015 included in the Buyer Parent SEC Documents filed or furnished prior to the date hereof, (b) for liabilities incurred in the ordinary course of business since March 31, 2015, which would not, individually or in aggregate, reasonably be likely to have a Buyer Material Adverse Effect, (c) as expressly contemplated by this Agreement to be incurred and (d) for Liabilities which have been discharged or paid in full in the ordinary course of business, neither Buyer Parent nor any Buyer Parent Subsidiary has any Liabilities other than those which, individually or in the aggregate, would not reasonably be expected to have a Buyer Material Adverse Effect. For purposes of this Section 5.5, the term "liabilities" shall not include obligations of Buyer Parent or any Buyer Parent Subsidiary to perform under or comply with any applicable Law, action, judgment or Contract, but would include such liabilities and obligations if there has been a default or failure to perform or comply by Buyer Parent or any Buyer Parent Subsidiary with any such Law, action, judgment or Contract if such default or failure would, with or without the giving of notice or passage of time or both, reasonably be expected to result in a monetary obligation.
- 5.6 <u>Compliance with Law.</u> Buyer Parent and each of Buyer Parent's Subsidiaries are in compliance with and are not in default under or in violation of any Laws, applicable to Buyer Parent, such Subsidiaries or any of their respective properties or assets, except where such non-compliance, default or violation would not reasonably be expected to prevent or materially delay the consummation of the Transactions or to have, individually or in the aggregate, a Buyer Material Adverse Effect.

5.7 Absence of Certain Changes or Events.

- (a) From December 31, 2014 through the date of this Agreement, there has not occurred any event that has had, or would reasonably be expected to have, individually or in the aggregate, a Buyer Material Adverse Effect.
- (b) From December 31, 2014 through the date of this Agreement, (i) the business of Buyer Parent and its Affiliates has been conducted in the ordinary course of business and (ii) neither Buyer Parent nor any of its Subsidiaries have taken any action that would constitute a breach of Section 6.1(e) had such action been taken after the execution of this Agreement.
- 5.8 Investigations; Litigation. As of the date hereof, (a) there is no investigation or review pending (or, to Buyer Parent's Knowledge, threatened) by any Governmental Authority with respect to Buyer Parent or any of Buyer Parent's Subsidiaries or any of their respective properties, rights or assets, and (b) there are no claims, actions, suits or proceedings pending (or, to Buyer Parent's Knowledge, threatened) against Buyer Parent or any of Buyer Parent's Subsidiaries or any of their respective properties, rights or assets before, and there are no orders, judgments or decrees of, any Governmental Authority, which, in the case of clause (a) or (b), would reasonably be expected to prevent or materially delay the consummation of the Transactions or to have, individually or in the aggregate, a Buyer Material Adverse Effect.
- 5.9 Consents and Governmental Approvals. Assuming the truth and accuracy of the representations and warranties made by Sellers in Article IV, and except for (a)(i) any HSR

Filing with the FTC or the Antitrust Division or (ii) any filing under any Foreign Antitrust Laws set forth on Schedule 5.9, in each case to the extent required, and (b) as would not be reasonably be expected to have, individually and in the aggregate, a Buyer Material Adverse Effect, no Consent from any Governmental Authority in connection with the execution and delivery of this Agreement or any Ancillary Agreements or the consummation or performance of the Transactions is required for Buyers to consummate the Transactions.

5.10 Tax Matters.

- (a) Except as would not, individually or in the aggregate, reasonably be expected to have a Buyer Material Adverse Effect:
- (i) all Tax Returns that are required to be filed by or with respect to Buyer Parent or any of its Subsidiaries have been timely filed (taking into account any extension of time within which to file), and all such Tax Returns are true, complete and accurate;
- (ii) since July 1, 2013, neither Buyer Parent nor any of its Subsidiaries has constituted a "distributing corporation" or a "controlled corporation" (in each case, within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code (or any similar provision of state, local, or non-U.S. Law);
- (iii) to Buyer Parent's Knowledge, Buyer Parent is neither (i) a passive foreign investment company, nor (ii) a controlled foreign corporation, each within the meaning of the Code;
- (iv) Buyer Parent and its Subsidiaries have paid all Taxes due and owing by any of them, including any Taxes required to be withheld from amounts owing to any employee, creditor, or third party (in each case, whether or not shown on any Tax Return), other than Taxes for which adequate reserves have been established in accordance with GAAP on the financial statements of Parent and its Subsidiaries; and
- (v) none of Buyer Parent or any of its Subsidiaries is a party to any Tax-Sharing Agreement or has any liability for Taxes of any Person (other than Buyer Parent or any of its Subsidiaries) under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local, or non-U.S. Law) or as transferee or successor.
 - (b) Buyer Parent is, and at all times since its formation has been, treated as a foreign corporation for U.S. federal income tax purposes.
- 5.11 <u>Required Vote</u>. No vote of the holders of securities of Buyer Parent is required for Buyer Parent to enter into or consummate the Transactions.

5.12 Financing.

(a) As of the date of the Debt Commitment Letter, neither Buyer Parent nor any of its Affiliates will have entered into any agreement, side letter or other arrangement relating to the financing of the Transactions that could affect the availability of the Debt

Financing on the Closing Date, other than as described in the Debt Commitment Letter and any fee letters or engagement letters related to the Debt Commitment Letter. As of the date of the Debt Commitment Letter, the Debt Commitment Letter will be in full force and effect and will represent a valid, binding and enforceable obligation of Buyer Parent and to Buyer Parent's Knowledge each other party thereto, to provide the financing contemplated thereby subject only to the satisfaction or waiver of the Debt Financing Conditions and subject to the qualification that such enforceability may be limited by bankruptcy, insolvency, reorganization or other laws of general application relating to or affecting rights of creditors and that equitable remedies, including specific performance, are discretionary and may not be ordered. Buyer Parent will have fully paid (or caused to be paid) any and all commitment fees and other amounts that are due and payable on or prior to the date of the Debt Commitment Letter in connection with the Debt Financing. As of the date of the Debt Commitment Letter, assuming the accuracy of the representations and warranties set forth in Article IV such that the condition set forth in Section 10.3(a) is satisfied, no event will have occurred which, with or without notice, lapse of time or both, would reasonably constitute a breach or default on the part of Buyer Parent or, to Buyer Parent's Knowledge, any other party thereto under the Debt Commitment Letter. As of the date of the Debt Commitment Letter, assuming the accuracy of the representations and warranties set forth in Article IV such that the condition set forth in Section 10.3(a) is satisfied and performance by Seller Parent in all material respects of its obligations under Section 6.1, Section 9.4 and Section 9.14 of this Agreement, Buyer Parent will have no reason to believe that it or any other party thereto will be unable to satisfy on a timely basis its obligations under the Debt Commitment Letter. As of the date of the Debt Commitment Letter, there will be no conditions precedent related to the funding of the full amount of the Debt Financing, other than the Debt Financing Conditions. Assuming the accuracy of the representations and warranties set forth in Article IV such that the condition set forth in Section 10.3(a) is satisfied and the satisfaction of the conditions set forth in Section 10.3(c) and performance in all material respects by Sellers of their obligations under Section 6.1, Section 9.4 and Section 9.14 of this Agreement, as of the date of the Debt Commitment Letter, Buyer Parent will have no reason to believe that (i) any of the Debt Financing Conditions will not be satisfied or (ii) the Debt Financing will not be made available to Buyer Parent on the Closing Date.

(b) Assuming (i) the accuracy of the representations and warranties set forth in Article IV such that the condition set forth in Section 10.3(a) is satisfied, (ii) the satisfaction of the conditions set forth in Section 10.3(c), (iii) the performance in all material respects by Sellers of their obligations under Section 6.1, Section 9.4 and Section 9.14 of this Agreement Buyer Parent will have at the Closing, directly or through one or more affiliates, all funds necessary to consummate the Transactions at the Closing Date, including the making of all required payments in connection with the Transactions at the Closing Date, including payment of the Cash Consideration and all other amounts to be paid pursuant to this Agreement and associated costs and expenses of the Transactions on the Closing Date. Notwithstanding anything to the contrary contained herein, in no event shall the receipt or availability of any funds or financing by Buyer Parent or any of its Affiliates be a condition to any of Buyer Parent's obligations hereunder.

5.13 No Other Representations. Except for the representations and warranties contained in Article IV, Buyer Parent acknowledges that neither Sellers nor any Representative of Sellers makes, and Buyer Parent acknowledges that it has not relied upon or otherwise been induced by, any other express or implied representation or warranty with respect to Sellers or any of their Affiliates.

ARTICLE VI

COVENANTS

6.1 Conduct of the Business and Buyer Parent Prior to the Closing.

(a) Unless Buyer Parent consents in writing (which consent shall not be unreasonably withheld, delayed or conditioned), and except (x) as required by Law or expressly contemplated by this Agreement or the Ancillary Agreements, (y) as is necessary or advisable to implement the Pre-Closing Reorganization Plan that does not result in any impairment to Acquired Assets or the Transferred Group other than those that are immaterial or (z) exclusively with respect to the Excluded Assets or the Excluded Liabilities in a manner that does not result in any Assumed Liabilities or other adverse effect on the Business or the Acquired Assets or adversely affect the ability of Sellers in any material respect to promptly consummate the Transactions contemplated by the Agreement, from and after the date of this Agreement until the earlier of (i) the Closing Date, or (ii) the termination of this Agreement pursuant to Section 11.1 (the "Pre-Closing Period"), Sellers shall, and shall cause each of their respective Subsidiaries to, (A) conduct the operations of the Business in the Ordinary Course of Business (including, for the avoidance of doubt, any "at-risk" launches of Generic Drugs in the ordinary course to the extent prior written notice is provided to Buyer Parent) and in compliance in all material respects with all applicable Law and FDA requirements, (B) use commercially reasonable efforts to maintain and preserve intact the Business in all material respects, (C) use commercially reasonable efforts to maintain the ordinary and customary relationships of the Business with its material customers, distributors, suppliers, licensors, licensees, employees, regulators and independent contractors, (D) use commercially reasonable efforts (including payments of the filing, issue, registration, renewal, maintenance or other official registry fees for the registered Transferred Intellectual Property as any of such fees becomes due and payable and avoiding extension fees) to maintain the existence of, validity, enforceability and rights of Sellers in, to or under the Transferred Intellectual Property, Permits and Regulatory Registrations (including any pending applications for new or renewed Permits or Regulatory Registrations), (E) except with respect to the disposition of any Inventory and equipment in the Ordinary Course of Business, use commercially reasonable efforts to maintain the tangible Acquired Assets in reasonable order and condition (normal wear and tear excepted) and (F) except in the Ordinary Course of Business, as part of the Pre-Closing Reorganization (to the extent that it does not result in any impairment to the Acquired Assets or the Transferred Group other than those that are immaterial), or as required by Law or a Governmental Authority, not make or change any material Tax election with respect to the Acquired Assets, Transferred Group or the Business, change any Tax accounting period for purposes of a material Tax or material method of Tax accounting with respect to the Acquired Assets, Transferred Group or the Business, file any material amended Tax Return with respect to the Acquired Assets, Transferred Group or the Business, settle or compromise any audit or proceeding relating to a material amount of Taxes with respect to the Acquired Assets, Transferred Group or the Business, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes relating to the Acquired Assets, Transferred Group or the Business, enter into any "closing agreement" within

the meaning of Section 7121 of the Code (or any similar provision of state, local, or non-U.S. Law) with respect to any material Tax relating to the Acquired Assets, Transferred Group or the Business, or surrender any right to claim a material Tax refund with respect to the Acquired Assets, Transferred Group or the Business, in each case only to the extent such action would affect the Acquired Assets, Transferred Group or the Business in a Post-Closing Tax Period.

(b) Without limiting the generality of Section 6.1(a), unless Buyer Parent consents in writing (which consent shall not be unreasonably withheld, delayed or conditioned), or (i) except as required by Law or as expressly contemplated by this Agreement or the Ancillary Agreements, (ii) as is necessary or advisable to implement the Pre-Closing Reorganization Plan or any integration plan which has been approved by the Seller Parent prior to the date hereof (in each case to the extent that it does not result in any impairment to the Acquired Assets or the Transferred Group other than those that are immaterial), (iii) exclusively with respect to the Excluded Assets or the Excluded Liabilities in a manner that does not result in any Assumed Liabilities or other adverse effect on the Business or the Acquired Assets or adversely affect the ability of Sellers in any material respect to promptly consummate the Transactions contemplated by the Agreement or (iv) as set forth on Schedule 6.1(b), during the Pre-Closing Period, Sellers shall not, and shall cause each of their respective Subsidiaries to not, take any of the following actions:

(i) transfer, sell, lease, enter into any sale lease back (or similar arrangement), license or otherwise convey or dispose of, or create any Liens (other than Permitted Liens) on any of the Acquired Assets, the Transferred Group Assets, other than (A) licenses entered into in the Ordinary Course of Business and (B) sales of Inventory in the Ordinary Course of Business and sales of obsolete equipment;

(ii) except as required by Law, as contemplated by this Agreement or the Ancillary Agreements, as required by the terms and conditions of any Benefit Plan in existence as of the date hereof, or as set forth on Schedule 6.1(b)(ii), directly or indirectly, (A) change the position or job title of any Business Employee who is (i) at the level of Senior Director or above, (ii) has aggregate annual salary or wage rate or consulting fees and target cash bonus opportunity in excess of \$250,000 or (iii) is one of the top 200 most highly paid Business Employees (based on aggregate annual base salary and target annual bonus opportunity) (any such Business Employee, a "Restricted Business Employee"), (B) grant to any Restricted Business Employee any increase in compensation or benefits, other than increases in base salary of up to 5% per individual and corresponding increases in target annual bonus opportunity, (C) grant to any Business Employees who is not a Restricted Business Employees (a "Non-Restricted Business Employee") any increase in compensation or benefits, other than increases of not more than 10% in the aggregate per Non-Restricted Business Employee, in the Ordinary Course of Business, (D) increase or grant to any Business Employee any severance, termination, change in control, stay put or retention payment, (E) grant any Seller Parent Incentive Award to any Business Employee other than to newly-hired Business Employees or in connections with promotions of Business Employees in the Ordinary Course of Business (such additional awards, "Permitted Awards"), (F) enter into or amend any employment, consulting, indemnification, severance, collective bargaining or termination agreement with any Business Employee, (G) establish, adopt or enter into or materially amend, any Transferred Entity Benefit Plan or any Seller Benefit Plan as it applies to the Business Employees to the extent such action could

reasonably be expected to result in an additional Liability to the Transferred Entities, (H) without Buyer Parent's prior consent, which shall not be unreasonably withheld, (i) hire any employee or engage any independent contractor (who is a natural person) to provide services to the Business who, upon such employment or engagement, would be a Restricted Business Employee, or (ii) terminate the employment or engagement of any Restricted Business Employee, other than for cause, (I) without Buyer Parent's prior consent, which shall not be unreasonably withheld, transfer any employee other than pursuant to Section 7.2 or change an employee's duties so as to cause them to become or cease to be a Business Employee, (J) take any action to accelerate any payments, rights or benefits, or make any material determinations under any Seller Benefit Plan to the extent such action or determination affects a Business Employee, or fund or in any other way secure the payment of compensation or benefits under any Seller Benefit Plan or Transferred Entity Benefit Plan, (K) materially change any actuarial or other assumptions used to calculate funding obligations with respect to any Transferred Entity Benefit Plan that is required by applicable Law to be funded or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by U.S. GAAP, (L) become a party to, establish, adopt, amend, commence participation in or terminate any collective bargaining agreement or other Contract with a labor union, works council or similar organization or (M) loan or advance money or other property to any Business Employee, other than employee expense advances or 401(k) plan loans made in the Ordinary Course of Business. For the avoidance of doubt, the foregoing restrictions shall not restrict or prohibit Sellers from entering into with, or making available to, newly hired employees or to employees in the context of promotions based on job performance or workplace r

(iii) make any change in accounting principles (including for Tax purposes) materially affecting the reporting of the Acquired Assets, Liabilities or results of operations of any member of the Transferred Group or the Business, other than as required by Law, a Governmental Authority, U.S. GAAP or any interpretation thereof;

(iv) make any loans, advances or capital contributions to, or investments in, any unaffiliated Person (other than pursuant to any existing obligations under any Contract) other than the advancement of trade credit to customers and distributors or expenses to employees in the Ordinary Course of Business;

(v) authorize for issuance, issue, sell, deliver or agree or commit to issue, sell or deliver any Equity Participations in any Transferred Entity (other than in each case, to another member of the Transferred Group)(whether through the issuance or granting of options, warrants, convertible or exchangeable securities, commitments, subscriptions, rights to purchase or otherwise);

(vi) amend any of the Transferred Entities' Organizational Documents except as may be required by Law (and in such event, only after providing five (5) days' prior notice of such amendment, including a copy of the proposed amendment, to Buyer Parent);

(vii) (x) materially modify, materially amend or terminate any Seller Material Contract or waive, release, assign or fail to exercise or pursue any material rights or Claims thereunder, (y) enter into a Contract that would have been a Seller Material Contract if entered into prior to the date hereof or (z) enter into a Contract that would have been a Seller Material Contract if entered into prior to the date hereof and contains a non-compete or exclusivity provision that would apply to Affiliates of a Transferred Entity (other than the Transferred Group), except, in the case of clause (x) or (y), where (A) such action is in the Ordinary Course and Business and (B) Seller Parent has consulted with Buyer Parent before taking such action;

(viii) (A) other than in relation to ANDA litigation in the Ordinary Course of Business pay, discharge, settle, compromise, satisfy or consent to any entry of any judgment with respect to, any Claim that (I) results in any material restriction on the Business or any member of the Transferred Group after the Closing greater than \$10 million individually or \$50 million in the aggregate, provided, that Seller Parent will consult with Buyer Parent prior to taking any such action with respect to the matters set forth on Schedule 6.1(b)(viii) (B) cancel any Indebtedness owed to any member of the Transferred Group or the Business to the extent such Indebtedness would constitute an Acquired Asset, (C) waive, release or assign any material Claims or rights of the Business, (D) waive any benefits of, or agree to modify in any respect, or, subject to the terms hereof, fail to enforce, or consent to any matter with respect to which Consent is required under, any confidentiality or similar Business Contract or (E) commence, join or make an appeal with respect to a Claim relating to the Business or any member of the Transferred Group other than (I) for the routine collection of bills, (II) in such cases where Seller Parent in good faith determines that failure to commence suit would result in the material impairment of a valuable aspect of the Business, or (III) pursuant to this Agreement;

(ix) incur any Indebtedness, or enter into any financing or guarantee arrangement, agreement or undertaking with any customer of the Business or any financial institution, leasing company or similar business, which would constitute an Assumed Liability or be a Liability of a member of the Transferred Group, other than in the Ordinary Course of Business;

(x) make any commitment for capital expenditures with respect to the Business which are in excess of \$10 million individually or \$50 million in the aggregate in any six (6) month period;

(xi) other than in the Ordinary Course of Business, (A) accelerate the delivery or sale of Products, or (B) offer discounts on sale of Products or premiums on purchase of raw materials;

(xii) except in the Ordinary Course of Business, offer any rebates, discounts, promotions or credits, make any change to any promotional programs or make any change in the manner in which Sellers generally extend rebates, discounts or credit to, or otherwise similarly deal with, customers with respect to the Products or the Business;

- (xiii) other than in the Ordinary Course of Business, change its collection practices in respect of accounts receivable or payment practices in respect of accounts payable, in each case with respect to the Products or the Business;
- (xiv) vary any inventory practices with respect to any Product (including Product inventory held by or on behalf of any of the Sellers' or their respective Affiliates' wholesalers) in any respect materially inconsistent with past practice, or fail to produce and maintain inventory levels and amounts substantially consistent with past practice; or
 - (xv) agree, in writing or otherwise, to take any of the foregoing actions.
- (c) Seller Parent shall not knowingly take or knowingly permit any of its Subsidiaries to take any action that is reasonably likely to prevent or materially interfere with or delay the consummation of the Transactions or result in any of the conditions to the Closing set forth in <u>Article X</u> not being satisfied.
- (d) Nothing contained in this Agreement shall give Buyer Parent, directly or indirectly, the right to control or direct Seller Parent's operations prior to the Closing. Prior to the Closing, the management of Seller Parent shall exercise, consistent with and in accordance with the terms and conditions of this Agreement, complete control and supervision over the operations of Seller Parent.
- (e) Buyer Parent shall not knowingly take or knowingly permit any of its Subsidiaries to take any action that is reasonably likely to prevent or materially interfere with or delay the consummation of the Transactions or result in any of the conditions to the Closing set forth in <u>Article X</u> not being satisfied.
- (f) Unless Seller Parent consents in writing (which consent shall not be unreasonably withheld, delayed or conditioned), except as required by Law or as expressly contemplated by this Agreement, the Ancillary Agreements or the Pre-Closing Reorganization Plan, during the Pre-Closing Period, Buyer Parent shall not:
- (i) amend or modify the Organizational Documents of Buyer Parent in any manner that would reasonably be expected to adversely affect the value of the Stock Consideration;
- (ii) split, combine, reduce or reclassify any of its issued or unissued shares, or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for, its shares in any manner that would reasonably be expected to adversely affect the value of the Stock Consideration:
- (iii) authorize or pay any dividends on or make any distribution with respect to its outstanding shares (whether in cash, assets, stock or other securities of Buyer Parent), except as set forth on Schedule 6.1(f):
- (iv) except in the Ordinary Course of Business, as part of the Pre-Closing Reorganization, or as required by Law or a Governmental Authority, make or change any material Tax election, change any Tax accounting period for purposes of a material Tax or

material method of Tax accounting, file any material amended Tax Return, settle or compromise any audit or proceeding relating to a material amount of Taxes, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes, enter into any "closing agreement" within the meaning of Section 7121 of the Code (or any similar provision of state, local, or non-U.S. Law) with respect to any material Tax, or surrender any right to claim a material Tax refund;

(v) except in the ordinary course of business consistent with past practice, repurchase, redeem or otherwise acquire any Equity Participations of Buyer Parent or any Buyer Parent Subsidiary, or any other equity interests or any rights, warrants or options to acquire any such Equity Participations, other than (A) the acquisition by Buyer Parent of shares of Buyer Parent Stock in connection with the surrender of shares of Buyer Parent Stock by holders of Equity Participations in order to pay the exercise price thereof, (B) the withholding of shares of Buyer Parent Stock to satisfy Tax obligations with respect to awards granted pursuant to any Buyer Parent Benefit Plan (or any successor thereto) or pursuant to individual equity compensation award agreements, (C) the acquisition by Buyer Parent of Equity Participations of Buyer Parent in connection with the forfeiture of such Equity Participations or (D) as required by Buyer Parent's equity or equity-based incentive compensation plans or awards as in effect on the date of this Agreement; or

(vi) agree, in writing or otherwise, to take any of the foregoing actions.

6.2 Antitrust Filings and Actions.

(a) Filings and Other Actions. Subject to the proviso in the penultimate sentence of this Section 6.2(a), Seller Parent and Buyer Parent shall cooperate with each other and use (and shall cause their respective Affiliates to cooperate and use) their respective reasonable best efforts to take or cause to be taken all actions, and to do or cause to be done all things necessary, proper or advisable on its part under this Agreement and applicable Law to consummate and make effective the Transactions as soon as reasonably practicable, including preparing and filing as promptly as reasonably practicable and advisable all documentation to effect all necessary notices, reports and other filings and to obtain as promptly as reasonably practicable all Consents necessary or advisable to be obtained from any Person, including any Governmental Authority, and to lift any injunction or other legal bar in order to consummate the Transactions. Without limiting the foregoing, each of Seller Parent and Buyer Parent undertakes and agrees to file (or cause their respective Affiliates to file, as applicable) as soon as reasonably practicable and advisable, a Notification and Report Form regarding the Transactions as and to the extent required by the HSR Act (the "HSR Filing") with each of the U.S. Federal Trade Commission (the "FTC") and the Antitrust Division of the U.S. Department of Justice (the "Antitrust Division") and to promptly submit any other filings required to be made under any Foreign Antitrust Laws. Each of Seller Parent and Buyer Parent shall (and shall cause their respective Affiliates to) (i) respond as promptly as reasonably practicable and advisable to any inquiries received from any Governmental Authority for additional information or documentation and to all inquiries and requests received from any Governmental Authority to delay the Transactions, except (x) in the case of Seller Parent, with the prior written Consent of

Buyer Parent or (y) in the case of Buyer Parent, with prior discussions with Seller Parent as described in Section 6.2(b). Buyer Parent shall (and shall cause each of its Affiliates to) offer to take (and if such offer is accepted, commit to take) all steps which it is capable of taking to avoid or eliminate impediments under any antitrust, competition, or trade regulation Law that may be asserted by a Governmental Authority with respect to the Transactions so as to enable the Closing prior to the Outside Date, and shall defend through litigation on the merits any Claim asserted in any court by any Governmental Authority, including appeals. Without limiting the foregoing, Buyer Parent shall (and shall cause each of its Affiliates to) propose, negotiate and offer to commit and effect (and if such offer is accepted, commit to and effect) by consent decree, hold separate Order, or otherwise, the sale, divestiture, licensing or other disposition of such assets or businesses of Buyer Parent (or such Affiliates) or of any of the Acquired Assets, or otherwise offer to take or offer to commit to take any action which it is capable of taking and if the offer is accepted, take or commit to such action that limits its freedom of action with respect to, or its ability to retain, any of the businesses, services or assets of Buyer Parent (or such Affiliates) or of any of the Acquired Assets (any of the foregoing, a "Divestiture Action"), in each case, to the minimum extent necessary so as to permit and cause the condition set forth in Section 10.1(b) to be satisfied by the Outside Date; provided, however, that Buyer Parent shall not be required pursuant to the terms of this Agreement to take or commit to take any Divestiture Action other than (i) with respect to any and all Overlap Products, and in addition and not in limitation, (ii) such further Divestiture Actions as would not in the aggregate be of greater economic significance than the value of the Overlap Products that are not the subject of a Divestiture Action; provided, however, that this Section 6.2(a)(ii) shall not require Buyer Parent or its Controlled Affiliates to divest Specialty Products (other than Specialty Products that are Overlap Products, if any). Each of Buyer Parent and Seller Parent shall promptly notify the other of any written communication to that party from a Governmental Authority or any other Person (whether or not a Governmental Authority) in connection with antitrust matters relating to the Transactions and, subject to applicable Law, and the instructions of any Governmental Authority, permit the other to review in advance any material proposed written communication to any of the foregoing.

(b) Cooperation. Buyer Parent and Seller Parent shall jointly control all communications with any Governmental Authority relating to Antitrust Laws, and determine and direct the strategy and process by which the Parties will seek required approvals relating to Antitrust Laws. If the Parties initially disagree upon any such proposed communication, strategy or process, the Parties agree to work together in good faith to resolve the disagreement and endeavor to implement such communication, strategy or process in a mutually acceptable manner; provided that to the extent that a disagreement is unresolved after good faith discussions between Siggi Olafsson of Buyer Parent and Bob Stewart of Seller Parent and other such executive officers as they may deem appropriate of Buyer Parent and Seller Parent, the implementation of such communication, strategy or process will be controlled by Buyer Parent after full consideration to the views of Seller Parent. Each of Buyer Parent and Seller Parent shall, upon request by the other, furnish the other with all information concerning itself, its Affiliates and their respective directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of a Party or any of their respective Affiliates to any Person, including any Governmental Authority, in connection with the Transactions insofar as they pertain to antitrust matters.

(c) <u>Notification</u>. Subject to applicable Law and the instructions of any Governmental Authority, each of Buyer Parent and Seller Parent shall keep the other apprised of the status of matters relating to consummation of the Transactions, including promptly furnishing the other with copies of notices or other communications received by it, or by any of its Affiliates, from any Person, including any Governmental Authority, with respect to the Transactions insofar as they pertain to antitrust matters.

(d) <u>Meetings</u>. Subject to applicable Law, neither Seller Parent nor Buyer Parent (nor any of their respective Affiliates) shall permit any of its officers or any other representatives or agents to participate in any substantive meeting or discussion with any Governmental Authority with jurisdiction over enforcement of any applicable antitrust or competition Laws ("Government Antitrust Authority") in respect of any filings, investigation or other inquiry unless it consults with the other in advance and, to the extent permitted by such Government Antitrust Authority, gives the other the opportunity to attend and participate.

(e) Other Actions. Without limiting the generality of the undertakings pursuant to this Section 6.2 but subject to the proviso in the penultimate sentence in Section 6.2(a), each of the Parties agrees to take or cause to be taken the following actions: (i) the prompt provision to each and every foreign, federal, state or local court or any Governmental Authority of non-privileged information and documents reasonably requested by any of them or that are necessary, proper or advisable to permit consummation of the Transactions and (ii) the prompt use of its reasonable best efforts to avoid the entry of any permanent, preliminary or temporary injunction or other Order, decree, decision, determination or judgment that would restrain, prevent, enjoin or otherwise prohibit consummation of the Transactions by the Outside Date. The Parties' obligations under this Section 6.2 shall include the obligation to use its reasonable best efforts to defend against any lawsuits or other legal Proceedings, whether judicial or administrative, challenging the consummation of the Transactions, including seeking to have any stay or other injunctive relief which would prevent or impair the consummation of the Transactions by the Outside Date entered by any court or other Governmental Authority reversed on appeal or vacated. The Parties agree that if advisable to facilitate obtaining the Consent of any Governmental Authority that is a condition to Closing the Agreement, the Parties shall promptly meet and use their good faith efforts to agree to a modification or modifications to the terms of this Agreement and/or any Ancillary Agreement that the Parties agree will facilitate obtaining such Consent provided such modifications in the aggregate shall not alter the economic values of the Transactions other than in immaterial respects.

6.3 Reasonable Best Efforts.

(a) Subject to the terms and conditions of this Agreement (including the proviso in the penultimate sentence in Section 6.2(a)) and without limiting the express obligations hereunder, each of the Parties agrees to cooperate fully with each other and to use its reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective, at the time and in the manner contemplated by this Agreement, the Transactions, including to obtain all approvals required under applicable Laws, and to execute and deliver such documents and other papers and any other agreements, as may be necessary to, as soon as practicable, carry out the provisions of this Agreement and consummate and make effective the transactions contemplated by this

Agreement. At the Closing, Buyer shall, and shall cause each of its applicable Subsidiaries to, execute and deliver each Ancillary Agreement, and Seller shall, and shall cause each of its applicable Subsidiaries to, execute and deliver each Ancillary Agreement to which it is a party.

(b) Buyer Parent and Seller Parent agree that promptly following the date of this Agreement they shall organize a transition team, cochaired by a representative of Buyer Parent a representative of Seller Parent and including equal representation of Buyer Parent and Seller Parent, for the purposes of identifying: (i) services to be provided by Seller Parent to Buyer Parent and its Subsidiaries after the Closing pursuant to the terms of the Transition Services Agreement; and (ii) services to be provided by Buyer Parent to Seller Parent and its Subsidiaries after the Closing pursuant to the terms of the Reverse Transition Services Agreement and, in each case, the applicable periods for such services, it being understood that (A) Seller Parent shall be obligated to provide, or cause to be provided, all services that (I) are being provided by or on behalf of Seller Parent to the Business as conducted at any time during the period starting one year prior to the date of this Agreement through the Closing Date (the "Lookback Period") or (II) are otherwise necessary for the Buyers to operate the Business in the same manner as conducted at any time during the Lookback Period in all material respects, (B) Buyer Parent shall be obligated to provide or cause to be provided, all services that (I) are being provided by or on behalf of the Business to the Retained Business as conducted at any time during the Lookback Period or (II) are otherwise necessary for Sellers to operate the Retained Business in the same manner as conducted at any time during the Lookback Period in all material respects, based on services provided by the Business to the Retained Business prior to Closing and Buyer Parent and Seller Parent shall identify any services that are included in the Transferred Assets for which Seller will need services from Buyer and (C) the fees for such services shall be based upon the costs of Seller Parent and its Subsidiaries in the case of clause (A), or Buyer Parent and its Subsidiaries in the case of clause (B), to provide such services with no mark-up or element of profit. Buyer Parent shall provide transition services to the Retained Business in respect of the Brand R&D Assets for a period of eighteen (18) months following the Closing Date on the pricing terms set forth in the immediately preceding sentence. The Parties shall use their reasonable best efforts to agree upon all matters in the Transition Services Agreement and Reverse Transition Services Agreement. The cost for all such services shall be determined in the manner set forth in the Transition Services Agreement or the Reverse Transition Services Agreement, as applicable. During the Pre-Closing Period, the Parties will negotiate in good faith and agree on a fair allocation of assets that are related both to the Business and the Retained Business, with such allocation to take into account and be based on the relative historical use of such assets by each business and the allocation of costs associated with such assets. In the case of any assets of Seller Parent or its Subsidiaries that are required for the Business but are not included in the Acquired Assets, the Parties shall use commercially reasonable efforts to ensure that Buyer Parent and its Subsidiaries are granted access or rights to such assets to the extent required for the Business. In the case of any Acquired Assets that are required for the Retained Business as in effect on the date of this Agreement, the Parties shall use commercially reasonable efforts to ensure that Seller Parent and its Subsidiaries are granted access or rights to such assets to the extent required for the Retained Business.

(c) From time to time, as and when requested by a Party, the other Party shall, and shall cause its Affiliates to, at the requesting Party's expense (except as otherwise expressly provided in this Agreement), execute such Transfer Documents (including such instruments of

assignment as may be necessary in order to have the assignment of the Transferred Intellectual Property recorded in the name of a Buyer or an Affiliate of a Buyer at the relevant patent and trademark offices) and take such further actions as may be reasonably required to carry out the provisions hereof and consummate and evidence the Transactions, including executing and delivering or causing to be executed and delivered to such Party such Transfer Documents as such Party or its counsel may reasonably request as necessary for such purpose.

- (d) Prior to Closing, the Parties shall negotiate in good faith and agree on the terms of a distribution arrangement, expiring on December 31, 2016 between the Business, on the one hand, and the Anda Business, on the other hand, providing distribution and related services that the Anda Business provided to the Business at any time during the twelve (12) months prior to the date of this Agreement (the "Anda Lookback Period"). The pricing and other terms of such arrangement shall be on terms at least as favorable to the Business as during the Anda Lookback Period and reflected in the financial results contained in the Performance Financial Statement.
- 6.4 Access; Confidentiality. During the Pre-Closing Period, each Seller shall and shall cause its Representatives to, upon reasonable prior notice, free of charge, give Buyers, their officers, their authorized Representatives and a reasonable number of their employees, reasonable access during normal business hours to the Business Contracts, Books and Records, analysis, projections, plans, systems, management and other personnel, the Seller's Representatives, commitments, offices and other facilities and properties to the extent related to the Business, the Acquired Assets and the Assumed Liabilities. The terms of the Confidentiality Agreement shall apply to any information provided to Buyer Parent pursuant to this Section 6.4. The right of Buyers to access pursuant to this Section 6.4 shall not modify in any way any representation or warranty in Article IV. Notwithstanding anything to the contrary set forth herein, no Seller shall be required to provide access to, or to disclose information, where such access or disclosure would (a) jeopardize the attorney-client privilege of such Seller, (b) contravene any applicable Law or (c) give a third party the right to terminate or accelerate the rights under a contract to which a Seller is a party as of the date of this Agreement or otherwise bound as of the date of this Agreement; provided that in each case, Seller Parent shall: (A) give reasonable notice to Buyer Parent with sufficient detail of the reason for such restriction or prohibition, and (C) cause the applicable Seller to use its reasonable best efforts to cause the documents or information that are subject to such restriction or prohibition to be provided in a manner that would not reasonably be expected to violate such restriction or prohibition.
- 6.5 Special Contracts. Between the date of this Agreement until the Closing, Seller Parent shall, and shall cause its Affiliates and the Representatives of Seller Parent and its Affiliates to, use their reasonable best efforts to diligently search and identify to Buyer Parent Contracts that contain any non-compete, exclusivity or similar provision which would restrict the conduct of the business of Buyer Parent and its Affiliates (including the Transferred Group) post Closing, the Business or the use of the Acquired Assets after consummation of the Transactions (such terms, "Restrictive Terms" and such Contract, "Restrictive Contract"). Seller Parent shall promptly notify Buyer Parent in writing after the identification of such Restrictive Contracts and include a complete and correct copy of such Restrictive Contract with the written

notice (the "Seller Contract Notice"). At the election of Buyer Parent, the Seller Parent shall, and shall cause its Controlled Affiliates to, use their commercially reasonable efforts to (i) remove Restrictive Terms from the Restrictive Contracts, (ii) amend such Restrictive Terms to the satisfaction of Buyer Parent (at its sole discretion) or (iii) if unable to remove or amend Restrictive Terms to either (a) enter into arrangements to provide to Buyer Parent the rights, benefits and interests of any such Restrictive Contract and to the extent the full benefits of any such Restrictive Contract cannot be provided to Buyer Parent after the Closing, then Seller Parent shall, and shall cause its Affiliates to, enter into such arrangements to provide to Buyer Parent the economic and operational equivalent of such Restrictive Contract or (b) request Seller Parent to terminate the Restrictive Contract provided any such termination fees or penalties arising therefrom will be the sole responsibility of the Buyer Parent.

6.6 Shared Business Contracts.

(a) The Parties acknowledge that the Shared Business Contracts relate to both the Business and the Retained Business. Following the Closing, the Parties desire for themselves and for the benefit of Sellers and Buyers, respectively, to have and obtain the rights and benefits under each Shared Business Contract to the extent related to the continuing business of the Buyers and Sellers. Immediately after the date hereof, the Parties agree to cooperate together to provide Sellers and Buyers with their applicable rights and benefits under each Shared Business Contract by assisting the respective Sellers and/or Buyers in entering into a new Contract or Contracts with the applicable third party on substantially similar terms (a "Separated Contract"). The costs of entering into a new Contract or Contract(s) shall be borne by Seller Parent, in the case of a Shared Business Contract that is a Business Contract or Buyer Parent, in the case of a Shared Business Contract that is not a Business Contract. If any Shared Business Contract cannot be separated into a Separated Contract at Closing, Seller Parent and Buyer Parent shall, and shall cause each of their respective Affiliates to, use their reasonable best efforts to cause, for the period after the Closing until such Shared Business Contract is separated into a Separated Contract or such Separated Contract expires pursuant to its terms, (i) the rights and benefits under each Shared Business Contract to the extent relating to the Business to be enjoyed by Buyer Parent; (ii) the Liabilities under each Shared Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract to the extent relating to the Retained Business Contract

(b) Without limiting Section 6.6(a), if any Shared IP License Agreement is not separated in accordance with Section 6.6(a), effective as of the Closing, (i) if such Shared IP License Agreement is a Business Contract, Buyer Parent shall grant at or prior to Closing (to the extent Buyer Parent or any member of Buyer Group has a right to, and subject to the terms and conditions of the applicable Shared IP License Agreement), effective as of the Closing Date, to an entity in the Retained Group designated by Seller Parent, a perpetual, irrevocable (i) exclusive within the field of brand pharmaceuticals and (ii) non-exclusive outside the field of brand pharmaceuticals, right and sublicense under the Shared IP License Agreements and solely for purposes of developing, manufacturing, using, selling, offering to sell, distributing, importing, supporting and otherwise disposing of brand pharmaceutical products (the "Buyer Licensed Shared IP Sublicenses"). and (ii) if such Shared IP License Agreement is not a Business

Contract, Seller Parent shall grant at or prior to Closing (to the extent Seller Parent or any Retained Entity has a right to, and subject to the terms and conditions of the applicable Shared IP License Agreements) to an entity in the Transferred Group designated by Buyer Parent, a perpetual, irrevocable (i) exclusive within the field of generic pharmaceuticals and (ii) non-exclusive outside the field of generic pharmaceuticals, right and sublicense under the Shared IP License Agreements and solely for purposes of developing, manufacturing, using, selling, offering to sell, distributing, importing, supporting and otherwise disposing of the Products or any other generic pharmaceutical product (the Seller Licensed Shared IP Sublicenses and, together with the Buyer Licensed Shared IP Sublicences, the "Licensed Shared IP Sublicenses"). The Seller Licensed Shared IP Sublicenses shall be non-assignable (provided that such Seller Licensed Shared IP Sublicense shall, to the extent permitted by and consistent with the terms of the applicable underlying Shared IP License Agreement, be assignable, without the consent of Buyer Parent, by the applicable Retained Entity (i) to Seller Parent or any Affiliate of Seller Parent if (A) such Affiliate agrees in writing to be bound by the terms of such Seller Licensed Shared IP Sublicense, (B) such assignee continues to be an Affiliate of Seller Parent and (C) the assigning Retained Entity shall remain primarily liable for the performance of all obligations of such Person under such Seller Licensed Shared IP Sublicense and (ii) to any Person in connection with the sale by Seller Parent or its Affiliates, as applicable, to such Person of (A) all or a relevant portion of the Retained Business, whether by merger, consolidation, combination, reorganization or similar transaction or the transfer, sale, lease, conveyance or asset sale or (B) the product or products of Seller Group to which such Seller Licensed Shared IP Sublicense relates). The Buyer Licensed Shared IP Sublicenses shall be non-assignable (provided that such Buyer Licensed Shared IP Sublicense shall, to the extent permitted by and consistent with the terms of the applicable underlying Shared IP License Agreement, be assignable, without the consent of Seller Parent, by the applicable entity in the Transferred Group (i) to Buyer Parent or any Affiliate of Buyer Parent if (A) such Affiliate agrees in writing to be bound by the terms of such Seller Licensed Shared IP Sublicense, (B) such assignee continues to be an Affiliate of Buyer Parent and (C) the assigning entity in the Transferred Group shall remain primarily liable for the performance of all obligations of such Person under such Buyer Licensed Shared IP Sublicense and (ii) to any Person in connection with the sale by Buyer Parent or its Affiliates, as applicable, to such Person of (A) the Business, whether by merger, consolidation, combination, reorganization or similar transaction or the transfer, sale, lease, conveyance or disposition of all or substantially all of the assets of the Business or (B) the Product or Products to which such Buyer Licensed Shared IP Sublicense relates). Notwithstanding anything to the contrary in this Section 6.6, nothing in this Section 6.6 shall require Buyer Parent to assume any obligation that purports to bind or restrict any business, activity or asset other than the Business or Acquired Assets.

6.7 <u>Intercompany Arrangements</u>. Immediately prior to the Closing, Seller Parent shall, and shall cause its Controlled Affiliates to, (i) terminate all agreements or arrangements, written or unwritten, of any kind (other than any Ancillary Agreements), between Sellers or any Retained Entity, on the one hand, and any entity in the Transferred Group, on the other hand, and (ii) settle or otherwise extinguish any amounts (other than any amounts under any Ancillary Agreements) owed to or by Sellers or any Retained Entity, on the one hand, and any entity in the Transferred Group, on the other hand.

6.8 Pre-Closing Reorganization. Within 90 days following the date of this Agreement, Seller Parent shall deliver to Buyer Parent a draft Pre-Closing Reorganization Plan. Pursuant to the principles set forth on Exhibit C and upon the terms and subject to the conditions set forth in this Agreement (it being understood that in the event of any inconsistencies or conflicts between the terms of this Agreement and the terms set forth on Exhibit C, the terms of this Agreement shall prevail, except to the extent the Parties have mutually agreed otherwise in writing), between the date hereof and the Closing: (x) Seller Parent and Buyer Parent shall use their reasonable efforts to agree a definitive steps plan for the sale and purchase of the Business, in accordance with the Pre-Closing Reorganization Plan, cooperating in good faith with respect to the transactions set forth in such steps plan; and (y) Seller Parent shall, and shall cause its Affiliates, as applicable, to, take such steps as are required to effect the Pre-Closing Reorganization in compliance in all respects with the terms of Exhibit C. The Parties agree to work together in good faith to finalize and implement the Pre-Closing Reorganization Plan in a mutually acceptable manner. Each of Buyer Parent and Seller Parent shall, upon request by the other, furnish the other with all information reasonably requested in connection with the Pre-Closing Reorganization Plan and such other matters as may be reasonably necessary or advisable. Seller Parent shall make any modification to the steps plan referred to in clause (x) of the previous sentence and the Pre-Closing Reorganization that is reasonably requested by Buyer Parent ("Buyer-Requested Modifications"). The details of and the implementation of the Pre-Closing Reorganization Plan will be controlled by Seller Parent after full consideration to the views of Buyer Parent. Unless a different timing is called for in the Pre-Closing Reorganization Plan, the Seller Parent shall commence all necessary steps to implem

6.9 Transaction Committee.

(a) In order to facilitate the consummation of the transactions contemplated by this Agreement on a timely basis, including the preparation and negotiation of the Ancillary Agreements, the separation of the Shared Business Contracts contemplated by Section 6.6, and the transfer of the Acquired Assets and the assumption of the Assumed Liabilities at Closing, prior to the Closing, Buyer Parent and Seller Parent shall establish a committee (the "Transaction Committee") to be managed by Siggi Olafsson of Buyer Parent and Bob Stewart of Seller Parent, with such other members as they shall mutually agree, which Transaction Committee shall have responsibility for (i) coordinating and directing the efforts of the Parties with respect to (A) the preparation, negotiation and finalization of the Ancillary Agreements, (B) the separation of the Shared Business Contracts contemplated by Section 6.6, subject to Section 6.2, (C) obtaining all Consents, Permits and Regulatory Registrations from third-parties that are necessary or desirable in connection with the consummation of the Transactions, and (D) coordinating and directing the efforts of the Parties with respect to obtaining the Debt Financing in accordance with Section 9.4 and Section 9.5, (ii) overseeing the transition team established pursuant to Section 6.3(b) and the related process of identifying services to be provided by the Parties following Closing, (iii) communications, public relations and investor relations strategy and approach of the Parties regarding this Agreement and the transactions contemplated hereby (other than any actions of Seller Parent taken in respect of a Company Proposal), and (iv) overseeing other business and operational matters relating to this Agreement and the

transactions contemplated hereby, including transitional plans of the Buyer Group and the Seller Group following the Closing, to the extent not in violation of applicable Laws, including Laws regarding the exchange of information and other laws regarding competition.

(b) If the Parties disagree upon any matter subject to the oversight of the Transaction Committee, the members of the Transaction Committee shall work together in good faith to resolve the disagreement in a mutually acceptable manner. In the event that the Transaction Committee is unable to resolve such disagreement in a timely manner, and in any event within 5 days of written notice of such disagreement by one Party to the other, the matter in dispute shall be elevated to Siggi Olafsson of Buyer Parent and Bob Stewart of Seller Parent for further good faith discussion, and, if they are unable to resolve such disagreement within an additional 5 days, to the Chief Executive Officers of each of Buyer Parent and Seller Parent. During the course of all such discussions, the Parties shall cooperate with each other and all reasonable requests made by one Party to the other for information, including requests for copies of relevant documents, will be honored. In the event that any disagreement is not resolved by the Parties within 15 days following delivery of the written notice mentioned above, the Parties may seek any remedies to which they may be entitled in accordance with the terms of this Agreement, provided that nothing herein shall prevent either Party from initiating proceedings in accordance with this Agreement if such Party would be substantially harmed by a failure to act during the time that such good faith efforts are being made to resolve the disagreement through negotiation or if the consummation of the Transactions would reasonably be expected to be delayed. In the event that any proceeding is commenced under this Section 6.9(b), the Parties agree to continue to attempt to work in good faith to resolve any disagreement according to the terms of this Section 6.9 during the course of such proceeding.

ARTICLE VII

EMPLOYEE MATTERS

7.1 Principles. The Parties intend that the Transactions shall result in the Transferred Entities' retention of each Transferred Entity Benefit Plan (together with all corresponding assets and Liabilities). Prior to the Closing, Seller shall take all necessary actions to cause (1) any Excluded Liabilities under the Transferred Entity Benefit Plans or with respect to Non-Business Employees to be transferred to and assumed by Sellers (other than the Transferred Entities) and (2) all assets related to Assumed Liabilities under Transferred Entity Benefit Plans or with respect to Business Employees or, to the extent specified in Section 2.3(f), Former Business Employees, to be transferred to the Transferred Entities. To the extent any Assumed Liability or Excluded Liability cannot be transferred to or from, respectively, a Transferred Entity due to applicable Law, the Parties will work together in good faith to provide for a transfer of an equivalent value in cash or other mutually agreed upon form. For the avoidance of doubt, Buyer Parent may meet its obligations pursuant to this Article 7 under the Transferred Entity Benefit Plans or under new or existing Benefit Plans of Buyer, in Buyer Parent's sole discretion.

7.2 <u>Transfer of Employees</u>. Prior to the Closing, Sellers shall take all necessary actions to cause (1) the employment of any Business Employee who is employed by the Sellers

and their Subsidiaries (other than the Transferred Group) currently employed at an entity other than a member of the Transferred Group to be transferred to a member of the Transferred Group and (2) the employment of any Non-Business Employee who is employed by a member of the Transferred Group to be transferred to an entity other than a member of the Transferred Group. Seller Parent shall be responsible for any severance obligations that arise as a result of such transfer (or termination) and shall be solely responsible for, and shall indemnify and hold Buyer Parent and its Affiliates (including the Transferred Entities) harmless from all Liabilities concerning the Non-Business Employees whether arising prior to, on or after the Closing. Prior to Closing, Seller Parent and Buyer Parent shall work together to determine the exact list of Business Employees.

- 7.3 Compensation and Benefits. Effective as of the Closing and through the first anniversary of the Closing, or such longer period as required by applicable Law, Buyer Parent shall provide, or shall cause its Affiliates to provide, to the Business Employees who continue to be employed by Buyer Parent or an Affiliate thereof immediately following the Closing (the "Continuing Employees"), (i) a base salary or wage rate that is not less than the base salary or wage rate in effect for each such Continuing Employee immediately prior to the Closing, (ii) annual cash bonus and long-term incentive opportunities (including any equity awards) which may be payable in cash that, in the aggregate, are no less favorable than those in effect for each such Continuing Employee immediately prior to the Closing, (iii) employee benefits (including defined benefits plan) that are, in the aggregate, no less favorable than those (including pursuant to a defined benefit plan) provided to the Continuing Employees immediately prior to the Closing and (iv) severance benefits that are no less favorable than the greater of (A) the severance benefits that would have been applicable to the Continuing Employees under a Seller Benefit Plan or a Transferred Entity Benefit Plan in effect as of immediately prior to the Closing (excluding change in control severance benefits triggered prior to the Closing pursuant to previous transactions by or with Seller Parent or its Affiliates), or (B) the severance benefits applicable to similarly-situated employees of Buyer Parent or its Affiliates.
- 7.4 Service Credit. Effective as of the Closing and thereafter, Buyer Parent shall provide, or shall cause an Affiliate to provide, that periods of employment with a member of the Transferred Group (including any current or former Affiliate of a member of the Transferred Group or any predecessor of a member of the Transferred Group) shall be taken into account for vesting and eligibility and level of benefits purposes under all employee benefit plans maintained by Buyer Parent or Affiliate thereof for the benefit of the Continuing Employees, including vacation or other paid-time-off plans or arrangements, 401(k), pension or other retirement plans and any severance or health or welfare plans (other than for purposes of determining any accrued benefit under any defined benefit pension plan or as would result in a duplication of benefits).
- 7.5 Welfare Plans. Effective as of the Closing and thereafter, Buyer Parent shall make commercially reasonable efforts to, or to cause its Affiliates to, (i) ensure that no eligibility waiting periods, actively-at-work requirements or pre-existing condition limitations or exclusions shall apply with respect to the Continuing Employees under the applicable health and welfare benefits plan of Buyer Parent or any Affiliate thereof (except to the extent applicable under Seller Benefit Plans immediately prior to the Closing), (ii) waive any and all evidence of insurability requirements with respect to such Continuing Employees to the extent such evidence

of insurability requirements were not applicable to the Continuing Employees under the Seller Benefit Plans immediately prior to the Closing, and (iii) credit each Continuing Employee with all deductible payments, out-of-pocket or other co-payments paid by such employee under the Seller Benefit Plans prior to the Closing during the year in which the Closing occurs for the purpose of determining the extent to which any such employee has satisfied his or her deductible and whether he or she has reached the out-of-pocket maximum under any health benefit plan of Buyer Parent or an Affiliate thereof for such year. The Transactions shall not affect any Continuing Employee's accrual of, or right to use, in accordance with applicable policy as in effect immediately prior to the Closing, any personal, sick, vacation or other paid-time-off accrued but unused by such Continuing Employee immediately prior to the Closing.

- 7.6 <u>Labor and Employment Law Matters</u>. Sellers and Buyer Parent shall, and shall cause their respective Affiliates to, cooperate to take all steps, on a timely basis, as are required under applicable Law to notify, consult with or negotiate the effect, impact or timing of the Transactions with each works council, union, labor board, employee group or Governmental Authority where so required under applicable Law. Sellers and Buyer Parent shall, and shall cause their applicable Affiliates to, comply with all applicable Laws, directives and regulations relating to the Business Employees. Buyer Parent or its applicable Affiliate shall become a party to any collective bargaining (including national, sector or local), works council or similar agreement with respect to any Business Employee. Buyer Parent shall, or shall cause its Affiliates to, join any industrial, employer or similar association or federation if membership is required for the currently applicable collective bargaining, works council or similar agreement to continue to apply.
- 7.7 <u>Defined Contribution Plans</u>. Buyer Parent shall cause each Continuing Employee who is a participant in a Seller Benefit Plan intended to qualify under Section 401(a) of the Code that includes a cash or deferred arrangement intended to satisfy the provisions of Section 401(k) of the Code (the "Seller 401(k) Plans") to be allowed to participate, effective as of the Closing, in a tax qualified plan which includes a cash or deferred arrangement intended to satisfy the provisions of Section 401(k) of the Code that is sponsored and maintained by Buyer Parent or an Affiliate thereof (the "Buyer Parent 401(k) Plan") and such Continuing Employee shall be credited with eligibility service and vesting service for all periods of service with Sellers or their respective Affiliates to the extent so credited with such service under the applicable Seller 401(k) Plan. In addition, Buyer Parent shall use commercially reasonable efforts to cause the Buyer Parent 401(k) Plan to accept rollover contributions of "eligible rollover distributions" (within the meaning of Section 401(a)(31) of the Code, inclusive of loans) from the Seller 401(k) Plan.
- 7.8 <u>Communications</u>. Prior to making any written or oral communications to the directors, officers or employees of the Sellers or any of their Subsidiaries pertaining to compensation or benefit matters that are affected by the transactions contemplated by this Agreement, Seller Parent shall provide Buyer Parent with a copy of the intended communication, Buyer Parent shall have a reasonable period of time to review and comment on the communication, and Seller Parent shall consider any such comments in good faith.

7.9 Treatment of Business Employee Equity Awards.

(a) As of the Closing, without any action on the part of the holders thereof, each Seller Parent Option granted prior to the date of this Agreement (or which is a Permitted Award) that is outstanding and unvested and held by a Business Employee immediately prior to the Closing shall be assumed by Buyer Parent and shall be converted into an option (a "Buyer Parent Option") to acquire (A) that number of shares of Buyer Parent Stock (rounded down to the nearest whole share) equal to the product obtained by multiplying (1) the number of shares of Seller Parent Stock subject to such Seller Parent Option immediately prior to the Closing by (2) the Exchange Ratio, (B) at an exercise price per shares of Buyer Parent Stock (rounded up to the nearest whole cent) equal to the quotient obtained by dividing (1) the exercise price per share of Seller Parent Stock of such Seller Parent Option by (2) the Exchange Ratio; provided, however, that each such Seller Parent Option (I) which is an "incentive stock option" (as defined in Section 422 of the Code) shall be adjusted in accordance with the foregoing in a manner consistent with the requirements of Section 424 of the Code and (II) shall be adjusted in a manner which complies with Section 409A of the Code and that causes the resulting Buyer Parent Option not to constitute the grant of a new option or a change in the form of payment of an option, as provided under Treasury Regulation section 1.409A-1(b)(5)(v)(D). Except as otherwise provided in this Section 7.9, each such Buyer Parent Option assumed and converted pursuant to this Section 7.9 shall continue to have, and shall be subject to, the same terms and conditions as applied to the corresponding Seller Parent Option immediately prior to the Closing.

(b) As of the Closing, without any action on the part of the holders thereof, each outstanding and unvested Seller Parent Restricted Share Award granted prior to the date of this Agreement (or which is a Permitted Award) held by a Business Employee shall be assumed by Buyer Parent and shall be converted into an award of restricted stock corresponding to Buyer Parent Stock (each, a "Buyer Parent Restricted Share Award") with respect to a number of shares of Buyer Parent Stock (rounded up or down to the nearest whole share) equal to the product obtained by multiplying (A) the applicable number of shares of Seller Parent Stock subject to the Seller Parent Restricted Share Award as of immediately prior to the Closing by (B) the Exchange Ratio. Except as otherwise provided in this Section 7.9, each Buyer Parent Restricted Share Award assumed and converted pursuant to this Section 7.9 shall continue to have, and shall be subject to, the same terms and conditions as applied to the corresponding Seller Parent Restricted Share Award immediately prior to the Closing.

(c) As of the Closing, without any action on the part of the holders thereof, each outstanding and unvested Seller Parent RSU Award granted prior to the date of this Agreement (or which is a Permitted Award) held by a Business Employee that vests solely based on the passage of time shall be assumed by Buyer Parent and shall be converted into an award of restricted stock units corresponding to Buyer Parent Stock (each, a "Buyer Parent Time-Vesting RSU Award") with respect to a number of shares of Buyer Parent Stock (rounded up or down to the nearest whole share) equal to the product obtained by multiplying (A) the applicable number of shares of Seller Stock subject to the Seller Parent RSU Award as of immediately prior to the Closing by (B) the Exchange Ratio. Except as otherwise provided in this Section 7.9, each Buyer Parent Time-Vesting RSU Award assumed and converted pursuant to this Section 7.9 shall continue to have, and shall be subject to, the same terms and conditions as applied to the corresponding Seller Parent RSU Award immediately prior to the Closing.

- (d) As of the Closing, without any action on the part of the holders thereof, each outstanding and unvested Seller Parent RSU Award granted prior to the date of this Agreement (or which is a Permitted Award) held by a Business Employee that vests subject to the achievement of one or more performance goals shall be assumed by Buyer Parent and shall be converted into an award of restricted stock units corresponding to Buyer Parent Stock (each, a "Buyer Parent Performance RSU Award"). Each performance goal applicable to any such Seller Parent RSU Award shall be deemed achieved as of the Closing Date at the greater of (i) the target performance goal and (ii) the level achieved based on actual performance through the Closing Date (assuming that the Closing Date is the last day of the applicable performance period), and the number of shares of Seller Parent Stock covered by such Seller Parent RSU Award as of immediately prior to Closing shall be determined in accordance with this sentence. Each Buyer Parent Performance RSU Award shall cover a number of shares of Buyer Parent Stock (rounded up or down to the nearest whole share) equal to the product obtained by multiplying (A) the applicable number of shares of Seller Parent Stock subject to the Seller Parent RSU Award as of immediately prior to the Closing (determined in accordance with the immediately prior sentence) by (B) the Exchange Ratio. Except as otherwise provided in this Section 7.9, each Buyer Parent Performance RSU Award assumed and converted pursuant to this Section 7.9 shall continue to have, and shall be subject to, the same terms and conditions, including any time-vesting conditions (but not performance-vesting conditions), as applied to the corresponding Seller Parent Performance RSU Award immediately prior to the Closing.
- (e) As of the Closing, without any action on the part of the holders thereof, each outstanding and unvested Seller Parent Cash Incentive Award granted prior to the date of this Agreement (or which is a Permitted Award) held by a Business Employee that vests subject to the achievement of one or more performance goals shall be assumed by Buyer Parent (each, a "Buyer Cash Incentive Award"). Each performance goal applicable to any such Seller Parent Cash Incentive Award shall be deemed achieved as of the Closing Date at the target performance goal, and the cash award payable pursuant to such Seller Parent Cash Incentive Award as of immediately prior to Closing shall be determined in accordance with this sentence. Except as otherwise provided in this Section 7.9, each Buyer Parent Cash Incentive Award assumed pursuant to this Section 7.9 shall continue to have, and shall be subject to, the same terms and conditions, including any time-vesting conditions (but not performance-vesting conditions), as applied to the corresponding Seller Parent Cash Incentive Award immediately prior to the Closing.
 - (f) [Reserved].
- (g) To the extent the assumption and conversion of the Seller Parent Incentive Awards cannot be effected under the Buyer Parent's equity incentive plans in the manner contemplated by this Section 7.9, or as otherwise determined by Buyer Parent, Buyer Parent shall grant an award of equivalent value (based on the value of the Seller Parent Incentive Award on the Closing Date and, in the case of Seller Parent Options, the spread value) with the same payment and vesting terms and conditions.
- 7.10 <u>Annual Bonuses</u>. Notwithstanding anything in the applicable Seller Benefit Plans and/or Transferred Entity Benefit Plans to the contrary, to the extent the Closing Date occurs prior to payment of the 2015 annual cash bonuses and 2015 annual cash incentive payments to

the Business Employees under any such Seller Benefit Plans and/or Transferred Entity Benefit Plans, payment of the 2015 annual cash bonuses and cash incentive payments shall be made to the Business Employees by Seller Parent or its Affiliates as soon as reasonably practicable following the Closing Date, but in no event later than March 15, 2016. In the event the Closing occurs during 2016, Buyer Parent and its Affiliates shall provide full year annual bonuses to the Continuing Employees under bonus plans maintained by Buyer Parent and/or its Affiliates in a manner consistent with Section 7.3 of this Agreement. The payment of the 2015 annual cash bonuses and 2015 annual cash incentive payments shall constitute a Current Liability

- 7.11 2016 Annual Equity Awards. In the event the Closing occurs before December 31, 2016, Buyer Parent and its Affiliates shall provide full year 2016 equity incentive opportunities under equity incentive plans maintained by Buyer Parent and/or its Affiliates in a manner consistent with Section 7.3 of this Agreement, with such grants to be made at the later of (i) such time as Buyer Parent regularly grants 2016 annual long-term equity awards and (ii) within 30 days following the Closing. In the event Closing does not occur before December 31, 2016, Seller may grant 2016 equity awards to Business Employees in the Ordinary Course of Business in aggregate and per person amounts consistent with 2015 (and such awards will constitute Permitted Awards).
- 7.12 Third Party Beneficiaries. Nothing in this Agreement shall confer upon any Continuing Employee any right to continue in the employ or service of Buyer Parent or any Affiliate of Buyer Parent, or shall interfere with or restrict in any way the rights of Buyer Parent or any Affiliate of Buyer Parent, which rights are hereby expressly reserved, to discharge or terminate the services of any Continuing Employee at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise in a written agreement between Buyer Parent, a member of the Transferred Group or any Affiliate thereof and the Continuing Employee or any severance, benefit or other applicable plan or program covering such Continuing Employee. Notwithstanding any provision in this Agreement to the contrary, nothing in this Agreement shall (i) be deemed or construed to be an amendment or other modification of any Seller Benefit Plan, or (ii) create any third party rights in any current or former service provider of the Company or its affiliates (or any beneficiaries, dependents or collective bargaining representative thereof).

ARTICLE VIII

TAX MATTERS

8.1 <u>Tax Returns</u>. Buyer Parent shall file or cause to be filed when due all Tax Returns that are required to be filed by or with respect to the Acquired Assets, any member of the Transferred Group and the Business after the Closing, and such Tax Returns, if relating to a Pre-Closing Tax Period or a Straddle Period to the extent they relate to the Acquired Assets, any member of the Transferred Group and the Business, shall be prepared consistent with past practices except as otherwise required by applicable Law. Buyer Parent shall deliver or cause to be delivered any such Tax Returns that relate to a Pre-Closing Tax Period or a Straddle Period to Seller Parent for its review at least thirty (30) calendar days prior to the due date for filing, and Seller Parent shall provide written notice of its comments at least fifteen (15) calendar days prior to filing, such comments not to be unreasonably rejected by Buyer Parent. Notwithstanding

anything to the contrary in this Section 8.1, Seller Parent shall file any Tax Return relating to a Pre-Closing Tax Period or a Straddle Period with respect to a member of the Transferred Group for any taxable period during which such member joins a Seller as a member of an affiliated, consolidated, combined or unitary group, and such Tax Returns, to the extent they relate to the Acquired Assets, any member of the Transferred Group and the Business (any such portion of such Tax Returns, a "Component Return"), shall be prepared consistent with past practices except as otherwise required by applicable Law. Seller Parent shall deliver or cause to be delivered any Component Returns that relate to a Pre-Closing Tax Period or a Straddle Period to Buyer Parent for its review at least thirty (30) calendar days prior to the due date for filing, Buyer Parent shall provide written notice of its comments at least fifteen (15) calendar days prior to filing, such comments not to be unreasonably rejected by Seller Parent.

8.2 Tax Proceedings. From and after the Closing, Buyer Parent shall notify Seller Parent in writing within thirty (30) calendar days of receipt by Buyer or any of its Affiliates (including any member of the Transferred Group) of notice, with respect to a Pre-Closing tax Period or a Straddle Period, of (i) any pending or threatened Tax audits or assessments that may give rise to Liabilities for Taxes and (ii) any Claims that may give rise to amounts, in each case of (i) and (ii), for which Sellers could reasonably be expected to indemnify the Buyer Indemnified Parties pursuant to Section 12.2 or which would otherwise reasonably be expected to result in material adverse Tax consequences to a Seller (a "Tax Contest Claim"), provided that any failure to comply with this provision shall not affect any Buyer Indemnified Party's right to indemnification hereunder to the extent such failure does not materially prejudice Seller Parent's ability to defend against such Tax Contest Claim. Sellers shall notify Buyer Parent in writing within thirty (30) days of receipt by any Seller or any Affiliate of any Seller of notice of any pending or threatened Tax audit, assessment or other Proceeding regarding (x) the Acquired Assets (y) any member of the Transferred Group and (z) the Business, provided that any failure to comply with this provision shall not affect any Seller Indemnified Party's right to indemnification hereunder to the extent such failure does not materially prejudice Buyer Parent's ability to defend against such audits, assessments, and Claims. Seller Parent shall, at its own cost and expense, control the defense employing counsel of its choice of any Tax Contest Claim to the extent that such Tax Contest Claim would not reasonably be expected to materially adversely affect the Tax liability of Buyer Parent or any of its Affiliates (including any member of the Transferred Group), provided that Seller Parent shall keep Buyer Parent reasonably informed of the progress of any such Tax Contest Claim (and permit Buyer Parent to participate in such Tax Contest Claim at Buyer Parent's own expense) and shall not agree to any settlement without receiving the Buyer Parent's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed. To the extent a Tax Contest Claim would reasonably be expected to materially adversely affect the Tax liability of Buyer Parent or any of its Affiliates (including any member of the Transferred Group), Buyer Parent shall, at its own cost and expense, control the defense employing counsel of its choice, provided that (A) Buyer Parent shall use its best efforts to separate the issues in all material respects into those for which the Sellers would be liable under Section 12.2 or which would otherwise reasonably be expected to result in material adverse Tax consequences to the Seller and all other issues and (B) Seller Parent (along with counsel and other advisors of its choice) shall be entitled to participate at their sole cost and expense in the defense with respect to the issues for which the Sellers would be liable under Section 12.2 or which would otherwise reasonably be expected to result in material adverse Tax consequences to the Seller. From and after the Closing, neither Buyer Parent nor

any of its Affiliates (including any member of the Transferred Group) shall agree to settle any Tax Contest Claim that would reasonably be expected to be the subject of indemnification by the Sellers under Section 12.2 or which would otherwise reasonably be expected to result in material adverse Tax consequences to the Seller without the prior written consent of the Sellers, which consent shall not be unreasonably withheld, conditioned or delayed.

- 8.3 Tax Cooperation. Buyer Parent and Seller Parent agree to furnish or cause to be furnished to the other, if reasonably requested, as promptly as practicable, such information and assistance relating to the Acquired Assets, Assumed Liabilities, any payment made under this Agreement, and the receipt and ownership of the Stock Consideration by the Sellers, including access to books and records, as is reasonably necessary for the filing of all Tax Returns by any Buyer or Seller as provided herein, the making of any election relating to Taxes, the preparation for any audit by any Governmental Authority and the prosecution or defense of any Proceeding relating to any Tax (subject to the other provisions of this Article VIII). Any expenses incurred in furnishing such information or assistance pursuant to this Section 8.3 shall be borne by the party requesting it. Notwithstanding anything to the contrary in this Section 8.3, no Buyer shall be entitled to review any Tax Return that a Seller joins as a member of an affiliated, consolidated, combined or unitary group except to the extent that such Tax Return relates to a member of the Transferred Group and no Seller shall be entitled to review any Tax Return that a Buyer or any of its Affiliates (including any member of the Transferred Group) joins as a member of an affiliated, consolidated, combined or unitary group except to the extent that such Tax Return relates to a member of the Transferred Group and would reasonably be expected to be the subject of indemnification by the Buyers under Section 12.3.
- 8.4 <u>Post-Closing Conduct</u>. After the Closing, except as required under applicable Law, Buyer Parent shall not, and shall not cause or permit any of its Subsidiaries (including, after Closing, any member of the Transferred Group) to (i) make, change or revoke any Tax election with respect to the Acquired Assets or any member of the Transferred Group that has any retroactive effect in the portion of any Pre-Closing Tax Period ending on or prior to the Closing, (ii) grant an extension of any applicable statute of limitations that relates to Taxes with respect to the Acquired Assets that relates to a Pre-Closing Tax Period, (iii) amend or cause to be amended any Tax Return of any member of the Transferred Group that relates to a Pre-Closing Tax Period, (iv) take any action on the Closing Date after the Closing that is outside the Ordinary Course of Business of any member of the Transferred Group, or (v) fail to comply with the terms and conditions of the Puerto Rico Grant, in each case that would reasonably be expected to materially increase the Liability of Sellers for Taxes (including by decreasing the amount of any credits or deductions otherwise available to Sellers), without the prior written consent of the Seller Parent, such consent not to be unreasonably withheld.
- 8.5 <u>Section 338 Elections</u>. No Buyer shall make or cause to be made any election under Section 338 of the Code (or any analogous provisions of state, local or non-United States income Tax Law) with respect to the purchase of any member of the Transferred Group by any Buyer without the prior written consent of Sellers, which consent may be withheld in the sole discretion of Sellers. A Buyer shall make or cause to be made any election under Section 338 of the Code (or any analogous provisions of state, local or non-United States income Tax Law) with respect to a member of the Transferred Group as is reasonably requested by Seller Parent (a "Seller-Requested Section 338 Election"), provided that a Buyer shall not be required to make any such election unless, in Buyer Parent's reasonable judgment, such election is not contrary to applicable Law.

- 8.6 Property Taxes. Seller Parent (or its applicable Affiliate) shall be liable for the proportionate amount of Property Taxes levied with respect to the Acquired Assets that is attributable to the Pre-Closing Tax Period as described in the definition of "Straddle Period," and Buyer Parent (or its applicable Affiliate) shall be liable for the proportionate amount of Property Taxes levied with respect to the Acquired Assets that is attributable to the Post-Closing Tax Period, as described in the definition of "Straddle Period." Upon receipt of any bill for such Property Taxes, Buyer Parent (or its Affiliate) or Seller Parent (or its Affiliate), as applicable, shall present a statement to the other Party setting forth the amount of reimbursement to which each is entitled under this Section 8.6 together with such supporting evidence as is reasonably necessary to calculate the proration amount. The undisputed proration amount shall be paid by the Party owing it to the other Party within thirty (30) days after delivery of such statement. In the event that Buyer Parent (or its Affiliate) or Seller Parent (or its Affiliate) makes (or causes to be made) any payment for which it is entitled to reimbursement under this Section 8.6, the applicable Party shall make such reimbursement promptly but in no event later than thirty (30) days after the presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement.
- 8.7 <u>Transfer Taxes</u>. All sales, transfer (including German and United Kingdom stamp duty land tax and any other real estate transfer tax), stamp, excise, consumption, documentary, filing, notarization, registration, land value gains, recordation and other similar Taxes (but not including VAT) together with any notarial and registry fees and recording costs ("**Transfer Taxes**") resulting from the Pre-Closing Reorganization shall be borne one hundred percent (100%) by Sellers (the "**Non-Shared Transfer Taxes**"). All Transfer Taxes resulting from the transfer by Sellers of the Acquired Assets to Buyers, including the transfer of the Regulatory Registrations and the Transferred Intellectual Property (collectively, "**Shared Transfer Taxes**"), shall be borne fifty-percent (50%) by Buyers and fifty-percent (50%) by Sellers *provided*, that interest, additions and penalties that arise as a result of a Person's failure to timely and properly pay its portion of the Transfer Taxes shall be borne exclusively by such Person. In the event one Party (or its Affiliate) receives a refund or credit of Shared Transfer Taxes paid pursuant to this <u>Section 8.7</u>, such Party (or its Affiliate) shall remit to the other Party (or its Affiliate) an amount equal to fifty-percent (50%) of the refund or credit received, and in the event Buyer (or its Affiliate) receives a refund or credit of Non-Shared Transfer Taxes paid pursuant to this <u>Section 8.7</u>, Buyer (or its Affiliate) shall pay an amount equal to one hundred percent (100%) of the refund or credit received to Seller. Sellers shall file or cause to be filed all Tax Returns with respect to any Transfer Taxes, and each Buyer agrees to cooperate with Sellers in the filing of any such Tax Returns with respect to the Shared Transfer Taxes, including by promptly supplying any information in its possession that is reasonably necessary for the Sellers to complete such Tax Returns.
 - 8.8 VAT on the Acquired Assets. All VAT shall be handled as follows:

(a) Subject to Section 8.8(c) below, for those jurisdictions that have adopted Art. 19 and Art. 29 of the VAT Directive or, for non EU countries, a Law to the same effect, the

Sellers and the Buyer agree that they shall use their reasonable endeavors to ensure that the transfer of Acquired Assets is treated as a "transfer of a totality of assets or part thereof" (in the UK a "transfer of a business as a going concern") in line with Art. 19 and Art. 29 of the VAT Directive or, for non EU countries, a Law to the same effect, such that no supply of goods or services will take place and the transfer will be out of the scope of the relevant VAT Law, provided that if the relevant Tax authority asserts that Art. 19 or Art. 29 does not apply and such transfer of Acquired Assets is a taxable transaction in line with Title IV of the VAT Directive or, for non EU countries, a Law to the same effect, paragraph (b) below shall apply to such transfer of Acquired Assets.

(b) Subject to Section 8.8(c) below, for those jurisdictions that have not adopted Art. 19 or Art. 29 of the VAT Directive or, for non EU countries, a Law to the same effect, Sellers and Buyers shall agree to a just and reasonable allocation of the portion of the Global Purchase Price minus the Seller Payment, as adjusted pursuant to Section 3.3, and plus Assumed Liabilities corresponding to each particular jurisdiction treating the transfer as a taxable transaction. Such allocation shall be the VAT exclusive consideration for the Acquired Assets in such jurisdiction and VAT shall (to the extent applicable) be payable by the relevant Buyer in addition to such consideration (unless the transfer of the Acquired Assets is subject to a reverse charge scheme in line with Art. 194 et seq. of the VAT Directive or, for non EU countries, a Law to the same effect), upon delivery of a valid VAT invoice. Where the allocation in a particular jurisdiction is further increased, Buyer Parent shall pay, on behalf of the relevant Buyer (where appropriate), an amount equal to any additional VAT that becomes due as a result of such increase, with payment to be made by Buyer Parent, on behalf of the relevant Buyer (where appropriate), on receipt of a valid VAT invoice and where the allocation in a particular jurisdiction is further decreased, Seller Parent, on behalf of the relevant Buyer (where appropriate), and shall, to the extent the excess VAT is actually recovered by it or a member of a group to which it belongs for VAT purposes, or is creditable by a Seller against any VAT liability of a Seller or a member of a group to which it belongs for VAT purposes, or is creditable by a Seller against any VAT liability of a Seller or a member of a group to which it belongs for VAT purposes, or is creditable by a Seller against any VAT liability of a Seller or a member of a group to which it belongs for VAT purposes, pay such excess VAT together with any interest to Buyer Parent on behalf of the relevant Buyer (where appropriate). For t

(c) The Sellers and the Buyers shall use their reasonable endeavors to ensure that the transfer of the Transferred Shares is treated as exempt for VAT purposes in compliance with the principles of Art. 135(1)(f) of the VAT Directive or, for non-EU countries, of a Law to the same effect.

(d) In case any Seller is required, in order to satisfy the statutory requirements of the relevant jurisdiction, or may in order to enable Buyers to recover any applicable VAT, to supply the relevant Buyer with an invoice drawn up in compliance with the principles of Title XI, Chapter 3 of the VAT Directive or, for non EU countries, of a Law to the same effect, Seller shall do so.

8.9 Notwithstanding this Article VIII or any other provision of this Agreement, neither Seller Parent nor any of its Affiliates shall be liable for any Taxes related to any actions, elections, or transactions entered into by any of the Buyers.

8.10 <u>Tax-Sharing and Procedures Agreement</u>. Prior to Closing, the Parties shall enter into an agreement which provides for the allocation, sharing, indemnity, and reimbursement of, as well as the conduct of procedures related to, Taxes (including related to claims and payments with respect to claims for UK Group Relief purposes) amongst the Transferred Entities and the other members of any affiliated, consolidated, combined or unitary group of which a Transferred Entity and a Seller are members (the "**Tax-Sharing and Procedures Agreement**"). The allocations and procedures provided for under the Tax-Sharing and Procedures Agreement shall be consistent with the principles set forth in Treasury Regulations Section 1.1552-1(a)(1) and this Agreement (including Sections 8.2 and 8.3) and, subject to being on arm's length terms, UK Group Relief legislation.

ARTICLE IX

OTHER AGREEMENTS

9.1 Books and Records; Access; Assistance.

(a) Subject to any limitations imposed by Law, including limitations that are required to preserve any applicable attorney-client privilege, after the Closing, as applicable, Buyers and Sellers shall make reasonably available to each other and their respective Affiliates and Representatives (as reasonably requested) and to any taxing authority or any other Governmental Authority, all books and records to the extent relating to the Business, the Acquired Assets and the Assumed Liabilities for all periods prior to the Closing Date and shall use reasonable best efforts to preserve, for at least six (6) years after the Closing Date or, if applicable and later, the expiration of the applicable statutes of limitations or extension thereof: (i) all such books and records, (ii) Tax work papers, Tax information statements, Tax documents and Tax Returns pertaining to the Business, the Acquired Assets or the Products that apply to Claims asserted for Taxes related to the matters addressed in such work papers, information statements, documents and returns and (iii) government Contract information, records or documents, to the extent relating to the Business or the Acquired Assets, for the required retention period. Buyers and Sellers shall also make available to each other, during normal business hours when reasonably requested, personnel responsible for preparing or maintaining information, records and documents, in connection with governmental Contracts, each as it relates to the Business, including product Liability and general insurance Liability. Sellers shall have the right to retain copies of all information and documents provided by Sellers to Buyers pursuant to Section 2.1. The right to access provided by this Section 9.1 shall include the right to make copies of accessed documents, provided that (A) all such copies shall be at the sole cost and expense of the requesting Party, and (B) the Party providing access shall have the right to reasonably redact all such documents.

(b) In the event of, and for so long as any Party hereto (or its Affiliates) is prosecuting, participating in, contesting or defending any Proceeding, whenever filed or made, in connection with or involving (i) any transaction contemplated under this Agreement or (ii) the

conduct or operation of the Business prior to or after the Closing, the other Party hereto shall, and shall cause its respective Affiliates to, (A) cooperate with it and its counsel in, and assist it and its counsel with, the contest or defense, (B) make available its personnel (including for purposes of fact finding, consultation, interviews, depositions and, if required, as witnesses), and (C) provide such information, testimony and documents, including, but not limited to, books and records, Tax Returns, Contracts, and commitments of the Business, whether stored in paper, electronic, or other format, during normal business hours and upon reasonable notice, in each case as shall be reasonably necessary in connection with the contest or defense, all at the sole cost and expense (not including employee compensation and benefits costs) of the prosecuting, participating, contesting or defending Party. The covenants in this Section 9.1(b) shall not be deemed to limit the access and cooperation to be provided to Sellers or Buyers pursuant to Section 6.4 or Section 9.1(a).

(c) Notwithstanding the foregoing, this Section 9.1 shall not provide any Buyer or Seller (or Representative of any thereof) any access rights to documents or information of the other Party which access would violate any Laws or obligations regarding the confidentiality thereof (unless any such violation could be and is avoided by the recipient's execution and delivery of an appropriate confidentiality agreement), or waive any attorney-client, work product, or like privilege, or for the purpose of use in connection with potential or actual litigation, arbitration or mediation between any Buyer or any Affiliate of any Buyer, on the one hand, and any Seller or any Affiliate of any Seller, on the other hand (nor, for the avoidance of doubt, shall any Buyer or any Seller or any of their respective Affiliates have any right to use any document or information obtained from the other pursuant to this Section 9.1 in any such Proceeding).

9.2 Privileged Matters.

(a) Each party hereto acknowledges that: (i) each party and its Affiliates has or may obtain Privileged Information; (ii) there are and/or may be a number of Litigation Matters affecting both of Buyer Parent and Seller Parent; (iii) both Buyer Parent and Seller Parent have a common legal interest in Litigation Matters, in the Privileged Information and in the preservation of the confidential status of the Privileged Information, in each case relating to the Business; and (iv) both Buyer and Seller intend that the transactions contemplated hereby and by the Agreement and the Ancillary Agreements and any transfer of Privileged Information in connection therewith shall not operate as a waiver of any potentially applicable privilege.

(b) Following the Closing, each of Buyer Parent and Seller Parent agrees, on behalf of itself and each of its Subsidiaries, not to disclose or otherwise waive any privilege attaching to any Privileged Information relating to the Business without providing prompt written notice to and obtaining the prior written consent of the other, which consent shall not be unreasonably withheld, conditioned or delayed and shall not be withheld, conditioned or delayed if the other party certifies that such disclosure is to be made in response to a likely threat of suspension or debarment or similar action; provided, however, that Buyer shall not be required to give any such notice or obtain any such consent and may make such disclosure or waiver with respect to Privileged Information if such Privileged Information relates solely to the Transferred Business (unless such information relates to matters for which Seller may have indemnification obligations under this Agreement or the Ancillary Agreements). In the event of a disagreement

concerning the reasonableness of withholding such consent, no disclosure shall be made prior to a resolution of such disagreement by a court of competent jurisdiction, provided that the limitations in this sentence shall not apply in the case of disclosure required by Law and so certified as provided in the first sentence of this paragraph.

(c) After the Closing, upon receipt of any subpoena or other compulsory disclosure notice from a court, other governmental agency or otherwise which requests disclosure of Privileged Information relating to the Transferred Business, to the extent permitted by Law, Seller Parent or Buyer Parent (in the case of information relating to matters for which Seller may have indemnification obligations under this Agreement or the Ancillary Documents, or in the case of Privileged Information not solely related to the Transferred Business and in which Seller may have an interest), as applicable shall as promptly as practicable provide to the other party (following the notice provisions set forth herein) a copy of such notice, the intended response, and all materials or information that might be disclosed and the proposed date of disclosure. In the event of a disagreement as to the intended response or disclosure, unless and until the disagreement is resolved as provided in paragraph (b) of this Section, the disclosing party shall, at the other party's expense, cooperate to the extent such other party seeks to limit such disclosure and take all reasonable steps to resist or avoid such disclosure, except as otherwise required by a court order requiring such disclosure.

9.3 Seller Non-Solicitation

(a) Seller Parent agrees that, except as provided in Section 9.3(b) and 9.3(d), it shall not, and it and its Subsidiaries shall use reasonable best efforts to cause its and their Representatives not to, directly or indirectly, solicit, initiate or knowingly encourage any inquiry with respect to, or the making or submission of, any Company Proposal. Seller Parent shall, and shall cause its Subsidiaries and its and their respective Representatives to, immediately cease and cause to be terminated all existing discussions or negotiations with any Person conducted heretofore with respect to any Company Proposal, or any inquiry or proposal that would reasonably be expected to lead to a Company Proposal, request the prompt return or destruction of all confidential information previously furnished in connection therewith and immediately terminate all physical and electronic data room access previously granted to any such Person or its Representatives.

(b) If Seller Parent receives a bona fide written Company Proposal or inquiry or proposal from a third party who is intending to make a Company Proposal and such Company Proposal, inquiry or proposal was made after the date of this Agreement and did not result from a breach of this Section 9.3, Seller Parent may take any or all of the following actions: (x) furnish non-public information to the third party (and any Persons working in concert with such third party and to their respective potential financing sources and Representatives) making or intending to make such Company Proposal (provided that all such information which relates to the Business has previously been provided to Buyer Parent or is provided to Buyer Parent substantially concurrently with the time it is provided to such third party), if, and only if, prior to so furnishing such information, Seller Parent receives from the third party an executed confidentiality agreement on terms no less restrictive of such third party than the Confidentiality Agreement and (y) engage in discussions or negotiations with the third party with respect to such Company Proposal. The limitations and obligations set forth in the first sentence of Section 9.3(a) and in this Section 9.3(b) and Section 9.3(c) shall be of no further force or effect following the Financing Contingency Release Date.

- (c) As promptly as practicable, and in any event within twenty-four (24) hours, following the receipt by the Company, any of its Subsidiaries or any of their respective representatives of a Company Proposal, Seller Parent shall provide Buyer Parent written notice thereof.
- (d) Prior to the Financing Contingency Release Date, Seller Parent shall be permitted to enter into a definitive agreement with respect to a Company Proposal (a "Company Transaction Agreement") so long as Seller Parent provided written notice of such Company Proposal to Buyer Parent at least two Business Days prior to entering into such definitive agreement and is not in breach of this Section 9.3.
- (e) As used in this Agreement, "Company Proposal" means any bona fide proposal or bona fide offer made by any Person for (i) the acquisition of Seller Parent by scheme of arrangement, takeover offer or business combination transaction as a result of which the holders of Seller Parent shares immediately prior to such transaction do not, in the aggregate, own at least 20% of the outstanding voting power of the surviving or resulting entity in such transaction immediately after consummation thereof; (ii) the acquisition by any Person (or the stockholders of any Person) of 80% or more of the outstanding Seller Parent Shares; or (iii) any merger, business combination, consolidation, share exchange, recapitalization or similar transaction involving Seller Parent as a result of which the holders of Seller Parent Shares immediately prior to such transaction do not, in the aggregate, own at least 20% of the outstanding voting power of the surviving or resulting entity in such transaction immediately after consummation thereof.
- (f) As used in this Agreement, "Financing Contingency Release Date" means the date that Buyer Parent either (i) delivers the Debt Commitment Letter or (ii) irrevocably waives the condition to its obligation to close in Section 10.3(e).
- (g) Seller Parent agrees that any violation of the restrictions set forth in this <u>Section 9.3</u> by any Representative of Seller Parent or any of its Subsidiaries shall constitute a breach by the Seller Parent of this Section 9.3.

9.4 Seller Parent's Financing Cooperation.

(a) Seller Parent agrees to, and to cause its Subsidiaries to, use reasonable best efforts to provide such assistance (and to use reasonable best efforts to cause its and their respective Representatives to provide such assistance) with respect to the Debt Financing, any Alternative Financing, any Replacement Financing or any equity or equity-linked offerings or placements (collectively, the "Financing") as is reasonably requested by Buyer Parent, including using reasonable best efforts with respect to: (i) upon request participating in, and assisting with, the due diligence and marketing efforts related to the Financing, including participating in a reasonable number of meetings (including one-on-one meetings or conference calls with the parties acting as agents, arrangers, underwriters or initial purchasers for, and prospective lenders, investors in, and purchasers of, the Financing), presentations, due diligence sessions, drafting sessions, road shows and sessions with prospective lenders, investors and rating agencies and other syndication activities; (ii) delivering to Buyer

Parent and its Financing Sources as promptly as reasonably practicable of documentation and other information reasonably requested by the Financing Sources under applicable "know-your customer" and anti-money laundering rules and regulations, including the USA Patriot Act; (iii) furnishing the Buyer Parent and its Financing Sources with (A)(I) the Financial Statements required pursuant to Section 9.14; and (II) any additional audited and unaudited financial statements with respect to the businesses acquired by the Business required by Rule 3-05(b)(2) of Regulation S-X under the Securities Act as would be applicable to a registration statement filed with the SEC on Form S-1 by Buyer Parent (assuming the consummation of the Transactions); and (B) as promptly as practical after requested by Buyer Parent, all financial statements, financial data, audit reports and other information with respect to the Business of the type required by Regulation S-X and Regulation S-K under the Securities Act and other rules and regulations of the SEC as may reasonably be requested by Buyer Parent and of the type and form that would be required in registration statements filed with the SEC on Form S-1 by Buyer Parent (assuming the consummation of the Transactions) (including all information regarding the Business reasonably requested by Buyer Parent to prepare pro forma financial statements meeting the requirements of Regulation S-X) (the information and documents in clauses (A) and (B), collectively, the "Required Information"); (iv) causing its independent auditors to cooperate with the Financing consistent with their customary practice, including by participating in a reasonable number of drafting sessions, providing customary "comfort letters" (including customary "negative assurances") and customary assistance with the due diligence activities of Buyer Parent and the Financing Sources (including by participating in a reasonable number of accounting due diligence sessions), and customary consents to the inclusion of audit reports in any relevant marketing materials, prospectuses, offering memoranda, registration statements and related government filings; (v) assisting Buyer Parent and the Financing Sources in the preparation of (A) offering documents, prospectuses, registration statements, syndication documents and materials, including bank information memoranda (confidential and public), private placement memoranda, offering memoranda, lender and investor presentations and similar documents for the Financing and (B) materials for rating agency presentations, and similar documents in connection with the Financing; (vi) cooperating reasonably with each Financing Source's due diligence including in relation to the provision of, and reliance on, any reports produced in connection with the transaction; (vii) executing and delivering customary definitive financing documents, including certificates and other documents, to the extent reasonably requested by Buyer Parent; and (viii) providing customary projected financial information relating only to the Business as reasonably requested by Buyer Parent to permit Buyer Parent to prepare customary projected financial information relating to Buyer Parent (to be prepared on a pro forma basis assuming the consummation of the Transactions) which are customarily required by financing sources for the syndication of credit facilities similar to those described in the Debt Commitment Letter or for purposes of obtaining corporate and debt ratings; provided that the effectiveness of any definitive documentation executed by the Seller Parent or any of its Subsidiaries shall be subject to the consummation of the Transactions. The Seller Parent hereby consents to the use of all of its and its Subsidiaries' logos in connection with the Financing, provided that such logos are used solely in a manner that is not intended to or reasonably likely to harm or disparage the Seller Parent or any of its Subsidiaries or the reputation or goodwill of the Seller Parent or any of its Subsidiaries. Notwithstanding any other provision set forth herein or in any other agreement between the Seller Parent and the Buyer Parent (or its affiliates), the Seller Parent agrees that Buyer Parent and its affiliates may share

customary projections with respect to the Business with the Financing Sources identified in the Debt Commitment Letter and to any existing lenders of the Buyer Parent, and that Buyer Parent, its affiliates and such Financing Sources may share such information with potential Financing Sources in connection with any marketing efforts in connection with the Financing, provided that the recipients of such information are subject to customary confidentiality arrangements between Buyer Parent and such parties (which need not include Seller Parent). Notwithstanding anything to the contrary in this Agreement, none of the Seller Parent, any of its Subsidiaries or any of its or their respective directors or officers or other personnel shall be required by this Section 9.4 (1) to take any action or provide any assistance that unreasonably interferes with the ongoing operations of the Seller Parent and its Subsidiaries; (2) to pass resolutions or consents to approve or authorize the execution of the Debt Financing or the Debt Financing Documents prior to the Closing (other than customary authorization and representation letters provided in connection with marketing or offering documents); or (3) to execute or deliver any certificate, document, instrument or agreement that is effective prior to the Closing or agree to any change or modification of any existing certificate, document, instrument or agreement that is effective prior to the Closing (other than customary authorization and representation letters and a payoff letter with respect to any credit agreement). Buyer Parent shall (1) promptly upon request by the Seller Parent, reimburse the Seller Parent for all reasonable and documented outof-pocket costs and expenses (including reasonable attorneys' fees) incurred by the Seller Parent or any of its Subsidiaries in connection with providing the assistance contemplated by this Section 9.4 other than accounting costs and expenses incurred in connection with preparing any financial statements or financial information required pursuant to Section 9.14 or included in the Required Information and assisting Buyer Parent with the preparation of pro forma financial statements and (2) indemnify and hold harmless the Seller Parent and its Subsidiaries and its and their respective directors, officers, personnel and advisors from and against any and all liabilities, losses, damages, claims, costs, expenses (including attorneys' fees), interest, awards, judgments and penalties suffered or incurred in connection with the Financing or any assistance or activities in connection therewith (other than arising from, fraud, intentional misrepresentation, misstatements or omissions on the part of the Seller Parent or any of its affiliates or Representatives).

(b) Seller Parent agrees to, and to cause its Subsidiaries to, use reasonable best efforts to provide such assistance (and to use reasonable best efforts to cause its and their respective Representatives to provide such assistance) with respect to obtaining the Debt Commitment Letter as is reasonably requested by Buyer Parent, including cooperating in a timely manner with the Financing Sources' diligence and otherwise taking in a timely manner such of the actions set out in Section 9.4(a) as are relevant to Buyer Parent obtaining the Debt Commitment Letter within the time period and in the amount contemplated in Section 11.1(e).

9.5 Buyer Parent's Financing Obligation.

(a) The Parties acknowledge that, except as provided in Section 10.3(e) there is no contingency on Buyer Parent's obligations under this Agreement related to Buyer Parent's ability to secure financing. Promptly after entering into the Debt Commitment Letter, Buyer Parent shall deliver to Seller Parent a true and complete copy of the executed Debt Commitment Letter and any related fee letters (redacted as to economic terms and other commercially sensitive numbers and provisions specified in any such fee letter (including any provisions

relating to "flex" terms or similar concepts) excluding in each case, any such provisions that could adversely affect the amount of the financing or the Debt Financing Conditions). On and after the date of the Debt Commitment Letter, Buyer Parent shall take, or use its reasonable best efforts to cause to be taken, all actions and do, or use its reasonable best efforts to cause to be done, all things necessary to obtain (x) the Debt Commitment Letter by August 10, 2015 on terms that would satisfy the condition set forth in section 10.3(e) and (y) the Debt Financing on or prior to the Closing Date on the terms and conditions set forth in the Debt Commitment Letter, including: (a) maintaining in effect and enforcing the Debt Commitment Letter and complying with its obligations thereunder, provided, that the Debt Commitment Letter may be amended, supplemented, modified and replaced as permitted pursuant to this Section 9.5; (b) participation by senior management of Buyer Parent in, and assistance with, the preparation of rating agency presentations and meetings with rating agencies; (c) satisfying on a timely basis all conditions to the Debt Financing (including the Debt Financing Conditions) set forth in the Debt Commitment Letter that are reasonably within Buyer Parent's control; (d) negotiating, executing and delivering Debt Financing Documents that reflect the terms contained in the Debt Commitment Letter (including any "market flex" provisions related thereto); and (e) subject to Section 3.1, drawing the full amount of the Financing, in the event that the conditions set forth in Section 10.1 and Section 10.3, and the Financing Conditions, have been satisfied. Without limiting Buyer Parent's other obligations under this Section 9.5, if a Financing Failure Event occurs Buyer Parent shall (i) promptly notify the Seller Parent of such Financing Failure Event of which Buyer Parent becomes aware and the reasons therefor, (ii) in consultation with the Seller Parent, use its reasonable best efforts to obtain alternative financing from alternative Financing Sources on terms (including relating to certainty of funding) not materially less favorable to Buyer Parent, with lenders reasonably satisfactory to Buyer Parent, in an amount sufficient, when added to the available cash of Buyer Parent and any portion of the Financing that is available, to consummate the Transactions, as promptly as practicable following the occurrence of such event (such financing, an "Alternative Financing"), and (iii) obtain, and when obtained, provide the Seller Parent with a copy of, a new financing commitment that provides for such Alternative Financing. Buyer Parent shall not, without the Seller Parent's prior written consent, agree to any amendment or modification to, or any waiver of any provision or remedy under, the Debt Commitment Letter or any Debt Financing Document unless the terms and conditions thereof, as so amended, modified or waived, are in the aggregate at least as favorable to the Seller Parent and Buyer Parent as those contained therein prior to giving effect to such amendment, modification or waiver; provided, that Buyer Parent may, without the Seller Parent's prior written consent, (x) enter into any amendment, replacement, supplement or other modification to or waiver of any provision of the Debt Commitment Letter or any Debt Financing Document that would not reasonably be expected to prevent, materially delay or materially impede the timely consummation of the Debt Financing or the Transactions, (y) replace or amend the Debt Commitment Letter solely to add lenders, lead arrangers, bookrunners, syndication agents or similar entities that have not executed the Debt Commitment Letter and (z) implement or exercise the "flex" provisions contained in one or more fee letters related to the Debt Financing. Notwithstanding the foregoing, the Parties hereto agree that the following would reasonably be expected to prevent, materially delay or materially impede the timely consummation of the Debt Financing or the Transactions: (i) any amendment, modification or waiver of the conditions to obtaining the Debt Financing, unless such amendment, modification or waiver results in conditions that are in the aggregate substantially

equivalent (or that are more favorable to the Seller Parent and Buyer Parent), (ii) any amendment, modification or waiver that reduces the amount of the Debt Financing or (iii) any amendment, modification or waiver that materially adversely affects the ability of Buyer Parent or its affiliates to enforce their rights against the other parties to the Debt Commitment Letter. Buyer Parent shall keep the Seller Parent reasonably informed on a reasonably current basis of the status of its efforts to obtain the Debt Financing.

(b) Notwithstanding any other provision in this Agreement, Buyer Parent shall have the right to substitute the proceeds of consummated equity or equity-linked offerings or debt offerings or other incurrences of debt (including unsecured notes) for all or any portion of the Debt Financing by reducing commitments under the Debt Commitment Letter or any Debt Financing Document; provided, that to the extent any such debt has a scheduled special or mandatory redemption right, such right is not exercisable prior to the earliest of the consummation of the Transactions on the Closing Date, the termination of this Agreement or the Outside Date as applicable (for the avoidance of doubt as it may be extended pursuant to this Agreement). Further, Buyer Parent shall have the right to substitute commitments in respect of other debt financing for all or any portion of the Debt Financing from the same and/or alternative bona fide third-party financing sources so long as all conditions precedent to effectiveness of definitive documentation for such debt financing have been satisfied and the conditions precedent to funding of such debt financing are in the aggregate, in respect of certainty of funding, substantially equivalent to (or no less favorable in any material respect to the Seller Parent than) the Debt Financing Conditions (any such debt financing which satisfies the foregoing, the "Replacement Financing"; the definitive documentation for any such Replacement Financing, the "Replacement Financing" Documents"). The representations, warranties, covenants and other restrictions of Buyer Parent and its affiliates contained in this Agreement with respect to the Debt Financing and the Debt Commitment Letter shall apply equally to any Replacement Financing and Replacement Financing Documents.

9.6 Buyer Conduct; Non-Use.

(a) Buyer Parent agrees that it shall not in any manner or in any location including through entities that Buyer Parent directly or indirectly controls, for the period beginning on the Closing Date until the fifth anniversary of the Closing Date, use the Information (as defined in Section 13.3(c)) to develop, market or commercialize any product that has the same active ingredient and is competitive with any Retained Product. Buyer Parent acknowledges that such Information provides fundamentally competitive information. Seller Parent shall be entitled to specific performance of this provision pursuant to Section 13.11 hereof.

(b) Seller Parent agrees that it shall not in any manner or in any location including through entities that Seller Parent directly or indirectly controls, for the period beginning on the Closing Date hereof until the fifth anniversary of the Closing Date, use the Information (as defined in Section 13.3(c)) to develop, market or commercialize any product that has the same active ingredient and is competitive with any Product. Seller Parent acknowledges that such Information provides fundamentally competitive information. Buyer Parent shall be entitled to specific performance of this provision pursuant to Section 13.11 hereof.

9.7 Wrong Pocket Assets.

(a) If any Acquired Assets remains vested in any Seller or any of their respective Affiliates following Closing, such Seller shall (or shall cause its applicable Affiliate to) transfer such Acquired Asset as soon as reasonably practicable to the Buyer Parent or its designee for no consideration (it being acknowledged and agreed that the Buyer shall have already paid good consideration for all Acquired Assets by paying the Global Purchase Price). The Seller shall notify the Buyer as soon as reasonably practicable upon becoming aware that that there are any Acquired Assets in its possession or control or that of any Affiliate of any Seller.

(b) If any Excluded Asset is vested in the Buyer Parent or any of its Affiliates following Closing, Buyer Parent shall (or shall cause its applicable Affiliate to) transfer such Excluded Asset as soon as reasonably practicable to the Seller Parent or its designee for no consideration (it being acknowledged and agreed that the parties have not agreed to sell such Excluded Asset). Buyer Parent shall notify the Seller Parent as soon as reasonably practicable upon becoming aware that that there are any Excluded Assets in its possession or control or that of any member of the Buyer Group.

9.8 Release of Indemnity Obligations.

(a) Seller Parent and Buyer Parent shall cooperate with each other with a view to entering into arrangements effective as of the Closing whereby Buyer Parent or its Affiliates would be substituted for Seller Parent or any Retained Entity in any guarantees, letters of comfort, indemnities or arrangements entered into by Seller Parent or the Retained Entities in respect of the Business (but only to the extent such guarantees, letters of comfort, indemnities or similar arrangements constitute Assumed Liabilities). If Buyer Parent or its Affiliates cannot enter into the arrangements referred to above, Seller Parent shall not terminate any such guarantee, letter of comfort, indemnity or arrangement without Buyer Parent's prior written consent; provided, however, that Buyer Parent shall enter into a separate guaranty with Seller Parent or the applicable Retained Entity to guarantee the performance of the obligations of Seller Parent or such Retained Entity, as applicable, under the Contract underlying such guarantee, letter of comfort, indemnity or arrangement to the extent such obligations constitute Assumed Liabilities or terms reasonably acceptable to Buyer Parent.

(b) Effective as of the Closing, Seller Parent shall, and shall cause its Affiliates to, cause all guarantees for money borrowed (under an indenture or otherwise) to be terminated with respect to any member of the Transferred Group. Prior to the Closing Buyer Parent shall and shall cause its Affiliates to use reasonable efforts to cause all other guarantees, letters of comfort, indemnities or arrangements entered into by the Transferred Group on behalf of Seller Parent or any of its Affiliates (other than the Transferred Group and other than to the extent included in Assumed Liabilities), to be unconditionally released or extinguished, together with any ancillary obligations thereto, without further recourse to the Transferred Group or Buyer Parent and its Affiliates.

(c) After the Closing, each of Seller Parent and Buyer Parent, at the request of the other, shall use, and shall cause their respective Affiliates to use, reasonable best efforts to

obtain any Consent, substitution or amendment required to novate or assign all Assumed Liabilities to Buyer Parent or its Affiliates and any Excluded Liabilities to Seller Parent or the Retained Entities, and obtain in writing the unconditional release of Seller Parent and its Affiliates with respect to the Assumed Liabilities and the unconditional release of Buyer Parent and its Affiliates with respect to the Excluded Liabilities, the costs of which shall be borne equally by Seller Parent and Buyer Parent.

(d) Seller Parent, on behalf of itself and its Controlled Affiliates, successors, and assigns (all such Persons, together with Seller Parent, the "Seller Release Parties"), as of the Closing, hereby releases and forever discharges Buyer Parent, and its Affiliates, successors and assigns (all such Persons (including the Transferred Entities), together with Buyer Parent, the "Buyer Release Parties"), from any and all Liabilities which the Buyer Release Parties may have or may have had, known or unknown, from the beginning of the world until the Closing, arising out of or against the Transferred Group, the Acquired Assets or the Business; provided that nothing herein constitutes a release from, waiver of, or otherwise applies to the terms of this Agreement or any Ancillary Documents or any enforcement thereof. Seller Parent, for itself and the Seller Release Parties, hereby irrevocably covenants to refrain from, directly or indirectly, asserting any claim or demand, or commencing, instituting or causing to be commenced or voluntarily aiding, any proceeding of any kind against any Buyer Release Party, based upon any matter purported to be released hereby.

(e) Buyer Parent, on behalf of the Transferred Group and their successors, and assigns, as of the Closing, hereby releases and forever discharges Seller Parent, and its Affiliates, successors and assigns (all such Persons (excluding the Transferred Group), together with Seller Parent, the "Transferred Group Release Parties"), from any and all Liabilities which the Transferred Group Release Parties may have or may have had, known or unknown, from the beginning of the world until the Closing, to the Transferred Group or otherwise arising out of or against the Excluded Assets and the Retained Business; provided that nothing herein constitutes a release from, waiver of, or otherwise applies to the terms of this Agreement or any Ancillary Documents or any enforcement thereof. From and after the Closing, Buyer Parent on behalf of the Transferred Group, hereby irrevocably covenants to refrain from, directly or indirectly, asserting any claim or demand, or commencing, instituting or causing to be commenced or voluntarily aiding, any proceeding of any kind against any Seller Release Party, based upon any matter purported to be released hereby.

9.9 Use of Name.

(a) None of the Transferred Group shall hold itself out as continuing to be owned by Allergan. Except as expressly provided herein, effective as of the Closing, Seller Parent hereby covenants and agrees, on behalf of itself and its Affiliates, that it shall not use, license, sublicense, or otherwise permit any Person to use, any trade or service marks, trade or service names or logos including the word "Warner Chilcott" or "Forest Labs" (collectively, the "Allergan Marks") in connection with the Products; provided, however, that Buyer Parent and its Affiliates shall use commercially reasonable efforts to remove the Allergan Marks and in any event within the period of up to three (3) years after the Closing Date (the "Allergan Transition Period") to change or remove external or internal signage using any of the Allergan Marks, remove the Allergan Marks from any business

names, transition any Internet sites to domain names not including any Allergan Marks and to sell, destroy or otherwise dispose of (subject to the terms of this Agreement) any materials or inventory bearing any of the Allergan Marks. For the avoidance of doubt, such activities during the Transition Period will only be for wind-down purposes, and Seller Parent will not, and will ensure that its Affiliates do not, actively use the Allergan Marks in marketing or promotional activities during the Transition Period. With immediate effect upon the expiration of the Transition Period, Seller Parent will, and will cause its Affiliates to, cease all such wind-down use of the Allergan Marks, but will retain the right to use the terminated Allergan Marks, as applicable, after the expiration of the Allergan Transition Period solely for record keeping or administrative purposes, including the retention and use of historical or archived documents (including customer contracts and promotional materials) containing or referencing the Allergan Marks, as applicable (and not for any marketing or promotional purposes). Notwithstanding the foregoing, in no event will Seller Parent or its Affiliates register any Trademark (including any Internet domain name) containing or confusingly similar to any Allergan Marks.

(b) Effective as of the Closing, Seller Parent shall cease, and shall cause its Affiliates to cease, the use, or display of any trade or service marks, trade or service names or logos including the word "Actavis" (collectively, the "Actavis Marks"); provided, however, that Seller Parent and its Affiliates shall have a period of up to three (3) years after the Closing Date (the "Actavis Transition Period") to change or remove external or internal signage to change or remove external or internal signage using any of the Actavis Marks, remove the Actavis Marks from any business names, transition any Internet sites to domain names not including any Actavis Marks and to sell, destroy or otherwise dispose of (subject to the terms of this Agreement) any materials or inventory bearing any of the Actavis Marks. For the avoidance of doubt, such activities during the Transition Period will only be for wind-down purposes, and Seller Parent will not, and will ensure that its Affiliates do not, actively use the Actavis Marks in marketing or promotional activities during the Transition Period. With immediate effect upon the expiration of the Transition Period, Seller Parent will, and will cause its Affiliates to, cease all such wind-down use of the Actavis Marks, but will retain the right to use the terminated Actavis Marks, as applicable, after the expiration of the Actavis Transition Period solely for record keeping or administrative purposes, including the retention and use of historical or archived documents (including customer contracts and promotional materials) containing or referencing the Actavis Marks, as applicable (and not for any marketing or promotional purposes).

Notwithstanding the foregoing, in no event will Seller Parent or its Affiliates register any Trademark (including any Internet domain name) containing or confusingly similar to any Actavis Marks.

9.10 Directors and Officers.

(a) Prior to the Closing Date, Seller Parent shall cause each entity in the Transferred Group to hold such corporate or other meetings as are necessary pursuant to applicable Laws to discharge the members of each board of directors or equivalent governing body of such entity with effect as of the Closing. Effective as of the Closing, Seller Parent, for and on behalf of itself and its Affiliates, hereby acquits, releases and discharges each entity in the Transferred Group from any and all Liabilities as of the Closing to Seller or any of its Controlled Affiliates (other than the Transferred Group) that arise out of or are in connection with the Transferred Group. Seller Parent shall cause its other Affiliates to use their reasonable best

efforts to take, or cause to be taken, all appropriate action and to execute and deliver such documents and other papers, as may be required to effect the release set forth in this Section 9.10(a).

- (b) Buyer Parent agrees that all rights to exculpation, indemnification and advancement of expenses for acts or omissions occurring at or prior to the Closing, whether asserted or claimed prior to, at or after the Closing, now existing in favor of the current or former directors, officers or employees, as the case may be, of the Transferred Group as provided in their respective Organizational Documents or in any agreement as in effect on the date hereof and which has prior to the date hereof been made available to Buyer Parent shall survive the Closing.
- 9.11 Ancillary Agreements. Each of Buyer Parent and Seller Parent shall use its reasonable best efforts to negotiate in good faith (to the extent not already in agreed form) and agree upon definitive forms for the Ancillary Agreements, which definitive forms shall incorporate usual and customary provisions for similar agreements, as applicable, unless otherwise agreed by Buyer Parent and Seller Parent. The terms of the Manufacturing Agreements shall be consistent with the terms set forth in Exhibit B and the terms of the Transition Services Agreement and Reverse Transition Services Agreement shall be consistent with the terms set forth in Section 6.3(b).

9.12 Employee Non-Solicit.

- (a) Beginning on the Closing Date, neither Seller Parent nor any of its Affiliates (including its directors, officers and agents) shall, prior to the first (1st) anniversary of the Closing Date, directly or indirectly, solicit for purposes of employment, offer to hire, hire or enter into any employment agreement with any Continuing Employee, or otherwise solicit, induce or otherwise encourage any Continuing Employee to discontinue, refrain from entering into any employment relationship (contractual or otherwise) with Buyer Group or during the period prior to Closing, reassign any such Person's duties and responsibilities to functions not Related to the Business; provided, however, that notwithstanding the foregoing, for purposes of this Agreement, Seller Parent and its Affiliates shall not be prohibited from placing general advertisements or conducting general employment solicitations (including via a search firm inquiry) that are not targeted at any employee of Buyer Group.
- (b) Beginning on the Closing Date, neither Buyer Parent nor any of its Affiliates (including its directors, officers and agents) shall, prior to the first (1st) anniversary of the Closing Date, directly or indirectly, solicit for purposes of employment, offer to hire, hire or enter into any employment agreement with any Non-Business Employee, or otherwise solicit, induce or otherwise encourage any Non-Business Employee to discontinue or refrain from entering into any employment relationship (contractual or otherwise) with Seller Parent or any Seller Parent Subsidiary; provided, however, that notwithstanding the foregoing, for purposes of this Agreement, Buyer Parent and its Affiliates shall not be prohibited from placing general advertisements or conducting general employment solicitations (including via a search firm inquiry) that are not targeted at any employee of Seller Parent or any Seller Parent Subsidiary.

9.13 Divestitures. Seller Parent shall agree if reasonably requested by Buyer Parent so as to permit (or as identified by Buyer Parent as reasonably likely to be necessary to permit) the expiration or termination of the applicable waiting periods under the HSR Act or the receipt of any other Consent under any other Foreign Antitrust Law, in each case as soon as practicable after the date of this Agreement (but in any event not later than the Outside Date), to effect and agree to any sale, divestiture, license, holding separate or other similar arrangement with respect to, or other disposition of or restriction on, any of the Acquired Assets or assets and Liabilities of the Transferred Group, and take such action or actions that would in the aggregate have a similar effect; provided, however, that any such sale, divestiture, license, holding separate or other similar arrangement, disposition, restriction or action or actions (each, a "Potential Sale Transaction") is conditioned on the occurrence of, and shall become effective only from and after, the Closing. Without limiting the foregoing, to the extent requested by Buyer Parent, Seller Parent shall, and shall cause its Subsidiaries to, cooperate with Buyer Parent to facilitate the Potential Sale Transaction. To the extent reasonably requested by Buyer Parent, Seller Parent shall and shall cause its Subsidiaries to (a) enter into confidentiality agreements containing customary terms with any Persons who Buyer Parent identifies to Seller Parent as potential purchasers in a Potential Sale Transaction (such potential purchasers to be referred to as "Potential Purchasers"), (b) permit Potential Purchasers to conduct (and cooperate with such Potential Purchasers') reasonable documentary and other investigations with respect to such Potential Sale Transaction (provided, that any such Potential Purchaser executes and delivers to Seller Parent a confidentiality agreement containing customary terms), (c) comply with any applicable right of first refusal, right of first offer, right of approval and similar provisions that may be applicable to a proposed transfer of a Potential Sale Transaction, and (d) deliver such notices, make such filings and execute such contracts relating to the Potential Sale Transaction as reasonably requested by Buyer Parent and at Buyer Parent's expense.

9.14 Financial Statements. At Seller Parent's sole cost and expense, Seller Parent shall (i) deliver to Buyer Parent, no later than 120 days following the date of this Agreement, (A) audited combined balance sheets, income statements and statements of cash flows and shareholder's equity (deficit) of the Business and prepared on a "predecessor" basis (which shall have been reviewed by Seller Parent's independent accountants in accordance with the Statement on Auditing Standards No. 100) on an historical basis taking into account adjustments required by Regulation S-X and prepared on a "predecessor" basis as of and for the years ending December 31, 2012, December 31, 2013 and December 31, 2014 (such audited combined financial statements, the "Audited Financial Statements") and (B) unaudited combined balance sheets, income statements and statements of cash flows and shareholder's equity (deficit) of the Business (which shall have been reviewed by Seller Parent's independent accountants in accordance with the Statement on Auditing Standards No. 100) as of and for the quarter ended March 31, 2015 and the quarter and the six-month period ended June 30, 2015 (and the corresponding periods from the previous year), (ii) after the date of this Agreement and prior to Closing, deliver to Buyer Parent no later than 40 days following the end of each fiscal quarter (except the fourth quarter of any fiscal year) unaudited combined balance sheets, income statements and statements of cash flows and shareholder's equity (deficit) of the Business on an historical basis taking into account adjustments required by Regulation S-X and prepared on a "predecessor" basis (which shall have been reviewed by Seller Parent's independent accountants in accordance with the Statement on Auditing Standards No. 100) as of and for such fiscal quarter and, if applicable, the six-month and nine-month period then ended (and the

corresponding fiscal quarter and interim period from the previous year) (together with the financial statements required pursuant to clause (i)(B) above, the "Interim Financial Statements"), and (iii) after the date of this Agreement and prior to Closing, deliver to Buyer Parent no later than 60 days following the end of each fiscal year audited combined balance sheets, income statements and statements of cash flows and shareholder's equity (deficit) of the Business as of and for the completed fiscal year taking into account adjustments required by Regulation S-X and prepared on a "predecessor" basis (together with an unqualified report of Seller Parent's independent accountants thereon) of the Business (together with the Audited Financial Statements and the Interim Financial Statements, the "Financial Statements"). The Financial Statements shall be prepared in accordance with U.S. GAAP and shall be in compliance in all material respects with all requirements of Regulation S-K and Regulation S-X under the Securities Act that would apply in order for a registration statement filed with the SEC on Form S-1 that contains such Financial Statements to be declared effective.

9.15 Seller Debt. Effective as of the Closing, no Transferred Group shall have, and Buyer Parent and its Affiliates shall not be required to assume, any indebtedness for borrowed money.

ARTICLE X

CONDITIONS TO CLOSING

- 10.1 Conditions to Each Party's Obligations. The respective obligations of each Party to consummate the Transactions at Closing are subject to the satisfaction, at or prior to the Closing, of each of the following conditions, any and all of which may be waived, in whole or in part in writing, by Seller Parent and Buyer Parent, as the case may be, to the extent permitted by law:
- (a) No Adverse Law or Order (other than any Antitrust Law) that is of material economic significance to the Transactions or would result in a criminal liability on the part of any officer or director of the Buyers or the Sellers shall be in effect; and
- (b) All applicable waiting periods (or extensions thereof) or necessary approvals relating to the Transactions under the HSR Act and the Antitrust Laws of the European Union shall have expired, been terminated or received, and no order, judgment or decree shall have been issued by, and no legal proceeding shall have been threatened in writing by or pending before, a Governmental Authority under any Antitrust Law of the United States or the European Union, against the Seller Parent or Buyer Parent that is reasonably likely to temporarily or permanently enjoin, restrain or prevent the consummation of the Transactions.
- 10.2 <u>Conditions to Sellers' Obligations</u>. The obligation of Sellers to consummate the Transactions contemplated by this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions, unless waived in writing by Seller Parent:
- (a) the representations and warranties set forth in (i) Section 5.7 [Absence of Certain Changes or Events] and Section 5.11 [Required Vote] shall be true and correct as of the date of this Agreement and at and as of the Closing as if made on the Closing Date (except that

representations and warranties that by their terms speak specifically as of the date of this Agreement or some other date shall be true and correct as of such date) and (ii) the remainder of Article V shall be true and correct as of the date of this Agreement and at and as of the Closing as if made on the Closing Date (except that representations and warranties that by their terms speak specifically as of the date of this Agreement or some other date shall be true and correct as of such date), except, in the case of this clause (ii), for inaccuracies that, individually or in the aggregate, have not and could not reasonably be expected to have a Buyer Material Adverse Effect (with such representations and warranties in this clause (ii) read for such purposes without any materiality or Buyer Material Adverse Effect qualifications);

- (b) Buyer Parent shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing;
- (c) Since the date of this Agreement, no Effects have occurred which, individually or in the aggregate, have had (and have continued to have) or would reasonably be expected to have, a Buyer Material Adverse Effect;
 - (d) Buyer Parent shall have delivered, or caused the applicable Buyer to have delivered, to Seller Parent:
 - (i) each of the documents required to be delivered to Buyer Parent or its Affiliates pursuant to clause (iii) of Section 3.2(a)(ii); and
- (ii) a certificate dated as of the Closing Date, signed by duly authorized officer of Buyer Parent, certifying that the conditions set forth in Section 10.2(a) and Section 10.2(b) have been duly satisfied in all respects.
- 10.3 <u>Conditions of Buyers' Obligations</u>. The obligation of Buyers to consummate the Transactions contemplated by this Agreement and to take the other actions required to be taken by Buyers at the Closing is subject to the satisfaction, at or prior to the Closing, of each of the following conditions, unless waived in writing by Buyer Parent:
- (a) the representations and warranties set forth in (i) Section 4.9(a) [Absence of Certain Changes or Events], Section 4.18 [Required Vote] and Section 4.22 [No Undisclosed Liabilities] shall be true and correct as of the date of this Agreement and at and as of the Closing as if made on the Closing Date (except that representations and warranties that by their terms speak specifically as of the date of this Agreement or some other date shall be true and correct as of such date) and (ii) the remainder of Article IV above shall be true and correct as of the date of this Agreement and at and as of the Closing Date as if made on the Closing Date (except that representations and warranties that by their terms speak specifically as of the date of this Agreement or some other date shall be true and correct as of such date), except, in the case of this clause (ii), for inaccuracies of the representations and warranties that, individually or in the aggregate, do not have and could not reasonably be expected to have a Seller Material Adverse Effect (with such representations and warranties in this clause (ii) read for such purposes without any materiality or Seller Material Adverse Effect qualifications);
- (b) Seller Parent shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing;

- (c) Since the date of this Agreement, no Effects have occurred which, individually or in the aggregate, have had (and have continued to have) or would reasonably be expected to have, Seller Material Adverse Effect; and
 - (d) Seller Parent shall have delivered, or caused the applicable Seller to have delivered, to Buyer Parent:
 - (i) each of the documents required to be delivered to Buyer Parent pursuant to clauses (iii) and (iv) of Section 3.2(a)(i); and
- (ii) a certificate dated as of the Closing Date, signed by a duly authorized officer of Seller Parent, certifying that the conditions set forth in Sections 10.3(a) and Section 10.3(b) have been duly satisfied in all respects
- (e) Buyer Parent (or one or more of Buyer Parent and the other Buyers) has obtained the Debt Commitment Letter (which shall be subject to a termination date for such financing commitment that is no sooner than the Outside Date), providing for debt commitments in an aggregate principal amount equal to at least \$33,750,000,000.
- 10.4 <u>Frustration of Closing Conditions</u>. Except as required by Law, neither Seller Parent nor Buyer Parent may rely on the failure of any condition set forth in <u>Section 10.1</u> or <u>Section 10.2</u> or <u>Section 10.3</u>, as the case may be, to be satisfied if such failure was caused by such Party's material breach of this Agreement.

ARTICLE XI

TERMINATION

- 11.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing:
 - (a) by the mutual written Consent of Seller Parent and Buyer Parent;
- (b) by Seller Parent or Buyer Parent if the Closing has not occurred by midnight, Eastern Time, at the end of the day on July 26, 2016 (as it may be extended pursuant to the first or third proviso of this Section 11.1(b), the "Outside Date"); provided, however, that if the Marketing Period has commenced on or prior to the Outside Date, but has not concluded as of the Outside Date, the Outside Date shall be extended to the fifth Business Day following the conclusion of the Marketing Period; provided, further, that the right to terminate this Agreement pursuant to this Section 11.1(b) shall not be available to any Party whose breach of any representation, warranty, covenant or agreement set forth in this Agreement has been the cause of, or resulted in, the Closing not occurring prior to the Outside Date; provided, however, either Buyer Parent or Seller Parent may, within three (3) business days immediately prior to July 26, 2016, elect to extend the Outside Date by delivering a written notice to the other Party stating that if on the Outside Date, the conditions set forth in Section 10.1(b) have not been satisfied or waived but all other conditions to the Closing set forth in Article X have been satisfied or waived (other than those conditions that by their nature are to be satisfied at the Closing, which conditions shall be capable of being satisfied on July 26, 2016), then the Outside Date shall be extended by three (3) months until October 26, 2016;

(c) by Seller Parent by written notice to Buyer Parent, if Buyers shall have breached any of their representations or warranties or failed to comply with any of their covenants or agreements contained in this Agreement, which breach or failure (i) would give rise to the failure of the conditions set forth in Section 10.2(a) or Section 10.2(b) and (ii) is incapable of being cured, or is not cured, by Buyers within thirty (30) days following receipt of written notice of such breach or failure to comply from Seller Parent; provided, however, that the right to terminate this Agreement under this Section 11.1(c) shall not be available to Seller Parent if Seller Parent has breached any of its representations or warranties or failed to comply with any of its covenants or agreements contained in this Agreement such that the conditions set forth in Section 10.3(a) or Section 10.3(b) could not then be satisfied;

(d) by Buyer Parent by written notice to Seller Parent, if Sellers shall have breached any of their representations or warranties or failed to comply with any of their covenants or agreements contained in this Agreement, which breach or failure (i) would give rise to the failure of the conditions set forth in Section 10.3(a) and Section 10.3(b) and (ii) is incapable of being cured, or is not cured, by Sellers within thirty (30) days following receipt of written notice of such breach or failure to comply from Buyer Parent; provided, however, that the right to terminate this Agreement under this Section 11.1(d) shall not be available to Buyer Parent if Buyer Parent has breached any of its representations or warranties or failed to comply with any of their covenants or agreements contained in this Agreement such that the conditions set forth in Section 10.2(a) or Section 10.2(b) could not then be satisfied;

(e) by Seller Parent if by August 10, 2015, Buyer Parent shall not have obtained and provided to Seller Parent the Debt Commitment Letter (which shall be subject to a termination date for such financing commitment that is no sooner than the Outside Date), providing for debt commitments in an aggregate principal amount equal to at least \$33,750,000,000; provided that the termination right under this Section 11.1(e) shall expire when such a Debt Commitment Letter is delivered;

(f) by either Seller Parent or Buyer Parent by written notice to the other, in the event of any final and nonappealable Adverse Law or Order which would result in the conditions set forth in Section 10.1(b) failing to be satisfied; or

(g) by Seller Parent prior to the Financing Contingency Release Date, in order to enter into a Company Transaction Agreement; provided that (i) substantially concurrent with the termination of this Agreement, Seller Parent enters into such Company Transaction Agreement, (ii) Seller Parent has not materially breached its obligations in Section 9.3, (iii) the Company Transaction Agreement expressly requires Seller Parent to terminate this Agreement and (iv) Seller Parent pays Buyer Parent the Seller Break Fee pursuant to Section 11.2 (for the avoidance of doubt, Seller Parent shall not be obligated to terminate this Agreement as a result of entering into a Company Transaction Agreement.

11.2 Buyer Break Fee.

(a) In the event that (i) this Agreement is terminated by Seller Parent pursuant to Section 11.1(e), and (ii) on or prior to such termination one or more Sellers were not in material breach of the terms of Section 9.4(b), then Buyer Parent shall pay, or cause to be paid, to Seller Parent, an amount equal to \$2,500,000,000.

(b) In the event that (i) this Agreement is terminated by Seller Parent or Buyer Parent pursuant to Section 11.1(b) or Section 11.1(f), (ii) on or prior to such termination, one or more Sellers were not in material breach of the terms of Section 6.2 and (iii) the conditions set forth in Section 10.1(a) (in the case of an Adverse Law or Order relating to an Antitrust Law) or Section 10.1(b) have not been satisfied as of the date of termination, then Buyer Parent shall pay, or cause to be paid, to Seller Parent, an amount equal to \$1,000,000,000.

(c) The amounts payable pursuant to this Section 11.2(c) (each, a "Buyer Break Fee") shall be paid by wire transfer of immediately available funds to one or more accounts (but no more than three) specified by Seller Parent in writing to Buyer Parent on the second Business Day following termination of this Agreement and shall be paid by the fifth Business Day following termination of this Agreement. For the avoidance of doubt, in no event shall Buyer Parent be obligated to pay or cause to be paid a Buyer Break Fee on more than one occasion. Notwithstanding anything to the contrary in this Agreement, in the event a termination pursuant to Section 11.1(b), Section 11.1(f) or Section 11.1(e) occurs, Seller Parent's right to receive payment of a Buyer Break Fee shall be the sole and exclusive remedy (whether at law, in equity, in contract, tort or otherwise) of Sellers and their Affiliates against Buyer Parent, its Affiliates and the Financing Sources.

11.3 Seller Break Fee.

(a) In the event that this Agreement is terminated by Seller Parent pursuant to Section 11.1(g), then Seller Parent shall pay, or cause to be paid, to Buyer Parent, an amount equal to \$2,500,000,000 (the "Seller Break Fee"). The Seller Break Fee payable pursuant to this Section 11.3 shall be paid by wire transfer of immediately available funds to one or more accounts specified by Buyer Parent in writing to Seller Parent on the second Business Day following termination of this Agreement pursuant to Section 11.1(g), without offset or deduction of any kind. For the avoidance of doubt, in no event shall Seller Parent be obligated to pay, or cause to be paid, the Seller Break Fee pursuant to this Section 11.3 on more than one occasion. Notwithstanding anything to the contrary in this Agreement, in the event a termination pursuant to Section 11.1(f), Section 11.1(f) occurs, Buyer Parent's right to receive payment of the Seller Break Fee shall be the sole and exclusive remedy (whether at law, in equity, in contract, tort or otherwise) of Buyers and their Affiliates against Seller Parent and its Affiliates.

11.4 <u>Procedure and Effect of Termination</u>. In the event of the termination of this Agreement and the abandonment of the Transactions pursuant to <u>Section 11.1</u>, written notice thereof shall forthwith be given by the terminating Party to the other Party, Buyer Parent shall return to Seller Parent or destroy all documents, work papers and other materials of Sellers relating to the Business and to the Transactions, whether so obtained before or after the execution hereof (other than, in the case of Buyer Parent or its Affiliates, to its counsel, accountants, financial advisors or lenders, and in the case of Seller Parent or its Affiliates, to its counsel, accountants or financial advisors), and no Party to this Agreement shall have any Liability under this Agreement to any other Party except (i) as otherwise provided in this <u>Article XI</u> or (ii) for any Liability of any Party then in breach of this Agreement; *provided*, *however*, that

the confidentiality provisions contained in Article XIII shall survive the termination of this Agreement except that Section 13.3 shall have no effect and the Confidentiality Agreement shall govern the confidentiality obligations of each Party.

ARTICLE XII

INDEMNIFICATION

12.1 Survival; Effect of Materiality Qualifiers.

(a) The representations and warranties of Sellers and Buyer Parent contained in this Agreement shall survive the Closing for the period set forth in this Section 12.1. All representations and warranties contained in this Agreement and all claims with respect thereto shall terminate upon the expiration of fifteen (15) months after the Closing Date, except that (a) the representations and warranties contained in Section 4.1 [Qualification, Organization, Subsidiaries, Etc.], Section 4.2 [Corporate Authority Relative to this Agreement; No Violation], Section 4.6 [Transferred Entities] (together, the "Seller Fundamental Representations"), Section 5.2 [Share Capital] and Section 5.3 [Corporate Authority Relative to this Agreement; No Violation] (together, the "Buyer Fundamental Representations") and all claims with respect thereto shall survive forever, and (b) the representations and warranties contained in Section 4.14 [Tax Matters] and all claims with respect thereto shall survive until the expiration of the applicable statute of limitations, giving effect to any extensions thereof. In the event that notice of any claim for indemnification under this Article XII has been given pursuant to Section 12.4 or Section 12.5, as the case may be, within the applicable survival period, the representations and warranties that are the subject of such indemnification claim (and the right to pursue such claim) shall survive with respect to such claim until such time as such claim is finally resolved. It is the intention of the Parties that the survival periods and termination date set forth in this Section 12.1 supersede a statute of limitation applicable to such representations and warranties or claim with respect thereof. The right of a Person to any remedy pursuant to this Article XII shall not be affected by any investigation or examination conducted, or any knowledge possessed or acquired (or capable of being possessed or acquired), by such Person at any time concerning any circumstance, action, omission or event relating

(b) In determining whether any representation or warranty in this Agreement was true and correct as of any particular date and the amount of any Losses in respect of the failure of any such representation or warranty to be true and correct as of any particular date, any qualification or limitation as to materiality (whether by reference to material adverse effect or otherwise) contained in such representation or warranty shall be disregarded; provided that this Section 12.1(b) shall not apply to the representations and warranties in Sections 4.3 [Financial Information], 4.9 [Absence of Certain Changes or Events], 4.10(a) [Business Contracts] 4.20 [Acquired Assets] and 4.22 [No Undisclosed Liabilities].

12.2 Indemnification by Sellers.

(a) From and after the Closing and subject to the provisions of this Section 12.2, Seller Parent shall indemnify, defend and hold harmless Buyers, their Affiliates

and their respective officers, directors, employees, agents, successors and permitted assigns, (collectively, the "Buyer Indemnified Parties") from, against and in respect of any and all Losses imposed on, sustained, incurred or suffered by, or asserted against, any of the Buyer Indemnified Parties, whether in respect of third party claims, claims between the Parties, or otherwise, directly or indirectly relating to, arising out of, resulting from, based upon, with respect to or by reason of:

- (i) the breach of any representation or warranty made by Sellers in this Agreement or any certificate delivered pursuant to this Agreement;
- (ii) the breach of any covenant or agreement made by Sellers in this Agreement or any certificate delivered pursuant to this Agreement;
- (iii) any Excluded Liability; and
- (iv) any Indebtedness of the Business or the Transferred Group at Closing to the extent not taken into account in Closing Net Cash.
- (b) Notwithstanding anything to the contrary contained in this Agreement other than Section 12.2(a):
- (i) the indemnification provided in Section 12.2(a)(i) shall be the sole and exclusive post-Closing remedy available to Buyer Indemnified Parties, for any Losses arising out of or in connection with any breach or alleged breach of any representation or warranty contained in this Agreement;
- (ii) Sellers shall have no liability for any claim for indemnification pursuant to <u>Section 12.2(a)(i)</u> unless (A) the Losses for which the Sellers would be responsible for such claim and all related claims exceed the De Minimis Amount and (B) the aggregate amount of Losses (excluding all Losses associated with claims less than the De Minimis Amount) exceeds the Deductible; provided further, that, each claim with respect to Taxes arising out of a particular subject matter, set of facts, events or circumstances in a taxing jurisdiction in any Tax period shall be treated as a separate claim with respect to such Taxes for purposes of this <u>Section 12.2</u>;
- (iii) subject to Section 12.2(c), the maximum aggregate amount of indemnifiable Losses arising out of or resulting from the causes enumerated in Section 12.2(a)(i) that may be recovered from Sellers shall not exceed \$500,000,000 (the "Cap"); and
- (iv) the limitations in Section 12.2(b)(ii) and (iii) shall not apply to any Losses as a result of inaccuracies in the representations and warranties contained in Section 4.6(a) [Transferred Entities] and any such Losses shall not be counted in determining the thresholds or the Cap.
- (c) Notwithstanding the limitations in <u>Section 12.2(b)(iii)</u>, if the aggregate amount of indemnifiable Losses arising as a result of inaccuracies in <u>Section 4.20(a)</u> and (b) [Acquired Assets] (excluding all Losses associated with claims less than the De Minimis

Amount and excluding all Losses to the extent less than the Deductible and including any Damages only in excess thereof) exceeds \$1,000,000,000 in excess of the Cap, (the "Acquired Assets Threshold"), Sellers shall then be liable for any such Losses in excess of the Acquired Assets Threshold, up to an aggregate maximum amount of \$1,000,000,000 (exclusive of any amounts paid up to the Cap).

(d) For the avoidance of doubt, any amounts taken into account in Closing Net Cash or Closing Date Net Working Capital shall not be taken into account under this <u>Section 12.2</u>.

12.3 Indemnification by Buyer.

- (a) From and after the Closing and subject to the provisions of this Section 12.3, Buyer Parent shall indemnify, defend and hold harmless Sellers, their respective Affiliates and their Affiliates' respective officers, directors, employees, agents, successors and permitted assigns (collectively, the "Seller Indemnified Parties", and each of the Buyer Indemnified Parties and the Seller Indemnified Parties, an "Indemnified Party") from, against and in respect of any and all Losses imposed on, sustained, incurred or suffered by, or asserted against, any of the Seller Indemnified Parties, whether in respect of third party claims, claims between the Parties, or otherwise, directly or indirectly relating to, arising out of, resulting from, based upon, with respect to or by reason of:
 - (i) the breach of any representation or warranty made by Buyers in this Agreement or any certificate delivered by Buyers pursuant to this Agreement;
 - (ii) the breach of any covenant or agreement made by Buyers in this Agreement or any certificate delivered pursuant to this Agreement;
 - (iii) any Assumed Liability; and
 - (iv) all Liabilities and Claims relating to the operation of the Acquired Assets (including those assets of the Transferred Entities that would be Acquired Assets if not owned by a member of the Transferred Group) to the extent such Liabilities or Claims are not Excluded Liabilities.
 - (b) Notwithstanding anything to the contrary contained in this Agreement other than Section 12.3(a):
 - (i) the indemnification provided in Section 12.3(a)(i) shall be the sole and exclusive post-Closing remedy available to the Seller Indemnified Parties, as against Buyer for any Losses arising out of or in connection with any breach or alleged breach of any representation or warranty contained in this Agreement;
 - (ii) Buyers shall have no liability for any claim for indemnification pursuant to Section 12.2(a)(i) unless (A) the Losses for which the Sellers would be responsible for such claim and all related claims exceed the De Minimis Amount and (B) the aggregate amount of such Losses (excluding all Losses associated with claims less than the De Minimis Amount) exceeds the Deductible; provided further, that, each claim with respect

to Taxes arising out of a particular subject matter, set of facts, events or circumstances in a taxing jurisdiction in any Tax period shall be treated as a separate claim with respect to such Taxes for purposes of this Section 12.23;

- (iii) the maximum aggregate amount of indemnifiable Losses arising out of or resulting from the causes enumerated in <u>Section 12.2(a)(i)</u> that may be recovered from Buyers shall not exceed the Cap; and
- (iv) the limitations in Section 12.3(b)(ii) and (iii) shall not apply to any Losses as a result of inaccuracies in the representations and warranties of Buyer Parent in Section 5.2(e) [Share Capital], and any such Losses shall not be counted in determining the thresholds or the Cap.

12.4 Third Party Claim Indemnification Procedures.

(a) Except as provided in Section 8.2 with respect to Tax Contest Claims, in the event that any claim or demand for which an indemnifying party (an "Indemnifying Party") may have liability to any Indemnified Party hereunder is asserted against or sought to be collected from any Indemnified Party by a third party (a "Third Party Claim"), such Indemnified Party shall promptly, but in no event more than ten days following such Indemnified Party's receipt of a Third Party Claim, notify the Indemnifying Party in writing of such Third Party Claim, the amount or the estimated amount of damages sought thereunder to the extent then ascertainable (which estimate shall not be conclusive of the final amount of such Third Party Claim), any other remedy sought thereunder, any relevant time constraints relating thereto and, to the extent practicable, any other material details pertaining thereto (a "Claim Notice"); provided, however, that the failure timely to give a Claim Notice shall affect the rights of an Indemnified Party hereunder only to the extent that such failure has a prejudicial effect on the defenses or other rights available to the Indemnifying Party with respect to such Third Party Claim. The Indemnifying Party shall have 90 days (or such lesser number of days set forth in the Claim Notice as may be required by court proceeding in the event of a litigated matter) after receipt of the Claim Notice (the "Notice Period") to notify the Indemnified Party that it desires to defend the Indemnified Party against such Third Party Claim.

(b) In the event that the Indemnifying Party notifies the Indemnified Party within the Notice Period that it desires to defend the Indemnified Party against a Third Party Claim, the Indemnifying Party shall have the right to defend the Indemnified Party by appropriate proceedings and shall have the sole power to direct and control such defense, with counsel reasonably satisfactory to the Indemnified Party at its expense. Once the Indemnifying Party has duly assumed the defense of a Third Party Claim, the Indemnified Party shall have the right, but not the obligation, to participate in any such defense and to employ separate counsel of its choosing at its own cost and expense. The Indemnifying Party shall not, without the prior written consent of the Indemnified Party (such consent not to be unreasonably withheld, delayed or conditioned), settle, compromise or offer to settle or compromise any Third Party Claim on a basis that would result in (i) the imposition of a consent order, injunction or decree that would restrict the future activity or conduct of the Indemnified Party or any of its Affiliates, (ii) a finding or admission of a violation of Law or violation of the rights of any Person by the Indemnified Party or any of its Affiliates, (iii) a finding or admission that would have an adverse

effect to no more than a de minimis extent on other claims made or threatened against the Indemnified Party or any of its Affiliates, (iv) except to the extent within the De Minimis Amount, any monetary liability of the Indemnified Party that will not be promptly paid or reimbursed by the Indemnifying Party or (v) any non-monetary condition or obligation being imposed on any Indemnified Party or any of its Affiliates that would restrict the future activity or conduct of the Indemnified Party or any of its Affiliates.

- (c) If the Indemnifying Party elects not to defend the Indemnified Party against a Third Party Claim, whether by not giving the Indemnified Party timely notice of its desire to so defend or otherwise, the Indemnified Party shall have the right but not the obligation to assume its own defense, it being understood that any right of the Indemnified Party to indemnification for a Third Party Claim shall not be adversely affected by assuming the defense of such Third Party Claim. The Indemnifying Party shall have no liability with respect to a Third Party Claim settled without its consent, which consent shall not be unreasonably withheld or delayed.
- (d) The Indemnified Party and the Indemnifying Party shall cooperate in order to ensure the proper and adequate defense of a Third Party Claim, including by providing access to each other's relevant business records and other documents and employees, it being understood that the costs and expenses of the Indemnified Party relating thereto shall be considered Losses. The Indemnified Party and the Indemnifying Party shall keep each other fully informed with respect to the status of such Third Party Claim.
- (e) The Indemnified Party and the Indemnifying Party shall use reasonable best efforts to avoid production of confidential information (consistent with applicable Law), and to cause all communications among employees, counsel and others representing any party to a Third Party Claim to be made so as to preserve any applicable attorney-client or work-product privileges.
- (f) Each of Buyer Parent and Seller Parent hereby consents to the non-exclusive jurisdiction of any court in which a Third Party Claim is brought for purposes of any claim for indemnification or reimbursement with respect to such Third Party Claim or the matters alleged therein.
- 12.5 <u>Direct Claims</u>. Except as provided in <u>Section 8.2</u> with respect to Tax Contest Claims, if an Indemnified Party wishes to make a claim for indemnification hereunder for a Loss that does not result from a Third Party Claim (a "**Direct Claim**"), the Indemnified Party shall notify the Indemnifying Party in writing of such Direct Claim, the amount or the estimated amount of damages sought thereunder to the extent then ascertainable (which estimate shall not be conclusive of the final amount of such Direct Claim), any other remedy sought thereunder, any relevant time constraints relating thereto and, to the extent practicable, any other material details pertaining thereto. The Indemnifying Party shall have a period of 30 days within which to respond to such Direct Claim. If the Indemnifying Party does not respond within such 30-day period, the Indemnifying Party will be deemed to have accepted the Direct Claim. If the Indemnifying Party rejects all or any part of the Direct Claim, the Indemnified Party shall be free to seek enforcement of its rights to indemnification under this Agreement with respect to such Direct Claim.

12.6 Adjustments to Losses.

- (a) <u>Insurance</u>. In calculating the amount of any Loss, the proceeds actually received by the Indemnified Party or any of its Affiliates under any insurance policy or pursuant to any claim, recovery, settlement or payment by or against any other Person, in each case relating to the Third Party Claim or the Direct Claim, net of any actual costs, expenses or premiums incurred in connection with securing or obtaining such proceeds (including any increased premiums resulting therefrom), shall be deducted, except to the extent that the adjustment itself would excuse, exclude or limit the coverage of all or part of such Loss.
- (b) <u>Taxes</u>. In calculating the amount of any Loss, there shall be deducted an amount equal to any net Tax benefit actually realized (including the utilization of a Tax loss or Tax credit carried forward) as a result of such Loss by the Party claiming such Loss, and there shall be added an amount equal to any Tax imposed on the receipt of any indemnity payment with respect thereto. A Party will be deemed to realize a net Tax benefit in a taxable period in respect of any Loss incurred by such Party to the extent that the cumulative liability for Taxes of such Party for such taxable period through the end of the taxable period, calculated with such Loss excluded, exceeds the actual cumulative liability for Taxes of such Party for such taxable period through the end of such taxable period, calculated with such Loss included.
- (c) <u>Reimbursement</u>. If an Indemnified Party recovers an amount from a third party in respect of a Loss that is the subject of indemnification hereunder after all or a portion of such Loss has been paid by an Indemnifying Party pursuant to this <u>Article XII</u>, the Indemnified Party shall promptly remit to the Indemnifying Party the excess (if any) of (i) the amount paid by the Indemnifying Party in respect of such Loss, plus the amount received from the third party in respect thereof, *less* (ii) the full amount of the Loss.
- (d) Other Recovery. The provisions of this Article XII are not intended to permit duplicate recoveries on the same matters, to the extent that any payment is made under Section 3.3 or Article VIII, no recovery in respect of the same Claim will be available under this Article XII.
- 12.7 Payments. The Indemnifying Party shall pay all amounts payable pursuant to this Article XII, in immediately available funds, to an account specified by the Indemnified Party following receipt from an Indemnified Party of a bill, together with all accompanying reasonably detailed supporting documentation, for a Loss that is the subject of indemnification hereunder, unless the Indemnifying Party in good faith disputes the Loss, in which event it shall so notify the Indemnified Party. In any event, the Indemnifying Party shall pay to the Indemnified Party the amount of any Loss for which it is liable hereunder, in immediately available funds, to an account specified by the Indemnified Party no later than three days following any Final Determination of such Loss and the Indemnifying Party's liability therefor.
- 12.8 <u>Characterization of Indemnification Payments</u>. Except as otherwise required by applicable Law, all payments made by an Indemnifying Party to an Indemnified Party in respect of any claim pursuant to <u>Section 12.2</u> or <u>Section 12.3</u> shall be treated as adjustments to the Global Purchase Price for all applicable Tax purposes.

ARTICLE XIII

MISCELLANEOUS

- 13.1 <u>Assignment</u>. Subject to <u>Section 1.4</u>, this Agreement may not be assigned or otherwise transferred by either Party without the consent of the other Party. Any purported assignment in violation of the preceding sentence shall be void; *provided*, *however*, that Buyer Parent may, without obtaining the prior written consent of Seller Parent, (i) assign its rights or delegate any of its obligations under this Agreement or any Local Transfer Agreement to any Affiliate or Subsidiary of Buyer Parent, or (ii) collaterally assign its rights under this Agreement to its secured lenders or to any administrative agent acting on behalf of its secured lenders; *provided* that Buyer Parent shall, in each case, remain obligated for performance of such obligations. Prior to the Closing, except in the case of Seller Parent contemporaneously terminating this Agreement in accordance with <u>Section 11.1(g)</u>, the ultimate parent entity of any acquirer of Buyer Parent or Seller Parent shall expressly agree to be bound by and perform the pre-Closing obligation of Buyer Parent or Seller Parent, as the case may be, under this Agreement.
- 13.2 <u>Public Announcements</u>. Neither Party nor any of their respective Affiliates shall issue any press release or make any public announcement relating to the subject matter of this Agreement (prior to the Closing Date, with respect to Buyer Parent and its Affiliates, and at all times, with respect to Seller Parent and its Affiliates) without the prior written consent of the other Party, which Consent shall not be unreasonably withheld, delayed or conditioned; *provided*, *however*, that any Party and any of their respective Affiliates may make any public disclosure (i) it believes in good faith is required by applicable Law or any listing or trading agreement concerning its publicly traded securities (in which case the disclosing Party or Affiliate will provide reasonable advance notice, to the extent reasonably practicable, to the other Party prior to making the disclosure and will in good faith consider the reasonable comments of the other Party on such disclosure) or (ii) that is substantially consistent with a prior press release or public announcement made with the consent of the other Party.

13.3 Confidentiality.

- (a) Buyer Parent and Seller Parent agree the Confidentiality Agreement, as it relates to Information (as defined below) shall, as of the Closing Date, be terminated and of no further force and effect.
- (b) Subject to Section 13.2, Seller Parent agrees that, after the Closing, Sellers and its Representatives shall keep confidential and exercise the same degree of care with respect to maintaining the confidentiality of any Information (as defined below) in any of their possession that Sellers exercise with respect to similar types of their own proprietary information, but in no event less than a reasonable degree of care, except that if any Information is required by Law or legal or administrative process to be disclosed, Seller Parent shall promptly (and in any event prior to making such disclosure, to the extent permitted by Law) notify Buyer Parent of such disclosure requirement so that Buyer Parent or its Affiliates may seek a protective Order or other appropriate remedy. In the event that no such protective Order or other remedy is obtained, or Buyer Parent does not waive compliance with this Section 13.3(b), and the

applicable Seller or Representative is nonetheless legally compelled to disclose such information, such Seller or its Representatives, as the case may be, will furnish only that portion of the information which such Seller or Representatives are, advised by counsel is legally required to be furnished and will give Buyer Parent written notice of the information to be disclosed as far in advance as practicable and exercise all reasonable efforts to obtain reliable assurance that confidential treatment will be accorded the information. For purposes of this Section 13.3(b), the term "Information" means (i) all information, knowledge and data of Seller and its Affiliates as of immediately prior to the Closing to the extent related to the Business, the Acquired Assets, the Products and/or the Assumed Liabilities and (ii) all information, knowledge and data provided by Buyer Parent or any of its Affiliates to any Seller in connection with the Transactions other than any information contemplated by clause (i), other than any such information that (A) only with respect to (ii) above, is known to any Seller prior to receipt thereof from Buyer Parent or any of its Affiliates, (B) is disclosed to a Seller by a third Person which has, or is reasonably believed by such Seller to have, a legal right to make such disclosure without requiring such Seller to maintain the confidentiality thereof, (C) is or becomes part of the public domain through no fault of any Seller, or (D) is independently developed by or for any Seller as evidenced by its written records, without reliance or reference to any information contemplated by clauses (i) or (ii).

(c) Subject to Section 13.2, Buyer Parent agrees that, after the Closing, Buyer Parent and its Representatives shall keep confidential and exercise the same degree of care with respect to maintaining the confidentiality of any Information (as defined below) in any of their possession that Buyer Parent exercises with respect to similar types of its own proprietary information, but in no event less than a reasonable degree of care, except that if any Information is required by Law or legal or administrative process to be disclosed, Buyer Parent shall promptly (and in any event prior to making such disclosure, to the extent permitted by Law) notify Seller Parent of such disclosure requirement so that Sellers or their Affiliates may seek a protective Order or other appropriate remedy. In the event that no such protective Order or other remedy is obtained, or Seller Parent does not waive compliance with this Section 13.3(c), and Buyer Parent or its Representative is nonetheless legally compelled to disclose such information, Buyer Parent or its Representatives, as the case may be, will furnish only that portion of the information which Buyer Parent or its Representatives are, advised by counsel is legally required to be furnished and will give Seller Parent written notice of the information to be disclosed as far in advance as practicable and exercise all reasonable efforts to obtain reliable assurance that confidential treatment will be accorded the information. For purposes of this Section 13.3(c), the term "Information" means all information, knowledge and data provided in connection with the Transactions by Seller Parent or its Affiliates to Buyer Parent to the extent related to the Excluded Assets and Excluded Liabilities, other than any such information that (A) is known to any Buyer prior to receipt thereof from Seller Parent or any of its Affiliates, (B) is disclosed to a Buyer by a third Person which has, or is reasonably believed by such Buyer to have, a legal right to make such disclosure without requiring such Buyer to maintain the confidentiality thereof, (C) is or becomes part of the public domain through no fault of Buyer, or (D) is independently developed by or for any Buyer as evidenced by its written records, without reliance or reference to Information received from Seller Parent or any of its Affiliates.

13.4 Expenses. Whether or not the Transactions contemplated by this Agreement are consummated, and except as otherwise specified herein, each Party shall bear its own costs and expenses in connection with this Agreement and the Ancillary Agreements and with respect to the transactions contemplated hereby and thereby.

- 13.5 Severability. In the event that any provision of this Agreement or the application thereof, becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other Persons or circumstances will be interpreted so as reasonably to effect the intent of the Parties. Further, the Parties agree to replace such void or unenforceable provision of this Agreement with a valid and enforceable provision that will achieve, to the extent possible, the economic, business and other purposes of such void or unenforceable provision.
- 13.6 Entire Agreement; Amendment. This Agreement and the Ancillary Agreements (including the Local Transfer Agreements and Transfer Documents after their execution and delivery hereunder) contain the entire agreement of the Parties with respect to the Transactions, superseding all negotiations, prior discussions and preliminary agreements made prior to the date hereof. This Agreement may not be amended, supplemented or otherwise modified except by an instrument in writing signed by both Seller Parent and Buyer Parent. No such supplement, amendment or addition shall be evidence, in and of itself that the representations and warranties in the corresponding section are no longer true and correct in all material respects. Notwithstanding anything to the contrary herein, this sentence of Section 13.6, Section 13.7 and Section 13.10, shall not be amended or otherwise modified without the prior written consent of the Financing Sources to the extent such amendment or modification would have an adverse effect on the Financing Sources' rights thereunder.
- 13.7 No Third-Party Beneficiaries. This Agreement is solely for the benefit of the Parties and, to the extent set forth herein, their respective Affiliate and the Indemnified Parties, and no provision of this Agreement shall be deemed to otherwise confer upon any other third parties any remedy, Claim, Liability, reimbursement, Claim of action or other right in excess of those existing without reference to this Agreement; provided that the Financing Sources are intended beneficiaries of, and shall be entitled to enforce, Section 11.2(c), Section 13.6, this Section 13.7, Section 13.10(b), Section 13.10(c) and Section 13.10(d).
- 13.8 Waiver. The waiver by a Party of any breach of any of the terms, covenants or conditions of this Agreement or of any right or privilege conferred by this Agreement shall not be construed as a subsequent waiver of any such terms, covenants, conditions, rights or privileges or as a waiver of any other terms, covenants, conditions, rights or privileges. No waiver shall be effective unless it is in writing and signed by an authorized Representative of the waiving Party.
- 13.9 Governing Law. This Agreement (and any Claim or controversy arising out of or relating to this Agreement) shall be governed by and construed in accordance with the Laws of the State of New York without giving effect to any choice or conflict of law provision or rule that would cause the application of the Laws of any jurisdiction other than the State of New York.

13.10 Consent to Jurisdiction; Waiver of Jury Trial; No Recourse to Financing Sources.

(a) Each Party hereto hereby irrevocably and unconditionally submits, for itself and its property, to the non-exclusive jurisdiction of any New York State court, or Federal court of the United States of America, sitting in New York, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement and any Ancillary Agreement delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each Party hereto hereby irrevocably and unconditionally:
(a) agrees not to commence any such action or proceeding except in such courts, (b) agrees that any claim in respect of any such action or proceeding may be heard and determined in such New York State court or, to the extent permitted by applicable Law, in such Federal court, (c) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such New York State or Federal court and (d) waives, to the fullest extent permitted by applicable Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such New York State or Federal court. Each Party hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law. Each Party hereto irrevocably consents to service of process in the manner provided for notices in Section 13.16. Nothing in this Agreement will affect the right of any Party hereto to serve process in any other manner permitted by applicable Law.

(b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT AND THE ANCILLARY AGREEMENTS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE ANCILLARY AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS, INCLUDING, BUT NOT LIMITED TO, ANY DISPUTE ARISING OUT OF, OR RELATING TO, THE DEBT COMMITMENT LETTER OR THE FINANCING. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (C) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 13.10(b).

(c) Notwithstanding anything herein to the contrary, each Party hereto agrees that (i) it will not bring or support any action, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against the Financing Sources in any way relating to this Agreement or any of the Transactions, including any dispute arising out of or relating in any way to the Debt

Commitment Letter, the Debt Financing or the performance thereof or of services related thereto, in any forum other than the Supreme Court of the State of New York, County of New York or, if under applicable Law jurisdiction is vested in the Federal Courts, the United States District Court for the Southern District of New York (and appellate courts thereof) and (ii) any such action, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, shall be governed by the laws of the State of New York.

- (d) Notwithstanding anything to the contrary contained herein, Seller Parent on behalf of itself and each of its Affiliates and Representatives (collectively, the "Seller Related Parties") agrees that neither it nor any Seller Related Party shall have any rights or claims against any Financing Source or any of their respective Representatives in connection with this Agreement (including the Transactions), the Financing or the transactions contemplated hereby or thereby and no Financing Source shall have any rights or claims against any Seller or its Affiliates in connection with this Agreement, the Financing or the transactions contemplated hereby or thereby, in each case, whether at law or equity, in contract, in tort or otherwise; provided, that the foregoing will not limit the rights of the parties to the Financing under the Debt Commitment Letter or the definitive documentation related to the Financing.
- 13.11 <u>Specific Performance</u>. The Parties acknowledge that, in view of the uniqueness of the Acquired Assets and the transactions contemplated by this Agreement, no Party would have an adequate remedy at Law for money damages in the event that this Agreement has not been performed in accordance with its terms, and therefore each Party agrees that the other Parties shall be entitled to specific enforcement of the terms hereof, in addition to any other remedy to which they may be entitled (in accordance with <u>Section 13.10</u>), at Law or in equity.
- 13.12 Representation by Counsel. Each Party hereto represents and agrees with each other that it has been represented by or had the opportunity to be represented by, independent counsel of its own choosing, and that it has had the full right and opportunity to consult with its respective attorney(s), that to the extent, if any, that it desired, it availed itself of this right and opportunity, that it or its authorized officers (as the case may be) have carefully read and fully understand this Agreement and the Ancillary Agreements in their entirety and have had them fully explained to them by such Party's respective counsel, that each is fully aware of the contents thereof and their meaning, intent and legal effect, and that it or its authorized officer (as the case may be) is competent to execute this Agreement and has executed this Agreement free from coercion, duress or undue influence.
- 13.13 <u>Bulk Transfers</u>. Buyer Parent waives compliance with the provisions of all applicable Laws relating to bulk transfers in connection with the transfer of the Acquired Assets.
- 13.14 <u>Headings</u>. The headings of the sections and subsections of this Agreement are inserted for convenience only and shall not be deemed to constitute a part hereof.
- 13.15 <u>Counterparts; Signature Pages</u>. The Parties may execute this Agreement in one or more counterparts, each of which will be deemed an original and all of which, when taken together, will be deemed to constitute one and the same agreement. Any signature page hereto

delivered by facsimile machine or by e-mail (including in portable document format (pdf), as a joint photographic experts group (jpg) file, or otherwise) shall be binding to the same extent as an original signature page, with regard to any agreement subject to the terms hereof or any amendment thereto and may be used in lieu of the original signatures for all purposes. Any Party that delivers such a signature page agrees to later deliver an original counterpart to any Party that requests it.

13.16 Notices. All notices, requests, Claims, demands or other communications required or permitted to be given hereunder shall be in writing and may be delivered by hand, by air mail, by nationally recognized private courier (for delivery in no fewer than two (2) Business Days with return receipt requested), or by facsimile. Except as provided otherwise herein, notices delivered by hand shall be deemed given upon receipt; notices delivered by air mail shall be deemed given ten (10) days after being deposited in the mail system, postage prepaid with return receipt requested; notices delivered by nationally recognized private courier shall be deemed given upon receipt; and notices delivered by facsimile shall be deemed given twenty-four hours (24) after the sender's receipt of confirmation of successful transmission. If a notice deemed given upon receipt is given after 5:00 p.m. in the place of receipt (the parties understand and agree that the foregoing applies only to notice and not to copies), such notice will be deemed given on the next succeeding Business Day. All notices shall be addressed as follows:

If to Seller Parent:

Allergan PLC 1 Grand Canal Square Docklands Dublin 2 Ireland

Attention: Chief Legal Officer and Secretary

Facsimile: +1 (862) 261-8043

with copies to (which shall not constitute notice):

Allergan plc Morris Corporate Center III 400 Interpace Parkway Parsippany, New Jersey 07054

Attention: Chief Legal Officer and Secretary

Facsimile: +1 (862) 261-8043

and:

Latham & Watkins LLP 885 Third Avenue New York, NY 10022-4834 Attn: Charles K. Ruck

R. Scott Shean

Facsimile: +1 (212) 751-4864

If to Buyer Parent:

Teva Pharmaceutical Industries Ltd. 5 Basel Street Petach Tikva 4951033 Israel

Attention: Chief Legal Officer Facsimile: +11 972 3 926-7896

with a copy (which shall not constitute notice) to:

Sullivan & Cromwell LLP
125 Broad Street
New York, NY 10004
Attn: Joseph B. Frumkin
Eric M. Krautheimer
Krishna Veeraraghavan

Facsimile: (212) 558-3588

and/or to such other respective addresses and/or addressees as may be designated by notice given in accordance with the provisions of this Section 13.16.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their respective duly authorized officers as of the date first above written.

SELLER PARENT ALLERGAN PLC

By: /s/ Brenton L. Saunders

Name: Brenton L. Saunders

Title: President and Chief Executive Officer

[Signature Page to Master Purchase Agreement]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their respective duly authorized officers as of the date first above written.

BUYER PARENT

TEVA PHARMACEUTICAL INDUSTRIES, LTD

By: /s/ Erez Vigodman

Name: Erez Vigodman

Title: President & Chief Executive Officer

By: /s/ Eyal Desheh

Name: Eyal Desheh

Title: Group EVP and Chief Financial Officer

By: /s/ Sigurdur Olafsson

Name: Sigurdur Olafsson

Title: President & CEO, Global Generic Medicines

[Signature Page to Master Purchase Agreement]

SETTLEMENT AGREEMENT AND MUTUAL RELEASES

This Settlement Agreement and Mutual Releases (the "Agreement") is entered into as of January 31, 2018 (the "Effective Date") by and between Teva Pharmaceutical Industries Ltd. ("Teva") and Allergan plc ("Allergan"). Teva and Allergan shall be referred to collectively as the "Parties" and individually as a "Party."

RECITALS

WHEREAS, on July 26, 2015, the Parties entered into a Master Purchase Agreement which was amended by the First Amendment to the Master Purchase Agreement dated as of June 9, 2016, the Second Amendment to the Master Purchase Agreement dated as of July 5, 2016 and the Third Amendment to the Master Purchase Agreement dated as of July 11, 2016 (the "MPA") through which Teva acquired the Business (the "Transaction") (capitalized terms used herein and not otherwise defined shall have the meanings ascribed thereto in the MPA);

WHEREAS, on August 2, 2016 (the "Closing Date"), the Transaction closed and Teva became the owner of the Business;

WHEREAS, after the Closing Date, the Parties initiated arbitration under Section 3.3 of the MPA in response to Teva's claim for a purchase price adjustment of nearly \$1.5 billion (inclusive of all Claims by either Party under Section 3.3 of the MPA, the "Working Capital Dispute");

WHEREAS, Teva and Allergan have made submissions to the Reporting Accountants in connection with the Working Capital Dispute (the "Submissions");

WHEREAS, on October 30, 2017, Teva asserted several claims for indemnification under Section 12.2 of the MPA ("October 2017 Notice"), including reiterating, restating, and updating claims for indemnification made on November 30, 2016 (such claims for indemnification, together with the claims for indemnification in the October 2017 Notice, the "Teva Asserted Claims") (the Teva Asserted Claims, collectively with any indemnification claims that Teva potentially could assert now or in the future under Section 12.2(a)(i) or Section 12.2(a)(iv) of the MPA, are referred to as the "Teva Indemnification Claims");

WHEREAS, on November 2, 2017, Allergan asserted several claims for indemnification under Section 12.3 of the MPA (the "November 2017 Notice"), including reiterating, restating and updating claims for indemnification made on November 18, 2016 and July 13, 2017 (such claims for indemnification in the November 2017 Notice, the "Allergan Asserted Claims") (the Allergan Asserted Claims, collectively with any indemnification claims that Allergan potentially could assert now or in the future under Section 12.3(a)(i) of the MPA, are referred to as the "Allergan Indemnification Claims");

WHEREAS, by this Agreement, the Parties desire to resolve any and all disputes arising out of, relating to, or in any way connected to the MPA, including but not limited to the Working Capital Dispute, the Teva Indemnification Claims and the Allergan Indemnification Claims, and to avoid future disputes under the MPA; it is the Parties' intention that, on and after the date hereof, (i) the only remedies available to Teva under the MPA are (A) indemnification under

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

ALLERGAN_MDL_SUPP_00000132

Section 12.2(a)(ii) of the MPA for unknown breaches by Allergan of covenants that were intended to be performed by Allergan after the Closing, (B) indemnification under Section 12.2(a)(iii) of the MPA (Excluded Liability), and (C) specific enforcement of Allergan's ongoing covenants; and (ii) the only remedies available to Allergan under the MPA are (A) indemnification under Section 12.3(a)(ii) of the MPA for unknown breaches by Teva of covenants that were intended to be performed by Teva after the Closing, (B) indemnification under Section 12.3(a)(iii) (Assumed Liability) or Section 12.3(a)(iv) of the MPA (Liabilities and Claims relating to the operation of the Acquired Assets), and (C) specific enforcement of Teva's ongoing covenants;

WHEREAS, this Agreement is entered into for purposes of compromise and settlement only;

NOW, THEREFORE, in consideration of the foregoing, and the mutual promises and representations contained in this Agreement, and in exchange for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

AGREEMENT AND MUTUAL RELEASES

- 1. <u>No Admissions</u>. This Agreement is being entered into solely to avoid lengthy, costly and time-consuming disputes. By entering into this Agreement, no Party is admitting any liability or wrongdoing whatsoever, and each Party continues to deny any and all liability and wrongdoing. This Agreement shall not be construed as an admission by either Party as to the merits of any position adopted by the other Party.
- 2. Dismissal of the Working Capital Dispute. Within two (2) Business Days of the Effective Date, the Parties shall jointly notify the Reporting Accountants that the Parties have reached an agreement in principle for the resolution of the Working Capital Dispute and that the Reporting Accountants should cease any and all activities relating to the Working Capital Dispute pending further instructions from the Parties. Within one (1) Business Day of the payment contemplated in Section 3 hereof, the Parties shall jointly notify the Reporting Accountants that the Working Capital Dispute has been finally and fully resolved and that the arbitration is terminated. The Parties shall split evenly any external costs or expenses associated with the Working Capital Dispute, including the fees and disbursements of the Reporting Accountants, but excluding the fees and expenses of the Parties' respective advisors. Upon payment of the Settlement Amount, Allergan's obligations under Section 3.3(g) and Section 3.3(h) of the MPA shall be fully satisfied.
- 3. <u>Payment</u>. Within thirty (30) days following the Effective Date, Allergan shall pay Teva the sum of US \$700,000,000 (the "Settlement Amount"). The Settlement Amount shall be paid by wire transfer to the following Teva account:

MIZRAHI TEFAHOT BANK LTD IL92 0204 6100 0000 0198 781 Main Branch, Tel Aviv Branch No: 461

Account No: 198781

Swift: MIZBILIT

Beneficiary: Teva Pharmaceutical Industries Ltd.

4. Agreed Liabilities and Indemnification; Third Party Claim Indemnification Teva agrees, on behalf of itself and each of its successors-in-interest and Procedures. assigns, that it shall assume, and shall be or become responsible for (i) any Liabilities or Losses arising from the Third Party Claims listed on Exhibit A hereto, (ii) any Liabilities or Losses arising from the Third Party Claims listed on Exhibit B hereto or arising from any other Third Party Claim, in each case to the extent such Liabilities or Losses are based upon generic opioid drugs that are Products, and (iii) any Liabilities, Losses or Claims that are, directly or indirectly, jointly or severally, asserted against or imposed on Allergan, its respective Affiliates and their respective officers, directors, employees, agents, successors and permitted assigns (the "Allergan Parties") to the extent such Liabilities, Losses or Claims are based on parent or control liability or a substantially similar theory in connection with any Proceeding involving (1) a member of the Transferred Group and (2) a Product or the Business (collectively, (i), (ii) and (iii), the "Teva Agreed Liabilities"). For the avoidance of doubt, any Liabilities or Losses arising from the Third Party Claims listed on Exhibit B hereto or arising from any other Third Party Claim, in each case to the extent such Liabilities or Losses are based upon branded opioid drugs of the Retained Business that are not Products, are Excluded Liabilities under the MPA for which Teva is entitled to indemnification under Section 12.2(a)(iii) of the MPA. Teva further agrees that it will indemnify, defend and hold harmless the Allergan Parties, from, against and in respect of any and all Losses imposed on, sustained, incurred or suffered by, or asserted against, any of the Allergan Parties, whether in respect of third party claims, claims between the Parties, or otherwise, directly or indirectly relating to, arising out of, resulting from, based upon the underlying facts of, with respect to or by reason of the Teva Agreed Liabilities. Teva shall have 90 days from the Effective Date to notify Allergan that it desires to defend Allergan against any of the matters listed on Exhibit A hereto in accordance with the terms of Section 12.4 of the MPA. Unless otherwise agreed by the Parties, (i) Allergan shall be responsible for the defense of Third Party Claims involving opioid drugs to the extent such Third Party Claims are based upon branded opioid drugs of the Retained Business that are not Products and (ii) Teva shall be responsible for the defense of Third Party Claims involving opioid drugs to the extent such Third Party Claims are based upon generic opioid drugs that are Products. In the case of Third Party Claims that involve both (i) branded opioid drugs of the Retained Business that are not Products and (ii) generic opioid drugs that are Products, the Parties shall (x) each be responsible for the defense of such Third Party Claims in accordance with the immediately prior sentence and (y) cooperate with each other to enable the proper and adequate defense of such Third Party Claim. Each Party further agrees to provide the other Party by no later than February 28, 2018 a supplemental list which includes all additional known Third Party Claims based upon opioid drugs received by such Party on or before February 25, 2018 ("Supplemental Opioid Case List"), which shall be in substantially the same format as Exhibit B; any Third Party Claims appearing on Exhibit B (or the Supplemental Opioid Case List) shall be deemed to have been notified by each Party in compliance with Section 12.4 of the MPA. On or before the first Business Day of each month beginning after March 31, 2018, each Party shall provide the other Party with a list of additional Third Party Claims based upon opioid drugs that have been filed and served upon the Party on or prior to the third to last Business Day of the prior month ("Monthly Opioid

Case List"). The Monthly Opioid Case List shall be in substantially the same format as Exhibit B and the Supplemental Opioid Case List, and each Party may request a copy of a complaint listed thereon. The Parties' respective rights and obligations pursuant to Section 12.4 of the MPA shall otherwise remain unchanged, including but not limited to the Parties' obligations to cooperate following the date hereof to ensure the proper and adequate defense of a Third Party Claim.

5. Mutual Releases.

- Teva, for itself and its past and present parents, subsidiaries, affiliates, directors, (a) managers, officers, shareholders, employees, attorneys, agents, representatives, predecessors, successors and assigns, hereby fully and forever releases and discharges Allergan and its past and present parents, subsidiaries, affiliates, directors, managers, officers, shareholders, employees, attorneys, agents, representatives, predecessors, successors and assigns, from any and all claims, counterclaims, demands, damages, debts, liabilities, attorneys' fees, actions, causes of action, obligations and demands whatsoever, whether fixed or contingent, at law or in equity, and now known or unknown (each, a "Claim"), (i) arising from or in any way relating to (A) the Working Capital Dispute, (B) the Teva Indemnification Claims (except for any Liabilities or Losses arising from the Third Party Claims listed on Exhibit B hereto or arising from any other Third Party Claim, in each case to the extent such Liabilities or Losses are based upon branded opioid drugs of the Retained Business that are not Products), (C) the Teva Agreed Liabilities, (D) any breach or alleged breach by Allergan of any representation or warranty contained in the MPA, (E) any breach or alleged breach by Allergan of any covenant in the MPA that was intended to be performed by Allergan or its Affiliates on or prior to the Closing, (F) any breach or alleged breach by Allergan prior to the date hereof of any covenant in the MPA that was intended to be performed by Allergan or its Affiliates after the Closing (an "Allergan Post-Closing, Pre-Settlement Covenant Breach") other than any Allergan Post-Closing, Pre-Settlement Covenant Breach the material underlying facts of which are unknown to Teva as of the date hereof or (G) the historical financial statements of the Business or the Transferred Group, including any Claim that such financial statements do not comply with U.S. GAAP or any other applicable accounting standards or Laws, or (ii) for any Losses resulting from any potential Claims that are referenced in the Submissions (collectively, the "Teva Released Claims").
- (b) Allergan, for itself and its past and present parents, subsidiaries, affiliates, directors, managers, officers, shareholders, members, employees, attorneys, agents, representatives, predecessors, successors and assigns, hereby fully and forever releases and discharges Teva and its past and present parents, subsidiaries, affiliates, directors, managers, officers, shareholders, employees, attorneys, agents, representatives, predecessors, successors and assigns, from any and all Claims (i) arising from or in any way relating to (A) the Working Capital Dispute, (B) the Direct Claims specified in the November 2017 Notice, (C) the Third Party Claims for indemnification listed on Exhibit C hereto, (D) any breach or alleged

breach by Teva of any representation or warranty contained in the MPA, (E) any breach or alleged breach by Teva of any covenant in the MPA that was intended to be performed by Teva or its Affiliates on or prior to the Closing or (F) any breach or alleged breach by Teva prior to the date hereof of any covenant in the MPA that was intended to be performed by Teva or its Affiliates after the Closing (a "Teva Post-Closing, Pre-Settlement Covenant Breach") other than any Teva Post-Closing, Pre-Settlement Covenant Breach the material underlying facts of which are unknown to Allergan as of the date hereof, or (ii) for any Losses resulting from any potential Claims that are referenced in the Submissions (collectively, the "Allergan Released Claims").

- (c) Except as provided herein, (i) Teva shall continue to have rights to indemnification under Section 12.2(a)(ii) and Section 12.2(a)(iii) of the MPA; and (ii) Allergan shall continue to have rights to indemnification under Section 12.3(a)(ii), Section 12.3(a)(iii) and Section 12.3(a)(iv) of the MPA. For the avoidance of doubt, (i) Teva shall be prohibited from asserting any of the Teva Released Claims as Claims under Section 12.2(a)(iii) of the MPA, (ii) Allergan shall be prohibited from asserting any of the Allergan Released Claims as Claims under Section 12.3(a)(iii) or Section 12.3(a)(iv) of the MPA and (iii) the rights and obligations of the Parties under Section 9.1 of the MPA shall remain in effect.
- (d) The Parties acknowledge that the releases in this Agreement may include a release of claims, counterclaims, demands, damages, debts, liabilities, attorneys' fees, actions, causes of action, obligations and demands whatsoever, whether fixed or contingent, at law or in equity that are unknown or unsuspected. The Parties hereby waive any common law or statutory doctrine or provision that limits the effect of a release of unknown or unsuspected claims, counterclaims, demands, damages, debts, liabilities, attorneys' fees, actions, causes of action, obligations and demands whatsoever, whether fixed or contingent, at law or in equity. The releases in this Agreement are to be interpreted as broadly as the law allows.
- (e) Teva represents and warrants to Allergan that no Buyer Indemnified Party has received any Third Party Claim against a Buyer Indemnified Party other than (i) the Teva Indemnification Claims and (ii) any Third Party Claims based upon any branded or generic opioid drugs.
- (f) Allergan represents and warrants to Teva that no Seller Indemnified Party has received any Third Party Claim against a Seller Indemnified Party other than (i) the Allergan Indemnification Claims and (ii) the Third Party Claims listed on Exhibit A or Exhibit B hereto and any Third Party Claims based upon any branded or generic opioid drugs.

6. Covenant Not to Sue and Agreement to Indemnify.

(a) Teva agrees, on behalf of itself and each of its current and former directors, officers, employees, representatives, agents, controlling entities or persons, predecessors or successors-in-interest and assigns, (i) that it will neither initiate

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

nor continue any claims, suits, actions, arbitrations or proceedings that seek any relief based upon the Teva Released Claims or the Teva Agreed Liabilities and (ii) that it will not assign or otherwise transfer the Teva Released Claims to any party. Teva further agrees that it will indemnify Allergan for any and all costs, charges or expenses, including but not limited to reasonable attorneys' fees, incurred in connection with any breach of this Section 6(a).

- (b) Allergan agrees, on behalf of itself and each of its current and former directors, officers, employees, representatives, agents, controlling entities or persons, predecessors or successors-in-interest and assigns, (i) that it will neither initiate nor continue any claims, suits, actions, arbitrations or proceedings that seek any relief based upon the Allergan Released Claims and (ii) that it will not assign or otherwise transfer the Allergan Released Claims to any party. Allergan further agrees that it will indemnify Teva for any and all costs, charges or expenses, including but not limited to reasonable attorneys' fees, incurred in connection with any breach of this Section 6(b).
- 7. Representations and Warranties of the Parties. The Parties represent and warrant to one another that:
 - (a) Such Party has the legal right, capacity and authority to enter into this Agreement;
 - (b) Such Party has taken all necessary corporate and legal actions, as applicable, to duly approve the making and performance of this Agreement;
 - (c) This Agreement has been validly executed and delivered by such Party and constitutes its valid and binding obligation, enforceable against the Party in accordance with the terms hereof:
 - (d) Neither the execution nor performance of this Agreement by such Party constitutes or will constitute a violation or breach of such Party's charter or bylaws (or comparable documents, as applicable);
 - (e) Neither the execution nor the performance of this Agreement will constitute a violation or breach of any law, order, injunction, judgment, statute or regulation applicable to such Party or constitutes or will constitute a material default (or would, with the passage of time or the giving of notice, or both, constitute such a default) under any material contract, agreement or other instrument to which such Party is a party or by which it is bound;
 - (f) Such Party has not relied upon any document, statement, representation, promise, inducement, understanding or information made or provided by any other Party or its representatives except as expressly set forth in this Agreement, and such Party has relied solely upon its own due diligence and independent judgment concerning this Agreement and the Party's decision to enter into this Agreement;
 - (g) Such Party has read this Agreement and fully understands all of its terms, covenants, conditions, provisions and obligations and such Party believes that this

- Agreement is a fair, just and reasonable resolution of the Working Capital Dispute, the Teva Indemnification Claims and the Allergan Indemnification Claims;
- (h) Such Party specifically acknowledges that this Agreement shall not be subject to any claim of mistake of fact, that it expresses a full and complete settlement between the Parties, and that regardless of the adequacy or inadequacy of the consideration described herein, this Agreement is intended to be a final and complete settlement of claims and obligations between the Parties described herein as covered by this Agreement; and
- (i) Such Party has not assigned or transferred any Claim or interest in any claim that is the subject of the releases in this Agreement.
- 8. <u>Multiple Counterparts</u>. This Agreement: (i) may be executed in one or more counterparts, each of which shall be deemed an original but all of which together will constitute one and the same instrument and shall be binding upon the person or entity executing the same; and (ii) may be executed by a signature page delivered by facsimile or email, in which case the person or entity so executing this Agreement shall promptly thereafter deliver its originally executed signature page (but the failure to deliver an original shall not affect the binding nature of such person's or entity's signature).
- 9. <u>Governing Law</u>. This Agreement shall be governed by the laws of the State of New York without regard to its conflict of laws provisions.
- 10. **Dispute Resolution**. Any dispute, controversy or claim relating to the interpretation or construction of Sections 4 or 5 of this Agreement, or to the determination of whether a claim for indemnification made by a Party under Sections 12.2 or 12.3 of the MPA is, in fact, subject to indemnification under the MPA, shall be finally resolved by arbitration in accordance with the International Institute for Conflict Prevention and Resolution ("CPR") Rules for Non-Administered Arbitration ("Rules") as in effect on the date of the Agreement, or such other rules and procedures as the Parties may agree. The arbitration will be conducted before a panel of three arbitrators, to be selected in accordance with the screened selection process provided in the Rules. The place of arbitration shall be New York, New York. The language of the arbitration shall be English. Except as otherwise agreed by the Parties, the arbitrators shall issue an award within ninety (90) days of the filing of the notice of intention to arbitrate, and the arbitrators shall agree to comply with this schedule before Any claims for indemnification sought by Allergan involving accepting appointment. allegations against the Transferred Group that relate to Claims based upon (i) contracts for services related to generic drugs that are Products or (ii) alleged or actual violations of competition or antitrust Laws in the generic drug market involving Products (other than any such violation of competition or antitrust Laws relating to any litigation settlement agreement between Allergan and Teva (or between their respective Affiliates)), shall be subject to a rebuttable presumption by the arbitrators that such claims are subject to indemnification by Teva under the MPA. Any claims for indemnification sought by Teva involving allegations against the Transferred Group that relate to Claims based upon (i) contracts for services related to branded drugs of the Retained Business that are not Products or (ii) alleged or

actual violations of competition or antitrust Laws in the branded drug market involving products of the Retained Business that are not Products (other than any such violation of competition or antitrust Laws relating to any litigation settlement agreement between Allergan and Teva (or between their respective Affiliates)), shall be subject to a rebuttable presumption by the arbitrators that such claims are subject to indemnification by Allergan under the MPA. In the event the arbitrators determine that the rebuttable presumption is inapplicable, the arbitrators will then proceed to determine whether the claim for indemnification is subject to indemnification under the MPA. Any award issued by the arbitrators shall be final, binding and conclusive on the Parties hereto and shall constitute an arbitral award upon which a judgment may be entered in any court having jurisdiction thereof. The prevailing party in any arbitration conducted under this provision will be entitled to an award of all fees, costs and expenses of the arbitrators and the arbitration (including, for the avoidance of doubt, reasonable attorneys' fees).

- 11. **No Effect on Manufacturing Agreements**. Nothing in this Agreement shall modify or in any way affect the parties' rights and obligations under any manufacturing or supply agreements between Allergan and Teva (or between their respective Affiliates).
- 12. Kadian Agreement. Teva shall, and shall cause its Controlled Affiliates to, (i) cooperate with Allergan to assign the Asset Purchase Agreement, dated December 17, 2008, by and between Actavis Elizabeth, LLC and King Pharmaceuticals, Inc. (the "Kadian Agreement") to Allergan or its Affiliate, such assignment to be effectuated by an agreement mutually satisfactory to Teva and Allergan, and (ii) prior to the assignment of the Kadian Agreement to Allergan or its Affiliate, cooperate with Allergan to provide Allergan with the benefits of the Kadian Agreement, including cooperation in asserting any indemnification rights of Actavis Elizabeth, LLC (or its successors and assigns) under the Kadian Agreement. Following the assignment of the Kadian Agreement to Allergan or its Affiliate, Allergan shall, and shall cause its Controlled Affiliates to, cooperate with Teva to provide Teva with the benefits of the Kadian Agreement that relate to the authorized generic of Kadian®, including cooperation in asserting any indemnification rights of Teva or its Controlled Affiliates under the Kadian Agreement with respect to any Liabilities or Losses to the extent such Liabilities or Losses are based upon or related to the authorized generic of Kadian®.
- 13. **No Modification**. This Agreement may only be modified or amended by a writing dated after the date hereof and signed by each of the Parties.

14. Construction.

- (a) This Agreement shall be construed so that the word "including" means "including without limitation;" and the singular shall include the plural and vice versa.
- (b) For the avoidance of doubt, "Products" as used in this Agreement shall exclude any products that are Excluded Assets.
- (c) Titles or headings contained in this Agreement are included only for ease of reference and will have no substantive effect.

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

- (d) None of the Parties will be entitled to have any language contained in this Agreement construed against another because of the identity of the drafter.
- 15. Confidentiality. Neither of the Parties hereto shall issue, make or cause to be made any disclosures regarding the terms of this Agreement without the written consent of the other Party, except that the Parties (i) may disclose the terms of this Agreement to attorneys, accountants and other advisors retained by the Party; (ii) may make such disclosures as may be required by applicable laws or regulations, provided that the disclosing Party notifies the other Party in writing of any such requirement and the intended disclosure at least two (2) Business Days in advance of any such disclosure; and (iii) may disclose that they entered into a "Settlement Agreement" without disclosing its terms. Either of the Parties may disclose the terms and conditions of this Agreement if such Party receives a subpoena or other process or order to produce this Agreement, provided that such Party shall, prior to any disclosure to any third party, promptly notify the other Parties to this Agreement so that each Party has a reasonable opportunity to respond to such subpoena, process or order. The Party receiving a subpoena, process or order shall (in the first instance) take no action contrary to the confidentiality provisions set forth above, and shall make reasonable efforts to respond only subject to the confidentiality designation available under a protective order in litigation. The Party objecting shall have the burden of defending against such subpoena, process or order. The Party receiving the subpoena, process or order shall be entitled to comply with it, except to the extent that any other Party is successful in obtaining an order modifying or quashing it.
- 16. **Entire Agreement**. This Agreement constitutes the full and entire understanding and agreement among the Parties with regard to the subject hereof and supersedes any prior negotiations, representations or agreements, written or oral, with respect to such subject matter; provided, however, that nothing herein shall amend, modify, or supersede the Tax Settlement and Resolution Agreement dated October 15, 2017, which the Parties intend to remain in full force and effect.
- 17. **Severability**. If any term or provision of this Agreement is held to be invalid, illegal or contrary to public policy, such term or provision shall be modified to the extent necessary to be valid and enforceable and shall be enforced as modified; provided, however, that if no modification is possible such provision shall be deemed stricken from this Agreement. In any case, the remaining provisions of this Agreement shall not be affected thereby.
- 18. **No Waiver**. Any waiver of any Party's rights under this Agreement is only effective if in writing signed by the Party to be charged or its duly authorized representative, and any such waiver shall only be effective for the specific matter waived and shall not be deemed to apply to any other conduct, provision or other matter.
- 19. **No Assignment**. The Parties agree that they have not, and will not, sell, transfer or assign, or purport to sell, transfer or assign, any Claim or interest in any claim that is the subject of the releases in this Agreement.
- 20. Allocation of Global Purchase Price. Within thirty (30) days following the Effective Date, Allergan shall deliver to Teva the final allocation of the Global Purchase Price (which, for the avoidance of doubt, shall be reduced by the entire amount of the Settlement Payment)

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

among the Acquired Assets (the "Final PPA"). Teva agrees to treat the Final PPA as the Global Purchase Price Allocation in accordance with the MPA.

21. <u>Notices</u>. All notices and other communications hereunder shall be in writing, shall be sent by Federal Express or other expedited courier service, and shall be deemed effective and duly given upon delivery to the other Party at the following addresses or to such other addresses as the Parties may notify one another of in accordance with the provision of this Section:

If to Teva:

Teva Pharmaceutical Industries Ltd. 5 Basel Street Petach Tikva 4951033 Israel Attention: Chief Legal Officer

Attention: Chief Legal Officer Facsimile: +11 972 3 926-7896

With a copy (which does not constitute notice) to:

Vinson & Elkins LLP 666 Fifth Avenue New York, NY 10103 Attention: Ari Berman

Facsimile: +1 (917) 849-5368

If to Allergan:

Allergan PLC Clonshaugh Business and Technology Park Coolock Dublin, D17 E400 Ireland

Attention: Chief Legal Officer and Secretary

Facsimile: +1 (862) 261-8223

With copies to (which shall not constitute notice):

Allergan plc 5 Giralda Farms Madison, New Jersey 07940

Attention: Chief Legal Officer and Secretary

Facsimile: +1 (862) 261-8223

and:

Latham & Watkins LLP 885 Third Avenue New York, NY 10022-4834 Attention: Charles K. Ruck R. Scott Shean Facsimile: +1 (212) 751-4864

22. <u>Independent Legal Advice</u>. This Agreement was negotiated between the Parties at arm's length. Teva and Allergan acknowledge that they have been advised by their own independently selected counsel and other advisors in connection with this Agreement. Teva and Allergan further acknowledge that they enter into this Agreement solely on the basis of advice from independently selected counsel and on the basis of their own independent investigation of all of the facts, laws and circumstances material to this Agreement or any provision hereof, and not in any manner or to any degree based upon any statement or omission by any other party hereto or its counsel. As such, Teva and Allergan agree that they shall have no basis to challenge, set aside or void this Agreement on grounds of fraud, fraudulent inducement or related legal theories.

[SIGNATURE PAGES FOLLOW]

ALLERGAN_MDL_01396697

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be duly executed in their respective names by their duly authorized representatives as of the date and year written below.

TEVA PHARMACEUTICAL INDUSTRIES,	ALLERGAN PLC
LTD.	
Name: Michael McClellan	Name:
Date:	Date:
TEVA PHARMACEUTICAL INDUSTRIES, LTD. Name: / h/con /	

SIGNATURE PAGE TO SETTLEMENT AGREEMENT AND MUTUAL RELEASES

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

ALLERGAN_MDL_SUPP_00000143

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be duly executed in their respective names by their duly authorized representatives as of the date and year written below.

TEVA PHARMACEUTICAL INDUSTRIES,	ALLERGAN PLC
LTD.	
Name:	Name: Chief Legal Officer by Corporate Secretare
Date:	Date: January 31, 2018
TEVA PHARMACEUTICAL INDUSTRIES, LTD.	
Name:	
Date:	

Exhibit A

- U.K. Competition and Markets Authority investigations relating to hydrocortisone tablets and any related Claims, including any related Claims by the National Health Service
- U.K. Competition and Markets Authority investigations relating to carbimazole and any related Claims, including any related Claims by the National Health Service
- U.K. Competition and Markets Authority investigations relating to nortriptyline and any related Claims, including any related Claims by the National Health Service
- U.K. Competition and Markets Authority investigations relating to fludrocortisone acetate and any related Claims, including any related Claims by the National Health Service
- U.K. Competition and Markets Authority investigations relating to dexamethasone and any related Claims, including any related Claims by the National Health Service
- U.K. Competition and Markets Authority investigations relating to amantadine and any related Claims, including any related Claims by the National Health Service
- Lanoel v. Teva Pharmaceutical Industries Ltd., et al., Derivative Action No. 2453-03-17 (Tel Aviv)
- Federal Trade Commission v. Allergan plc, et al., Case No. 3: 17-cv-00312 (N.D. Cal.)
- Floyd v. Feygin, et al., Case No. 507458/2017 (N.Y. Sup. Ct.)
- State of California v. Watson Laboratories. Inc., et al., Case No. 3:17-CV-00562 (N.D. Cal.)
- Benta SAL v. Actavis (MEEA) FZE, et al. (Lebanon)

Exhibit B

Case Name	Case Number	Where Case Was Filed
Adams County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00894	Southern District of Ohio
American Federation of State, County, and Municipal Employees District Council 37 Health & Security Plan, individually and on behalf of all those similarly situated v. Purdue Pharma, L.P.	1:17-cv-02585	Northern District of Ohio
Amite County, Mississippi v. AmerisourceBergen Drug Corporation et al.	5:18-cv-00009-DCB- MTP	Southern District of Mississippi
Ashland County Board of County Commissioners v. AmerisourceBergen Drug Corporation et al.	17-cv-02569	Northern District of Ohio
Baldwin County, Alabama v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00010	Southern District of Alabama
Baptist Hospital, Inc. and Jay Hospital, Inc. v. McKesson Corp., et al.	3:17-cv-00816	Northern District of Florida
Belmont County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00663	Southern District of Ohio
Benton County, Mississippi v. AmerisourceBergen Drug Corporation et al.	3:18-cv-00007-MPM- RP	Northern District of Mississippi
Berkely County Council v. Purdue Pharmaceutical Products, LP, et al.	3:17-cv-00143	Northern District of West Virginia
Board of County Commissioners of Sedgwick County, Kansas v. AmerisourceBergen Drug Corp. et al.	6:17-cv-01313 18-op-45025 (MDL)	District of Kansas

CONFIDENTIAL - SUBJECT TO PROTECTIVE ORDER

Case Name	Case Number	Where Case Was Filed
Bossier Parish v. AmerisourceBergen Drug Corp. et al.	3:17-cv-01815	Middle District of Louisiana
Brooke County Commission et al. v. Purdue Pharma, L.P. et al	17-c-258 through 255	Circuit Court of Marshall County, West Virginia
Brooke County Commission; Hancock County Commission; Harrison County Commission; Lewis County Commission; Marshall County Commission; Ohio County Commission; Tyler County Commission; And Wetzel County Commission,v. Purdue Pharma L.P. et al.	5:18-cv-00009-JPB	Northern District of West Virginia
Brown County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00664	Southern District of Ohio
Buchanan County, Missouri v. AmerisourceBergen Drug Corporation et al.	5:17-cv-06141	Western District of Missouri
Buncombe County v. AmerisourceBergen Drug Corporation et al.	1:17-cv-00310	Western District of North Carolina
Butler County Board of Commissioners v. Purdue Pharma, LP et al.	2:17-cv-01024	Southern District of Ohio
Campbell County, Tennesse v. AmerisourceBergen Drug Corporation et al.	3:18-cv-00006	Eastern District of Tennessee
Catawba County, North Carolina v. AmerisourceBergen Drug Corporation et al.	5:18-cv-00007	Western District of North Carolina
Cecil County, Maryland v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00063-GLR	District of Maryland

Case Name	Case Number	Where Case Was Filed
City of Alexandria, City of Elwood and Madison County v. Purdue Pharma L.P. et al.	1:18-cv-00123-TWP- MJD	Southern District of Indiana
City of Baton Rouge, parish of East Baton Rouge, Louisiana v. AmerisourceBergen Drug Corporation	3:18-cv-00047	Middle District of Louisiana
City of Birmingham v. Purdue Pharma LP et al.	2:17-cv-01360 17-op-45008 (MDL)	Northern District of Alabama
City of Chicago v. Purdue Pharma	1-14-cv-04361 17-op-45169 (MDL)	Northern District of Illinois
City of Chicago v. Purdue Pharma LP et al.	14-cv-60604	Northern District of Illinois
City of Cincinnati, Ohio v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00713	Southern District of Ohio
City of Connersville and Fayette County v. Purdue Pharma L.P. et al.	1:18-cv-00175-WTL- TAB	Southern District of Indiana
City of Dayton, Ohio v. Purdue	3:17-cv-00229	Southern District of Ohio (Originally in Montgomery County Court of Common Pleas)
City of Detroit, Michigan, a municipal corporation v. Purdue Pharma, L.P. et al.	2:17-cv-14075	Eastern District of Michigan
City of Escanaba, Michigan, a municipal corporation v. Purdue Pharma, L.P. et al.	2:17-cv-00208	Western District of Michigan

Case Name	Case Number	Where Case Was Filed
City of Fort Payne, Alabama, a Municipal Corporation v. AmerisourceBergen Drug Corporation et al.	4:17-cv-01877 17-op-45079 (MDL)	Northern District of Alabama
City of Gadsden, Etowah County, and Todd Entrekin, Sheriff of Etowah County, Alabama v. Purdue Pharma LP et al.	4:17-cv-01800 17-op-45101 (MDL)	Northern District of Alabama
City of Greenfield, aka Town of Greenfield v. AmerisourceBergen Drug Corporation et al.	3:17-cv-30175	District of Massachusetts
City of Greenville, Alabama v. Purdue Pharma L.P. et al.	2:17-cv-00836	Middle District of Alabama
City of Greenwood, Indiana v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00047	Southern District of Indiana
City of Hammond v. Purdue Pharma LP et al.	2:17-cv-00431	Southern District of Indiana
City of Indianapolis and Marion County v. Purdue Pharma, L.P. et al.	1:17-cv-04321 17-op-45091 (MDL)	Southern District of Indiana
City of Jacksonville v. AmerisourceBergen Drug Corporation et al.	7:18-cv-00002	Eastern District of North Carolina
City of Kokomo, Indiana v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00060-JMS- MJD	Southern District of Indiana
City of Lafayette v. Purdue Pharma LP, et al.	4:17-cv-00091	Northern District of Indiana

Case Name	Case Number	Where Case Was Filed
City of Lansing, Michigan, a municipal corporation v. Purdue Pharma, L.P. et al.	1:17-cv-01114	Western District of Michigan
City of Lorain, Ohio v. Purdue	1:17-cv-1639	Northern District of Ohio (originally in Lorain County Court of Common Pleas)
City of Manchester v. Purdue Pharma et al.	1:17-cv-00507 (originally 216-2017- CV-00649)	District of New Hampshire (originally in Hilsborough, SS Northern District until removed 10/19/2017)
City of Methuen v. AmerisourceBergen Drug Corporation et al.	1:18-cv-10035	District of Massachusetts
City of New Castle v. Purdue Pharma, LP et al.	1:17-cv-04591 18-op-45016 (MDL)	Southern District of Indiana
City of Noblesville, Indiana v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00048	Southern District of Indiana
City of Opp, Alabama, a municipal corporation v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00840 18-op-45011 (MDL)	Middle District of Alabama
City of Parma v. Purdue Pharma et al.	1:17-cv-01872	Northern District of Ohio (originally in Cuyahoga County Court of Common Pleas until 9/6/2017)
City of Philadelphia v. Allergan plc et al.	2718	Court of Common Pleas of Philadelphia County
City of Revere v. AmerisourceBergen Drug Corporation et al.	1:18-cv-10124	District of Massachusetts

Case Name	Case Number	Where Case Was Filed
City of Seattle v. Purdue Pharma LP et al.	17-2-25505-0 / 17-2- 25504-1	Superior Court of the State of Washington in King County
City of Shreveport v. Purdue Pharma, L.P. et al.	605608	1st Judicial District for the Parish of Caddo
City of Woburn v. AmerisourceBergen Drug Corporation et al.	1:18-cv-10032	District of Massachusetts
Claiborne County, Mississippi v. Purdue Pharma L.P. et al.	5:17-cv-00150	Southern District of Mississippi
Clark County v. Purdue Pharma, L.P. et al.	A-17-765828-C	District court of Clark County, Nevada
Clermont County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00662	Southern District of Ohio
City of Portsmouth v. Amerisource Bergen Corp. et al.	2-17-cv-00723	Southern District of Ohio
Columbia County v. Purdue Pharma, LP et al.	4:17-cv-02067	Middle District of Pennsylvania
Columbiana County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00883	Southern District of Ohio
Coshocton County Board of County Commissioners v. AmerisourceBergen Drug Corporation et al.	2:17-cv-01105	Southern District of Ohio

Case Name	Case Number	Where Case Was Filed
Count of Grand Traverse, Michigan v. Purdue Pharma, L.P. et al.	1:17-cv-01115	Western District of Michigan
Count of Macomb, Michigan v. Purdue Pharma, L.P. et al.	4:17-cv-14077	Eastern District of Michigan
Count of Saginaw, Michigan v. Purdue Pharma, L.P. et al.	1:17-cv-14076	Eastern District of Michigan
County of Armstrong v. Purdue Pharma LP et al.		Armstrong Court of Common Pleas
County of Beaver v. Purdue Pharma LP et al.	11326-2017	Beaver Court of Common Pleas
County of Bowie v. Purdue Pharma, LP et al.	5:17-cv-00168	U.S. District Court Eastern District of Texas
County of Cambria v. Purdue Pharma LP et al.	2017-04131	Cambria Court of Common Pleas
County of Camp v. Purdue Pharma L.P. et al.	2:18-cv-00002-JRG	Eastern District of Texas
County of Cascade v. Purdue Pharma L.P. et al.	4:17-cv-00130	District of Montana
County of Cherokee v. Purdue Pharma, L.P. et al.	6:17-cv-00659	Eastern District of Texas

Case Name	Case Number	Where Case Was Filed
County of Chippewa, Michigan v. Purdue Pharma, L.P. et al.	2:17-cv-00206	Western District of Michigan
County of Crawford, Michigan v. Purdue Pharma L.P. et al.	1:18-cv-10077-JEL- PTM	Eastern District of Michigan
County of Dallas v. Purdue Pharma L.P. et al.	DC-18-00290	116th District Court
County of Delta, Michigan v. Purdue Pharma, L.P. et al.	2:17-cv-00207	Western District of Michigan
County of Falls v. Purdue Pharma L.P. et al.	CV40015	82nd District Court, TX
County of Fayette v. Purdue Pharma L.P. et al.	2017-02676	Court of Common Please of Fayette County, Pennsylvania
County of Franklin v. Purdue Pharma, L.P. et al.	5:18-cv-00002	Eastern District of Texas
County of Franklin v. Purdue Pharma L.P. et al.	5:18-cv-00002	Eastern District of Texas
County of Fulton v. Purdue Pharma L.P. et al.	17EV005007	State Court of Fulton County
County of Genesee, Michigan v. Purdue Pharma, L.P. et al.	4:17-cv-14074	Eastern District of Michigan
County of Greene v. Purdue Pharma LP et al.	AD-791-2017	Greene Court of Common Pleas

Case Name	Case Number	Where Case Was Filed
County of Harrison v. Purdue Pharma, L.P. et al.	2:17-cv-00748	Eastern District of Texas
County of Hopkins v. Purdue Pharma, LP et al.	CV43486	62nd Judicial District Court
County of Jones v. Purdue Pharma L.P. et al.	1:18-cv-00003-C	Northern District of Texas
County of Lackawanna v. Purdue Pharma LP et al.	17-cv-5156	Lackawanna Court of Common Pleas
County of Lamar v. Purdue Pharma, L.P. et al.	4:17-cv-00814	Eastern District of Texas
County of Lawrence v. Purdue Pharma LP, et al.	1180-17	Lawrence County Court of Common Pleas
County of Leelanau, Michigan v. Purdue Pharma L.P. et al.	1:18-cv-00030	Western District of Michigan
County of Leon v. Purdue Pharma, LP et al.	6:17-cv-00318	Westen District of Texas
County of Manistee, Michigan v. Purdue Pharma L.P. et al.	1:18-cv-00033	Western District of Michigan
County of Marquette, Michigan v. Purdue Pharma L.P. et al.	2:18-cv-00001	Western District of Michigan
County of Mason, Michigan v. Purdue Pharma L.P. et al.	1:18-cv-00032	Western District of Michigan
County of Mitchell v. Purdue Pharma L.P. et al.	1:17-cv-00197	Northern District of Texas
County of Mora v. Purdue Pharm et al.	1:17-cv-01044 (originally D-430- CV-2017-00067)	District of New Mexico (originally in State of NM Fourth Judicial District Court until removed

Case Name	Case Number	Where Case Was Filed
		10/16/2017)
County of Morris v. Purdue Pharma, L.P., et al.	2:17-cv-00725	Eastern District of Texas
County of Multnomah v. Purdue Pharma L.P. et al	17CV33413	Circuit Court of the State of Oregon
County of Niagara v. Purdue Pharma LP et al.	E162958/2017	Supreme Court Niagara County
County of Nolan v. Purdue Pharma L.P. et al.	1:17-cv-00196	Northern District of Texas
County of Onondaga, New York v. Purdue Pharma, L.P.	5:18-CV-0100	Northern District of New York
County of Osceola v. Purdue Pharma L.P. et al.	64431661	Circuit Court of the Ninth Judicial Circuit and for Osceola County, Florida
County of Pike v. Purdue Pharma LP, et al.	17-CV-0128	Circuit Court of Pike County, Kentucky
County of Polk v. Purdue Pharma L.P. et al.	9:17-cv-00213	Eastern District of Texas
County of Red River v. Purdue Pharma, L.P., et al.	5:17-cv-00185	Eastern District of Texas
County of Roscommon, Michigan v. Purdue Pharma L.P. et al.	1:18-cv-10078-TLL- PTM	Eastern District of Michigan

Case Name	Case Number	Where Case Was Filed
County of Rusk v. Purdue Pharma, L.P. et al.	6:17-cv-00634	Eastern District of Texas
County of Smith, Texas v Purdue Pharma L.P. et al.	6:17-cv-00699	Eastern District of Texas
County of Throckmorton v. Purdue Pharma L.P. et al.	11125	62nd Judicial District Court
County of Throckmorton v. Purdue Pharma L.P. et al.	1:18-cv-00004-C	Northern District of Texas
County of Titus v. Purdue Pharma, L.P. et al.	5:17-cv-00189	Eastern District of Texas
County of Upshur v. Purdue Pharma LP et.al.	2:17-cv-00672	U.S. District Court Eastern District of Texas
County of Van Zandt v. Purdue Pharma L.P. et al.	17-00302	294th Judicial District Court
Crawford County Board of County Commissioners v. AmerisourceBergen Drug Corporation et al.	2:17-cv-00975	Southern District of Ohio
Darke County Board of County Commissioners	2:17-cv-01064	Southern District of Ohio
DeKalb County, Georgia v. Purdue Pharma, L.P. et al.	17CV12677	Superior Court of DeKalb County
Drew Memorial Hospital, Inc. v. Purdue Pharma L.P. et al.	CV-17-221-3	Circuit Court of Drew County, Arkansas
Eastern Band of Cherokee Indians v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00004	Western District of North Carolina
Erie County Board of County Commissioners v. AmerisourceBergen Corp et al.	2:17-cv-00918	Southern District of Ohio

Case Name	Case Number	Where Case Was Filed
Fairfield County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-01012	Southern District of Ohio
Flandreau Santee Sioux Tribe, Rosebud Sioux Tribe, and Sisseton-Wahpeton Oyate v. Purdue Pharma L.P. et al.	4:18-cv-04003-KES	District of South Dakota
Forrest County, Mississippi v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00009-KS- MTP	Southern District of Mississippi
Gallia County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00768	Southern District of Ohio
Greene County, Tennesse v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00002	Eastern District of Tennessee
Guernsey County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00886	Southern District of Ohio
Hancock County, Tennessee v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00010	Eastern District of Tennessee
Haywood County, Tennesse v. AmerisourceBergen Drug Corporation et al.	2:18-cv-02016	Western District of Tennessee
Hocking County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00769	Southern District of Ohio
Houston County, Alabama v. Purdue Pharma L.P. et al.	1:17-cv-00838	Middle District of Alabama
Humphreys County, Mississippi v. Purdue Pharma, LP et al.	4:17-cv-00190	Northern District of Mississippi

Case Name	Case Number	Where Case Was Filed
Huron County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00891	Southern District of Ohio
IBEW local 38 Health and Welfare Fund, on behalf of itself and all others similarly situated v. Purdue Pharma et al.	1:17-cv-02171 18-op-45069 (MDL)	Northern District of Ohio
Teamsters Local 493 Health Services and Insurance Plan v. Purdue Pharma, L.P. et al.	3:17-cv-02092 18-op-45074 (MDL)	District of Connecticut
IBEW Local 671 Health Services and Insurance Plan v. Purdue Pharma, L.P. et al.	3:17-cv-02093	District of Connecticut
Teamsters Local 677 Health Services and Insurance Plan v. Purdue Pharma, L.P. et al.	3:17-cv-02094 18-op-45071 (MDL)	District of Connecticut
IBEW Local 90 Benefits Plan v. Purdue Pharma, L.P. et al.	3:17-cv-02095	District of Connecticut
IN RE: National Prescription Opiate Litigation	MDL 2804	US Judicial Panel on Multidistrict Litigation
J. Paul Jones Hospital, a Non-Profit Alabama Corporation v. McKesson Corporation et al.	2:18-cv-00029	Southern District of Alabama
Jackson County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00680	Southern District of Ohio
Jefferson County Council v. Purdue Pharmaceutical Products, LP, et al.	3:17-cv-00144	Northern District of West Virginia
Jefferson Davis County, Mississippi v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00200	Southern District of Mississippi

Case Name	Case Number	Where Case Was Filed
Jennings County vs. Purdue Pharma L.P. et al	4:18-cv-00006-RLY- TAB	Southern District of Indiana
Johnson County, Tennesse v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00003	Eastern District of Tennessee
Laborers 17 Health Benefit Fund v. Purdue Pharma, L.P. et al.	1:17-cv-09877	Southern District of New York
Lake County v. Purdue Pharma L.P. et al.	2:18-cv-00024	Northern District of Indiana
Lewis v. Purdue Pharma L.P. et al	5:17-cv-05118 17-op-45076 (MDL)	Western District of Arkansas
Lexington-Fayette Urban County Government v. AmerisourceBergen Drug Corporation et al.	5:17-cv-00442	Eastern District of Kentucky
Licking County Board of County Commissioners v. AmerisourceBergen Drug Corp.et al.	2:17-cv-00904	Southern District of Ohio
Logan County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-01097	Southern District of Ohio
Louisiana Department of Health v. Purdue Pharma et al.	C 661 638 / U61638	East Baton Rouge 19th Judicial District Court
Louisiana Health Service & Indemnity Company d/b/a Blue Cross Blue Shield of Louisiana, and HMO Louisiana, Inc. v. Purdue Pharma, LP et al.	663226	East Baton Rouge 19th Judicial District Court
Louisville/Jefferson County Metro Government v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00508	Western District of Kentucky

Case Name	Case Number	Where Case Was Filed
Luzerne County v. Purdue Pharma, LP et al.	3:17-cv-02043	Middle District of Pennsylvania
Marion County, Mississippi v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00202	Southern District of Mississippi
Marshall County v. Purdue Pharma, L.P. et al.	3:18-cv-00046	Northern District of Indiana
Metropolitan Government of Nashville and Davidson County, Tennessee v. Purdue Pharma L.P. et al.	3:17-cv-01605	Middle District of Tennessee
Mobile County Emergency Medical Services System Rescue Squad, Inc., A nonprofit corporation; v. AmerisourceBergen Drug Corporation et al.	1:18-cv-00016-B	Southern District of Alabama
Morrow County Board of Commissioners v. Purdue Pharma L.P. et al.	2:17-cv-01126	Eastern District of Ohio
Mower County v. Purdue Pharma LP, et al.	0:17-cv-05263	District of Minnesota
MSP Recovery Claims, Series LLC v. Purdue Pharma, L.P.	1:18-cv-00040	Northern District of Ohio
MSPA Claims 1, LLC; MAO-MSO Recovery II, LLC; and MSP Recovery Claims, Series LLC; v. Purdue Pharma L.P. et al.	1:18-cv-00130	Northern District of Ohio
Municipality of Guayama, Puerto Rico v. Purdue Pharma L.P. et al.	3:17-cv-02356	District of Puerto Rico
Municipality of Guayanilla, Puerto Rico v. Purdue Pharma L.P. et al.	3:17-cv-02364	District of Puerto Rico

Case Name	Case Number	Where Case Was Filed
Municipality of Loiza, Puerto Rico v. Purdue Pharma L.P. et al.	3:17-cv-02371	District of Puerto Rico
Municipality of Sabana Grande and Municipality of Cayey on behalf of themselves and others similarly situated v. Purdue Pharma L.P. et al.	3:17-cv-02380	District of Puerto Rico
Muskingum County Board of County Commissioners v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00040-ALM- KAJ	Southern District of Ohio
New Hanover County v. AmerisourceBergen Drug Corp. et al.	7:17-cv-00241	Eastern District of North Carolina
Oklahoma State v Purdue Pharma et al.	CJ-2017-816	Cleveland County District Court
Onslow County v. AmerisourceBergen Drug Corporation et al.	7:18-cv-00001	Eastern District of North Carolina
Ottowa County Board of County Commissioners v. AmerisourceBergen Drug Corporation et al.	3:17-cv-02570	Northern District of Ohio
Pike County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00696	Southern District of Ohio
Ramsey County, Minnesota v. Purdue Pharma L.P. et al.	0:17-cv-05285	District of Minnesota
Regina Hapgood, Individually and on behalf of the Estate of Richard Coelho v. Purdue Pharma L.P. et al.	1:18-cv-10010	District of Massachusetts
Richland County Children's Services v. Purdue Pharma et al.	2017-cv-662; (removed 10/16/2017 to 1:17-cv-02185)	Northern District of Ohio (originally Court of Richland County)

Case Name	Case Number	Where Case Was Filed
Rockingham County v. AmerisourceBergen Drug Corporation et al.	1:17-cv-01114	Middle District of North Carolina
Ross County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00704	Southern District of Ohio
Rush Health Systems, Inc. v. McKesson Corp., et al.	3:17-cv-01012	Southern District of Mississippi
Scioto County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00682	Southern District of Ohio
Sheet Metal Workers Local No. 25 Health and Welfare Fund, on behalf of itself and all others similarly situated v. Purdue Pharma, LP et al.	2:17-cv-05079	Eastern District of Pennsylvania
Shelby County, by the Shelby County Board of Commissioners v. Purdue Pharma, LP et al.	CT-004500-17	Circuit Court for Shelby County, Tennessee
Southwest Mississippi Regional Medical Center; Infirmary Health Hospitals, Inc.; Monroe County Healthcare Authority d/b/a Monroe County Hospital, on behalf of themselves and all others similarly situated v. AmerisourceBergen Drug Corp. et al.	5:17-ev-00145	Southern District of Mississippi
St. Bernard Parish Government versus Purdue Pharma L.P. et al.	17-1532	34th Judicial District Court for the Parish of St. Bernard
St. Croix Chippewa Indians v. McKesson Corporation et al.	3:17-cv-00914	Western District of Wisconsin
State of California v. Purdue Pharma LP,	30-2014-00725287- CU-BT-CXC	Orange County Superior Court

Case Name	Case Number	Where Case Was Filed
State of Mississippi v. Purdue Pharma et al.	G-2015-1814	Hinds County Chancery Court
State of New Mexico, ex rel. Hector Balderas, Attorney General v. Purdue Pharma LP et al.	D-101-CV-201702541	First Judicial District Court, County of Santa Fe, NM
State of Ohio v. Purdue Pharma	17CI000261	Ross County Court of Common Pleas
Sunflower County, Mississippi v Purdue Pharma L.P. et al.	4:17-cv-00189	Northern District of Mississippi
Surry County v. AmerisourceBergen Drug Corp. et. a.	1:17-cv-01159	Middle District of North Carolina
Tallapoosa County, Alabama v. AmerisourceBergen Drug Corporation et al.	3:17-cv-00763 17-op-45097 (MDL)	Middle District of Alabama
Teamsters Health Services and Insurance Plan Local 404, on behalf of itself and all others similarly situated v. Purdue Pharma LP, et al.	1:17-cv-12342	District of Massachusetts
The Borough of Ridgefield v. Purdue Pharma, L.P. et al.	BER-L-007640-17	Superior Court of New Jersey
The City of Columbus v. Purdue Pharma, L.P. et al.	2:17-cv-01102	Southern District of Ohio
The City of Elyria v. Purdue Pharma, LP et al.	17-CV-193904	Lorain County Court of Common Pleas
The City of Nashua v. Purdue Pharma L.P. et al.	1:17-cv-00730	District of New Hampshire

Case Name	Case Number	Where Case Was Filed
The City of Toledo v. Purdue Pharma, LP, et al.	3:17-cv-02516	Northern District of Ohio
The County of Ashtabula v. Purdue Pharma, L.P. et al.	2017-cv-0821	Ashtabula Court of Common Pleas, Ohio
The County of Clinton v. Purdue Pharma L.P. et al.	2018-00000052	Supreme Court of the State of NY County of Clinton
The County of Cuyahoga v. Purdue Pharma, LP., et al.	CV-17-888099	Cuyahoga Court of Common Pleas
The County of Floyd v. Purdue Pharma LP et al.	17-CI-00797	Circuit Court of Floyd County
The County of Jefferson v. Purdue Pharma L.P. et al.	17-CV-554	Court of Common Pleas, Jefferson County, Ohio
The County of Jefferson v. Purdue Pharma L.P. et al.	2:18-cv-00037-ALM- CMV	Southern District of Ohio
The County of Knott v. Purdue Pharma L.P. et al.	17-CI-00257	Circuit Court of Knott County, State of Kentucky
The County of Knott v. Purdue Pharma L.P. et al.	7:18-cv-00006-GFVT	Eastern District of Kentucky
The County of Lake v. Purdue Pharma L.P. et al.	17CV001972	Court of Common Pleas, Lake County, Ohio
The County of Lorain v. Purdue Pharma L.P. et al.	1:18-cv-00145	Northern District of Ohio

Case Name	Case Number	Where Case Was Filed
The County of Rensselaer v. Purdue Pharma, LP et al.	00257543 (Until a judge is assigned)	Rensselaer County Supreme Court
The County of Schoharie v. Purdue Pharma, LP et al.	2017-395	Schoharie County Supreme Court
The County of Tompkins v. Purdue Pharma L.P. et al.	EF2018-0008	Supreme Court of the State of NY County of Tompkins
The County of Trumball v. Purdue Pharma L.P. et al.	4:18-cv-00147	Northern District of Ohio
The County of Trumbull v. Purdue Pharma L.P. et al.	2017 CV 02266	Court of Common Pleas, Trumball County, Ohio
The Fiscal Court of Rowan County, on behalf of Rowan County v. AmerisourceBergen Drug Corp. et al.	0:17-cv-00130	Eastern District of Kentucky
The Fiscal Court of Allen County, on behalf of Allen County v. AmerisourceBergen Drug Corp. et al.	1:17-cv-00196	Western District of Kentucky
The Fiscal Court of Anderson County, on behalf of Anderson County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00070	Eastern District of Kentucky
The Fiscal Court of Bell County, on behalf of Bell County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00246	Eastern District of Kentucky
The Fiscal Court of Boone County, on behalf of Boone County v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00157	Eastern District of Kentucky
The Fiscal Court of Boyd County, on behalf of Boyd County v. AmerisourceBergen Drug Corp. et al.	0:17-cv-00104	Eastern District of Kentucky

Case Name	Case Number	Where Case Was Filed
The Fiscal Court of Boyle County, on behalf of Boyle County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00367	Eastern District of Kentucky
The Fiscal Court of Bullitt County, on behalf of Bullitt County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00727	Western District of Kentucky
The Fiscal Court of Campbell County, on behalf of Campbell County v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00167	Eastern District of Kentucky
The Fiscal Court of Carlisle County, on behalf of Carlisle County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00136	Eastern District of Kentucky
The Fiscal Court of Christian County, on behalf of Christian County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00146	Eastern District of Kentucky
The Fiscal Court of Clark County, on behalf of Clark County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00473	Eastern District of Kentucky
The Fiscal Court of Clay County, on behalf of Clay County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00255	Eastern District of Kentucky
The Fiscal Court of Cumberland County, on behalf of Cumberland County v. AmerisourceBergen Drug Corp. et al.	1:17-cv-00163	Eastern District of Kentucky
The Fiscal Court of Fleming County, on behalf of Fleming County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00368	Eastern District of Kentucky
The Fiscal Court of Franklin County, on behalf of Franklin County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00071	Eastern District of Kentucky

Case Name	Case Number	Where Case Was Filed
The Fiscal Court of Garrard County, on behalf of Garrard County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00369	Eastern District of Kentucky
The Fiscal Court of Greenup County, on behalf of Greenup County v. AmerisourceBergen Drug Corp. et al.	0:17-cv-00105	Eastern District of Kentucky
The Fiscal Court of Harlan County, on behalf of Harlan County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00247	Eastern District of Kentucky
The Fiscal Court of Henderson County, on behalf of Henderson County v. AmerisourceBergen Drug Corp. et al.	4:17-cv-00130	Western District of Kentucky
The Fiscal Court of Henry County, on behalf of Henry County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00073	Eastern District of Kentucky
The Fiscal Court of Hopkins County, on behalf of Hopkins County v. AmerisourceBergen Drug Corp. et al.	4:17-cv-00157	Western District of Kentucky
The Fiscal Court of Jessamine County, on behalf of Jessamine County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00438	Eastern District of Kentucky
The Fiscal Court of Kenton County, On Behalf Of Kenton County v. AmerisourceBergen Drug Corporation et al.	2:17-cv-00182	Eastern District of Kentucky
The Fiscal Court of Knox County, on behalf of Knox County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00248	Eastern District of Kentucky
The Fiscal Court of Laurel County, on behalf of Laurel County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00269	Eastern District of Kentucky

Case Name	Case Number	Where Case Was Filed
The Fiscal Court of Leslie County, on behalf of Leslie County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00249	Eastern District of Kentucky
The Fiscal Court of Lincoln County, on behalf of Lincoln County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00370	Eastern District of Kentucky
The Fiscal Court of Madison County, on behalf of Madison County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00371	Eastern District of Kentucky
The Fiscal Court of Marshall County, on behalf of Marshall County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00147	Western District of Kentucky
The Fiscal Court of Nicholas County, on behalf of Nicholas County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00373	Eastern District of Kentucky
The Fiscal Court of Oldham County, on behalf of Oldham County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00590	Western District of Kentucky
The Fiscal Court of Pendleton County, on behalf of Pendleton County v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00161	Eastern District of Kentucky
The Fiscal Court of Perry County, on behalf of Perry County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00265	Eastern District of Kentucky
The Fiscal Court of Pulaski County, on behalf of Pulaski County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00264	Eastern District of Kentucky
The Fiscal Court of Scott County, on behalf of Scott County v. AmerisourceBergen Drug Corp. et al.	5:17-cv-00474	Eastern District of Kentucky

Case Name	Case Number	Where Case Was Filed
The Fiscal Court of Shelby County, on behalf of Shelby County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00072	Eastern District of Kentucky
The Fiscal Court of Spencer County, on behalf of Spencer County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00557	Eastern District of Kentucky
The Fiscal Court of Union County, on behalf of Union County v. AmerisourceBergen Drug Corp. et al.	4:17-cv-00120	Western District of Kentucky
The Fiscal Court of Whitley County, on behalf of Whitley County v. AmerisourceBergen Drug Corp. et al.	6:17-cv-00250	Eastern District of Kentucky
The Fiscal Court of Woodford County, on behalf of Woodford County v. AmerisourceBergen Drug Corp. et al.	3:17-cv-00727	Eastern District of Kentucky
The Leech Lake Band of Ojibwe v Purdue Pharma L.P. et al.	0:17-cv-05491	District of Minnesota
The People of Northampton County and Northampton County, Pennsylvania v. Purdue Pharma L.P. et al.	C48-cv-2017-11557	Court of Common Please of Northampton County, Pennsylvania
The People of the County of Jersey v. Purdue	2017 L 000011 D 001	Jersey County Circuit Court
The People of the County of Union v. Purdue	2017 L 17	Jersey County Circuit Court (Consolidated) - 9/29/2017
The People of the State of Illinois, The People of Bond County, and County of Bond v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01170 18-op-45004 (MDL)	Southern District of Illinois

Case Name	Case Number	Where Case Was Filed
The People of the State of Illinois, The People of Christian County, and County of Christian v. AmerisourceBergen Drug Corporation et al	17-cv-3262 17-op-45078 (MDL)	Southern District of Illinois
The People of the State of Illinois, The People of Coles County, and County of Coles v. AmerisourceBergen Drug Corporation et al.	1:18-cv-01022-JBM- JEH	Central District of Illinois
The People of The State of Illinois, The People of Edwards County, and The County of Edwards v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01340 18-op-45049 (MDL)	Southern District of Illinois
The People of the State of Illinois, The People of Gallatin County, and County of Gallatin v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01171 17-op-45152 (MDL)	Southern District of Illinois
The People of the State of Illinois, The People of Hamilton County, and County of Hamilton v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01185 17-op-45157 (MDL)	Southern District of Illinois
The People of the State of Illinois, The People of Hardin County, and County of Hardin v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01172 18-op-45003 (MDL)	Southern District of Illinois
The People of The State of Illinois, The People of Jaspar County, and The County of Jaspar v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01342 18-op-45026 (MDL)	Southern District of Illinois
The People of the State of Illinois, The People of Pulaski County, and County of Pulaski v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01255 17-op-45158 (MDL)	Southern District of Illinois
The People of The State of Illinois, The People of Shelby County, and The County of Shelby v. AmerisourceBergen Drug Corporation et al.	3:17-cv-03293 18-op-45007 (MDL)	Central District of Illinois
The People of the State of Illinois, The People of Wabash County, and County of Wabash v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01173 17-op-45103 (MDL)	Southern District of Illinois

Case Name	Case Number	Where Case Was Filed
The People of the State of Illinois, The People of Washington County, and Washington County v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01174 17-op-45151 (MDL)	Southern District of Illinois
The People of The State of Illinois, The People of White County, and The County of White v. AmerisourceBergen Drug Corporation et al.	3:17-cv-01338 18-op-45024 (MDL)	Southern District of Illinois
The Town of Sheridan v. AmerisourceBergen Drug Corp. et al.	1:17-cv-04651 18-op-45055 (MDL)	Southern District of Indiana
The Township of Bloomfield, New Jersey v. Purdue Pharma L.P. et al.	2:17-cv-13462	District of New Jersey
Tippah County, Mississippi v. AmerisourceBergen Drug Corporation et al.	3:18-cv-00006-MPM- RP	Northern District of Mississippi
Township of Irvington v. Purdue Pharma LP et al.	2:17-cv-12155	District of New Jersey
Transferred to NY-Suffolk Supreme Court (formerly The County of Nassau v. Purdue)	40000/2017 (Nassau is 40008/2017)	Suffolk County Supreme Court
Union County, Mississippi v. AmerisourceBergen Drug Corporation et al.	3:18-cv-00008-SA- JMV	Northern District of Mississippi
United Food and Commercial Workers Health and Welfare Fund of Northeastern Pennsylvania, on behalf of itself and all others similarly situated v. Purdue Pharma, LP et al.	2:17-cv-05078	Eastern District of Pennsylvania
Vigo County, Indiana v. AmerisourceBergen Drug Corporation et al.	2:18-cv-00010	Southern District of Indiana

Case Name	Case Number	Where Case Was Filed
Vinton County Board of County Commissioners v. AmerisourceBergen Drug Corp. et al.	2:17-cv-00665	Southern District of Ohio
Washington County v. Purdue Pharma LP et al.	0:17-cv-05287	District of Minnesota
Washington County, Mississippi v. Purdue Pharma, LP et al.	4:17-cv-00191	Northern District of Mississippi
Williamson County, Tennessee v. AmerisourceBergen Drug Corporation et al.	3:18-cv-00008	Middle District of Tennessee
Yadkin County v. AmerisourceBergen Drug Corp., et al.	1:17-cv-01085	Middle District of North Carolina